OFFSHORE HELICOPTER SAFETY INQUIRY January 11, 2010 Tara Place, Suite 213, 31 Peet Street St. John's, NL

January 11, 2010

PRESENT:

John F. Roil, Q.C./
Anne FaganInquiry Counsel
John Andrews/ Canada-Newfoundland and Labrador Offshore
Amy Crosbie Petroleum Board (C-NLOPB)
Ian Wallace/ Hibernia Management and
Cecily Strickland Development Company (HMDC)
Denis Mahoney/D. Blair PritchettSuncor (Petro-Canada)
Alexander C. MacDonald, Q.C./
Stephanie Hickman Husky Oil Operations Ltd.
Lowis Monning/
Nick Schultz
Rolf Pritchard/
Laura Brown LaengleGovernment of Newfoundland and Labrador
Norman J. Whalen, Q.CCougar Helicopters Inc.
Jamie Martin/Allison BattcockFamilies of Deceased Passengers
Kate O'BrienDavis Estate (Pilot) and
agent on behalf of Douglas A. Latto for Lanouette Estate (Co-pilot)
David F. Hurley, Q.C. Offshore Safety and Survival Centre, Marine Institute, MUN
V. Randell J. Earle, Q.CCommunications, Energy and Paperworkers Union
Local 2121

TABLE OF CONTENTSJanuary 11, 2010

Discussion	
MR GARY VOKEY (SWORN)	
MR. TREVOR PRITCHARD (AFFIRMED)	Pg. 19
MR. PAUL SACUTA (AFFIRMED)	Pg. 19
	D
Examination-in-chief by John Roil, Q.C.	
	D 051
Certificate	

January 11, 2010	Mult	i-Pa	ige [™]	Offshore Helicopter Safety Inquiry
	Page 1			Page 3
1 January 11, 2010	6	1		the manner of proceeding in this matter.
2 COMMISSIONER:		2		documents are provided on the website of the
3 O. Good morning, ladies	and gentlemen, and good	3		inquiry for access by parties with standing to
4 morning members of t	he panel. Now, Mr. Roil,	4		have passwords and this, that, and the other
5 you're ready.	1	5		thing, but they do not form a part of the
6 ROIL, O.C.:		6		evidence before the inquiry unless they are
7 Q. Mr. Commissioner, th	ank you very much. I have	7		entered as an exhibit.
8 a few opening comme	nts that I'd like to make,	8	COMN	MISSIONER:
9 and in addition Mr. Ea	urle has indicated to me,	9	Q.	That is true, yes.
10 Mr. Earle represents	CEP, that he has a	10	EARL	E, Q.C.:
11 preliminary issue that	he'd like to bring up	11	Q.	So we are faced not only with the proposition
12 before we get into the	evidence.	12		that there does not appear to be any
13 COMMISSIONER:		13		opportunity provided in the current schedule
14 Q. Oh, yes, to deal with t	hat now, is that the	14		for the asking of questions on these
15 ROIL, Q.C.:		15		documents, the express intent that this is
16 Q. Do you want to deal y	with that now or we can	16		going to be a limited area of inquiry, which I
deal with it after my		17		find quite surprising in the context that the
18 COMMISSIONER:		18		inquiry itself ruled as to whether the matter
19 Q. We may as well deal w	with it now.	19		was an appropriate matter for an undertaking
20 EARLE, O.C.:		20		in the first instance, and you will recall we
21 Q. Good morning, Mr. Co	ommissioner.	21		were given an opportunity and did take an
22 COMMISSIONER:		22		opportunity to explain to the inquiry why we
23 O. Good morning, Mr. Ea	arle.	23		thought it was important for the inquiry to
24 EARLE, O.C.:		24		have these documents. So we find ourselves in
25 O. You will recall that	when the Canadian	25		a situation at this point in time where these
	Dage 2			Page /
1 Association of Petr	oleum Producers Mr	1		1 age 4
2 Barnes was on the	stand we asked that a			it appears does not provide any consideration
2 Dames, was on the 3	provided on undertaking		-	of either CAPP or C-NI OPP returning to the
4 and on December 12	th of this year those items			stand and that the documentation received is
5 were provided At a	meeting of coursel about	5	1	not now even a part of the evidence of the
6 the same time, we we	are advised that counsel	6	1	inquiry and we are now proceeding into the
7 at least and I presup	he he was speaking for		1	evidence and presentations by the operators
the inquiry anticipat	red limitations on the			and I don't think it will come as any surprise
9 capacity of the partie	es to ask questions on	0	1	to you. Mr. Commissioner, that the matters
10 matters produced in	undertakings and it	10		which we sought undertakings from CAPP on are
11 appears to me that in	fact it's a bit more	11	,	matters which we think are appropriate to ask
12 than that now becau	se I'm looking at the	12	1	questions of the operators on as well because
13 latest schedule for the	is inquiry and I note	12		CAPP is just an umbrella organization and
14 that the inquiry is to 1	proceed through to the	14		anyone who's followed our line of questioning
15 18th of February at th	his Phase 1A and nowhere	15	1	knows that we feel that the process involved
15 Four of February at u	hedule any indication of	15	1	in the if you will commitment to the HUERA
17 time allotted for the	examination of parties	17	-	is an important matter for consideration by
18 who have given under	extanting on those documents	18	1	this inquiry because it illustrates how long
19 which have been prov	vided Now it is not only	10	ł	it has taken to put in place something which
20 the CFP which has r	requested documents by	20	-	is described in one of the letters by C-NI OPR
20 undertaking but the	families of the survivors	$\begin{vmatrix} 20 \\ 21 \end{vmatrix}$		as a mature and tested technology and we
22 requested documenta	tion from C-NI OPR and that	$\begin{vmatrix} 21 \\ 22 \end{vmatrix}$	•	think this is an important inquiry if you
22 similarly has been n	rovided There is no	$\begin{vmatrix} 22\\ 23 \end{vmatrix}$		will as to the nimbleness of this industry
24 indication of any tim	he in the schedule when	23	,	when it comes to making modifications related
25 these will be dealt wi	th Now as Lunderstand	24	,	to safety as they pertain to beliconters and
<u>125</u> mose will be usait wi	un now us i unuerstanu	25		to survey us mey pertain to nencopiers and

Ja	nuary 11, 2010	Multi-P	age	Offshore Helicopter Safety Inquiry
		Page 5		Page 7
1	transportation in helicopters. So I raise	1	р	arties, Mr. Earle included, is that simply
2	these matters with you because quite frankly	y 2	b	ecause something was asked for in an
3	CEP 2121, which as I keep reminding you,	3	u	ndertaking did not mean that it was
4	represents an awful lot of the people who ge	t 4	n	ecessarily something that would have to be
5	on these helicopters feels quite stymied in	5	q	uestioned about, it might be no questions
6	its efforts to bring forward those things	6	W	ould arise, in which case we would not put it
7	which are important in the minds of our	7	iı	nto evidence. The second is that he says
8	members as to helicopter safety, and I would	d 8	tł	nere's no spot, and we have always talked
9	ask you to consider and advise how we are i	in 9	a	bout on the tentative schedule that we've
10	the present context to have these document	s 10	d	eveloped from time to time that the response
11	which were undertakings become part of the	ne 11	0	pportunity would be down in the February 15th
12	evidence, and how we are to have an	12	to	the 18th time period, and that was where I
13	opportunity to ask questions, and I say that	13	u	nderstood, and I thought all counsel
14	knowing that there's a backdrop to all of	14	u	nderstood, that these issues would be dealt
15	this, if you will, the elephant in the closet	15	W	with, and if need more time, then we need more
16	in all this, is that everybody is concerned	16	ti	me. In addition, I sent an e-mail to all
17	about the fact that these inquiries somehow	17	С	ounsel recently where we talked about, or I
18	develop a life of their own, and it is not my	18	ta	alked about how these documents could be put
19	desire to have this inquiry go on any longer	19	iı	n, either by consent if everybody agreed,
20	than it should, but as a party we are here to	20	W	e'd simply put them in, or if necessary, we
21	be heard, we're not here to be a decoration,	21	W	ould call back a witness. So I'm a little
22	and we have matters that we feel should be	e 22	b	it stymied by all of that, but let me say
23	looked into and it seems at this point in time	23	tł	nat the HUEBA example that he gives is
24	that the schedule doesn't contemplate that,	24	p	erhaps a bit of a troubling one. Remember
25	and the manner of proceeding doesn't	25	tł	hat in Phase 1, we are identifying issues,
		Page 6		Page 8
1	contemplate that.	1	a	nd we have always said that in Phase 2 we
2	COMMISSIONER:	2	W	ould look at how we would solve those issues.
3	Q. Thank you. Just one or two comments	3	It	would seem to me at this point in time that
4	ROIL, Q.C.:	4	tł	ne timeliness or responsiveness of the
5	Q. Before we do that, before you make comm	ents, 5	ir	ndustry has become an issue already, so if
6	could I respond to my learned friend becaus	se I 6	h	is objective is to simply ask more questions
7	feel I have to.	7	a	bout the timeliness, I wonder whether that
8	COMMISSIONER:	8	W	ouldn't be something that we could look at or
9	Q. Yes. I don't want this mornings session,	9	s	hould look at in Part 2 rather than in Part
10	which we're prepared to move forward, to	end 10	1	, or in Phase 2 rather sorry, the next
11	up in a debate, but by all means, Mr. Roil.	11	р	art of Phase 1, which is called 1B. As I
12	ROIL, Q.C.:	12	S	ay, you know, Ms. Fagan and I constantly have
13	Q. I think I need the opportunity. Mr. Earle	13	k	ept in touch with parties and advised them as
14	referred to Inquiry Counsel and he referred	to 14	to	b how things were developing. As I say, I'm
15	him, and since Ms. Fagan would not be a h	im, 15	SI	urprised at the way in which the question has
16	it is me that was referred to clearly, and I	16	С	ome up. I understand his concern, and
17	want to put some context around this. Firs	t 17	С	ertainly there is no attempt to make sure
18	of all, you will remember our comments at	bout 18	tł	hat issues wouldn't get explored adequately
19	the collaborative approach. I'm a little bit	19	iı	n this phase, but the question of adequately
20	stymied by the fact that I was not given a	20	a	nd whether there's other issues that might
21	heads up from Mr. Earle about his concerns	. I 21	С	ome out of those undertakings, if there are,
22	thought in our collaborative approach we w	ould 22	tł	nen certainly they can come in and you would
23	have done that. First of all, he said that	23	b	e the Judge, sir, of that. So those are the
24	I'd made comments about the limited abilit	y to 24	0	nly comments I wanted to make in response to
25	probe on this. What I did say to all of the	25	W	hat he said.

Janu	ary 11, 2010	Multi-Page ^{TI}	M Offshore Helicopter Safety Inquiry
	Р	age 9	Page 11
1 CO	OMMISSIONER:	1	seemed to indicate that they did not directly
2	Q. I'm not going to invite anybody else to	2	regulate the helicopter provider. We also
3	comment on it now. My understanding was	at 3	heard from representatives of the Transport
4	the time that documents would be provided, a	ind 4	Canada and the TSB about what their areas of
5	I take it from what Mr. Earle has said, that	5	responsibility were. We heard from our
6	they have been provided. The question of ho	w 6	consultant at Aerosafe, and, of course, we did
7	they would be dealt with after they were	7	hear from Mr. Decker, the survivor, some very
8	provided is really a matter for counsel who	8	useful and compelling evidence. We also heard
9	may wish to ask questions. So it's not an	9	from CAPP, the Association of Oil Operators,
10	emergency thing, it's something that is to be	10	from Helly Hansen, and from Memorial
11	expected. I think the proper way to address	11	University, who was a trainer. So we now move
12	this is for counsel to meet you, Mr. Earle,	12	on to the oil industry itself, and I make
13	and any other counsel who might be involved	d, 13	these comments to make sure that everybody is
14	or Mr. Barnes, if he's here, to talk about the	14	aware that the issue here is not about any
15	extent of the questioning that's required or	15	particular company, but about how the industry
16	that you want to do, and what should be	16	itself responds to and deals with safety and
17	possibly admitted. Whether it will be	17	helicopter transit in the offshore. We have
18	admitted or not would have to be a matter that	t 18	chosen three operators as examples, but there
19	would come before me at the hearing, unless	it 19	are obviously many more who have been in the
20	could be decided before the hearing, but all	20	offshore and who will be in the offshore in
21	these things are there for consideration. So	21	the future. The three operators that we have
22	I would suggest that counsel at the earliest	22	are HMDC, who operate Hibernia; Suncor, and
23	opportunity, perhaps at the close of	23	Husky Energy, and these three operators are
24	proceedings on any day, today, tomorrow, th	e 24	being brought forward as examples. When we
25	next day, sit and talk about these things. It	25	started the inquiry there was a fourth company
	Pa	ge 10	Page 12
1	may not be a problem. If it is a problem,	1	out there called Statoil. They were then
2	we'll solve it, but it's not my wish that	2	called, I think, Statoil Hydro. They have now
3	people who want to ask questions that are	3	left. They were doing a drilling program. A
4	legitimate and relevant shouldn't be able to	4	new company has come in called Conoco-
5	ask them. So I think we need not delay the	5	Phillips. They are now doing a drilling
6	proceedings this morning further, but	6	program. Whether and how long they will stay
7	certainly I would suggest a meeting betweer	1 7	remains to be seen. These three companies
8	counsel within the next two or three days.	8	have been here for some period of time, so we
9	Okay, thank you.	9	have chosen them, but the evidence is not
10 RC	DIL, Q.C.:	10	being brought to show who is the best or who
11	Q. Thank you, Commissioner. I did have a num	ber 11	is better than the other, it is about three
12	of opening comments to put context into the	12	examples of how the industry, the individual
13	next few weeks of evidence, and so if you	13	companies, the entities, how they operate with
14	would give me a moment, I would like to ma	ke 14	respect to safety and helicopter transit in
15	those comments now.	15	the offshore. Ultimately, the whole inquiry
16 C0	OMMISSIONER:	16	is about how the C-NLOPB regulates the affairs
17	Q. Yes.	17	of these operators and how these companies
18 R0	DIL, Q.C.:	18	interpret and apply the rules and regulations
19	Q. These are as much for the public as they are	19	of the C-NLOPB. Now the evidence that we will
20	for those within the room. Just to remind us	20	have over the next few days, and, in fact, the
21	all where we have come from and where we	are 21	next few weeks, will really comprise four
22	going, you will recall that we heard first	22	separation presentations. First of all, we
23	from the C-NLOPB, and their evidence was ab	out 23	will have, as we do today, a panel of
24	the regime that they have created that	24	representatives of all three of those
25	regulates the oil operators. Their evidence	25	companies, and they will give some evidence

Janı	ıary 11, 2010 N	Multi-Page ^{TT}	M Offshore Helicopter Safety Inquiry
	Pag	ge 13	Page 15
1	about the things they do together, and the	1	is simply a word to indicate that something
2	things that they share in, and that they work	2	has been eliminated, either whited out or
3	together on, and they will also give some	3	blacked out of a document, and the process of
4	evidence on what they did working together	4	redaction takes place for a number of
5	following the incident on March 12th, because	5	different reasons, principally because
6	I think that there's a lesson for us in the	6	something is highly sensitive. For example,
7	work that was done at that time. After that -	7	the price that a company would pay for the
8	- that will take probably most of this week.	8	rental of an helicopter on an hourly basis, as
9	We will then have each of these three	9	between the three companies, they might be
10	companies talk individually about what they do	D 10	very different, so they would not want to
11	because although they are working together,	11	share with one another the price. The price
12	they are companies that are competing in the	12	has no relevance to us, so something like that
13	international oil industry, and so they have	13	would be redacted. Some individual people's
14	their own separate documentation, they each	14	names would be redacted because of privacy
15	have their own separate contracts, their own	15	concerns, where they're not relevant to who
16	separate safety plans, and they each contract	16	the person is is not relevant to our
17	separately for helicopter services, and they	17	undertakings. The third, and perhaps the
18	each are able to do audits and inspections.	18	larger place where you will now begin to see
19	So we'll have some evidence from them on that	at. 19	redactions, is where things come up that are
20	Remembering our focus, the purpose is not to	20	clearly within the jurisdiction of some other
21	criticize the past, but rather to look for	21	agency or entity like the Transportation
22	opportunities for improvement into the future.	22	Safety Board or Transport Canada. In that
23	There will be very few documents that will	23	case, we have taken the opportunity to take
24	come forward in the joint panel that we're	24	out of documents extracts that would clearly
25	hearing from today because, of course, most of	f 25	indicate that the issue was an issue that is
	Pag	ge 14	Page 16
1	the documents are individual to the companies.	, 1	not before us, but is one that is actually
2	so they will come out in the following panels	2	before the Transportation Safety Board. So
3	of the individual oil companies as we come out	t 3	with that little preview, I have a list of
4	later this month. A little bit more about	4	documents for exhibits for today. I said that
5	confidential documents and what not, and	5	there wouldn't be very many, and those in the
6	exhibits, because we are now moving into an	6	room who have received the list and have seen
7	area where there will be more use of this	7	the list will say there is a large number,
8	process of redaction that we have talked	8	however, the way that we download them onto
9	about, and more use of what we call	9	our electronic database means that one of the
10	confidential exhibits. First of all, there	10	documents have been broken into many small
11	are two types of exhibits that we have. There	11	parts. So the three individual documents just
12	are the public exhibits which go under our	12	by way of description are the Joint Operator
13	website and are freely accessible by all.	13	Panel Presentation, which they will go through
14	There are some documents which are being	14	today with the PowerPoint presentation. We've
15	designated as confidential documents. Those	15	had those before. Then there are a series of
16	documents are documents that generally a	16	documents that go to making up part of what
17	company or an entity has a proprietary	17	the oil industry has called the Helicopter
18	interest in, and so they would not want the	18	Operations Taskforce Report. It's a document
19	world to be able to look at these confidential	19	of this sort of size and it has a number of
20	accuments. The parties in the room and their	20	aus. Each of the tabs in it are a separate
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	they may not be evaluable, on everywhelts for		exhibit for our purposes. It allows us to be
22	nuey may not be available on our websile for public access. Within both of those types of	22	aute to designate some confidential, some as
23	documents or exhibits there is a process that	23	document which is called a Docling Charter
24	we have used called reduction and reduction	24	The panel presentation is a public document
125	the number of the reduction, and reduction	25	The puller presentation is a public document.

January 11, 2010 Mu		Aulti-P	age ^T	M Offshore Helicopter Safety Inquiry
	Pag	ge 17		Page 19
1	Some of what is in the Operations Taskforce	1	I TRE	VOR PRITCHARD (AFFIRMED)
2	2 final report is public, some of it is	2	2 PAU	L SACUTA (AFFIRMED)
3	confidential, and we'll see why as we go	3	8 ROII	, Q.C.:
4	through it, and some of it has been redacted	4	4 Q	. Commissioner, in preparation with the oil
5	5 out. The third document is the Pooling	4	5	operators for this, the presentation has been
6	6 Charter, and that is a confidential exhibit.	6	5	divided into various sections and each of the
7	So what I would ask you to do today is to	7	7	three presenters will deal with individual
8	there have been some changes, I would caution	1 E	3	sections. When I ask questions of them, I
9	the parties in the room, some of the exhibits	9)	would offer that all three could be entitled
10	that were originally set as "P" documents, or	10)	to answer if they feel that they are in a
11	as public, are now "C" documents or	11	l	better position or wish to answer a question,
12	2 confidential, and the explanation of that will	12	2	but in terms of the actual presentation, it
13	come in due course. We won't get to that	13	3	has been divided up so that one person will
14	document in today's evidence, that will be in	14	1	speak on various of the subjects that are
15	5 tomorrow. I would ask that you admit as	15	5	before us, and I understand that Mr. Sacuta is
16	exhibits, Exhibit P-116, and that is the Panel	16	5	to begin.
17	7 Presentation for today, and Exhibit P- 1	17 17	7 MR.	SACUTA:
18	through and that starts at 117-100 through	18	3 A	. I think Mr. Pritchard will begin.
19	206. Those are public documents. The 117- 2	207 19	ROI	, Q.C.:
20	and 208 are confidential documents. Sorry,	20) Q	. Mr. Pritchard is going to make a couple of
21	1 117-209 is a public document. 210, 11, 12,	21	l	comments ahead of time.
22	and 13 are confidential exhibits, and then 14,	22	2 MR.	PRITCHARD:
23	15, 16, and 17, right through to 20, 300, and	23	3 A	. Mr. Commissioner, we're here today on behalf
24	4 401 and 402 are public exhibits, and finally	24	1	of the three operators currently conducting
25	5 C-118, the Pooling Charter, is a confidential	25	5	producing and drilling operations in the
	Pag	ge 18		Page 20
1	exhibit. The Registrar has a revised list of	1	l	offshore Newfoundland, Grand Banks area. We
2	2 the status of these documents available and	2	2	were greatly saddened by the events on March
3	the pages that would have to be changed. As I	3	3	12th. We know there are family members
4	say, this exhibit will not, I don't think,	4	1	watching here today and perhaps on television.
5	5 find its way into discussion today. It will	4	5	First of all, we'd like to say that our
6	tomorrow, and the parties will have an	6	5	thoughts are never far from them. The
7	7 opportunity to meet with me and discuss those	e 7	7	accident was devastating for all the families,
8	changes, or to have a look at them themselves	8	3	and indeed life changed on that particular
9	overnight.	9)	day. It affected many people, including the
10) COMMISSIONER:	10)	three of us here today, our offshore
11	Q. So at any rate, these exhibits should be	11	l	workforce, and indeed the entire province was
12	2 marked as such.	12	2	affected. It seems no one was left untouched.
13	3 ROIL, Q.C.:	13	3	There's nothing more important than safety of
14	Q. Commissioner, the three gentlemen seated	14	1	our workforce, and this includes the safe and
15	before you today are as follows. Paul Sacuta	15	5	reliable transportation of our employees to
16	seated on your left. Mr. Sacuta is the	16	5	and from the offshore industry. We require
	President of Hibernia Management and	17	/	the helicopter operations to run our oil and
	bevelopment Corporation. Seated next to hir	n 18	5	gas business. This is based on the
19	Manager Operations for Health Free	4	<i>.</i>	environment in which we work in and the safety
$ ^{20}$	ivializer Operations for Husky Energy, and	u 20)	there are risks involved with trevel to and
$\begin{vmatrix} 2 \\ 2 \end{vmatrix}$	Managar for Tarra Nova with Sunagar Error	u 21	L N	from the offehore workforce workeless
	² Ivialized for Terra Nova with Suncor Energy	y. $\begin{bmatrix} 22\\ 22 \end{bmatrix}$	2	rather. The industry is committed to reducing
	affirmed as appropriate	23) 1	risk and on a continuous improvement basis
$\begin{vmatrix} 24 \\ 25 \end{vmatrix}$	$+ \qquad \text{annucl as appropriate.}$	24	+	We recognise there is tremendous interest from
14-		14.	,	The recognise more is itemendous interest nom

Januar	y 11, 2010 Mul	ti-Page ^{TT}	M Offshore Helicopter Safety Inquiry
	Page 21		Page 23
1	the public as well as many of the stakeholders	1	Then in 2000. I became the General Manager for
2	in our offshore safety, and the work of this	2	Blue Water Services in Aberdeen. At that
3	inquiry we will have an opportunity to	3	time I was responsible for three EPSOs
4	discuss our industry as it works our safety	4	working in the North Sea and one FPSO in
5	culture our safety knowledge and give	5	South Africa all of which required
6	information to assist the inquiry on a	6	transportation by helicopter During my time
7	continuous improvement basis of course This		in the North Sea I was involved in a number
8	allows us to in a go forward position to	8	of initiatives with UKOOA which is the United
0	improve the safety of the industry so that we	0	Kingdom Offshore Operators Association and
10	can travel to and from the offshore facilities	10	typically that would be one would be safety
10	safely We'll now give you some background	10	was an initiative undertaking and other EPSO
12	and contact to ourselves as individuals as	12	working groups. Lalso had regular interfaces
12	to why we represent our companies here today	12	with the certifying authorities which we'll
13	I have been in the cil production and	13	beer about on the regulators there. I then
14	transportation business for 36 years	14	ioinad Husky Energy in Juna 2005 as the
15	icined the Merchant News in 1073 as an	15	Operations Manager That was just a few
10	angingering codet with PD Shipping. I worked	10	months prior to sail away of the See Pose to
10	at son on many types of grude oil tankers and	10	the White Pose Field. In August last year I
10	product corriers. I worked through the ranks	10	was appointed as General Manager Operations
19	to Chief Engineer When Lbecame involved	19	Hucky Energy My responsibilities now are for
20	with a project with PD in 1086 this was one	20	the production facilities the drilling
21	of the first EDGOG New EDGO Mr		operations the marine, and the logistics
22	Commissioner is a floating production storage	22	which includes the belicopter operations
25	and offloading system. This is one of the	25	Although I have not worked on a regular
24	very first ones in 1986. It was at that time	24	rotation offshore for many years. I still do
2.5	very mist ones in 1960. It was at that time	2.5	Totation offshore for many years, 1 still do
	Page 22	2	Page 24
1	that I started to be introduced to actual oil	1	travel offshore approximately six times per
2	production. Previously, I was always involved	2	year. I would estimate that throughout the
3	with the transportation and ships of the crude	3	course of my career, I've travelled more than
4	oil and products. I joined that project in	4	100 times offshore, to and from offshore on
5	1986 and worked in the Belfast Shipyard	5	helicopters. As it happens, my son now works
6	Harland & wolff, and during that time I held	6	followed in my footsteps and he works
7	offshore of on offshore worker, hosping Chief	7	offshore in the North Sea too. I've been in
8	onshore as an onshore worker, becoming Unier	8	St. John's for more than four and a half
9	Engineer on board that particular project in	9	years. I find it somewhat similar to Aberdeen
10	ail production facility on the marine	10	fishing industry working together. Lariou
11	on production facility on the manne	11	living in St. John's for the people's attitude
12	project solid away in 1000 and Lieft that	12	and work athics. I arisy the outdoors
13	project in 1996 Throughout the course of	15	walking and golfing Mr. Commissioner at
14	that time I travelled to and from the	14	the time of the accident. I was the senior
15	installation by heliconter. In 1996 Ligined	15	representative for Husky Energy and that's
17	a company called Blue Water Services They	17	why Liepresent my company today. I'll now
18	provide FPSOs to the oil industry and at that	18	pass on to Mr. Vokey, who will give us some
19	time L became Offshore Installation Manager	19	hackground
20	so I worked offshore responsible for all	20 MR 1	/OKFY·
21	aspects of safety and helicopter	20 MR. V	Mr Commissioner Mr Roil my name is Gary
22	transportation on board In 1909 I became	22	Vokey I am the Asset Manager for the Terra
23	the Project Manager for an FPSO date	23	Nova Project, and I've been in the oil and gas
24	commissioning, bring that vessel to a safe	24	industry for some 29 years. I was born and
25	location in preparation for its next work.	25	raised in Newfoundland, with a Bachelor I
		1	· · · · · · · · · · · · · · · · · · ·

Janu	uary 11, 2010	Multi-P	age	Offshore Helicopter Safety Inquiry
		Page 25		Page 27
1	attended Memorial University for my form	nal 1		my co-workers, both offshore and onshore, have
2	education, and I graduated with a Bachelor	of 2	2	helped make a dream for a lot of
3	Engineering, specializing in civil	3	3	Newfoundlanders come true. I've spent a
4	engineering. I began my career in 1981 with	th 4	ļ	number of years working offshore in various
5	Petro Canada, or as we now know it as Sun	cor 5	5	places, and have flown numerous times of
6	Energy in their Drilling Department. My	6	<u>5</u>	helicopters in several jurisdictions. I have
7	offshore drilling career included working	7	7	flown offshore in excess of 75 times. I also
8	offshore Nova Scotia, the Grand Banks,	, 8	3	have a family member that continues to work a
9	Labrador, and the east coast of Africa. I'm	9)	regular rotation off the Grand Banks. There's
10	pleased to say that I was working offshore in	n 10)	no question the loss of our colleagues on 491,
11	1984 when the Terra Nova discovery was n	nade. 11		March 12th, has changed the way people view
12	The early part of my career included a numb	per 12	2	our industry, and indeed it was a tragedy. I
13	of moves between Alberta and Newfound	land, 13	3	hope through this inquiry that we learn and we
14	with two out of my three children being bo	m 14	Ļ	continue to assure ourselves, our co-workers,
15	in St. John's. In 1988, I moved into the	15	5	and our families, that we are committed to
16	production side of our business, and spent th	ne 16	5	safety and the offshore is a safe place to
17	next eight years working at various oil and	17	7	work. Thank you.
18	gas facilities in Alberta. In 1996, I moved	18	COMM	ISSIONER:
19	back into the offshore world, this time on th	e 19) Q.	Thank you.
20	production side of our business, and some of	of 20) MR. SA	ACUTA:
21	the highlights of my career since then	21	Α.	Mr. Commissioner, my name is Paul Sacuta. I'm
22	included a one year secondment to Norsk H	ydro, 22	2	currently the President of the Hibernia
23	a company you will now recognize as Statoi	1 in 23	3	Management and Development Company. I have
24	Norway, during the mechanical completion,	the 24	Ļ	over 25 years experience in the oil and gas
25	commissioning and the start up of an offsho	re 25	5	business. After graduating from the
]	Page 26		Page 28
1	production platform. I've also had a one ye	ar 1	1	University of Alberta with a Degree in
2	secondment to the Hibernia Manageme	ent 2	2 (Chemical Engineering, I started my career in
3	Development Company in the role of Produ	ction 3	3 1	Western Canada where I worked as a drilling
4	Supervisor on the Hibernia Platform, and, i	n 4	4 6	engineer and a facilities engineer at a number
5	fact, myself and Mr. Sacuta were back to ba	ck 5	5 (of locations in Alberta. After seven years in
6	Production Supervisors on that facility som	e 6	5 Y	Western Canada, myself and my family moved to
7	ten years ago. I was also one of the first	7	7 t	he Island of Sumatra in the country of
8	offshore installation managers on the Terra	1 8	3]	Indonesia, where I spent five years working at
9	Nova FPSO through the mechanical complet	tion, 9) (one of the world's largest onshore LUK
10	commissioning, hook up, and first oil on the	e 10) ((phonetic) gas producing facilities.
11	Terra Nova in early 2002. This was certain	y 11		Thereafter, my family and I moved to Doha,
12	one of the highlights of my career. Since	12	2 (Qatar, where I worked in various supervisory
13	2002, I've been working onshore, first in the	e 13	3 1	evel positions at a newly constructed LUK
14	role of Operations Manager, then in the role	e 14	L]	Financial gas facility. In late 1998, my
15	of Operations and Engineering Manager, an	d my 15	5 f	family and I moved to St. John's to begin an
16	current role as Terra Nova Asset Manage	r, 16	5 8	assignment on the Hibernia Platform. To say I
17	where I'm accountable for all aspects of	17		was thrilled would be an understatement. My
18	production, drilling, reservoir, and	18	8 1	wife was born and raised in St. John's, so it
19	engineering for the Terra Nova asset. As a	ı 19) 1	was a homecoming from my family's perspective,
20	Newtoundlander, it creates pride in me to se	e 20) 8	and working on the Hibernia Platform was a
21	what our industry has created over the cours	e 21	2	great opportunity from a career perspective.
22	of the last 30 odd years. When I graduated in	n 22		spent four and a half years working offshore
23	1981, this was a dream that many would n	ot 23	5 6	as a Production Supervisor, and lastly as the
24	nave reit would have been possible in their	24	ι (Justice Installation Manager. I worked a 21
25	careers. I sit nere today knowing that I and	25) (hay on and 21 day off rotation and stood

Ja	nuary 11, 2010	Multi-F	Pag	Offshore Helicopter Safety Inquiry
	Р	age 29		Page 31
1	beside a large number of personnel who		1	there is a glossary at the very back of your
2	continue to work offshore today. In the		2	presentation, which we perhaps won't go to and
3	summer of 2003, I moved to an onshore		3	spend any time with, but this industry, like
4	supervisory engineering position with	4	4	many in the world, are replete with acronyms,
5	Hibernia. In 2005, I moved I was asked to	4	5	so we'll try as much as possible to avoid
6	take a rotational assignment working in		6	using them
7	Equatorial Guinea, West Africa, supervising	5	7 М	R. SACUTA:
8	the operations of three offshore producing	8	8	A. Absolutely, we'll try to use the full term
9	facilities. During my time rotating back and	9	9	where we can. We may jump back to the acronym
10	forth to West Africa, my family stayed in St.	10	0	at some points.
11	John's, which was very important to me. In	. 11	1 C	OMMISSIONER:
12	2007, October of 2007, I returned to Hibernia	1 12	2	Q. One of the more difficult things when I
13	in my current position. In my career I've	13	3	started to read about all this were the
14	flown on helicopters in Western Canada,	14	4	acronyms everywhere.
15	offshore California, the UK sector of the	15	5 M	R. SACUTA:
16	North Sea, the Gulf of Mexico, West Africa	, 10	6	A. Yes. As Mr. Roil mentioned, we've broken up
17	Sable in Nova Scotia, and obviously I continu	ue 17	7	the presentation into 14 sections today, and
18	to fly offshore today to the Hibernia	18	8	we will be splitting the responsibility to
19	Platform. I would estimate I've taken well	19	9	discuss those sections throughout the next two
20	over 100 helicopter flights during my career.	20	0	days. Section 1 will provide an introduction
21	In my current role, I usually try to travel	21	1	to the concept of join ventures, as well as
22	offshore approximately 10 times per year, so	I 22	2	introduce us, the operators of the three
23	have a personal interest in helicopter safety.	23	3	offshore Newfoundland producing facilities.
24	As the President of HMDC, I am responsible for	or 24	4	Section 2 will provide a brief overview of the
25	the safety of the entire Hibernia workforce,	25	5	petroleum industry, discuss our unique
	P	age 30		Page 32
1	so I also have a professional interest in	1	1	environmental operating environment, and
2	helicopter safety. In the spirit of		2	provides a description of the offshore
3	continuous improvement, I'm here today	to 3	3	producing facilities, a very brief
4	support the work of the inquiry as we	4	4	description. Section 3 will describe the
5	collaborate to ensure we have reduced the ri-	sk í	5	regulatory environment under which we operate,
6	of helicopter transportation to as low as		6	as well as Cougar Helicopters. Sections 4 and
7	reasonable practicable. Regardless of where	e î	7	5 will describe the concept of safety
8	my career takes me in the future, this will	8	8	management systems which all of the operators
9	always be home for my family, and I have	a g	9	utilize, as well as safety participation and
10	very high desire to see Newfoundland and	d 10	0	communications, and how important they are to
11	Labrador continue to prosper and grow. I	1	1	us. Section 6 discusses the importance
12	believe safe operations are critical to the	12	2	operators place on contracted services. We
13	success of our industry.	13	3	all use extensive contract personnel in our
14	COMMISSIONER:	14	4	facilities. Section 7 will discuss helicopter
15	Q. Thank you, gentlemen.	15	5	operations in the Newfoundland area. Section
16	MR. SACUTA:	10	6	8 will discuss the personal protective
17	A. I'm going to do the presentation outline, if I	17	7	equipment related specifically to helicopter
18	could, Mr. Roil.	18	8	transportation. Section 9 will review the
19	ROIL, Q.C.:	19	9	specific qualifications and training required
20	Q. Yes, before that, I'm just going to say,	20	0	to work offshore. Section 10 discusses each
21	gentlemen, from time to time I may call upo	n 2	1	of the operator's emergency preparedness and
22	you to give us your opinion on something i	n 22	2	now it does interface with the Department of
$ ^{23}$	relation to your experience, so I'm glad that	1	3	National Defence. Section 11 discusses the
$ ^{24}_{2^{-}}$	we nave that before us. Mr. Sacuta, I would	1 24	4	Department of National Defence search and
125	say just for the public and for the media,	25	5	rescue capabilities and responsibilities.

Jai	nuary 11, 2010 Mult	i-P	age	ge [™] Offshore Helicopter Safety Inquiry
	Page 33			Page 35
1	Section 12 discusses the work of a	1		PetroCanada and is now Suncor The White Rose
	Heliconter Operations Task Force which was put	2)	is operated also by one of the project co-
3	in place by the area operators immediately	3	2	venturers. Husky Energy
	after the events of March 12th		, L R C	ROIL OC:
5	Section 13 will discuss the current	5	5	O I take it we'll look at each project in a
6	status of the follow-up recommendations which	6	, 5	little more detail a little later on?
	were identified in the Helicopter Operations	7	, 7 MI	AR SACUTA.
	Task Force Report And lastly we will have		2 1011	A Ves we will
9	some closing remarks	9	,) R(
10	ROIL OC:	10)	0 Good okay
11	0. Good, thank you. Okay. Perhaps then you can	11	M	MR. SACUTA:
12	take us to slide number four.	12)	A. We now have a short overview of the offshore
13	MR. SACUTA:	13	3	petroleum industry in the Newfoundland and
14	A. As the introduction today, I will discuss the	14	ł	Labrador area.
15	concept of joint ventures, as well as identify	15	5	Operations on the Grand Banks are exposed
16	the operators of the three projects in the	16	5	to: wind speeds in excess of 100 knots, which
17	Newfoundland area. Based on the financial	17	7	is approximately 190 kilometres per hour;
18	commitment required to obtain acreage, explore	18	3	waves as high as 24 metres or approximately 75
19	and develop this acreage in an offshore	19)	feet; air temperatures as cold as minus 17
20	environment, companies frequently form joint	20)	degrees Celsius; and sea water temperatures as
21	ventures. The companies participating in a	21	l	low as minus 2.4 degrees Celsius. It is one
22	joint venture are often referred to as co-	22	2	of the harshest operating environments in the
23	venturers or partners. The term is	23	3	world.
24	interchangeable. Each co-venturer's	24	RC	ROIL, Q.C.:
25	respective ownership interest in the joint	25	5	Q. I would actually stop you there and ask each
	Page 34			Page 36
1	venture is based on the percentage of the	1	l	one of you, in your own experience, either
2	assets and liabilities each co-venturer holds	2	2	personally or what you're aware of because of
3	in the particular project. The rights and	3	3	your professional responsibilities, are you
4	obligations of the co-venturers are usually	4	1	aware of an environment anywhere in the world
5	governed by a joint operating agreement or a	5	5	where oil is extracted or explored that has
6	JOA. Hibernia, Terra Nova and White Rose all	6	5	harsher environment than this particular one?
7	have joint operating agreements, the terms of	7	/ MI	MR. SACUTA:
8	which are specific to each individual project.	8	3	A. I think the combination of environmental
9	As previously mentioned, Hibernia is	9)	conditions under which we operate is one of
10	operated by the Hibernia Management and	10)	the most unique in the world. There are areas
11	Development Company or HMDC, which is a	11	l	that have certain aspects, but the combination
12	separately incorporated company, the shares of	12	2	of the seas, the temperatures, and in the next
13	which are owned by the Hibernia co-venturers	13	3	slide I'll talk a little bit about ice and
14	or partners.	14	ŀ	icebergs. It is very much a unique operating
15	ROIL, Q.C.:	15	5	environment.
16	Q. Okay. So I take it that HMDC was a company	16	5 RC	ROIL, Q.C.:
17	formed for the purpose of managing this	17	7	Q. Either one of you other Gentlemen?
18	project?	18	CC	COMMISSIONER:
19	MR. SACUTA:	19)	Q. If I may?
20	A. Absolutely.	20) R(ROIL, Q.C.:
21	ROIL, Q.C.:	21	l	Q. Yes, absolutely.
22	Q. Ukay.	22	2 CC	COMMISSIONER:
23	MR. SACUTA:	23	3	Q. Just a question on that. What area of the
$ ^{24}_{22}$	A. Terra Nova is operated by one of the project	24	ι -	world would be closest to our offshore area,
125	co-venturers, suncor, which at the time was	125	,	in terms of narsnness?

January 11, 2010 Mult		Multi-	-Page TM		Offshore Helicopter Safety Inquiry
	Pa	ge 37			Page 39
1	MR. SACUTA:		1	u	nique.
2	A. I think two areas come to mind and I'll ask		2 R	OIL, Q	C.:
3	the others, but I think Sokland Island in		3	Q. C	kay. Is there anything about the individual
4	Russia, similar environmental conditions, col	d	4	ty	pes of facilities, the GBS versus the FPSO,
5	weather and snow and ice, and then the North	h	5	a	nd we'll learn a bit more about them, is
6	Sea. Although the North Sea doesn't have the	e	6	tł	here anything about them that is impacted
7	same ice and iceberg issues to deal with, I		7	d	ifferently because of ice?
8	think the North Sea would be the other one.		8 M	IR. SAG	CUTA:
9	COMMISSIONER:		9	A. I	mean, we'll talk about it a little bit on
10	Q. I see, and my understanding is that conditions	3	10	S	ubsequent slides, but for example, the
11	are somewhat different in the northern North		11	Н	ibernia GBS was designed to withstand sea ice
12	Sea than the southern part.		12	a	nd impact of icebergs, whereas the Terra Nova
13	MR. SACUTA:		13	a	nd Sea Rose facilities aren't designed for an
14	A. Yes, absolutely.		14	ir	npact.
15	MR. VOKEY:		15 R	OIL, Q	C.:
16	A. Off the Shetland Islands would probably be the	ne	16	Q. S	o they move in the event of ice?
17	closest marine type of environment, in terms		17 M	IR. SAG	CUTA:
18	of wind and wave. As Mr. Sacuta indicated,	,	18	Α. Τ	hey have different actions that they would
19	they do not have the iceberg challenges that		19	ta	ke based on the threat of prevailing ice or
20	we have.	/	20	ic	ebergs.
21	MR. PRITCHARD:	/	21 R	OIL, Q	C.:
22	A. I think the other important aspect is the	/	22	Q.P	erhaps we can get into that when we get onto
23	distance that we travel between -	,	23	tł	ose particular slides.
24	COMMISSIONER:	,	24 N	IR. SAG	CUTA:
25	Q. Yes.	,	25	A. Iı	n the offshore Newfoundland and Labrador
	Pag	ge 38			Page 40
1	MR. PRITCHARD:		1	a	rea, there are currently three world class
2	A. The kilometres over sea. Some areas will		2	р	roducing facilities: Hibernia, Terra Nova and
3	travel that kind of distance, would have the		3	tł	e White Rose field. The Hebron oilfield is
4	capability of going over land before going		4	ir	its early design phase, with an expected
5	over sea, so a unique combination here.		5	st	art up of between 2016 and 2017. We
6	ROIL, Q.C.:		6	C	ollectively employ over 3400 people between
7	Q. Thank you. Okay, Mr. Sacuta.		7	0	ffshore and onshore, and currently,
8	MR. SACUTA:		8	aj	pproximately 700 people are based on our
9	A. As I mentioned, ice is a significant hazard		9	0	ffshore facilities on any given day. This
10	for the Grand Banks operators. The Jeanne		10	d	oes not include the Conoco rig which is
11	D'Arc Basin is located on the southern edge of	of	11	C	urrently drilling offshore Newfoundland which
12	the marginal ice zone. Both sea ice and		12	c	an have personnel on board as high as 180.
13	icebergs are prevalent between March and Ma	ay.	13 R	OIL, Q	C.:
14	Ice and icebergs are considered in both the		14	Q.C	kay. Just getting to that again and to
15	design and operating strategies of all three		15	re	emind ourselves, I see by the slide that you
16	operators. Each operator does have an ice		16	h	ave and by the grouping of the oilfields, the
17	management protocol in place. This includes	S	17	li	ttle red dots, that they're actually
18	surveillance by aircraft and vessels. It		18	re	elatively close to one another, the three
19	includes towing of icebergs using wire ropes		19	fa	cultures that you operate. In relation to
20	and nets. It also includes deflection using		20	th	e three of you, whereabouts is the
21	our vessels using water cannons or prop wash	I. [2	21	C	onocoPhillips process?
22	That's really only effective on the smaller	1	22 N	IR. SAC	CUTA:
23	icebergs, as opposed to the big ones. As I		23	A. C	onoco is not in the same direction at all.
$ ^{24}_{-}$	mentioned, I think the combination of ice and		24	lt m m-	s on the southern
25	weather makes our operating environment	ι ΄	25 N	IK. PRI	ICHAKD:

January 11, 2010 Mult		ti-P	age	e TM	Offshore Helicopter Safety Inquiry
	Page 4	1			Page 43
1 A. Sou	uthwest of Newfoundland.	1		A.	That's correct. It's a gravity base structure
2 ROIL, Q.C.	.:	2	2		which is set on the ocean floor.
3 Q. I th	ink we have this diagram forward which	3	8 RO	DIL.	Q.C.:
4 sho	we the Laurentian Basin being marked.	4	ļ.	Q.	Okay, and the depth in that area is
5 MR. SACU	JTA:	5	5	-	approximately 80 metres of water?
6 A. Yes	s, that's correct.	6	5 MF	R. S	ACUTA:
7 ROIL, Q.C.		7	7	A.	80 metres of water.
8 Q. And	d they're in that area?	8	8 RO	DIL,	Q.C.:
9 MR. SACU	JTA:	9)	Q.	Yeah, okay.
10 A. Yes	s, that's correct, so opposite direction to	10) MF	R. S	ACUTA:
11 our	flight path.	11		A.	The Hibernia platform, from the bottom of the
12 ROIL, Q.C.		12	2		GBS to its highest point, the platform is
13 Q. Rig	ht, okay. You also heard me mention in	13	3		approximately 224 metres high. This is
14 ope	ening comments about StatoilHydro when they	14	Ļ		equivalent to a 60-storey office building, to
15 wer	re here earlier, or Statoil, I think they're	15	5		put it in perspective. The platform weighs
16 call	led now.	16	5		approximately 1.2 million tons and has three
17 MR. SACU	JTA:	17	7		main components. The topsides, which includes
18 A. Rig	ght.	18	3		two drilling rigs, utility systems which
19 ROIL, Q.C.		19)		generate power and other essential services,
20 Q. Wh	nen they were when they had their drilling	20)		water, oil and gas handling facilities and
21 pro	gram, were they close to you or were they	21			living accommodations for up to 280 personnel.
22 som	newhere else again?	22	2		You can see the helideck on the southwest part
23 MR. SACU	JTA:	23	3		of the platform, just above the accommodations
24 A. The	ev were farther out. Same direction, but	24	ŀ		module.
25 mu	ch farther out than where we were located.	25	5 RO)IL.	0.C.:
	Page 4	2			Page 44
	1 age 4	2		0	That's the green area with the circle on it?
1 KOIL, Q.C	av	2	, ME	2. 2. 5	
3 MR SACUT	ау. ГА.	3	3	Δ	The green area with the circle on it. The
4 A Hib	pernia was the first field to begin		, . L	11.	second component is the gravity based concrete
5 proc	duction in the offshore Newfoundland area	5	5		structure It's 85 metres high It contains
6 As	discussed earlier it is operated by the	6	ń		two drilling shafts one for each of the
	ernia Management Development Company, which	7	, ,		individual drilling rigs. It also contains
8 is m	and up of six co-venturers ExxonMobil		2		the storage for the Hibernia platform
	avron Suncor Canada Hibernia Holding	0	,)		approximately 1.3 million barrels of storage
10 Cor	moration which is actually the Federal	10	,)		capacity and as I've mentioned it's designed
	vernment The Federal Government got	11	,		to withstand the impact of sea ice and
11 00v	olved in the early '90s when Gulf Oil	12)		icebergs
12 Invo	ked out of the project. It includes Murphy	12	2		Lastly we have an offshore loading
	and Statoil. The platform is located 315	13	,		system which we use to transfer oil from the
14 Oli	and Staton. The platon is located 515	14	r		CPS to tankers, for shipment to market, or to
15 KIIO	tres and it was discovered in 1979 and	15	,		the Newfoundland Transshipment Terminal
10 met	iovad first oil in November of 1007 So	10	, , DO	ш	
	can see the length of time it took from	11	$\frac{1}{2}$	лг, О	Okay
10 you	initial discovery until the first oil was	10	,) мс	لا. کد	
19 uic 1	rovimataly 18 years	20) IVII	х. э л	I'll be talking further about some of the
	TOATHACTY TO YEATS.	20	,	А.	safety related features of the Hibernia
$\begin{bmatrix} 21 & \text{KOIL}, Q.C.: \\ 22 & O & The$	at actual facility as we show it there in)		platform during the individual papal
$\begin{bmatrix} 22 & Q. & I \text{ IIIa} \\ 23 & \text{that} \end{bmatrix}$	diagram that actually sits on the bottom	22	-		presentations next week I'd like now to hand
$\begin{bmatrix} 2.5 \\ 2.4 \end{bmatrix}$ of \mathbf{f}	he ocean floor does it?	23	, I		over to Mr. Vokey if Loould
$\begin{vmatrix} 24 & 01 \\ 25 & MD \\ S & CUTT$		24	н с D О	лт	
23 MIK. SACUT	1A:	123	, KO	лL,	, ų.c.,

January 11, 2010 Multi-F		^M Offshore Helicopter Safety Inquiry
Page	e 45	Page 47
1 0. Before you do, you've indicated that there are	1	an FPSO. The FPSO is located in the centre of
2 drilling rigs, two drilling rigs -	2	our production field and is connected to the
3 MR. SACUTA:	3	subsea wellheads that are drilled, in the case
4 A. Yes.	4	of Terra Nova, by a separate drilling unit.
5 ROIL, Q.C.:	5	The majority of the Terra Nova wells were
6 Q that are a part of this particular facility.	6	actually drilled by the MODU Henry Goodrich
7 MR. SACUTA:	7	from 2000 to 2007.
8 A. That's correct.	8 ROIL	, Q.C.:
9 ROIL, Q.C.:	9 Q.	Okay. Now you've just used a MODU. What's a
10 Q. So drilling and production of oil both take	10	MODU?
11 place from the same one facility?	11 MR.	VOKEY:
12 MR. SACUTA:	12 A.	Sorry. It's a mobile offshore drilling unit,
13 A. That's correct. The design included two	13	in this case, a semisubmersible, the Henry
14 drilling rigs. We currently operate one	14	Goodrich. So our wells are drilled by a
15 drilling rig, but we jump back and forth	15	separate facility and then the wells are
16 between the rigs as slots become available.	16	actually tied back into the FPSO for
17 One of the rigs, the west rig, actually all	17	production.
18 the slots, 32 drilling slots have been fully	18	The field, similar to Hibernia, is in
19 drilled and in order for us to drill again	19	relatively shallow water. In the case of
20 from that side of the rig, we actually have to	20	Terra Nova, it's 90 to 100 metres, and our
21 deplete a well and then we're able to reclaim	21	subsea wellheads, because of the shallow water
22 the well to then drill again from that rig.	22	depth, and it doesn't have the protection of
23 So right now, we jump back and forth with a	23	an ice wall, similar to Hibernia, they're in -
24 single drilling through between the two	24	- those wellheads are in large excavations and
25 drilling rigs.	25	it's approximately 15 feet below the surface
Page	e 46	Page 48
1 ROIL, Q.C.:	1	of the water and that's to protect the
2 Q. Okay. With those two activities, does that	2	wellheads in the event of icebergs, if they
3 mean that you would generally have more	3	come in and ground in that water depth, if
4 persons on board at any one time?	4	they do become mobile again, they won't drop
5 MR. SACUTA:	5	down and hit the wellhead. So it's a
6 A. Yes, our POB right now would average anywher	re 6	mechanism for producing and protecting the
7 between 220 and 250, depending on the activity	y 7	subsea wellheads.
8 that's underway at the time.	8	Product on the FPSO is obtained from the
9 ROIL, Q.C.:	9	reservoir and is generally a mixture of oil,
10 Q. Okay. I think we renext going to go to Mr.	10	gas and water and it comes onto the FPSO
12 MD VOKEY.	11	heard the products are concreted. The cilic
12 MR. VORET:	12	cleaned. The gas is dried and reinjected and
A. I II taik, WI. Commissioner, a fittle bit		any water that is produced is cleaned all the
14 about Terra Nova. Terra Nova was the second 15 producing field in the Jeanne D'Arc Basin It	14	oil is taken out and it's discharged
15 producing field in the scaline D rate Dash. It 16 was discovered in 1984 and work to bring this	15	overboard
into develop was formally sanctioned in 1997	17 ROII	
and first oil was in 2002 PetroCanada now	18 0	So you say the gas is injected back into the
19 Suncor Energy, is the largest interest holder	19	earth?
20 in the field and is the operator. Partners in	20 MR	VOKEY:
the develop includes. vou can see them there.	21 A	The gas is reinjected back into dedicated
22 ExxonMobil, Statoil, Husky Energy. Murphy	22	wells and that is to help us keep pressure
23 Mosbacher and Chevron Texaco.	23	maintenance in the wells, and there's also
24 The field is developed using a floating	24	water, seawater is injected also to keep the
storage and offloading vessel, which we call	25	pressure up so the oil will freely flow.

Ja	nuary 11, 2010	Multi-l	Pa	ge TM	Offshore Helicopter Safety Inquiry
	I	Page 49			Page 51
1	The produced crude is offloaded, similar		1	A.	If there's no questions, I'll turn it over to
2	to Hibernia and White Rose. It's offloaded to	0	2		Mr. Pritchard.
3	purpose-built tankers. In the case of Terra		3 1	ROIL,	Q.C.:
4	Nova, the oil goes directly, for the most		4	Q.	Yeah.
5	part, to Whiffen Head and then there's second	nd	5 1	MR. PI	RITCHARD:
6	leg tankers that take it from there to various		6	A.	Husky has been involved on the east coast of
7	markets on the Eastern Seaboard.		7		Canada for over 25 years and the original
8	Like our counterparts at Hibernia and		8		White Rose field was discovered in 1984 and
9	White Rose, that diagram also shows a stand	lby	9		Husky and Suncor, they're the joint venture
10	or supply vessel and it shows a helicopter	1	0		partners in the core White Rose. Now Nalcor
11	there. As Mr. Pritchard indicated,	1	1		Energy is a partner with us in the North
12	helicopters are the primary means of our	1	2		Amethyst, which is a new area and a new
13	transportation for people offshore.	1	3		tieback to the FPSO. So Nalcor Energy has a
14	Terra Nova was the first purpose-built	1	4		five percent stake in this new development.
15	FPSO for the east coast basin, and it's the	1	5		We currently have four glory holes and
16	largest fully disconnectible FPSO in the work	d 1	6		these glory holes are the excavations that Mr.
17	today. In terms of dimensions, the vessel is	1	7		Vokey spoke to regarding protection of the
18	approximately 1,000 feet long or the	1	8		subsea equipment. So we operate with four,
19	equivalent size of three football fields.	1	9		three of which are associated with oil
20	It's about 150 feet wide and from the bottom	n 2	20		production, water injection and gas lift and
21	of the vessel to the helideck, it's	2	21		the northern drill centre is for the gas
22	approximately eight storeys. The additional	1 2	22		reinjection.
23	facility features, the vessel is ice classed.	2	23		Very much like the Terra Nova facility,
24	Just the helideck is on the forward part of	2	24		Mr. Vokey has described how it operates and
25	the vessel and just about a third of the way	2	25		how it works. The Sea Rose facility is very,
	I	Page 50			Page 52
1	back it's not a good depiction there.		1		very similar. So I'll go through some of the
2	We'll show it better in the Suncor but it's	S I	2		subtle differences, rather than go across
3	what we refer to as the turret and that part		3		where its similar.
4	of the vessel is actually linked to the sea		41	ROIL,	Q.C.:
5	floor and the vessel will rotate around it,		5	Q.	Yeah, I think particularly the differences,
6	depending on wind and current. So that pa	rt	6		particularly as they relate to helicopters
7	of the vessel is linked right to the sea		7		would be important.
8	floor. Other features of the FPSO is we have	e	8 1	MR. P	RITCHARD:
9	our own power generation. We can stor	re	9	А.	Absolutely.
10	upwards just under a million barrels of oil,	1	0 1	ROIL,	Q.C.:
11	and in terms of the facilities, the	1	1	Q.	Thank you.
12	accommodations, Terra Nova was origin	ally 1	2 1	MR. P	RITCHARD:
13	designed with 80 beds. During a major	r 1	3	А.	So one of the differences, the accommodation
14	maintenance outage in 2006, an additional	40 1	4		is at the stern of the vessel, very much like
15	beds were added. So during normal operation	ons, 1	5		a tanker orientated vessel, and we have the
16	we have capacity upwards of 120 beds for u	use. 1	6		helideck above the accommodation and slightly
17	Approximately 75 percent of the people	1	7		off to the port side, the left side.
18	that work on the Terra Nova FPSO work	a 1	8 1	ROIL,	Q.C.:
19	regular 21-day-on and 21-day-off rotation	n. 1	9	Q.	Okay. So as we look at that photograph there,
20	The rest are what we refer to as ad-hoc	2	20		you see the actually, you see a photograph
21	personnel who come out for various period	s of 2	21		I guess of a helicopter either standing or
22	time, depending on what their work scope i	s. 2	22		about to land.
23	ROIL, Q.C.:	2	23 1	MR. P	RITCHARD:
24	Q. Okay. That's fine, thank you.	2	24	A.	It's just about to land, I believe, and you
25	MR. VOKEY:	2	25		can see that there's a little bit of an

January 11, 2010 Mr		Iulti-Pa	age	Offshore Helicopter Safety Inquiry
	Pag	e 53		Page 55
1	overhang of the helideck on the installation	1		there's also a drilling program that would be
2	to the port side.	2		executed from a variety of vessels and
3	ROIL, Q.C.:	3		structures. The drilling programs are
4	Q. So it's to the rear of the ship, as opposed to	4		associated with exploration, initial finding
5	on the bow, and it's slightly set off to the	5		of oil, delineation and appraisal of those
6	port side?	6		finds and any water injection or gas injection
7	MR. PRITCHARD:	7		or gas lift that we need to be associated
8	A. That's correct.	8		with. So we see on the slides there, the
9	ROIL, Q.C.:	9		variety. There's four different types there
10	Q. Okay.	10		and we see the jack-up rig to the left, and
11	MR. PRITCHARD:	11		that attaches itself to the seabed with legs
12	A. We also offload to the offload tankers from	12		and is a fixed structure there. It has
13	the stern of the vessel. You can see a large	13		limited capabilities in the Newfoundland land
14	reel on the back end. That's how we offload	14		because of the ice and the iceberg situation.
15	our oil. We also have a disconnectible turret	15		so it has a limited period, around from June
16	system, similar to the Terra Nova. So in the	16		to November time, that that type of facility
17	event of icebergs that we are unable to	17		can be used.
18	manage, we have the facilities to move away.	18	ROIL	0.C.:
19	The Sea Rose is also different in respect	19	0.	Okay. Do jack-up or have jack-up rigs ever
20	of it is a weather vaning FPSO.	20	χ.	been used in Newfoundland?
$ _{21}^{-0}$	ROIL O.C.:	21	MR. P	RITCHARD:
22	O What does weather vaning mean?	22	A	Yes indeed We used the Rowan Gorilla 6 in
23	MR PRITCHARD	22	11.	2006
$ _{24}^{23}$	A The weather vaning is such that we will move	23	ROIL	
25	and turn wherever the wind and the current	25	0 NOIL,	And would that have a helideck or a facility
		2.5	ν.	Dogo 56
	rage	e 34		Fage 30
	would take the vessel. So we end up as a big			Tor fanding hencopters on board?
	Tame News esset has threaters that mould		MR. P	KIICHARD:
3	Terra Nova asset has thrusters that would	3	A.	Absolutely. All of these drilling
4	assist them to do heading control. So if they	4		arrangements have helidecks associated with
5	wish to orientate themselves in a particular	5		them. All these activities require people and
6	way, they have that facility.	6		therefore people require transportation, nence
7	we have a maximum person on board, POB,	17		the slide deck here for the different types.
8	of 90 and they generally work a three-and-	8		So the next picture there shows us a
9	three rotation.	9		semisubmersible or a mobile offshore drilling
10	Some of our development wells were	10		unit, a MODU we would describe it. The next
	predrilled prior to the arrival of the Sea			picture is snown to be a drill snip and that s
12	Rose on site. So very similar in the Terra	12		currently the style of vessel that
13	Nova asset, predrills in the glory holes prior	13		ConocoPhillips are using down in the
14	to the Sea Rose coming and getting hooked up	p 14		Laurentian Basin. And of course, the Hibernia
15	with the flow lines to start production.	15		is a fixed structure. Mr. Sacuta has
16	ROIL, Q.C.:	16		indicated how their drilling program works.
17	Q. Okay, and that drilling is done by other	17		So all of these activities require
18	facilities, other semisubmersibles or other	18		people, generally purpose-built for drilling
19	drill ships?	19		activities and therefore we need helicopter
20	MR. PRITCHARD:	20		transportation for these facilities.
21	A. That's correct, and my next section actually	21	ROIL,	Q.C.:
22	just describes a little bit about the various	22	Q.	And I take it that because these rigs travel
23	styles and types of drilling activities.	23		all over the world that generally helicopters
24	So along with the production operations	24		are the preferred means of transport to
25	from the three mentioned producing assets.	25		offshore facilities in other places in the

January 11, 2010 Mu		age	⁴ Offshore Helicopter Safety Inquiry		
Pa	age 57		Page 59		
1 world?	1		used and shared, we'll get onto in a lot more		
2 MR. PRITCHARD:	2		detail as we go through. Okay, Mr. Sacuta,		
3 A. Yes, indeed. So in support of the production	1 3		we'll continue, I think, for the next little		
4 operations and drilling activities and any	4		while. We'll have a break about quarter to		
5 project-related vessels, because we do also	5		11.		
6 bring in, from time to time, project-related	6	MR. SA	ACUTA:		
7 vessels that do have helidecks as well, to get	. 7	А.	Okay. Operators in the Newfoundland and		
8 those project personnel to and fro, there's a	8		Labrador area work in one of the most highly		
9 suite of logistical support craft and offload	9		regulated regions in the world. In all the		
10 tankers. This slide shows the resources that	10		areas of the world that I've worked, I would		
11 we have and later we'll be discussing a little	11		say it's the most regulated. I can't say it's		
12 bit about the sharing activities that we have.	12		the most regulated everywhere, but certainly		
13 It should be noted that it appears as if	13		in any of the areas that I've ever worked,		
14 Husky Energy operates quite a number of the	he 14		it's the most regulated. So on the next		
15 resources and assets and that's simply becau	se 15		section of slides, I'll be talking a little		
16 we are currently the operators of the GSF	16		bit about some of the regulatory requirements.		
17 Grand Banks and the Henry Goodrich as the	two 17		The operator of each offshore project is,		
18 drilling operations that would be underway	/ 18		as previously discussed, either one of the co-		
19 currently.	19		venturers, such as Suncor or Husky, or a		
20 ROIL, Q.C.:	20		separately incorporated company, such as HMDC.		
21 Q. So that is in support of three separate	21		They are appointed by the co-venturers to		
22 facilities?	22		conduct the project operations within the		
23 MR. PRITCHARD:	23		scope of the authority conferred by the joint		
24 A. That's in support of three separate	24		operating agreement. They're provided with		
25 facilities, that's correct. Each, of course,	25		oversight and direction by a management		
Pa	ige 58		Page 60		
1 have a dedicated standby vessel. We've	1		committee, usually made up of the		
2 indicated that each facility should have a	2	. 1	representatives of the individual co-		
3 dedicated standby vessel, so you see the	3		venturers. They're responsible for the health		
4 number of standby vessels there. The shuttle	e 4	. ;	and safety of personnel and also for		
5 tankers that we see is that Husky Energy	5		legislative and regulatory compliance on		
6 utilizes the two shuttle tankers that we have	6	1	behalf of the co-venturers.		
7 to go direct to market, and not necessarily	7	ROIL, Q	.C.:		
8 using the NTL terminal. So our shuttle	8	Q.	When you refer to management committee, would		
9 tankers would be more directed to market.	9	1	that be akin to a board of directors, if it		
10 ROIL, Q.C.:	10		was a single company operation? Is that the		
11 Q. So the two Knutson named vessels go to man	rket. 11		nature -		
12 The other three go to the -	12	MR. SA	CUTA:		
13 MR. PRITCHARD:	13	Α.	Akin, in Hibernia's case, responsible to		
14 A. Generally speaking, that is the arrangement.	14		Hibernia Executive Committee, which is made up		
15 MR. SACUTA:	15		of representatives from each of the co-		
16 A. The other three do have the opportunity,	16		venturers, for example.		
17 depending on the time and inventories, we d	lo 17	ROIL, Q	.C.:		
18 occasionally send one of our tankers direct to	o 18	Q	Again, I think we can hear more about the		
19 market as well, but the majority of them do g	go 19		details of that in the individual		
20 to the Newfoundland Transshipment termina	ıl. 20		presentations.		
21 MR. PRITCHARD:	21	MR. SA	CUTA:		
22 A. Okay. That's the end of this particular	22	А.	In the individual panels, that's correct.		
23 section.	23	ROIL, Q	.C.:		
24 ROIL, Q.C.:	24	Q.	Okay, good.		
25 Q. And the three helicopters and how they are	; 25	MR. SA	CUTA:		

Ja	nuary 11, 2010	Multi-	ılti-Page		Offshore Helicopter Safety Inquiry
		Page 61			Page 63
1	A. This slide outlines that the C-NLOPB ha	s	1	A.	That's right. We have regulatory requirements
2	iurisdiction over petroleum operations in t	he	2		that we'll expect Cougar to meet and Cougar
	Newfoundland and Labrador offshore area	The	3		has regulatory requirements to Transport
	guidelines respecting drilling program	s The	4		Canada in them providing services to the
	contains specific references related to	5	т 5		operators
6	heliconter operations However helicon	ter	6	POIL	
	operations are under the jurisdiction of		7	NOIL,	Right okay thank you
	Transport Canada, which has been previo	Juely	/ Q	VD C	
	testified at this Inquiry. The interface	Jusiy	0	MIK. 57	Under the Canada Nawfoundland Atlantic Accord
	between the two regulatory regimes and		2 10	А.	Implementation A at no one can correct out work
	between the operators and Cougar Helicot	tors 1	10		or activity related to the exploration or
	poly o C.		11		drilling for the production and concernation
$ _{12}^{12}$	KOIL, Q.C.:	· ^ 11	12		drilling for the production and conservation,
13	Q. Although the C-NLOPB actually has, as w		13		processing or transportation of any petroleum
$ _{17}^{14}$	see I think later, statements about what	1	14		in the offshore area unless they hold an
15	things should be on a hencopter and nov	va 1	15		operating license which has been issued by the
16	nelicopter snould be equipped?		16		C-NLOPB and they hold a work authorization
17	MR. SACUTA:		17		issued by the C-NLOPB. The Board shall,
18	A. That's correct.		18		before issuing an authorization for work or
19	ROIL, Q.C.:		19		activity, consider the safety of that work or
20	Q. Okay. So how does that work in terms	of 2	20		activity. It's important to note that the
21	Transport Canada?	2	21		Board may suspend or revoke any operating
22	MR. SACUTA:	2	22		license or work authorization for non-
23	A. The requirements are regulatory in nature.	So 2	23		compliance of the regulations.
24	it's up to the operators to meet those	2	24	ROIL,	Q.C.:
25	requirements.	2	25	Q.	Okay, just stop you there. You talked
		Page 62			Page 64
1	ROIL, Q.C.:		1		earlier, all three of you, about drilling and
2	Q. Okay.		2		production. So do I take it that before you
3	MR. SACUTA:		3		drill, you have to get a license. Before you
4	A. But Cougar Helicopters is basically regula	ited	4		produce, you have to get a license. You might
5	by Transport Canada and receive all the	ir	5		get two at the same time, but there are
6	approvals through Transport Canada, 1	not	6		licenses all over the place.
7	through the C-NLOPB.		7	MR. S	ACUTA:
8	ROIL, Q.C.:		8	Α.	Absolutely. There's a slide further on that
9	Q. Okay. So they get I see, so go back to)	9		talks about the various operations,
10	your slide then.	1	10		authorizations and various authorizations that
11	MR. SACUTA:	1	11		you're expected to obtain prior to doing those
12	A. Yeah.	1	12		activities.
13	ROIL, Q.C.:	1	13	ROIL,	Q.C.:
14	Q. They get their direction, in terms of	1	14	Q.	Good, okay.
15	operations, from Transport Canada?	1	15	MR. S	ACUTA:
16	MR. SACUTA:	1	16	А.	It's also important to mention that a non-
17	A. Correct.	1	17		compliance is considered an offense.
18	ROIL, Q.C.:	1	18	ROIL,	Q.C.:
19	Q. You get yours from the C-NLOPB?	1	19	Q.	What happens in terms of breach of an offence
20	MR. SACUTA:	2	20		or -
21	A. Correct.	2	21	MR. S	ACUTA:
22	ROIL, Q.C.:	2	22	А.	There is -
23	Q. And then how you interact with them is he	ow it 2	23	ROIL,	Q.C.:
24	gets -	2	24	Q.	- if there is even an offence?
25	MR. SACUTA:	2	25	MR. S	ACUTA:

January 11, 2010	Multi-Pa	ge TM	Offshore Helicopter Safety Inquiry	
	Page 65		Page 67	
1 A. There is a possibility you could be fined, f	or 1 M	MR. PRITCHA	RD:	
2 example.	2	A. So they	apply them in that manner.	
3 ROIL, Q.C.:	3 F	ROIL, Q.C.:		
4 Q. Are there possible personal consequences	on 4	Q. So if y	ou don't comply with them -	
5 senior management representatives?	5 N	MR. PRITCHA	RD:	
6 MR. SACUTA:	6	A. There's	an authorization that comes along with	
7 A. Absolutely.	7	the gui	delines.	
8 ROIL, Q.C.:	8 F	ROIL, Q.C.:		
9 Q. Okay.	9	Q. Yes, ol	ay. Mr. Vokey?	
10 MR. SACUTA:	10 M	MR. VOKEY:		
11 A. There are numerous active regulations tha	t the 11	A. Same f	or Suncor.	
12 Board has jurisdiction over. They includ	.e: 12 F	ROIL, Q.C.:		
13 the certificate of fitness regulations;	13	Q. Same f	or Suncor.	
14 petroleum installation regulations; drilling	g 14 M	MR. VOKEY:		
15 regulations; the production and conservat	ion 15	A. Wheth	er it's a draft, whether it's a	
16 regulations: in order to complete any divi	ng 16	regulat	ion or a guideline, all is applicable.	
17 activities in the area. you have to have	17	To us.	they're the rules.	
18 follow the diving regulations: the	18 F	ROIL O.C.:		
19 Occupational Health and Safety regulation	ons 19	O Okav a	and they're made the rules by virtue of	
20 which are in draft currently: and then the	20	being a	condition of your licenses?	
21 geophysical operations Access to all o	f 21 N	MR PRITCHA	RD.	
22 these regulations can be found on the Boar	rd's 21			
22 incsc regulations can be found on the Doal				
	23 F	O Or auth	orizations	
24 KOIL, Q.C.:	n 24	Q. OI auu	ionzations.	
25 Q. Okay. Just as an item that has come u		MR. SACUTA.		
	Page 66		Page 68	
before, and I'm sure it will come up agai	n, 1	A. Yeah.	As a matter of fact, if you didn't	
2 does the fact that the occupational health a	ind 2	comply	with the guidelines, you wouldn't get	
3 safety regulations, does the fact that they	3	your at	ithorization.	
4 are only in draft have any real impact on y	ou 4 F	ROIL, Q.C.:		
5 or your company I'd ask all three of you	1 to 5	Q. And yo	bu need your authorization before you can	
6 answer in terms of your operations?	6	do any	work?	
7 MR. SACUTA:	7 N	MR. SACUTA:		
8 A. Absolutely not. We're still expected to m	eet 8	A. Before	you can do any work.	
9 the regulations. The fact that they're in	9 F	ROIL, Q.C.:		
10 draft does not have any impact on how we	treat 10	Q. Right,	okay.	
11 them or the fact that we are measured again	nst 11 N	MR. SACUTA:		
12 those that are identified in the draft	12	A. There a	re also a number of other guidelines	
13 regulations.	13	which	include the safety plan guidelines,	
14 ROIL, Q.C.:	14	drilling	program guidelines, reporting and	
15 Q. Mr. Pritchard?	15	investi	gation of safety related incidents,	
16 MR. PRITCHARD:	16	physica	al environmental programs, expectations	
17 A. It's on the certification that we get from th	ie 17	guideli	nes and the geophysical geological	
18 Board.	18	enviror	mental and geotechnical guidelines.	
19 ROIL, Q.C.:	19	Mr. C	commissioner, I'd like to take this	
20 Q. I'm sorry?	20	opporti	inity to inform you that effective	
21 MR. PRITCHARD:	21	Decem	ber 31st, 2009, the Board has introduced	
22 A. They're enforceable because they're on	the 22	new g	bal-based, new goal-oriented offshore	
license that we get from the Board.	23	petrole	um drilling and production regulations.	
24 ROIL, Q.C.:	24	The B	pard has also posted the following	
25 Q. Yes.	25	docum	ents to eight operators in understanding	

January 11, 2010 Mult			Page	етм	¹ Offshore Helicopter Safety Inquiry
	Page 6	9			Page 71
1 the ne	w regulatory requirements, which	1	M	R. S	SACUTA:
2 include	s a draft drilling and production	2	2	A.	That's correct. It's Civil Aviation Authority
3 guidelir	ne, a draft safety plan guideline, a	3	3		and it's the Civil Aviation publication.
4 draft en	vironmental protection plan guideline,	4	1 RC	DIL,	, Q.C.:
5 and a d	raft data acquisition and reporting	5	5	Q.	In the UK?
6 guidelir	ne. The draft guidelines will be	6	5 MI	R. S	SACUTA:
7 issued f	or a one-year trial basis and will be	7	7	A.	In the UK.
8 revised	as necessary during this period, based	8	3 RC	DIL,	, Q.C.:
9 on the	feedback and experience gained from	9)	Q.	Yes, okay.
10 their us	se. Any of us that already have	10) MI	R. S	SACUTA:
11 current	authorizations do not have to meet the	11	l	A.	And then lastly, the Atlantic Canada Offshore
12 new gui	delines, but during the next renewal	12	2		Petroleum Industry, the standard practice for
13 process	, we will have to meet the new goal-	13	3		training and qualifications of personnel, and
14 oriented	l guidelines.	14	1		this was produced by CAPP, the CAPP, Canadian
15 ROIL, Q.C.:		15	5		Association of Petroleum Producers.
16 Q. And I th	hink in their evidence, they indicated	16	5 RC	DIL,	, Q.C.:
17 that the	re was some move in that direction.	17	7	Q.	Okay. We're going to take a break at this
18 So I gue	ess what you're telling us is it has	18	3		point, but before we do, you've used the
19 now ha	opened?	19)		expression guidance and guidelines and
20 MR. SACUTA:		20)		regulations.
21 A. It has r	low happened. We received a letter	21	M	R. S	SACUTA:
22 dated D	ecember 23rd identifying that it would	22	2	A.	Yes.
be in pl	ace December 31st of last year.	23	3 RC	DIL.	, O.C.:
24 ROIL, O.C.:	5	24	1	0.	To an oil operator, to a senior executive of
25 Q. Okay.	Well, we'll certainly get copies of	25	5		an oil company, are these guidances simply
	Page 7				Page 72
1 that from	m the C-NI OPR representatives Thank	<i>1</i>	I		hints and suggestions that what you should do
	n the e representatives. Thank		,		or do they have really a higher level of
3 MR SACUTA		3	2		direction to you?
1 1 1 1 1 1 1 1 1 1	also other guidance that's related to		, 1 мт	2 6	
5 helicon	ter operations which operators are	5	, 1411	Δ	They have a higher level of direction because
6 exposed	to when operating in the Newfoundland	6	5	11.	quite often when we're issued an operations
7 area T	hese include: the heliconter passenger		,		authorization or any authorization, there are
8 transpor	rtation suit system guidelines.		2		expectations the Board places on us, that we
o Transpor	wrt Canada's guidelines respecting		,)		meet the guidelines and when the authorization
10 helicon	ter facilities on ships: TP which	10	,)		is issued it's very specific that you shall
10 neneop	or Transport Publication 4/14 which	11	, I		meet these guidelines Although they're
11 stands I	hore helidecks are designed to and	12	,		called a guideline the Board enforces them
12 the ons	a little further about the	12	2		through their authorization process
13 In taik	ry requirements for inspection of	13	, 1 DC	ш	
14 logulat	vs: the UK Civil Aviation Authority's	14	+ KC	лL, О	And then they become a requirement?
15 nendeer	helicopter landing areas which	15	, 5 мі	γ. Σ σ	
10 Offstion	s guidance on standards, which is	17	7 IVII	A. 5	They become a requirement as a term for you
17 provide	CAP stands for Civil Aviation	18	2	A.	receiving whatever authorization you've
$\begin{bmatrix} 10 & CAP45/. \\ 10 & Dublice \end{bmatrix}$	tion There is quite an interface	10	,)		requested
20 between	the North Sea and ourselves and we do	19	΄) ΡΩ	ш	
21 Outre of	ten reference and look at that CAD427	20) KU	ль, О	Okay That might be as good a place as any
21 quite Of 22 quidelie	he if there's changes made)	ų.	for us to take our morning break Mr
23 ROIL OC	e ii mere s changes made.	22	2		Commissioner
23 KOIL, Q.C.:	nd that CAR437 has nothing to do with	23	, 1 CC	<u>۱</u> ۸/۱۸	COMMISSIONEL.
25 the org	inization called CAPP?	24	+ UU 5	0	Yes okay then Take 15 minutes
1-2 une orge		140	,	<u>v</u> .	res, onay mon. rand is minutes.

January 11, 2010 Mult	i-Page TM Offshore Helicopter Safety Inquiry
Page 73	Page 75
1 (BREAK)	1 O. Yes. So if you said you were doing -
2 ROIL O.C.:	2 MR SACUTA:
3 0 Thank you Commissioner Mr Sacuta Lthink	3 A Or a diving program
4 vou're still responsible for the next few	4 ROIL OC
5 pages so keep going please	5 0 If you said you were -
6 MR SACUTA	6 MR SACUTA
7 A Okay Work authorizations issued by the C-	7 A Diving is another good example
8 NI OPB or the Board specify: the type of	8 ROIL O.C.
a operation which is permitted: the vessels or	9 O Veah diving If you were doing diving for
installations authorized to conduct that work:	three months, you would only ask for the three
the time period for the work to be completed:	10 months would you?
and the conditions pertaining to the	12 MP SACUTA:
12 and the conditions pertaining to the	12 MR. SACOTA.
14 KOIL, Q.C	14 KOL, Q.C.,
different bring a different facility in take	15 Q. OKAY.
10 Unifield, offing a differently in, take	10 MR. SACUTA.
12 MD CACUTA.	17 A. So this shoe shows various work activities
10 MR. SACUTA:	18 which require authorizations. T in not going
A. Each time you do that, you would have to get	the left hand side, there's a number of
to bring in a new drilling rig to do a	20 the feft-find side, there is a fulfible of
drilling activity, you, would have to get an	21 activities that would require all
22 diffining activity, you would have to get all	22 authorization. On the right-hand side of the
25 authorization for that activity.	25 graph of of the side are any of the
24 KOIL, Q.C.:	24 requirements associated with actually 25 obtaining the authorization, the types of
25 Q. Okay. And even though you re doing the same	25 obtaining the authorization, the types of
Page 74	Page 76
1 drilling but with a different ship or	1 things you would have to submit to get the
2 different drill ship -	2 authorization. The safety plan box has been
3 MR. SACUTA:	3 highlighted because I'm going to discuss it a
4 A. That's correct.	4 little bit further in the next set of slides,
5 ROIL, Q.C.:	5 because the safety plan is one of the specific
6 Q you have to go back again?	6 things that is related to helicopter
7 MR. SACUTA:	7 transportation.
8 A. Correct.	8 ROIL, Q.C.:
9 ROIL, Q.C.:	9 Q. So if you're doing a diving program, you need
10 Q. Okay.	10 a safety plan?
11 MR. SACUTA:	11 MR. SACUTA:
12 A. The production operations authorization or the	12 A. Correct.
13 operations authorization and other required	13 ROIL, Q.C.:
authorizations are issued by the Board to each	14 Q. If you're doing a construction and
15 operator prior to commencement of the said	15 installation program, you need a safety plan?
16 operations and they are normally renewed every	16 MR. SACUTA:
17 Unree years.	17 A. That's correct.
18 KOIL, Q.C.:	18 ROIL, Q.C.:
19 Q. Okay. Would some autionizations be for a	19 Q. SO each time you do one of these things -
20 SHORE PERIOD?	20 Mik. SACUTA:
22 A You may have a short term drilling program	21 A. Each time you do any of the activities of the
that only requires an authorization for a	22 reft side of tills slide, you require a safety
24 shorter period than three years for example	
25 DOL O.C.	25 0 Okay thank you

January 11, 2010 Multi-		ti-P	age	^M Offshore Helicopter Safety Inquiry
	Page 7	7		Page 79
1	MR. SACUTA:	1	ROII	L. O.C.:
2	A. So the Board's Chief Safety Officer shall	2	Q	Okay. So when we look at the safety plans for
3	approve the safety plan and authorization is	3		each of you in your individual presentations
4	only granted to an operator if the safety plan	4		later this week and month, or this month
5	is deemed acceptable by the Board. The	5		really, we should be looking to see that it
6	operator must demonstrate to the Board that	6		complies with this?
7	the approved management system and safety plan	7	MR	SACUTA:
8	effectively identifies assesses and controls	8	Δ	Correct
9	risk posed to a worker or to worker health and	9	ROII	
	safety including the safe transport to and	10		Okay
11	from the installation. The safety plan which	11	MR	SACUTA
12	is re-submitted and revalidated every three	12	Δ	So the safety plan highlights the management
12	vers evolves as an integral component of a	12	А	systems and processes for safe operations in
13	continuously improving safety management	13		the offehore, area which includes beliconter
14	system So for the three operators on a	14		operations. The components of a safety plan
15	three yearly basis and sometimes more	15		include written policies programs and
10	frequently then that if we make changes to our	10		procedures in the following items: sefety
$ _{10}^{17}$	asfoty plan, but at a minimum overy, three	1/		management, a basis of sofe operations and the
10	safety plan, but at a minimum every timee	10		design your organizational structure the
19	the Roard for their approval	19		authorities and the command structure, the
20		20		authornes and the command structure, hazard
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	KULL, Q.C.:			and fisk identification and assessment,
$\begin{vmatrix} 22\\ 22 \end{vmatrix}$	Q. MIT. Sacuta, and to the others, my question. I think all three, of you, I have all three of	22		installation how you operate and mointain
23	unink an unee of you, I hope an unee of	23		these facilities the training and
24	you were here when the C-NLOPB gave his	24		unose facilities, the training and
25	evidence, when Mr. Andrews particularly said	25		quanneations required, the command structure
	Page 78	3		Page 80
1	"the Board is not primarily responsible for	1		and any contingency planning, and it also has
2	the safety of offshore workers. The operators	2		a requirement on physical environmental
3	are." Do you accept that as the	3		monitoring, weather monitoring, for example.
4	responsibility?	4		There are multiple references specific to
5	MR. SACUTA:	5		helicopter operations under the work
6	A. Absolutely. We are responsible for the safety	6		authorizations issued by the Board. They
7	of our workforce.	7		include the following: the safety plan
8	ROIL, Q.C.:	8		guidelines references contingency planning,
9	Q. All three of you agree with that?	9		the need for a standby helicopter, the need
10	MR. VOKEY:	10		for flight following and vessel watch, and
11	A. That's correct.	11		mutual aid agreements between operators.
12	MR. PRITCHARD:	12	ROII	L, Q.C.:
13	A. Absolutely.	13	Q	. What are mutual aid agreements?
14	MR. SACUTA:	14	MR.	SACUTA:
15	A. As a direct quote out of the safety plan	15	A	. Mutual aid agreements would be an agreement
16	guidelines, the safety plan guidelines state	16		between the three operators that we would
17	"the safety plan must clearly show how safety	17		share resources as required during an
18	management fits within the overall system of	18		emergency situation, for example.
19	management. It should define the roles of and	19	ROII	L, Q.C.:
20	relationships between the operator's executive	20	Q	And do you have such agreements?
21	level management personnel and the various	21	MR.	SACUTA:
22	line and statt functions in achieving safety	22	A	. We do.
23	related goals and objectives. This is a	23	ROII	L, Q.C.:
24	direct quote out of the safety plan	24	Q	. Okay, and have they ever been called into
25	guidelines.	25		play?

January 11, 2010 Multi			age ^T	^M Offshore Helicopter Safety Inquiry
	Page 81			Page 83
1	MR. SACUTA:	1	ROII	L, Q.C.:
2	A. We have called them into play. Whenever we've	2	Q	. No.
3	had situations that require one of us to	3	MR.	SACUTA:
4	support the other, one of the operators can	4	А	. That would never be allowed by the Board.
5	contact the other and request assistance,	5	ROII	L, Q.C.:
6	whether it's vessels or any other resources	6	0	Okay, and that's what I was looking for. Are
7	that they have available.	7		there any cracks there that need to be filled?
8	Next. the petroleum installation	8	MR.	SACUTA:
9	regulations has specific references to	9	А	Absolutely not.
10	immersion suits and the helideck design	10	ROII	L. O.C.:
11	requirements. Petroleum geophysical	11	Q	Okay, thank you.
12	regulations also has references to the	12	MR.	SACUTA:
13	helideck design and transportation suits. The	13	А	. The Board conducts an annual audit and three
14	draft OSH regulations, helicopter	14		quarterly inspections per year. They can also
15	transportation suits specifically referenced,	15		complete ad-hoc inspections, if required. An
16	and in the drilling regulations, the drilling	16		example could be after an incident offshore.
17	program guidelines, helicopter passenger	17		The Board may determine an ad-hoc visit may be
18	transportation suits. So you can see that	18		required or appropriate to evaluate the
19	there are a number of references in the	19		operator's response and follow up. During
20	various regulations and guidelines associated	20		these visits, the Board has full access to all
21	with helicopter transportation and the	21		personnel on board the facility.
22	requirements for transportation suits.	22	ROII	L, Q.C.:
23	ROIL, Q.C.:	23	Q	Now these audits you're talking about are not
24	Q. Okay. The fact that helicopter transportation	24		focused entirely on helicopter operations, are
25	suits, for example, or helideck design, as	25		they?
	Page 82	2		Page 84
1	another example, fits within more than one of	1	MR.	SACUTA:
2	the regulations, does that create any	2	А	. No. They're an audit to measure you against
3	duplicity or any operational challenges for	3		your commitments in your safety plans, to
4	you?	4		measure you against the regulations, that
5	MR. SACUTA:	5		you're meeting all the regulations that the
6	A. Not from my perspective.	6		Board governs. So it's an audit of your
7	MR. PRITCHARD:	7		compliance with the regulations and the
8	A. No, we just assess each individual	8		commitments in your safety plan. But the
9	requirement.	9		Board also has the opportunity, as I
10	ROIL, Q.C.:	10		mentioned, to do these independent or ad-hoc
11	Q. So where, for example, petroleum installation	11		if there's a situation that would warrant them
12	regulations don't make specific reference to a	12		coming offshore. So there's just not four
13	helicopter passenger transportation suit,	13		visits a year when the Board the Board can
14	would there still be if you were simply	14		come out as many times as they deem fit to
15	looking for a permit under the well, could	15		come to our installations.
16	you look for a permit under the petroleum	16	ROII	L, Q.C.:
17	installation regulations that would not	17	Q	And in your experience, do they come out on an
18	involve the use of helicopters?	18		ad-hoc basis?
19	MR. SACUTA:	19	MR.	SACUTA:
20	A. The Board would hold you to the standards in	20	А	. Absolutely. The scope of the Board's audit,
21	the other regulations. You would not be able	21		as I mentioned, and inspections is to monitor
22	to say "well, we're making an application	22		compliance with the regulations, any
23	under the installation regulations, so we	23		commitments or requirements at the time you
24	don't need to have helicopter transportation	24		receive your authorization or your approval,
25	suits."	25		any conditions of that approval. They may put

January 11, 2010 Multi-		lulti-P	age	• TM	^M Offshore Helicopter Safety Inquiry		
	Page	e 85			Page 87		
$\begin{vmatrix} 1 \\ 2 \end{vmatrix}$	conditions for compliance. They'll certainly monitor you against your own safety plan and	1	2	Q.	Okay. So the Hibernia gravity base structure has been certified by Lloyd's Register?		
3	your environmental plan commitments, as wel	1 3	3 MF	R. S .	ACUTA:		
4	as monitoring that you have an incident	4	ļ	A.	They are our certifying authority and they		
5	management process.	5	5		have granted us our certificate of fitness.		
6	Transport Canada surveys are normally	6	5		That's correct.		
7	delegated to the certifying authority.	7	RC)IL,	Q.C.:		
8	ROIL, Q.C.:	8	3	Q.	And they are a worldwide certifying authority?		
9	Q. Now stop, sorry.	9) MF	λ. S.	ACUTA:		
10	MR. SACUTA:	10)	A.	They are worldwide, that certifies very many		
11	A. Yes.	11	l		offshore petroleum installations.		
12	ROIL, Q.C.:	12	2 RC)IL,	Q.C.:		
13	Q. Transport Canada surveys of what?	13	3	Q.	Okay, and by way of another example then, the		
14	MR. SACUTA:	14	ŀ		White Rose facility is a DNV facility?		
15	A. Transport Canada surveys of the helideck.	15	5 MF	۲. S.	ACUTA:		
16	ROIL, Q.C.:	16	5	A.	That's correct, another certifying authority,		
17	Q. Ah, okay.	17	7		just -		
18	MR. SACUTA:	18	B RC)IL,	Q.C.:		
19	A. Are designated under the delegated statutory	19)	Q.	Yes.		
20	inspection program. They will periodically	20) MF	۲. S.	ACUTA:		
21	conduct monitoring surveys of the certifying	21	l	A.	Yeah, that's correct.		
22	authority. So they do do a cross check to	22	2 RC)IL,	Q.C.:		
23	make sure that the certifying authority is	23	3	Q.	Okay. If there was a drill ship that was		
24	doing what they're supposed to do as part of	24	ŀ		are there any other certifying authorities in		
25	that delegation.	25	5		the world that might certify a GBS, an FPSO or		
	Page	e 86			Page 88		
1	ROIL, Q.C.:	1	l		a MODU or some sort of a drill ship? Is there		
2	Q. Okay, and I think your next slide talks about	2	2		any other possible -		
3	who the certifying authorities are.	3	B MF	₹. S.	ACUTA:		
4	MR. SACUTA:	4	ł	A.	There is ABS.		
5	A. That's correct. So the Atlantic Accord Act	5	5 MF	ι . V	OKEY:		
6	requires that each production, drilling,	6	5	A.	The American Bureau of Shipping.		
7	diving and accommodations installation have a	. 7	7 MF	₹. S.	ACUTA:		
8	valid certificate of fitness, which has been	8	3	A.	American Bureau of Shipping.		
9	issued by a recognized certifying authority	9	RC	ЛL,	Q.C.:		
10	before the Board can authorize an activity in	10)	Q.	American Bureau of Shipping. So if a ship		
11	the offshore area. Each of the three offshore	11			came in that was certified by the American		
12	installations has a certifying authority	12	2		Bureau of Shipping, that would be another		
13	involved in their respective business. Both	13	3		delegated authority, would it?		
14	Hibernia and Terra Nova use Lloyd's Register	14	+ Mŀ -	ε. s.	ACUTA:		
15	out of the UK and white Rose uses DNV. Each	15)	A.	correct. So the role of the certifying		
10	audita and facility. Audita inspects and	10)		autionity is to provide certification of a		
10	scheduled quarterly and if required special	1/	,)		regulations The cortifying authority		
10	visits can occur. If there's a situation	10))		provides an independent third party evaluation		
19	where you've had some issue that's occurred o	n 20	,		of regulatory compliance and fitness for		
$\begin{vmatrix} 20 \\ 21 \end{vmatrix}$	your facility that requires the certifying	II 20	,		nurpose. The intent of the certification is		
$\begin{vmatrix} 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 $	authority to make a special visit to ensure)		to provide assurance that the installation		
$\begin{vmatrix} 22\\ 23 \end{vmatrix}$	that your certificate of fitness still remains	22	-		during the term of the certificate is fit for		
$\begin{vmatrix} 23 \\ 24 \end{vmatrix}$	valid they will do that	23	, L		purpose and remains in compliance with the		
25	ROIL, O.C.:	25	5		regulations. Before issuing a certificate of		

January 11, 2010	Mult	i-Pa	age™	⁴ Offshore Helicopter Safety Inquiry
	Page 89			Page 91
1 fitness, the 2 satisfied tha	certifying authority must be t the installation is designed	1		shipboard oil pollution emergency plan, and accident investigations, which are still
3 and constru	cted in accordance with the	3		maintained by Transport Canada.
4 regulations.	is fit for the intended purpose	4	ROIL	. O.C.:
5 and can be s	afely operated without polluting	5	Q.	Okay. So Transport Canada delegates out some
6 the environ	ment and will remain fit when	6		but continues to do personally or accept
7 maintained	n accordance with the approved	7		within itself responsibility for others?
8 inspection a	and monitoring maintenance and	8	MR. S	SACUTA:
9 weight cont	ol programs.	9	А.	Correct.
10 ROIL, Q.C.:		10	ROIL	, Q.C.:
11 Q. Now, as I ta	ke it, Mr. Sacuta, the certifying	11	Q.	Okay, we're now moving -
12 authority ad	tually certifies the entire	12	MR. S	SACUTA:
13 installation.		13	А.	At this point, we're going to move on.
14 MR. SACUTA:		14	ROIL	, Q.C.:
15 A. That's corre	ct.	15	Q.	- to safety management, and I think Mr.
16 ROIL, Q.C.:		16		Pritchard is going to take over.
17 Q. But for our	purposes, it is in terms of its	17	MR. F	PRITCHARD:
18 totality of	the installation, it is	18	A.	I am, indeed. Safety offshore is the
19 responsible	for the helideck portion as well?	19		responsibility of the operators. You may hear
20 MR. SACUTA:		20		me use the phrase line management
21 A. That's corre	ct.	21		responsibility. Safety is line management
22 ROIL, Q.C.:		22		responsibility. This takes the intent of the
23 Q. Okay.		23		CEO and the top executives through to everyone
24 MR. SACUTA:		24		working at tasks at the work site, and I will
25 A. Based on th	e delegation from Transport Canada.	25		go through a structure in future slides to
	Page 90			Page 92
1 ROIL, Q.C.:		1		show how that works. It just shows that
2 Q. Other aspec	ts of its work are important to	2		everyone has a role to play in safety.
3 other people	e, but not perhaps our focus here.	3		I'll be discussing three key areas of
4 MR. SACUTA:		4		people, processes and equipment, by which we
5 A. Absolutely.	So I'd like to talk a little bit	5		can create barriers to prevent incidents and
6 about the c	elegated statutory inspection	6		continuously look for improvements in all
7 program I m	Semial equipments equipment canada			three areas. There are various means of
8 has signed	of statutory increation and	8		improvements. That might he maintenance
9 delegation	functions under the Canada	10		reviews process document reviews safety
10 Certification	reference to specific recognized	11		alerts that would improve people's
11 Shipping F	as or ROs For the EPSOs both	12		understanding and potentially add that into
12 Organization	DNV are identified as ROs and have	12		the training. So we look for that continuous
14 been delega	ted by Transport Canada to perform	14		improvement via that level of feedback or
15 surveys and	issue certificates Llovd's	15		perhaps the IOHS recommendations
16 Register of	Shipping was delegated on December	16		Industry has been developing its
17 4th. 2000 ar	d DNV on April 22nd of 2002. The	17		management hazards and risk management for
18 certifying au	ithority is involved in ensuring	18		many years and uses a state of the art risk
19 that the hel	idecks on each installation	19		management system. They're not all identical,
20 conform to	Transport Canada Transport	20		however, the principles remain the same, and
21 Publication	4414. It is important to note	21		whilst we only have three producing assets
22 that this del	egation does not cover marine	22		working on the Grand Banks, we do have major
23 occupationa	l safe and health, the safety	23		world players involved at that management
24 manning le	vels, radio installation, the	24		committee level that we discussed a little bit
25 requirement	s for radio installation, the	25		earlier. So we have many operating companies

January 11, 2010		Multi-Page TM		¹ Offshore Helicopter Safety Inquiry	
		Page 93			Page 95
1	with many years experience being able to g	rive	1	So I wa	nt to discuss helideck operations and
2	feedback and input to how our operations y	vork,	2	perhaps	fuelling of helicopters. So at a
3	bringing best practices from around the wo	rld	3	project	concept, you need to select that
4	to our operations.		4	you're g	going to, you know, have a helideck,
5	5 Our safety focus is about people and		5	what ty	pe of operation it's going to be.
6	commitment. It's about commitment to y	our	6	Whethe	r it be a GBS or an FPSO, we need a
7	spouse, your partner, children or		7	helideck	to transport people to and from.
8	grandchildren and your team members. It's	all	8	There's	many factors that would influence that
9	about behaviours and reactions to those	;	9	decision	, whether it's at the front end of an
10	responsibilities that we have. People mus	t	10	FPSO or	at the stern of the FPSO, but no
11	work as a team and remain vigilant as w	ve 🛛	11	matter,	we need a helideck. Once decided, we
12	progress through our day-to-day activities a	and	12	need to	consider the operational aspects of
13	just understand how things are moving a	nd	13	the loca	ation, and then we work with the
14	changing, and I'll move on with that a little		14	enginee	ring designs, the standards, and the
15	bit later when I talk to the Swiss cheese		15	legislati	on in order to engineer the correct
16	slice model that we used previously in		16	helideck	facilities. So during that time, we
17	testimony here. So people need to know v	vhat	17	need to	use our engineering capabilities and
18	they are accountable for. It's really about		18	we usua	ally work with eliminating hazards,
19	what we are doing today, here at the Inquir	у,	19	controll	ing hazards, mitigating hazards and
20	and what we plan for tomorrow that's goin	g to	20	have a r	neasure of recovery should be if the
21	make a difference to safety in the future.		21	mitigati	ons fail.
22	Now this, I've introduced the equipment	,	22	So in	the context of this, we need to
23	processes and people. These are the area	s	23	conside	the helideck and we need to refuel.
24	that we can create barriers with. So during	3	24	So if w	e have a bulk fuel storage system
25	any large project, you know, a large or sma	.11	25	located	close to the helideck, that will
		Page 94			Page 96
1	type of project, they're three categories th	at	1	create a p	potential hazard if a hard landing of
2	we can manage risk with and reduce the ri	sk.	2	a helicop	pter should occur. So we should
3	It's the people that play a significant		3	eliminate	that bulk fuel storage area and move
4	role in all areas of that equipment, process	ses	4	it somew	here else. We would then, of course,
5	and people. So from an equipment poin	t of	5	need to a	ssess if the movement of that storage
6	view, it's the people that do the design.		6	creates a	nother hazard somewhere else, but
7	It's the people that produce the processes	and	7	nonethel	ess, we should move it away from the
8	have input and feedback to that, and from	the	8	helideck	area.
9	people's point of view, in this slide here,		9	In tern	ns of control, we should control
10	it's about leadership and accountability	,	10	that heli	-refuelling system by having the
11	attitudes and behaviours that all contribute	es	11	correct st	andards of pipe work and equipment
12	to the reduction of the risk.		12	and fit-fo	pr-purpose equipment to manage that
13	ROIL, Q.C.:		13	fuel to t	he helideck. And in terms of
14	Q. Okay. So I take it that while you've go	t	14	mitigatio	ns, we should be able to mitigate by
15	people, processes and equipment separa	ted,	15	having th	e appropriate trained people on the
16	they are, in fact, interconnected because		16	maintena	ince issues applied to the systems to
17	people are involved with all aspects of		17	ensure fr	t for purpose.
18	designing equipment, designing processes	?	18	So wh	ere we've got to there, and then, of
19	MR. PRITCHARD:		19	course, w	ve should have a means of recovery.
$ ^{20}$	A. Absolutely, and I'd like to kind of step		20	Should a	Il of those elements fail in some way,
$ ^{21}_{22}$	through a bit of example there to give a	n 1	21	we shoul	a nave a means of recovery, and so
$ ^{22}_{22}$	understanding of now that integration wor	KS.	22	even in e	hy funding around the area. We have
$ ^{23}$	KUIL, U.C.:		25	fireficient	by running around the area. We have
$ _{2^{5}}^{24}$	ψ Q. I call, picase uo.		24	compater	ing systems. We mave maneu,
143			$ \Delta J $	competer	in people working the heliutek allu

Janu	Multi-Page TM		e TM	Offshore Helicopter Safety Inquiry	
		Page 97			Page 99
1	therefore we have a means of recovery.	-	1 R(OIL,	Q.C.:
2	The training and competence of the		2	Q.	So yeah, just slow us down a little bit here,
3	people. So a special ingredient to safety		3		so that we all follow with you. So this is a
4	management is, of course, measuring an	nd	4		descending responsibility, starting at the
5	monitoring. So the measuring and monitor	ring	5		highest of management, in terms of setting
6	of the people is compliance to the training	5	6		policies, and then we cascade down to
7	matrix. We also look at the maintenance	e	7		procedures that must comply with the policies.
8	issues and we look for compliance again	st	8		Is that the way it works?
9	maintenance to ensure our equipment is fit	for	9 M	R. PI	RITCHARD:
10	purpose and we also have our documenta	tion 1	10	A.	That's exactly the way it works. So we've got
11	reviews to make sure our processes are	; 1	11		a high level statement from the CEO, and all
12	controlled correctly.	1	12		companies would have that high level policy
13	So to reduce the risk to an acceptable	1	13		statement, and now we'll work to a procedure
14	level, we have well-designed equipmen	. t , 1	14		that would apply the intent of the policy
15	correct procedures, operated by motivate	d 1	15		statement. So to work safely is an intent
16	people with the right attitude and the	1	16		from the CEO. We have, just one I'm going to
17	training competence. So we can see there t	he !	17		choose here is the permit-to-work procedure,
18	integration of how the equipment design, t	he !	18		and I'll follow that through in an analogy
19	processes that we use, both in terms of	1	19		shortly as well.
20	maintenance and operations, and the attitud	de 2	20 RC	OIL,	Q.C.:
21	of the people all need to come together the	e 2	21	Q.	Okay.
22	from concept selection at the very front en	d 2	22 M	R. PI	RITCHARD:
23	and operational use, as we use it on a day-t	0- 2	23	А.	So the permit-to-work procedure is in support
24	day basis.	2	24		of the policy statement of safe work. The
25	So each operator's management system	may 2	25		permit is a procedure to ensure hazards for
		Page 98			Page 100
1	look slightly different. However, they all		1	t	he entire scope have been identified and
2	have goal setting, planning and performan	ice	2	C	controls are in place to manage all the
3	measures, have a systematic method of ens	uring	3	8	associated risks. The system experts and
4	work is conducted in a safe, environmental	lly	4	١	workers involved in the task and sometimes not
5	responsible manner. The structure of the	•	5	ľ	necessarily workers involved in the task, but
6	safety management system starts with a hi	igh	6	S	some other individual can be associated with
7	level policy that sets out the intent of the		7	t	he review of the hazard identification and
8	company and the chief executive officers a	ınd	8	ľ	nitigations. This review is once that
9	top level executives and it cascades in a mo	ore	9	ľ	eview is completed, it's also reviewed by the
10	comprehensive and explicit terms as it reac	hes	10	(on board supervision. So one part of the
11	the workforce where the work is carried ou	it.	11	S	scope is obviously to enter the tank and
12	So from the intent of the policy, which mak	ce a 1	12	t	here's a generic safe work practice now, so
13	statement that, you know, work should b	be 1	13	r	narrowed down to a practice or a generic tank
14	performed safely, the procedures are develo	oped 1	14	e	entry. So the permit would recognize that we
15	and for this analogy, I'd like to use the	1	15	ł	have to enter the tank and now we work to the
16	entry into a bulk fuel system, so the helided	2 k 1	16	Ę	generic safe confined space entry. There will
17	bulk fuel system that I had previously	1	17	ł	be more specific work instructions that
18	described we should move away from	the 1	18	C	lescribe the specific nature of inspection and
19	helideck area. We still have maintenance a	ind 1	19	ľ	requirements of this specific tank and it
20	inspection to carry out there.	2	20	١	would identify any further requirements
21	So looking to the policy, the high level	2	21	8	associated with that tank entry. So we're now
22	policy stating that we should work safely, w	ve 2	22	8	at the work instruction level.
23	have procedures, and I'll talk to the permit-	- 2	23		There would also be a checklist to ensure
24	to-work procedure as being this particula	r 2	24	t	hat all of the mitigations that apply to this
25	procedure in the cascade effect.		25	t	particular tank entry are in fact completed.

January 11, 2010 Mul			ge TM	Offshore Helicopter Safety Inquiry
	Page 10	1		Page 103
1	So once you identify it, we need to ensure	1	I	understand what the hazards and potential
2	that it's complied with and we need a	2	1	mitigation measures would be. There's no such
3	checklist to do that. Then we work with any	3	1	thing as don't question it in this group here
4	special arrangements that are identified and	4	1	because any question stimulates conversation
5	controlled accordingly and we might find this	5	1	and it's conversation and discussion around
6	from the drawings that are associated with	6		what the work scope is about and the hazards
7	this particular vessel. So we'll get more	7	1	involved and the mitigations that we look
8	information from the drawings and forms and	8	1	towards.
9	the thickness, measurement if that was what	9		So this slide shows us some of the tools
10	the requirement was the scope to opter the	10	1	Now this is not the full suite of tools, but
	tank and make a thickness measurement for	11	1	ives an example of the types of tools, but
12	instance. So we're now into the forms and	12		might use
13	we've discussed there the drawings	13 14 P(
14	So we can see there the hierarchial	14 K		Okay, but would each of these tools, he used
16	structure from policy intent procedures that	16	Q. (every time or only some in certain fact
17	we work with I'll say generic practices	17		situations?
18	specific work instructions the checklist that	18 M	IR PR	PITCHARD.
19	we would use to perform the work, and any	19	A. (Only some in certain situations, and once
20	forms that we need to make, any recordings.	20		again, if I could just potentially step you
21	and of course, the drawings that are	21	1	through an analogy to how we might use some of
22	associated with the specific work, and that's	22	1	these tools and who would use the tools in
23	what we see in the right-hand slide of this	23	(certain situations.
24	slide.	24		So a typical scope of work using some of
25 1	ROIL, Q.C.:	25	1	the tools, let's say we're going to change out
	Page 10	2		Page 104
1	Q. So can we conclude then, if we looked at any	1	1	the jet fuel pumps. So once again, we're at
2	one of your individual safety management	2	1	the helideck and the fuel transfer system and
3	systems and we looked at any particular	3		we want to change out this particular pump. A
4	activity, we should be able to see this kind	4]	permit to work would be raised and that would
5	of cascading going on. Perhaps not all of it	5	i	identify the full scope of the work.
6	would get down to the level of drawings, but -	6 R0	OIL, (Q.C.:
7 1	MR. PRITCHARD:	7	Q. `	Who raises the permit to work? Is that
8	A. Generally drawings would be probably there,	8	5	somebody in management?
9	but not necessarily the forms, for instance.	9 M	IR. PR	RITCHARD:
10	You know, if it's not recordings, if you don't	10	A	It would come up on a maintenance card. Then
11	have it purely down to that level for whatever	11	1	the, I'll call the area authority might raise
12	the work scope would be, but in a general	12	1	that scope or it might be the lead mechanic.
13	sense, you would see all of those principles	13	L	Many people can raise permit to works
14	applied.	14	i	in fact, we would plan, that work beforehand
15	identification and mitigations and hazard	15		so the permit to work would actually start
17	identification is a very important step in	17	1	notentially a few days before the scope would
18	ensuring safety at the workplace. If you do	18	1	actually be executed
19	not identify the hazards and associated risk	19 R	OIL. (
20	then you are unable to mitigate against the	20	0.	Yeah. So the answer would be that the
21	effect of that. Effective hazard	21	с. (appropriate person might be at different
22	identification and mitigation requires the	22]	levels of the organization, depending on what
23	involvement, not only of the subject matter	23	i	it was you were going to do?
24	experts, but also a diverse group that is	24 M	IR. PR	RITCHARD:
25	brought together to discuss the full scope and	25	A. (Correct.

Ja	nuar	y 11, 2010 Mu	lti-P	age	M Offshore Helicopter Safety Inquiry
		Page 10)5		Page 107
1	ROIL	. O.C.:	1		talk and that toolbox talk is associated with
2	Q.	Okay.	2		the group of workers that are about to perform
3	MR. F	PRITCHARD:	3		the task and we review all of the hazards that
4	А.	So the scope of the work would identify the	4		have been identified and what the mitigation
5		pump itself. So we have tag numbers so we can	5		factors are during the toolbox talk. So we
6		physically identify the pump itself and the	6	i	usually have a leader of the work team that
7		location that it's at, and the safe job	7		would perform that duty.
8		analysis would then step through the scope,	8	ROIL	, Q.C.:
9		identifying the hazards. So we've got the top	9	Q.	Again, would that be at the rank and file
10		one there, the safe job analysis, and we would	10)	offshore worker or is it up at management?
11		look there towards the hazards of, say,	11		What's the level of engagement here?
12		electricity, jet fuel, the weight of the pump,	12	MR.	PRITCHARD:
13		perhaps an exposed location or working at	13	A.	Normally the team leaders can be the
14		height. All of those hazards are reviewed and	14		discipline leads or it can indeed just be one
15		identified and then we mitigate against those	15		of the workforce themselves. Usually somebody
16		hazards. I should rather say here that it's	16		who's, you know, very familiar with the
17		the we mitigate against the hazardous	17		systems, comfortable with the scope and the
18		event. Height might be recognized as a	18		supervision on board and comfortable that he
19		hazard, but it's actually fall from height	19		will be the lead for that scope.
20		which is the event that you should need to	20)	So during the year, the course of the
21		associate with.	21		work scope and perhaps an audit of the permit-
22	ROIL	Q.C.:	22		to-work system would be applied. Now not all
23	Q.	Okay. And who does the safe job analysis?	23		tasks will have an audit performed on them, so
24		Again, what persons or levels of persons are	24		we monitor our permit-to-work system by doing
25		involved in that process?	25		audits, but not every every task is not
		Page 10)6		Page 108
1	MR. P	RITCHARD:	1		audited. This will determine compliance to
2	Α.	It pretty much depends on the complexity of	2		the permit-to-work procedures, the practices
3		the work scope itself. So if we have a very	3		and work instructions. So the permit will
4		complex operation, we may actually perform a	4		look at those activities and ensure that they
5		process hazardous analysis or analysis onshore	5		are in compliance. Also through the course of
6		and we may feed that to the offshore group to	6		this work, perhaps a behavioural observation
7		get their input and there would be a combined	7		may be made by anyone. In our individual
8		effect there for a large scope job to ensure	8		operator's panel, we will go into each
9		that the full scope is identified, all the	9)	individual operator's ways and means of doing
10		hazards are identified and the correct	10)	behavioural observations, but this is just an
11		mitigations are in place to allow us to go	11		opportunity for someone to reflect on how the
12		forward with that scope. But in general, day-	12		operation is going, perhaps look for areas of
13		to-day activities, the offshore workforce deal	13		improvement or indeed just say, you know, this
14		with the safe job analysis at site. Certainly	14		job is going extremely well and give the work
15		the onshore group cannot stand alone produce a	15		team a pat on the back for a job well done.
16		risk assessment that would be applied	16		So that's the type of feedback that the
17		offshore. There's local conditions and local	17		workforce get from the behavioural observation
18		knowledge that needs to be applied on each	18		program.
19		occasion. So bigger scope supported by	19		So the next slide continues with risk
20		onshore. The day-to-day activities are more	20)	management. Effective risk management
21		associated with the offshore workforce.	21		requires that persistent application and
22		So the permit would be issued. Having	22		continuous improvement of the safety
23		identified all the hazards and put in place	23		management processes to reduce risk to as low
24		all the mitigations, we issue the permit, and	24		as may be practical. The philosophy of how
25		before the job starts, we would have a toolbox	25		projects develop has already been discussed in

January 11, 2010 Mu		Multi-P	age	Offshore Helicopter Safety Inquiry
	Page	e 109		Page 111
1	terms of eliminating any hazards, controlling	1	MR	. PRITCHARD:
2	the hazards, mitigate against hazards and	2	A	A. Absolutely. That's the intent of it. So we
3	provide a means of recovery if situations	3		can apply some mitigations in some places and
4	develop. To do this, hazards need to be	4		then recognize that the job still needs to
5	identified for all situations and evaluated	5		have further mitigations or to do the job a
6	with consideration as to the likelihood of	6	i	different way, if we consider that the risk
7	such hazards becoming a reality and the	7		which is the consequence and likelihood
8	associated consequence, what can be done to	8		elements are not suitable.
9	reduce the risk by either reducing the	9	ROI	L, Q.C.:
10	likelihood or the consequence. So this can be	10		2. What do you do when you have an activity that
11	in the form of the equipment design, the	11		presents a likelihood of being frequent and a
12	procedures or the people. So we see once	12		consequence of being high? In other words,
13	again those three categories that I described	13		it's going to happen and the consequences are
14	earlier.	14		going to be dramatic and you can't move it,
15	We also need to determine how practical	15		what do you do?
16	the new measures may be and if they are	16	MR	. PRITCHARD:
17	reasonable, efficient, then we need to track	17	A	A. We have to do something different and we
18	the implementation of that plan to a	18		either engineer something different or take a
19	conclusion. So we need to establish the	19		totally different approach, if we can't
20	hazards, the mitigations, and then plan to	20)	achieve, through various means of mitigations
21	ensure they are installed.	21		to bring that back into the green or yellow
22	So the next slide is a typical risk	22		sectors.
23	matrix and it's used for the process of	23	ROI	L, Q.C.:
24	determining risk. We do it initially without	24	. (2. Have you ever I don't know if you've ever
25	any mitigations and then we look to what the	25		run into a situation where you plan to do a
	Page	e 110		Page 112
1	mitigation effect would be. So when we	1		certain activity and then just found we can't
2	identify a hazard, we say what would the	2		do it because it's too risky and we're going
3	effect of that and the likelihood and then	3		to have to do something different entirely?
4	when we apply our mitigations, we also look	5 4	MR	. PRITCHARD:
5	towards the matrix to see what improvements	we 5	A	A. Yes, we have, and I can't give you that great
6	have made by virtue of the risk matrix. So	6		example, but I do know from time to time when
7	the likelihood and consequence axis.	7		we would do the risk assessments, we say no,
8	ROIL, Q.C.:	8		we've got to stop here, pull back, reassess
9	Q. I think we saw a similar risk matrix in the	9)	and potentially do something different.
10	Aerosate presentation.	10	ROI	L, Q.C.:
	MR. PRITCHARD:	11	Ç	Q. Okay. Perhaps during the lunch break, if you
12	A. I'm pretty sure, yes, that's correct. It's	12		want to have another reflection on an example
13	very widely used in the certainly our	13		of that, where that's happened to either one
14	industry and I believe the aviation industry.	14		of you, that would be fine.
15	ROIL, Q.C.:	15	MR	. PRITCHARD:
10	Q. This isn't a new invention of the offshore industry anywh	10		A. I WIII do.
1/	in the world?		ROI	L, Q.C.:
18		18		
19	MR. PRITCHARD.	19	WIK.	. PRITCHARD:
$\begin{vmatrix} 20 \\ 21 \end{vmatrix}$		20	r F	square it could be a seven by seven people
$\begin{vmatrix} 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 $	0 And I gather the objective is to plot your			can expand this type of risk matrix tool to
$\begin{vmatrix} 2^{2} \\ 2^{3} \end{vmatrix}$	risks and then take stens to bring those risks	22		their own use but what it does do is it
$ _{24}^{23}$	out of the red and into the green as much as	23		allows for a consistent approach that is
25	possible?	25		documented that would allow people to

January 11, 2010 Mult			Page	¹ Offshore Helicopter Safety Inquiry
		Page 113		Page 115
1	understand why the risk has been reduced	. how	1	other and in physical size.
2	it's been reduced and why the job would	l be	2	So as an analogy for this, I'd like to
3	allowed to go ahead or not, as the case n	nav	3	consider the performance of the installation
4	be. The hazard is identified, the likelihoo	d	4	fire foam system on the helideck. So for foam
5	of that hazardous event occurring and t	he	5	to be discharged on the helideck should we
6	consequence of that event is what we a	ire	6	have an incident, we need fire water from the
7	looking for.		7	fire pumps. So the fire pumps are designed to
8	So we've seen from the previous slides		8	discharge a certain volume of water and by
9	that it's the people, process and equipme	nt	9	maintenance routines, we measure and monitor
10	that have the potential to create the barrier	rs	10	on a regular basis how much the pumps
11	from hazardous events. So we see once a	igain	11	discharge. So if a pump is brand new and we
12	here the Swiss cheese slice model with	h i	12	get a good consistence, perhaps a little bit
13	equipment, process and people. Now v	ve've	13	more than manufacturer's expectations from the
14	shown two slices of cheese per kind of	of	14	pumps, perhaps the hole in the cheese would be
15	category there, but obviously we can have	many	15	relatively small. As time goes on and the
16	more barriers in each of the categories an	nd	16	pump wears a little, then perhaps the hole in
17	the more barriers you have, that should	be	17	the cheese grows a little bit more, and if we
18	applied.		18	get to the point of having less water from the
19 I	ROIL, Q.C.:		19	pump than the expected requirements for the
20	Q. So the best example would see many equ	ipment 2	20	foam, making of the foam, then certainly the
21	defences, many process defences, and r	nany	21	hole in the cheese gets a lot wider.
22	people defences?		22	The fire pumps have, by design, certain
23 I	MR. PRITCHARD:	2	23	features that would protect the fire system,
24	A. That's exactly right, yes. We would look	to 2	24	because we need the availability of the pumps.
25	have as many, but for this model, we look	ed at	25	So we need to do maintenance on the pump, so
]	Page 114		Page 116
1	this. So this slide takes us back to the	-	1	we have maintenance processes that would
2	potential barriers we can create and wh	at	2	ensure availability and also, by design, we
3	happens if the barriers are not robust and	an	3	have multiple number of fire pumps to give us
4	incident occurs or we come through the h	oles	4	that availability and redundancy built in.
5	in the cheese slices.		5	The quality of the foam dispensed is also
6	It should be noted that the holes in the		6	checked on a regular basis and so therefore
7	cheese are constantly moving, both in rela	tive	7	there's a tolerance to the quality of the
8	terms to each other and in physical size, a	nd	8	foam. So once again, we can see if the foam
9	I'll give you some examples as to why t	hat	9	quality is not quite there. If it's good,
10	happens and what we should be vigilant a	bout.	10	then the hole in the cheese would be
11 H	ROIL, Q.C.:	,	11	relatively small. If it's getting to the edge
12	Q. Okay. So Swiss cheese is not an ideal and	alogy	12	of the tolerance, perhaps opening up, and if
13	because Swiss cheese, the holes are alway	rs the	13	it's beyond tolerance, then the hole in the
14	same once you have the piece of cheese.	What	14	cheese opens a little bit wider.
15	you're saying is here everything is dynan	nic,	15	The helideck team, so this is the people
16	is it?	-	16	side of it now, the helideck team have trained
17 I	MR. PRITCHARD:	-	17	for events through refresher courses and, you
18	A. Yes, the holes in the cheese when they'	re	18	know, perhaps if we have a well-trained
19	lined up, people can get a concept that y	ou 1	19	nelideck team that have just been for
20	have a barrier and it you have the barrier	;	20	retresher and have worked as a team for many
$ ^{21}$	the incident arrow doesn't pass through.	But 2	21	years together, we could consider the hole in
$ ^{22}$	when you put the barrier in place or the he	Die 2	22	the cheese a little bit smaller. However, if
23	is lined up that something can physically :	move	23	we ve got a relatively new team awaiting that
$ ^{24}$	through. But in real terms, those holes an	e 2	24	refresher course, then perhaps the hole in the
25	moving in dimensions both relative to e	ach 2	25	cneese could open up slightly more.

January 11, 2010	Multi-Page ^T	^M Offshore Helicopter Safety Inquiry
	Page 117	Page 119
1 So as you can see, there's a myriad of 2 ways and means that the holes in the cha- 3 can open and close and in relative terr 4 together, they have some movement. So 5 only when the barriers all break down tha 6 would have an incident. 7 At all junctures, you would recognize 8 that if it's perhaps people, and they are th 9 most important aspect to this, they are 10 looking after the equipment and doing the 11 maintenance programs. They're the one 12 are going through the process of checking 13 quality of the foam and indeed, they are 14 people on the helideck who would be tra-	Page 117 1 eese 2 ns 3 o it's 4 at we 5 6 7 re 8 9 10 that 11 ag the 12 ROII 13 Q 14	Page 119 describe that in too much detail, but there are communications from the safety and environment and various other sub-committees and task forces that report back to the Executive Committee, who are the responsible decision makers at CAPP. L, Q.C.: . Have either or all of you sat personally on CAPP committees? PRITCHARD: . I've attended the CAPP meetings. L, Q.C.: . The regular CAPP meetings, but have you sat on any of the sub-committees?
 and have to be used in the event of at incident. ROIL, Q.C.: Q. So I think as you said in one of your earlier 	n 15 MR. 16 A 17 ROI ier 18 Q	PRITCHARD: . Not on the sub-committees, no. L, Q.C.: . Okay.
 19 slides, the people aren't segregated just t 20 that side of the line up of cheese. Peopl 21 are involved with putting the equipment 22 place, putting the processes in place. 23 MR. PRITCHARD: 24 A. Correct. 25 ROIL, Q.C.: 	to 19 MR. e 20 A 21 22 23 24 25	PRITCHARD: The committees and task forces typically have input to regulations and guidelines and prepare standard practises and training qualification guidelines for considerations to be adopted on basin-wide approach. The committees and task forces can comprise of
 Q. So it's a very people oriented focus, I wo think? MR. PRITCHARD: A. We have an influence over each and e barrier that we can create, whether it b equipment or processes. So it's really ab the attitude and leadership of everyone ensure that we have that level of complia with those pieces of equipment and t processes. I want to move on now to a li bit about the safety participation, so I wat to establish there that people are very important to all aspects of creating barrie 	Page 118 uld 1 uld 1 2 3 overy 4 e 5 out 6 to 7 ance 8 he 9 ttle 10 nt 11 rs. 13 ROII	Page 120 subject matter experts, HSC professionals, and potentially offshore workers seconded onshore, or indeed people who have worked offshore and are now in an onshore role. So the committees are kind of a diverse group of people, generally in the health and safety and environmental side. So there's a well established communication in the JOHS Committees offshore, so the Joint Occupational Health and Safety Committees which are established offshore, there's a good communication route to them. L, Q.C.:
14 It's the level of communications that we 15 with people as well that is so important. 16 I'll start with the CAPP side of things here 17 So industry participates at all CAPP levels 18 the east coast, and CAPP is a national 19 recognized organization, and for this I w 20 be speaking more in specific terms to the 21 coast operations. Whilst Husky has defe 22 its membership, we are still at the table a 23 the committee and task force levels. 24 believe Mr. Paul Barnes described the wo	have 13 ROII have 14 Q So 15 15 c. 16 16 son 17 18 rill 19 MR. east 20 A erred 21 21 at 22 23 orkings 24 ROII	 Okay, let me stop you there because I know that others in the room would want to pursue that as well, how and in what way is the Joint Occupational Health and Safety Committees, how do they get engaged in CAPP? PRITCHARD: They can be given reviews of say, typically the Rigging and Slinging Guidelines, which was a documentation that CAPP are going to coordinate L, Q.C.:

January 11, 2010 Multi		ti-P	age TM	Offshore Helicopter Safety Inquiry	
	Page 12	1		Page 123	
1	MR. PRITCHARD:	1		particular aspect.	
2	A. Those guidelines in preparation would be	2	MR. S	ACUTA:	
3	issued offshore to get feedback from the JOHS	3	А.	And we did have JOHS Committee members attend	
4	Group and indeed the subject matter experts	4	,	that water ingress testing.	
5	offshore as well, those people that would use	5	ROIL,	Q.C.:	
6	that particular guideline. So we get feedback	6	Q.	Were the ones jumping in the water or were	
7	from that guideline through the JOHS Committee	7		they observers?	
8	back to the level of CAPP Committee level.	8	MR. S.	ACUTA:	
9	ROIL, Q.C.:	9	A.	Some of them were the ones jumping in the	
10	O. Okav.	10)	water, yes.	
11	MR. PRITCHARD:	11	ROIL,	Q.C.:	
12	A. So CAPP related issues such as Training and	12	Q.	They became the guinea pigs for the testing,	
13	Qualification Standards or the Rigging and	13		did they?	
14	Slinging Guidelines are typical of	14	MR. S	ACUTA:	
15	communications with the JOHS Committees.	15	A.	Yes.	
16	There's also a communication with the Board at	16	MR. P	RITCHARD:	
17	twice yearly meetings. There would be topics	17	A.	So on the occupational health and safety	
18	during the quarterly Board meetings that the	18		regulations, the C-NLOPB, or the Board, is	
19	operators have with the Board, and indeed if	19)	joint provincial and federal and does not have	
20	required, we can have ad hoc meetings with the	20)	legislation covering the following; employer	
21	Board regarding CAPP issues. So during recent	21		and employee duties, right to refuse dangerous	
22	times there's been very frequent updates on	22		work, health and safety program policy, and	
23	helicopter transportation with reference to	23		OSH committees. The Newfoundland Offshore	
24	such items as the HUEBA, flight suits, and	24		Health and Safety Act describes those four	
25	general helicopter operations.	25		requirements, as does the Canadian Labour	
	Page 12	2		Page 124	
1	ROIL O.C.	- 1		Code Part II which is a federal	
$\begin{vmatrix} 1 \\ 2 \end{vmatrix}$	O Have that actually increased since the		ę.	iurisdiction So we have the Newfoundland	
3	incident of March 12th or had it increased	3		Offshore Health and Safety Act the provincial	
4	before then?	4		iurisdiction, and the Canadian Labour Code.	
5	MR PRITCHARD	5	í	Part II is a federal jurisdiction Both of	
6	A. Substantially increased in terms of helicopter	6	i	those actually mention those four areas I've	
7	transportation.	7	,	iust described, and the way that the Board	
8	ROIL O.C.:	8		picks that up is by describing that as other	
9	O. Since the March 12th incident?	g	,	requirements for our work authorizations, and	
10	MR. PRITCHARD:	10)	that's how they become applied.	
11	A. Absolutely.	11	ROIL	0.C.:	
12	ROIL, O.C.:	12	0.	Is there a conflict between the Newfoundland	
13	0. You mentioned one of the bullets, helicopter	13		Occupational Health and Safety Act and the	
14	flight suit water ingress testing. I take it	14		Canada Labour Code, or does one apply to	
15	this is the testing that took place after the	15		certain workers and the other apply to other	
16	suits were in place and in use, or was this	16	;	workers or other situations?	
17	before they were put into place?	17	MR. F	PRITCHARD:	
18	MR. PRITCHARD:	18	A.	I think they all apply very similar, and	
19	A. No, this was after the incident, and when the	19)	indeed there's a planned amendment to the	
20	helicopter flight suit water ingress became an	20)	Atlantic Accord which would allow the Board to	
21	issue, we did some extensive training or some	21		actually prescribe those four areas by	
22	additional training not training, testing	22		regulation. I'll discuss a little more about	
23	in Halifax with respect to the E452 suit and a	23		the Offshore Health and Safety Committees	
24	new suit that we'll hear more about in future	24		offshore. The Committee is comprised of both	
25	testimony. So we have got some slides on that	25		management workers and have a co-chair system,	

January 11, 2010 Mul		Multi-	i-Page TM		Offshore Helicopter Safety Inquiry		
	P	Page 125			Page 127		
1	so that they both work the chair.		1		coming offshore, it would create that one		
2	ROIL, O.C.:		2		constituency if they were of sufficient size.		
3	0. Before you go into that, we heard from ear	lier	3		but the drilling group		
4	witnesses about the fact that the committee	es	4	ROIL.	0.C.:		
5	offshore transcend the various contractors	and	5	0.	This is perhaps something we should look at in		
6	employers that are out there, and there's o	ne	6	χ.	more detail in the individual presentations		
	committee that represents all of the worke	ers	7		but just to set up that issue so that having		
	of the different employers?		, 8		different employers because you know that		
	MR PRITCHARD		0		onshore an employer will have its own each		
	A That's correct		10		employer will have its own IOHS Committee		
			11		Out on the offshore all the employers work		
$\begin{bmatrix} 1 \\ 1 \\ 1 \end{bmatrix}$	O Vou're familiar with that?		11		together to have one joint if you will using		
12	Q. Tou re fammar with that:		12		the word joint in a different way		
13	MR. FRITCHARD.	int	13	MD D	DITCHARD.		
14	A. This an comes together under JOHS, the JO.	IIIt	14		Vas vaab		
15			15	A.			
10	A gain does that create any problems for w		10	KUIL,	Q.c.: One joint committee So operationally that		
$ _{10}^{17}$	Q. Again does that create any problems for yo	Ju as	17	Q.	works for you, you're satisfied, with the way		
10	are different employers that are a part of the		10		it works?		
19	are different employers that are a part of th		19	MDD			
$ _{21}^{20}$			20	MK. P.	RIICHARD: It works for us because sefety does not		
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	MR. PRITCHARD:		21	А.	transcend only of the boundaries. We're all		
$\begin{vmatrix} 22\\ 22 \end{vmatrix}$			22		going to work together as one to the sefety of		
23	ROIL, Q.C.:		23		going to work together as one to the safety of		
24	Q. Different employers and different employers	yees .	24 25	DOII			
23	fioni different employers that are an		23	KOIL,	Q.C		
	P	Page 126			Page 128		
1	represented on the one?		1	Q.	And representation doesn't depend on the size		
2	MR. PRITCHARD:		2		of my work force, it's more done by work area,		
3	A. Not at all. We're all have a mind for safet	У	3		is it?		
4	as our top priority, so we all come togethe	er	4	MR. P	RITCHARD:		
5	collectively, and companies and where you	u work	5	А.	It's by departmental, but the size of the work		
6	is no matter to us.		6		force suggests that if a large group were to		
7	ROIL, Q.C.:		7		come on board as construction group, you know,		
8	Q. Does every company get a representative.	? If	8		on a new project to install something, they		
9	you have five contractors on board the	e	9		could in their own right create a new		
10	facility that you have, and the JOHS Comm	ittee	10		constituency and elect a member to sit on the		
11	on that facility, would all five of those		11		JOHS Committee.		
12	contractors have a representative necessari	ly	12	ROIL,	Q.C.:		
13	or does it move back and forth at differen	nt 🛛	13	Q.	And then if they withdraw their services and		
14	times to different people?		14		they're finished their job, they would just go		
15	MR. PRITCHARD:		15		away again?		
16	A. It works in general terms, that it become	s	16	MR. P	RITCHARD:		
17	more departmental. So if you are related	to	17	А.	Correct.		
18	the marine group, the maintenance group	, or	18	ROIL,	Q.C.:		
19	indeed if you come on as a construction gr	oup,	19	Q.	Okay, thank you.		
20	if there is sufficient numbers within that		20	MR. P	RITCHARD:		
21	construction group, you would create a n	new	21	Α.	So the members of the committee are elected		
22	constituency, and, therefore, an elected		22		and represented for a two year term, and all		
23	member from that new constituency wou	ld be	23		departments are represented. The meetings are		
24	applied to the JOHS Committee. So depen	ding	24		held once every three weeks, or once every		
25	on the size if it's one specific group		25		rotation, and the Minutes of the meetings are		

January 11, 2010 Multi			Offshore Helicopter Safety Inquiry			
	Page 12	29	Page 131			
1	issued to management and the C-NLOPB. The	1 Q.	The same bodies, the same individuals?			
2	Occupational Health and Safety Committee	2 MR. PR	ITCHARD:			
3	members receive training in the organization	3 A.	Yes.			
4	and function of the company, company specific	4 ROIL, Q	Q.C.:			
5	health and safety programs, and the company	5 Q.	So how do they know what the second group has			
6	specific investigation techniques. So they	6	done?			
7	can assist in any investigations that may be	7 MR. PR	ITCHARD:			
8	required on board.	8 A.	There's Minutes of the meeting, and as I can,			
9	ROIL, Q.C.:	9	they can converse with the people from the			
10	Q. I have another question arising out of the	10	first group, the first meeting. There will be			
11	work regime which we've heard about and the	11	new people arriving offshore that will			
12	way, if at all, that impacts the operation of	12	actually take part in the second meeting, so			
13	the Occupational Health and Safety Committees	13	there's time there for communications and they			
14	on board. We understand that workers work on	14	can have their own meetings, if they so wish.			
15	a three week rotation, but that they all don't	15	If there's some real burning issues, then the			
16	go out on the 1st of January and come back on	16	JOHS people can get together at any time.			
17	the 21st, that there are people moving all the	17 ROIL, Q	Q.C.:			
18	time. You know, groups go out on the 1st, and	18 Q.	Okay, so the first group will have Minutes of			
19	another group might start their 21 day	19	their meeting?			
20	rotation on the 7th, and some more might start	20 MR. PR	ITCHARD:			
21	on the 14th, and so on.	21 A.	Correct.			
22	MR. PRITCHARD:	22 ROIL, Q).C.:			
23	A. By virtue of the fact that there is that	23 Q.	Those Minutes, will they go back to the same			
24	rotation and people change out every week, the	24	group six weeks later, or will those Minutes			
25	JOHS Committee members can see each other and	1 25	go to the next group that were three weeks			
	Page 17	30	Page 132			
1	can communicate with each other such that		later?			
	there isn't a Group "A" and a Group "B"		RITCHARD.			
	Group "A" mingles and converses with Group	2 MIC Π	They're just consecutive Minutes of the IOHS			
	"B" so that there's not two distinct groups	4	meetings			
5	there they are but one committee and they can	5 ROIL				
6	converse	6 0	So each group knows what the other group is			
	ROIL O.C.	7	going?			
	O So if you have a meeting now and a meeting in		Sound . RITCHARD			
9	three weeks time would there be different		Correct			
10	people attending the first and the second	10 MR S				
	meeting?	10 MIC 57	One of the agenda items in the standard			
12	MR PRITCHARD	12	meeting is to review the previous IOHS			
13	A There will be different people by virtue of	13	meeting so the crew that comes on board will			
14	the rotation but in the subsequent three	14	review all the issues that were raised by the			
15	weeks if there's issues arising from the first	15	previous meeting as part of the standard			
16	meeting that can be discussed and worked with	16	meeting agenda			
17	the new group that are going to be joining	17 ROIL				
18	that would actually sit to the second meeting	18 0	Right okay thank you			
19	ROIL OC:	19 MR PF	RITCHARD.			
20	O Okay so the first group the second group	20 A	Part of the employee rights is the employees			
21	the third group, are they the same people as	21	have the right to participate and become			
22	the first group more or less?	22	elected members if they so wish or just			
$ _{23}^{-2}$	MR. PRITCHARD:	23	simply participate in safety departmental			
$ _{24}$	A. Correct.	24	meetings. The right to know means that they			
25	ROIL O.C.:	25	should be aware of the hazards and mitigations			
January 11, 2010 Mult			Multi	-Page ^{TI}	Offshore Helicopter Safety Inquiry	
-----------------------	-------	---	--------	---------------------	--	--
		Pa	ge 133		Page 135	
1		associated with the tasks on board, as well a	.S	1	will also be given that level of instruction	
2		the knowledge of the company specific safe	ety	2	and orientation for helicopter transportation.	
3		systems and the safety culture of that	2	3	So open communication is constantly encouraged	
4		organization. The right to refuse dangerous	3	4	between the workers and the management. This	
5		work is if the worker considers the task		5	slide show is a number of meeting types, by no	
6	i	assigned to that person or team is considered	f	6	means all, whereby safety is discussed in	
7	,	dangerous, then there's a process that follow	vs	7	terms of proactive mitigation type, as well	
8		until the dangerous aspects of the work are	:	8	as, I'll say, historical safety trends, but	
9)	either changed or is deemed safe to comple	te	9	typically town hall meetings with senior	
10		the task. The right to refuse process		10	management or indeed for Husky themselves, the	
11		involves the workers and supervision, and the	ne	11	OIM discusses safety statistics with the	
12		JOHS Committee, and then ultimately the Bo	ard	12	workforce. The permit to work and toolbox	
13		if it cannot be resolved on board. So there's		13	talk, we've discussed that a little bit in the	
14		a process of understanding the risks of		14	previous testimony. The workforce and	
15		dangerous work and whether the workers	are	15	supervision and potentially the onshore group,	
16	i	willing to go ahead and proceed, kind of an	ı	16	engineering group, can get involved in that?	
17	,	elevation of that.		17 ROII	, Q.C.:	
18	ROIL,	Q.C.:		18 Q	. And how often would those kinds of things	
19	Q.	Have either one of you or all of you ever		19	happen, are they once a day, once a week, once	
20)	known a situation where a worker has refus	ed	20	a month, once a year, what sort of frequency	
21		to do work in the offshore because he or she	e	21	can you assign to that kind of communication?	
22		believed that it was unsafe?		22 MR.	PRITCHARD:	
23	MR. S	ACUTA:		23 A	. Certainly the permit to work is going to be on	
24	· A.	I've been involved yes, and most recently v	ve	24	a daily basis. If we've got tasks, we will	
25		had three late last year.		25	have a permit to work. Once you have a permit	
		Pa	ge 134		Page 136	
1	ROIL,	Q.C.:		1	to work, you will by virtue of that have a	
2	Q.	Any of those in relation to helicopter		2	toolbox talk because you're going to talk	
3		transportation?		3	about the mitigations against any hazards	
4	MR. S	ACUTA:		4	identified. Departmental safety meetings,	
5	Α.	All three.		5	they can be requested at any time, but	
6	ROIL,	Q.C.:		6	generally for Husky here, we have departmental	
7	Q.	All three were helicopter transportation		7	meetings every three weeks. So that three	
8		related, and all since March?		8	weekly departmental meetings	
9	MR. S	ACUTA:		9 ROII	, Q.С.:	
10	Α.	Yes, all three of them occurred after Robbie	e	10 Q	. These are on board departmental meetings?	
11		Decker's testimony.		11 MR.	PRITCHARD:	
12	MR. P	RITCHARD:		12 A	Yes, they'll roll up departmental safety	
13	А.	As mentioned, communications is so impor	tant	13	meeting can roll up to the JOHS Committee	
14		and we have people we need to be able to)	14	level. So there's a departmental individual	
15		communicate to ensure the integrity of the	•	15	there that's usually an elected member for the	
16	i	systems. So along with the company		16	JOHS meeting, and, therefore, any issues	
17		orientations that reflect the company's safet	У	17	arising from the departmental meeting that	
18		culture and understanding of the safety		18	cannot be satisfied at that level there	
19)	management tools and processes to ensure	the	19	will be some supervision at that departmental	
20		hazards are controlled and mitigated, there's	3	20	meeting, so if they cannot be satisfied at	
21		also continuous orientation for helicopter		21	that, they would rise to the JOHS level, and	
22		transportation every time someone travels	•	22	if the JOHS themselves can't come up with a	
23		Every time you go to the heliport, you will		23	conclusion to any issues, it will be raised to	
24		see an induction video and get instructions.		24	the onshore management to aid in resolving	
25		Conversely from the offshore to onshore, y	ou	25	issues. Town hall meetings with management,	

January 11, 2010	Multi-Page	• TM Offshore Helicopter Safety Inquiry
	Page 137	Page 139
1 certainly for the Husky again our Offs	hore 1	explain the way in which each of these
2 Installation Manager has a meeting ever	v trin 2	contractors integrate and help us deliver our
3 so every three weeks there's a town h	$\begin{array}{c c} \text{all} & 2 \\ \text{all} & 3 \end{array}$	business. The section is not intended to
s so every linee weeks liefe s a town in		provide a comprehensive list of contracted
4 presentation.	4	services. I think it's fair to say each
5 KOIL, Q.C.:	5	services. I think it's fair to say each
6 Q. That expression town hall presentation	1S 6	company represented here would probably have
7 probably one that you re very familiar v	vith, 7	over 100 different contracts in support of our
8 and the one that I know from other worl	k that 8	business, but to give you an example of one
9 I've done in life, what is a town hall mee	eting 9	that we use independently, and some more
10 when you don't have a town hall?	10	collaboratively. Contractors are a
11 MR. PRITCHARD:	11	significant part of our business, and as such
12 A. We get into the biggest area we can	to 12	are part of our safety culture. Contractors
13 facilitate as many people, and everyone	e on 13	are expected to have their own safety
14 board is invited to come along to the to	own 14	management systems, but those systems must be
15 hall to listen to whatever the news is	. 15	aligned with the safety management systems of
16 Generally it's, you know, new topics, n	naybe 16	the respective operator. While working on our
17 safety statistics, and it may be any issue	es 17	offshore facilities, the contractors must
that are burning for the group themselve	s. 18	adhere to all the operator policies while on
19 ROIL O.C.:	19	board their installations.
20 0 0 Okay so these are community meeting	s the 20 RO	
20 Q. Okuy, so these are community incerting	20 KO	O So who's responsible to ensure that their
22 facilities?		cafety systems mesh with yours? Is it C-NI OPR
22 Includes:	22	or is it yours, or who's responsible for that
23 MR. PRITCHARD:	23	of is it yours, of who s responsible for that
24 A. Correct.	24	governance, il you will?
25 ROIL, Q.C.:	25 MF	R. VOKEY:
	Page 138	Page 140
1 Q. Okay.	1	A. From Suncor's perspective, it would be Suncor.
2 MR. PRITCHARD:	2	I know the other companies do it too. We all
3 A. And shift handovers, we do a 12 hour	shift 3	do independent and sometimes collaborative
4 rotation, so they occur twice daily as the	ne 4	audits of our suppliers and contractors, and
5 shift handover occurs. Safety is also	a 5	in the case of Suncor, we will probably do 12
6 discussion point at the daily meetings t	hat 6	to 15 a year and they are typically of the
7 occur between onshore and offshore, une	der the 7	contractor management systems. Contract
8 departmental meetings as well.	8	services are audited by the operators, as I've
9 ROIL O.C.:	9	indicated, through our respective supplier
10 0 So onshore/offshore have daily meeti	ngs 10	audit programs
11 Would they be done electronically or bo	11 RO	
12 they		O So you say there in your slide "are subject
12 MD DDITCHADD.	12	to audit" and I think your avidance what
15 MR. PRITCHARD.	15	you're telling me is that not only are they
A. Just telephone conference. We dial in,	we 14	you le tennig me is that not only are they
15 review reports, and any issues of the day	. 15	subject to it, those audits are actually done?
16 ROIL, Q.C.:	16 MF	R. VOKEY:
17 Q. Un-hm.	17	A. That's correct.
18 MR. PRITCHARD:	18 RO	ML, Q.C.:
19 A. I'll now hand over to Mr. Vokey to ste	p us 19	Q. So it we look, we should find some audits of
20 through the contracted services.	20	the helicopter provider by somebody prior to
21 MR. VOKEY:	21	March 12th?
22 A. As an earlier part of our presentation,	we 22 MF	R. VOKEY:
23 described a number of services, like sup	nnly 23	A Ves
1	pp1y 25	A. 105.
24 vessels and helicopters. In this section	i, 24 RO	A. 105. ML, Q.C.:

January 11, 2010 Mu		ti-Page TM		Offshore Helicopter Safety Inquiry
	Page 14			Page 143
1	MR. VOKEY:	1	MR. V	/OKEY:
2	A. Both of a technical and both of a management	2	A.	That's correct, and quite often it's done in
3	system nature. As each development is	3		the pre-qualification phase of the bid, and
4	planned, the operator will determine the	4		then through the invitation to tender, you dig
5	aspects of its business which are part of its	5		further into it, and then for ongoing
6	core competency or area of expertise. The	6		compliance, you would do that through our
7	areas which are a core area of expertise to	7		audits.
8	the company, the operator will typically hire	8	ROIL,	Q.C.:
9	those people directly to perform the work.	9	Q.	What you're telling us is that each company
10	ROIL, Q.C.:	10		has its own way of doing that, but all of you
11	Q. Okay, what kind of people would be core, for	11		do it in that manner?
12	example?	12	MR. P	RITCHARD:
13	MR. VOKEY:	13	А.	Notwithstanding that, Mr. Roil, as the C-NLOPB
14	A. As an example, production operators. You're	14		audit the operators, they can recognize the
15	going to need production operators through the	15		function that we go out and audit our
16	life of the field while you're producing.	16		contracting services.
17	It's highly technical, it's a safety critical	17	MR. V	/OKEY:
18	position.	18	Α.	That's correct.
19	ROIL, Q.C.:	19	ROIL,	Q.C.:
20	Q. The Offshore Installation Manager, that kind	20	Q.	So in the audit of you, they are effectively
21	of person?	21		auditing your contractors?
22	MR. VOKEY:	22	MR. P	RITCHARD:
23	A. Typically supervisory now. As I indicated	23	А.	That's correct.
24	earlier in testimony in the bio, there are	24	ROIL,	Q.C.:
25	occasions where supervision can be seconded,	25	Q.	Okay, thank you.
	Page 142	2		Page 144
1	but typically they will be company people.	1	MR. V	/OKEY:
2	For those areas that are not within the	2	А.	On the east coast, there's a number of common
3	operator's expertise, these types of services	3		services for which all the operators contract,
4	are contracted. While each of the operators	4		and I'll just read off a few of them there.
5	may contract for the same service from a	5		Shore-based services, marine offshore service
6	common supplier, most are done through	6		and support vessels, drilling, catering, ice
7	separate contracts directly between the	7		and weather observation and forecasting, and
8	service provider and the operator. When a	8		one that I will talk in more detail in a later
9	contract for service isn't entered into, the	9		session, helicopter services or helicopter
10	operator is responsible for ensuring that the	10		operations.
11	contractors are in compliance with all	11	ROIL,	Q.C.:
12	applicable legislative requirements, including	12	Q.	I think we actually of course, part of
13	those of the C-NLOPB and Transport Canada.	13		working this program out, this presentation, I
14	ROIL, Q.C.:	14		think we actually deal with shore-based marine
15	Q. So if they're your contractor, you have the	15		and helicopter, but just take for the moment
16	responsibility to supervise them, direct them,	16		the other three; drilling, catering, and more
17	and audit them, not necessarily the C-NLOPB?	17		particularly interesting to us, I guess, is
18	MR. VOKEY:	18		ice and weather observation. So drilling, do
19	A. Say that again, please?	19		you share that, or does each company tend to
20	ROIL, Q.C.:	20		have its own drilling expertise, or a drilling
$\begin{vmatrix} 21 \\ - 1 \end{vmatrix}$	Q. Yes. If it's a contractor that you've hired,	21	100	contractor?
$\begin{vmatrix} 22 \\ a 2 \end{vmatrix}$	then the responsibility to do the auditing of	$ ^{22}$	MR. V	/UKEY:
23	them, to check on them, to make sure that they	23	A.	In the case of drifting, there is two ways of doing it. There is the traditional entry is the
$ ^{24}_{27}$	nave safety policies and safety practices, is	24		doing it. I nere is the realized and way to do
25	your responsibility primarily?	25		it. In particular, in the early development

January 11, 2010			Multi-Page TM		⁴ Offshore Helicopter Safety Inquiry	
	Η	Page 145			Page 147	
1	stages of a project where an operator, as a	in	1	А.	Yes, it does, particularly the weather	
2	example, has ten of fifteen wells to drill,		2		forecasting piece in terms of, you know, what	
3	they will contract directly with a drilling		3		the weather is going to look like between,	
4	rig service provider, and in the case of		4		say, St. John's and the offshore, but we also	
5	Husky, the Glomar Grand Banks would b	be one	5		provide weather information, real time weather	
6	that they contracted for, White Rose or S	ea	6		information, through each installation to the	
7	Rose type of work support. Having said the	nat,	7		pilots while they're in route, whether it's	
8	three operators; Statoil, Husky, and Petro	0	8		wind speed, fog, or otherwise, sea states,	
9	Canada, now Suncor, a couple of years	ago	9		vessel movement.	
10	contracted as a rig-share agreement betw	een	10	ROIL,	Q.C.:	
11	the three operators to bring the Henry		11	Q.	So these ice and weather observers, and I	
12	Goodrich back into the basin because non	e of	12		think Mr. Decker indicated that he was one of	
13	the operators had a large enough scope of	work	13		those persons, they provide information that	
14	to bring it in independently, so we brough	t it	14		is relied upon by Cougar Helicopters in terms	
15	in on a more collaborative basis. Husk	у	15		of the provision of their service to the	
16	requested so many days, as did Suncor, as	did	16		offshore facilities?	
17	Statoil.		17	MR. V	OKEY:	
18	ROIL, Q.C.:		18	А.	That's correct, and just a point of note, we	
19	Q. The short answer is sometimes it's individ	ual	19		don't always have weather observers on board	
20	contracting, sometimes you share and try	to	20		the installations. Typically that would be	
21	use resources efficiently between you?		21		contracted for the winter months in	
22	MR. SACUTA:		22		conjunction with the ice, but we do have	
23	A. As far as Hibernia is concerned, HMDC act	ually	23		individuals on each of the installations that	
24	owns both of the drilling rigs on the Hiber	nia	24		are qualified in meteorology, such that they	
25	Platform, and they contract the operations	and	25		can report weather conditions.	
	I	Page 146			Page 148	
1	maintenance of those drilling rigs to Nob	le	1	ROIL,	Q.C.:	
2	Drilling Canada.		2	Q.	So they report to the base that is your	
3	ROIL, Q.C.:		3		weather contracted provider?	
4	Q. Okay. Again we don't want to spend too	much	4	MR. V	OKEY:	
5	time on it, but catering is done		5	А.	That's correct.	
6	MR. VOKEY:		6	ROIL,	Q.C.:	
7	A. Catering is typically done individually.		7	Q.	And they use that information in creating	
8	ROIL, Q.C.:		8		their weather forecasts and their predictions?	
9	Q. Individually, okay. Ice and weather		9	MR. V	OKEY:	
	observation and forecasting?		10	A.	It's a data point for their weather forecast.	
	MR. VOKEY:		11	ROIL,	Q.C.: Vesh	
12	A. To the best of my knowledge, we all utility	ze	12	Q.		
13	POIL O.C.		13	MR. V	OKEY: The convice providers we're talking shout	
14	KOIL, Q.C.:		14	А.	they have you know Transport Canada	
15	Q. But you each have your own contracts?		15		certified meteorologists	
10	MR. VOKEL.	rc	10	DOII		
11/	A. We each It's like with Cougar Hencopu	as,	17	KOIL,	Each of them has their own meteorologist and	
10	have our own individual contracts	20011	10	Q.	they get data from various sources?	
$\begin{vmatrix} 1 \\ 20 \end{vmatrix}$	ROLL OC:		20	MR V	OKEV	
$\begin{vmatrix} 20 \\ 21 \end{vmatrix}$	O Does ice and weather observation an	d	20	Δ	That's correct	
$\begin{vmatrix} 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 $	forecasting does that service have any		21	ROII		
$\left \begin{array}{c} 23 \\ 23 \end{array} \right $	particular impact on the provision of		23	0	Including whatever facilities or whatever	
24	helicopter services?		24	κ.	resources vou have on board the three or	
25	MR. VOKEY:		25		whatever facilities are out there?	

January 11, 2010		Multi-Page TM		Offshore Helicopter Safety Inquiry
	Page	149		Page 151
1	MR. VOKEY:	1		does work very well.
2	A. That's correct.	2 R	OIL,	Q.C.:
3	ROIL, Q.C.:	3	Q.	So when we hear in the or see in the C-
4	Q. Okay, thank you.	4		NLOPB guidance that a standby vessel must
5	MR. VOKEY:	5		standby when a helicopter is landing or taking
6	A. I just want to talk a little bit about shore-	6		off from an installation, these are the
7	based services and some of the support	7		vessels that provide that service?
8	operations would be shore-based facilities,	8 M	IR. V	OKEY:
9	cargo vessel coordination, provision of local	9	A.	I guess there's vessels are used for two
10	road transport, things of that nature, and A.	10		things in our core business. They're used for
11	Harvey and Company is under a contract to	11		support supply. We call those supply vessels
12	provide that. Marine support services is a	12		or support vessels, but there's always a
13	significant portion of our business.	13		dedicated standby vessel at each of the
14	ROIL, Q.C.:	14		operated assets. So Hibernia always has a
15	Q. Okay, that's those supply boats that we see	15		standby vessel in attendance, as does the Sea
16	down at the A. Harvey facility in St. John's	16		Rose, and as does the Terra Nova FPSO.
17	Harbour?	17 R	OIL,	Q.C.:
18	MR. VOKEY:	18	Q.	So while we see them going back and forth, we
19	A. Yes, some of them, I guess each of the	19		know that there's always one that is stationed
20	companies here, we contract individually for	20		out at each platform?
21	our supply vessels, but in the case of Suncor,	21 M	IR. V	OKEY:
22	we use ATL primarily and Secunda; Hibernia	22	A.	Yes, if you see them, they're not on standby
23	uses Maersk, and Husky would use for the mos	st 23		duty. Let s talk a little bit more about
24	part, AIL and Maersk. So a significant	24		standby vessel. We use boats for a variety of
25	portion of our business relates to, as 1	25		purposes in the orisitore environment, as 1
	Page	150		Page 152
	indicated, our marine operations. Whether	1		indicated, including the transport of cargo
$\begin{vmatrix} 2 \\ 2 \end{vmatrix}$	it's a standby requirement associated with the	2		and people, and from March, I guess, until
	offshore facilities, or the need for vessels	3		June, ice management. The guidelines
	to support anchor handling or towing, or the	4		to provide the C N OPP with a latter of
5	requirement for ice monitoring, and that is,	5		compliance from Transport Canada's Marine
	the year. Each of the operators contract for	0		Safety Group indicating that the vessels that
	the provision of marine support services. We	0		are in use by the operators meet the standard
	talked earlier, about mutual aid, agreements	0		for standby vessels and that's TD 7020 and
10	and as Mr. Pritchard indicated right now	10		for a standby complaint vessel, they'd have
	Husky because of the type of operation	10		the following requirements. They need
12	they're into with two drill rigs they do have	12		emergency response canabilities including
13	the bulk of the vessels but I think it is	12		fast rescue craft and these are essentially
14	fair to say that the three operators on an	14		crafts that are launched and recovered from
15	ongoing basis, while we've contracted for "x"	15		the standby vessels. Personnel recovery
16	particular vessels, we do share vessels for a	16		equipment to assist standby vessels in
17	number of things. We'll use it for personal	17		retrieving individuals from the water.
18	transport in the event that we can't fly. If	18 R	OIL,	Q.C.:
19	we've got some extra beds on board, we'll give	19	Q.	The word "recovery" means to get them out of
20	it up to Hibernia or White Rose, and likewise	20	-	the water, is it?
21	if there's equipment to be gotten offshore and	21 M	IR. V	OKEY:
22	we don't have a vessel onshore, whereas	22	A.	To get them out of the water in the event that
23	Hibernia do, we'll use space on their vessel	23		they're in the water. They're also required
24	to get our equipment offshore, so and	24		to have firefighting and medical first aid
25	that's primarily on an ad hoc basis, but it	25		support capability on board, and they need

J	January 11, 2010	Multi-Page	Offshore Helicopter Safety Inquiry
Γ	Р	age 153	Page 155
	1 passenger carrying capabilities. However, as	1	that were concerning for them, and that would
	2 I indicated previously, not all the vessels	2	be the clearance to remove this spool in the
	3 are the same and passenger carrying capability	3	congested area that we had, the stops that
	4 may range probably anywhere from 12 to 30, 1	2 4	were physically on the crane itself, so this
	5 to 32, in that range. These vessels are also	5	is a crane that would lift up and go along a
	6 designed to operate in our offshore	6	crane wheel and some physical stops that were
	7 environment similar to our installations, and	7	perhaps inadequate, and the weather conditions
	8 because they do operate on a year around	8	that were prevailing for the rest of the
	9 basis, they do have ice strengths in hulls,	9	course of that time. So they reported back to
	and the vessels are also equipped with what we	e 10	the supervision, who also then visited the
	11 refer to as DP, or dynamic positioning, so	11	site and agreed, as they reviewed that, that
	12 that allows it to come in alongside	12	looking at the Risk Matrix, it would be an
	13 installations and to hold a position without	13	area that would be considered between a kind
	14 being affected by wind, wave, or current.	14	of remote and occasional for the pipe work to
	15 ROIL, Q.C.:	15	be physically dropped as a dropped object, how
	16 Q. Okay, I think that's probably as good a place	16	often does that occur, you know, within Husky
	17 as any for us to take our lunch break. It's	17	or within the industry, so there are occasions
	18 now 12:30.	18	within the industry where there are dropped
	19 COMMISSIONER:	19	objects. So that would be between occasional
	20 Q. Two o'clock then.	20	and remote, and the consequence of the
	21 (RECESS)	21	physical pipe work, where it was a three ton
	22 COMMISSIONER:	22	weight dropping in that area would be quite
	23 Q. Okay, Mr. Roil.	23	significant in terms of consequence. So to
	24 ROIL, Q.C.:	24	date we have not completed that scope of work,
	25 Q. Thank you, Commissioner. Before we go ahea	id, 25	and we are looking at different ways and
┢	p	Page 154	Page 156
	1 I think that two of our panellists want to g		means of course to go back to the crane
	2 back to items that we dealt with already 0	ine 2	review the crane arrangement and see what
	2 back to herits that we dealt with already. 6 3 was Mr. Pritchard, who Lunderstand wante	ed to 3	else we can do to mitigate against that So
	4 speak to the Risk Matrix example that was	here 4	we've had that work on our books I would
	4 speak to the Kisk Matrix example that was 5 at slide number 42. So we have that up a	nd 5	think for around a year awaiting kind of
	6 soon as I get it there and running we'll be		further assessment. It's a piece of pipe work
	 ready to go 		that gives us flexibility it doesn't restrict
	8 MR PRITCHARD	8	us in any way in general operations. We would
	 A I've been reminded of a scope of work that 	twe 0	like to get it replaced and changed out for
	10 were trying to perform on the EPSO Sea Ro	ose 10	flexibility purposes and usage there. So
	and this was the removal of a very large sr		that's just an example whereby the work part
	12 from the turret area. The turret area is	12	was created onshore and reviewed there. Not
	13 quite a congested area and the analysis fo	r 13	everything can be fully assessed it needs to
	14 that ioh and the work part the preparation	n 14	go offshore. The workforce were involved and
	15 for that was completed onshore. So we ha	vea 15	the subject matter experts and they
	16 work part that understands the equipment	We 16	determined that you know we're too close to
	17 do that from the two day drawings that we	have 17	the likelihood and the consequence
	18 and try to work out the work plan that wo	uld 18	arrangements prevent us from going ahead with
	19 be associated with the removal of that spor		that job as we currently stand today
	That work part was then issued to offshore	and 20 CC	MMISSIONER.
	21 prior to any even permit to work being rais	sed. 20 cc	0. Just on that point then I take it production
	22 the crane driver who would be the operator	r of 22	is not held up?
1	the crane, and the area authority went out	to 23 MF	R PRITCHARD:
ļ	24 review the job. Now when they went ou	t to $\begin{vmatrix} 25 \\ 24 \end{vmatrix}$	A. No.
	25 review the job. they identified some area	s 25 CC	MMISSIONER:

Page 157 Page 157 1 0. So you can take the time to study this? 1 degree of risk attached to them should 2 MR.PRITCHARD: 3 a. That's correct. 3 in the area, or are the actual workers the 4 COMMISSIONER: 3 in the area, or are the actual workers the 5 Q. Another questions, if you don't mind, Mr. 5 Sim the subject, we taked 7 this morning about individuals doing 5 A. In general, the people doing the jobs? 9 a certain thing. Let's sub, top Ill an example 6 A. In general, the people doing the jobs? 10 out of the air, the refuelling of helicopters. 10 advisor on board who will monitor jobs and 11 Twis is refuelling helicopters. 10 understand the operations at hand, and will go 12 doing that all day long, or sho. Do people do 12 to the sidely person? 13 A. Indeed. They vary from installation to 9 0 When necessary. 14 COMMISSIONER: 10 0 When necessary. 14 MR PRITCHARD: 10 N. Mercreta.	January 11, 2010 Multi		ulti-P	age	⁴ Offshore Helicopter Safety Inquiry
1 Q. So you can take the time to study this? 1 degree of risk attached to them should 2 MR. PRITCHARD: 2 something go wrong, is there a study person 3 A. That's correct. 2 something go wrong, is there a study workers the 4 COMMISSIONES: 3 in the are, or are the actual workers the 5 Q. Another questions, if you don't mind, Mr. 6 A. In general, the people doing the job? 6 N. In general, the people doing the jobs 7 recognize the hazards and the mitigations that 8 a certain thing, Lef's say, to pull an example 9 themselves. We do, however, have a safety 10 doing that all day long, or she. D people do 11 understand the operations at hand, and will go 13 A. They do, yes. 14 COMMISSIONER: 14 COMMISSIONER: 14 N. Reprice and example 14 COMMISSIONER: 18 A. Correct. 14 A. They do, yes. 14 COMMISSIONER: 20 Q. When necessary. 15 A. Mr. Vokey, Commissioner, on behalf of Terra 18 A. Correct. 19 CoMMISSIONER: 16 norm necessary and when selected by t		Page	157		Page 159
2 MR. PRITCHARD: 2 something go wrong, is there a safety person 3 A. That's correct. 3 in the area, or are the actual workers the 4 COMMISSIONE: 5 Q. Another questions, if you don't mind, Mr. 5 5 Q. Another questions, if you don't mind, Mr. 5 6 A. In general, the people, as well as doing the job? 7 Q. Another questions, if you don't mind, Mr. 5 MR. PRITCHARD: 8 particular work and having responsibility for a certain thing. Lefs say, top all a example 6 A. In general, the people doing the job? 10 out of the air, the refuelling of helicopters. 10 advisor on board who will monitor jobs and 11 If "X": is refuelling helicopters. 10 understand the operations as hand, and will go 12 doing that all day long, or she. Do people do 12 to the sites, the work sites, to review the 13 A. They do, yes. 15 O. They eneral. 16 CoMMISSIONER: 13 oper tailing officer and refuelling 1 MR. PRITCHARD: 18 A. Correct. 14 O. And they're trained in the multiple functions? 18 A. Correct. 19 0. OMothey obsectife sites. <tr< td=""><td>1</td><td>0. So you can take the time to study this?</td><td>1</td><td></td><td>degree of risk attached to them should</td></tr<>	1	0. So you can take the time to study this?	1		degree of risk attached to them should
3 A. That's correct. 3 in the acca, or are the actual workers the safety people, as well as doing the job? 4 COMMISSIONER: 5 0. Another questions, if you don't mind, Mr. 6 Roil, while we're on the subject, we talked the generation thing, Let's say, to pull an example 6 A. In general, the people doing the job? 9 a certain thing, Let's say, to pull an example 9 themselves. We do, however, have a safety 10 out of the air, the refuelling of helicopters, he's not 11 a divisor on board who will monitor jobs and 11 11 "s'' is refuelling, of helicopters, he's not 11 understand the operations at hand, and will go 13 A. They do, yes. 15 0. I see, so the actual worker gets some support 14 COMMISSIONER: 15 0. I see, so the actual worker gets some support 15 0. I see, so the actual worker gets some support 16 from the safety person? 17 0. And they're trained in the multiple functions? 18 A. Correct. 19 A. Indeed. They yeary from installation to actual workers gets some support 10 10 10 A. Mr. Yokey, Commissioner, on behalf of Terra 20 Q. When necessary. 21 MR. PRITCHARD. </td <td>2 M</td> <td>IR. PRITCHARD:</td> <td>2</td> <td></td> <td>something go wrong, is there a safety person</td>	2 M	IR. PRITCHARD:	2		something go wrong, is there a safety person
4 COMMISSIONFR: a cafety people, as well as doing the job? 5 Q. Another questions, if you don't mind, Mr. a cafety people, as well as doing the job? 5 Q. Another questions, if you don't mind, Mr. b Commission participation of the subject, we talked 7 this morning about individuals doing f a metral, the repeople doing the job? 9 a certain thing, Let's say, to pull an example out of the air, the refuelling of helicopters, the refuelling of helicopters, the soft of the air the refuelling of helicopters helicopter landing of helicopters here. 10 multiple jobs? I presume they do, do the? to the sites, the work sites, to review the 11 m. PRITCHARD to the sites, the work sites, to review the 12 to the do, yes. to the sites, the work sites, to review the 13 A. They do, yes. to from the safety peorson? 17 Q. And they're trained in the multiple functions? ta A. Mc. VOKEY: 23 Nova, we use our vessel crews as part of the to safety advisor to go to specific sites. 24 ME. VOKEY: A. Mc. Vokey, Commissioner, on behalf of Terra 25 A. Mc. Vokey. Commissioner, on behalf of Terra Something like refuelling a helicopter soft adam? 21 MS. PRITCHABD ta canne operator 22 Mo. Would we a support team of firefighters soft adam? 3 A. That is fairly standard. to the safet	3	A. That's correct.	3		in the area or are the actual workers the
5 Q. Another questions, if you don't mind, Mr. 5 MR. PRITCHARD: 6 Norway and having responsibility for a certain thing. Let's say, to pull an example out of the air, the refuelling of helicopters. 6 A. In general, the people doing the jobs or recorgize the hazards and the mitigations that are there, so they are all encompassing a certain thing. Let's say, to pull an example out of the air, the refuelling of helicopters. 11 H'''s is refuelling of helicopters. 6 A. In general, the people doing the jobs or recorgize the hazards and the mitigations that are there, so they are all encompassing a certain thing. Let's say, to pull an example out of the air, the refuelling of helicopters. 12 doing that all day long, or she. Do people do 10 to the sites, the work sites, to review the 13 a. They do, yes. 11 team physically working. 14 MR. PRITCHARD 13 team physically working. 15 A. Indeed. They vary from installation to in installation as to who you take on board as or who their qeneding on own thetice; 10 10 10 21 A. Mee use our vessel crew as part of the helicopter landing officer and refuelling the conter side of a sks or critical procedures. 2 A. Whe necessary. 21 MR. VOKY: 25 A. We generally have, and 1 think for all 22 A. When necessary. 2 <td>4 C</td> <td>OMMISSIONER:</td> <td>4</td> <td></td> <td>safety people, as well as doing the job?</td>	4 C	OMMISSIONER:	4		safety people, as well as doing the job?
a Roil, while we're on the subject, we talked f Roil, while we're on the subject, we talked f particular work and having responsibility for a certain thing, Let's say, to pull an example iii f iiii f iiiiii f iiiiii f iiiiiiii f iiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii	5	O Another questions if you don't mind Mr	5	MR I	PRITCHARD
1 Norming about individuals doing a particular work and having responsibility for a certain thing. Let's say, to pull an example out of the air, the refuelling of helicopters. 1 1 If "x" is refuelling helicopters. 1 10 out of the air, the refuelling of helicopters. 1 11 If "x" is refuelling helicopters. 1 12 doing that all day long, or she. Do people do multiple jobs? I presume they do, do they? 1 13 A. They do, yes. 1 14 M.R. PRITCHARD: 15 15 A. Indeed. They vary from installation to 0 installation as to who you take on board at wour helicopter landing officer and refuelling 21 1 24 MR. VOKEY: 2 A. Mr. Vokey, Commissioner, on behalf of Terra 3 2 25 A. Mr. Vokey, Commissioner, on behalf of Terra 4 Page 158 1 installations, we have two types of tasks. We 2 14 hewould have a support team of firefighters 5 a crutice task, and that means here's a 3 a crutice task, and that means here's a 4 Something like refuelling a helicopter 3 Page 160 1 Page 158 1 installations, we have two types of tasks. We 4 Something like refuelling a helicopter so 4 2 A. We Nee: 3 <t< td=""><td>6</td><td>Roil while we're on the subject we talked</td><td>6</td><td>Δ</td><td>In general the people doing the jobs</td></t<>	6	Roil while we're on the subject we talked	6	Δ	In general the people doing the jobs
Image of the initial and the initinitial and the initial and the initial and the initia		this morning about individuals doing	7	11.	recognize the hazards and the mitigations that
a certain thing. Let's say, to pull an example a curter, so they are and encompassing 9 a certain thing. Let's say, to pull an example 9 10 out of the air, the refuelling of helicopters. 10 11 If "x" is refuelling helicopters. 10 12 doing that all day long, or she. Do people do 11 13 multiple jobs? I presume they do, do they? 13 14 MR. PRITCHARD: 13 15 A. They do, yes. 15 16 COMMISSIONER: 15 17 Q. And they're trained in the multiple functions? 18 18 MR. PRITCHARD: 19 19 A. Indeed. They vary from installation to 20 20 Wen necessary. 21 21 team of whatever, depending on what their 23 23 A. Mr. Vokey. Commissioner, on behalf of Terra 24 24 MR. VOKEY: 25 A. We generally have, and I think for all 25 and baggage handlers during helicopter 3 safety advisor to go to specific sites. 24 MR. VOKEY: 25 A. We generally have, and I think for all		narticular work and having responsibility for			are there so they are all encompassing
2 a Context, and a struction of a context, and context, a	0	a certain thing. Let's say to pull an example	0		themselves. We do however have a safety
10 Out of the strength of the leicopters, he's not 11 If "x" is refuelling helicopters, he's not 12 doing that all day long, or she. Do people do 13 multiple jobs? I presume they do, do they? 14 MR. PRITCHARD: 15 A. They do, yes. 16 COMMISSIONER: 17 Q. And they're trained in the multiple functions? 18 MR. PRITCHARD: 19 A. Indeed. They vary from installation to installation as to who you take on board as 21 team of whatever, depending on what their 23 other duties on board may be. 24 MR. VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 26 A. Mr. Vokey. Commissioner, on behalf of Terra 27 A. We are our vessel crews as part of the 2 helicopter landing officer, and 3 conturent tasks, while a helicopter is and baggage handlers during helicopter 4 hewould have a support team of firefighters 5 acrae operator 10 Q. Because I heard of that principle before that 11 a crane operator 12 A. That is fairly st	10	out of the air, the refuelling of heliconters	10		advisor on board who will monitor jobs and
11 In a link all ally long, or she. Do people do 13 multiple jobs? I presume they do, do they? 14 MR.PRTCHARD: 15 A. They do, yes. 16 COMMISSIONER: 17 Q. And they're trained in the multiple functions? 18 MR.PRTCHARD: 19 A. Indeed. They vary from installation to 10 oinstallation as to who you take on board as 21 your helicopter landing officer and refuelling 21 team or whatever, depending on what their 23 other duties on board may be. 24 AR. VOKEY: 25 A. Mr. Vokey. Commissioner, on behalf of Terra 26 helicopter landing, and the crane operator 3 would be the helicopter landing officer, and 4 he would have a support team of frengthers 5 and baggage handlers during helicopter 6 operations. So they wouldn't be doing 7 concurrent tasks while a helicopter is down. 8 That would be their primary task. 9 Q. He won't be operating the crane when a 16 OCMMISSIONER: 17 Q. On	11	If "x" is refuelling heliconters he's not	11		understand the operations at hand, and will go
Part of the stars in dry long to the poly object to multiple jobs? Part of the stars in dry long to the poly object to the stars in dry long to the poly object to the stars in dry long to the poly of the poly object to the stars in dry long to the poly object to the stars in dry long to the poly object to the stars in dry long to the poly object to the stars in dry long to the poly object to the stars in dry long to the poly object to the poly poly the poly the poly poly the poly the poly the poly the poly the poly poly the poly the poly the poly poly the poly the poly poly the poly poly the poly the poly the poly the poly the poly t	12	doing that all day long or she. Do people do	12		to the sites, the work sites to review the
 13 MR. PRITCHARD: 14 MR. PRITCHARD: 15 A. They do, yes. 16 COMMISSIONER: 17 Q. And they're trained in the multiple functions? 18 MR. PRITCHARD: 19 A. Indeed. They vary from installation to 10 installation as to who you take on board as 11 your helicopter landing officer and refuelling 12 team or whatever, depending on what their 13 d. That would be the helicopter landing officer, and 14 helicopter landing, and the crane operator 15 and baggage handlers during helicopter is 16 commissioner, on behalf of Terra 17 Nova, we use our vessel crews as part of the 18 helicopter landing, and the crane operator 20 operations. So they wouldn't be doing 21 commissionER: 22 commissionER: 3 concurrent tasks while a helicopter is 3 a. That would be their primary task. 3 COMMISSIONER: 4 COMMISSIONER: 5 de cause I heard of that principle before that 14 commissionER: 15 de won't be operating the crane when a 14 commissionER: 14 commissionER: 15 Nova, we use our vessel crews as part of the 16 helicopter is during helicopter 6 set sequence for doing everything. So the 10 conscurrent tasks while a helicopter is down. 14 commissionER: 15 a. That is fairly standard. 14 commissionER: 15 A. The crane needs to be actually in its cradle 16 between critical tasks, and helicopter 16 helicopter operations. 16 commissionER: 17 MR. PRITCHARD: 18 A. The crane driver becomes the helicopter 19 by procedure, so consequent it's a good fit 19 between critical tasks, and not so critical asks and not so critical tasks, and not so critical tasks and not so critical tasks	12	multiple jobs? I presume they do do they?	12		team physically working
 In Information (Information) In Commission (Information) A. They do, yes. A. They do, yes. GOMMISSIONER: A. Indeed. They vary from installation to installation as to who you take on board as your helicopter landing officer and refuelling to that would be the helicopter landing officer, and helicopter landing and the crane operator so usual be the helicopter landing officer, and to concurrent tasks while a helicopter is down. Commission (Information) Commission (Information)	113 14 M		14	COM	MISSIONED.
 10 So They only cell in the multiple functions? 11 G COMMISSIONER: 12 A. Indeed. They vary from installation to installation as to who you take on board as 21 your helicopter landing officer and refuelling 22 team or whatever, depending on what their 23 other duties on board may be. 24 MR, VOKFY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra Page 158 1 Nova, we use our vessel crews as part of the 2 helicopter landing, and the crane operator 3 would be the helicopter landing officer, and 4 he would have a support team of firefighters 4 me would be the helicopter landing officer, and 3 aggage handlers during helicopter is down. 8 That would be their primary task. 9 COMMISSIONER: 10 Q. Because I heard of that principle before that 1 a crane operator 1a helicopter 1a helicopter 1a helicopter 1a helicopter comes along? 17 MR. PRITCHARD: 18 A. Thet crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 10 w. The trane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 (D. He won't be operating the crane when a helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 anding officer for that very reason. 19 COMMISSIONER: 10 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 anding officer for that very reason. 10 COMMISSIONER: 11 A. The crane needs to be actually in its cradle 20 that the crane driver becomes the helicopter 21 anding officer for that very reason. 12 COMMISSIONER: 13 A. The crane helicopter so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 anding officer for that very reason. 14 COMMISSIONER: <	15	A They do yes	14		I see so the actual worker gets some support
10 COMMISSIONER: 10 India dra statistical tasks, and bargage handlers during helicopter 11 India dra statistical tasks, and bargage handlers during helicopter 12 A. Medeal, heave our vessel crews as part of the 13 Nova, we use our vessel crews as part of the 14 Nova, we use our vessel crews as part of the 15 and baggage handlers during helicopter 16 mean of firefighters 17 Nova, we use our vessel crews as part of the 18 helicopter landing officer, and 14 he would have a support team of firefighters 15 and baggage handlers during helicopter 16 operations. So they wouldn't be doing 10 concurrent tasks while a helicopter is down. 18 A. That is fairly standard. 11 a crane operator 12 MR. PRITCHARD: 13 A. That is fairly standard. 14 COMMISSIONER: 15 Q. One other question that occurred to me this 16 helicopter comes along? 17 MR. VOKEY: 12 MR. WOKEY: 13 A. The cra	15 16 C	OMMISSIONED	15	Q.	from the safety person?
 A. That is fairly standard. A. That is fairly standard. COMMISSIONER: Comments in the interpretation of the pretation of the pretation of the pretation of the pretation. A. That is fairly standard. ComMISSIONER: 	17	• And they're trained in the multiple functions?	17	MRI	PRITCHARD.
10 instruct-instruction 10 installation as to who you take on board as 20 installation as to who you take on board as 21 your helicopter landing officer and refuelling 24 me or whatever, depending on what their 23 other duties on board may be. 24 MR. VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 26 Nova, we use our vessel crews as part of the 21 helicopter landing, and the crane operator 3 would be the helicopter landing officer, and 4 he would have a support team of firefighters 5 and baggage handlers during helicopter 6 operations. So they wouldn't be doing 7 concurrent tasks while a helicopter is down. 8 That would be their primary task. 9 correction create 10 Q. Because I heard of that principle before that 11 a crane operator 12 MR. VOKEY: 13 A. That is fairly standard. 14 COMMISSIONER: 19 belicopter comes along? 17 MR. PRTICHARD: </td <td>18 M</td> <td>Q. The decy retrained in the induple functions:</td> <td>18</td> <td>Δ</td> <td>Correct</td>	18 M	Q. The decy retrained in the induple functions:	18	Δ	Correct
19 A Intext and your helicopter landing officer and refuelling 19 Commissioner, on whatever, depending on what their 21 your helicopter landing officer and refuelling 20 Q. When neccessary, and when selected by the 23 other duties on board may be. 21 MR. PRITCHARD: 24 MR. VOKEY: 25 A. Wr. Vokey, Commissioner, on behalf of Terra 26 24 MR. VOKEY: 25 A. We generally have, and I think for all 25 A. Mr. Vokey, commissioner, on behalf of Terra 26 A. We generally have, and I think for all 26 Nova, we use our vessel crews as part of the 1 installations, we have two types of tasks. We 2 hew ould be the helicopter landing officer, and 1 installations, we have two types of tasks. We 3 would be the helicopter is down. 5 set sequence for doing everything. So the 6 operations. 5 set sequence for doing everything. So the 7 conturrent tasks while a helicopter is down. 8 a rotitical task, and that means there's a 6 operations. 6 set sequence for doing everything. So the 7 conturrent tasks while a helicopter operations. 1 <td>10</td> <td>A Indeed They vary from installation to</td> <td>10</td> <td>COM</td> <td>MISSIONED.</td>	10	A Indeed They vary from installation to	10	COM	MISSIONED.
21 your helicopter landing officer and refuelling 21 WR. PRITCHARD: 22 team or whatever, depending on what their 21 MR. NOKEY: 23 A. Mr. Vokey; commissioner, on behalf of Terra 22 A. When necessary, and when selected by the 23 safety advisor to go to specific sites. 24 MR. VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 25 A. We generally have, and I think for all Page 158 1 Nova, we use our vessel crews as part of the 1 installations, we have two types of tasks. We 2 helicopter landing officer, and 3 critical tasks or critical tasks, and that means there's a 3 would be the helicopter team of firefighters 4 Something like refuelling a helicopter would 5 and baggage handlers during helicopter 6 set sequence for doing everything. So the 7 concurrent tasks while a helicopter is down. 8 a routine task. In addition to that, 9 corkmissioner 9 everybody that's on the helideck or supports 10 Q. Because I heard of that principle before that 1 landings or takeoff, they actually undergo 12	20	installation as to who you take on board as	20		When necessary
21 team or whate lanching which and leftering 22 team or whate lanching on what their 23 other duties on board may be. 24 MR. VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 26 Page 158 1 Page 158 1 Page 158 2 helicopter landing, and the crane operator 3 would be the belicopter landing officer, and 4 he would have a support team of firefighters 5 and baggage handlers during helicopter 6 operations. So they wouldn't be doing 7 concurrent tasks while a helicopter is down. 8 That would be their primary task. 9 concurrent tasks while a helicopter is down. 8 a croutical task, and that means there's a 9 concurrent tasks while a helicopter is down. 10 Q. Because I heard of that principle before that 11 a crane operator 12 MR. VOKEY: 13 A. That is fairly standard. 14 COMMISSIONER: 15 A. The crane needs to be actually in its cradle	20	your helicopter landing officer and refuelling	20	V.	
23 other duties on board may be. 24 A. Wrich intersally, and wich selected by the	$\begin{vmatrix} 21 \\ 22 \end{vmatrix}$	team or whatever depending on what their	$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	MIX. 1	When necessary and when selected by the
 24 MR, VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 24 MR, VOKEY: 25 A. Mr. Vokey, Commissioner, on behalf of Terra 24 MR, VOKEY: 25 A. We generally have, and I think for all Page 158 Page 158 Page 160 1 installations, we have two types of tasks. We 2 helicopter landing, and the crane operator 3 would be the helicopter tam of firefighters 4 he would have a support team of firefighters 5 and baggage handlers during helicopter 6 operations. So they wouldn't be doing 7 concurrent tasks while a helicopter is down. 8 That would be their primary task. 9 COMMISSIONER: 10 Q. Because I heard of that principle before that 11 a crane operator 12 MR. VOKEY: 13 A. That is fairly standard. 14 COMMISSIONER: 14 COMMISSIONER: 14 COMMISSIONER: 15 Q. He won't be operating the crane when a 16 helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 10 that the crane driver becomes the helicopter 11 landing officer for that very reason. 12 COMMISSIONER: 13 A. The crane needs to be actually in its cradle 14 Dy procedure, so consequent it's a good fit 15 by procedure, so consequent it's a good fit 16 COMMISSIONER: 12 COMMISSIONER: 13 A. The crane needs to be actually in its cradle 14 So they are trained specifically in how to 15 Community. When tasks like that, fairly routine 14 So they are trained specifically in how to 15 consequent it's a good fit 14 So they are trained specifically in the crane driver becomes the helicopter 14 So they are trained specifically one-third of all our t	22	other duties on board may be	22	А.	safety advisor to go to specific sites
25 A. Mr. Vokey, Commissioner, on behalf of Terra 2 A. We generally have, and I think for all 21 Page 158 Page 160 1 Nova, we use our vessel crews as part of the 2 have routine type of tasks and we have 2 helicopter landing, and the crane operator 3 critical tasks or critical procedures. 4 he would have a support team of firefighters 5 and baggage handlers during helicopter 5 and baggage handlers during helicopter 5 be a critical tasks or critical procedures. 6 operations. So they wouldn't be doing 6 set sequence for doing everything. So the 7 concurrent tasks while a helicopter is down. 7 knowledge required is different than if it was 8 That would be their primary task. 9 everybody that's on the helideck or supports 10 Q. Because I heard of that principle before that 10 those type of operations during helicopter 12 MR. VOKEY: 12 every body that's on the helideck or supports 10 D. Because I heard of that principle before that 10 those type of operations during helicopter 13 A. That is fairly standard. 13 training course,	23 24 M	P VOKEV	23	MR	VOKEV.
Page 158Page 1581Nova, we use our vessel crews as part of the12helicopter landing, and the crane operator23would be the helicopter landing officer, and34he would have a support team of firefighters35and baggage handlers during helicopter56operations. So they wouldn't be doing57concurrent tasks while a helicopter is down.8That would be their primary task.9COMMISSIONER:10Q. Because I heard of that principle before that11a crane operator12MR. VOKEY:13A. That is fairly standard.14COMMISSIONER:15Q. He won't be operating the crane when a16helicopter comes along?17MR. PRTCHARD:18A. The crane needs to be actually in its cradle19by procedure, so consequent it's a good fit10that the crane driver becomes the helicopter11landing officer for that very reason.12COMMISSIONER:13that the crane driver becomes the helicopter14So they are trained specifically in how to15handle and refuel helicopters.16COMMISSIONER:17Q. One other question that occurred to me this24morning, when tasks like that, fairly routine25tasks, but tasks nevertheless, have some	25	A Mr Vokey Commissioner on behalf of Terra	25	Δ	We generally have and I think for all
Page 158Page 1601Nova, we use our vessel crews as part of theinstallations, we have two types of tasks. We2helicopter landing, and the crane operatorinstallations, we have two types of tasks. We3would be the helicopter landing officer, andcritical tasks or critical procedures.4he would have a support team of firefightersSomething like refuelling a helicopter would5and baggage handlers during helicopterbe a critical task, and that means there's a6operations. So they wouldn't be doing67concurrent tasks while a helicopter is down.78That would be their primary task.89conditions that principle before that1010Q. Because I heard of that principle before that1011a crane operator1112MR. VOKEY:1213A. That is fairly standard.1314COMMISSIONER:1415A. The crane needs to be actually in its cradle1516helicopter comes along?1617MR. PRITCHARD:1718A. The crane needs to be actually in its cradle1919by procedure, so consequent it's a good fit1920that the crane driver becomes the helicopter2021landing officer for that very reason.2122COMMISSIONER:2123Q. One other question that occurred to me this2324morning, when tasks like that, fairly routine24 <t< th=""><th>23</th><th>A. MI. Vokey, Commissioner, on benan of ferra</th><th>1.50</th><th>71.</th><th></th></t<>	23	A. MI. Vokey, Commissioner, on benan of ferra	1.50	71.	
1Nova, we use our vessel crews as part of the2helicopter landing, and the crane operator3would be the helicopter landing officer, and4he would have a support team of firefighters5and baggage handlers during helicopter6operations. So they wouldn't be doing7concurrent tasks while a helicopter is down.8That would be their primary task.9COMMISSIONER:10Q. Because I heard of that principle before that11a crane operator12MR. VOKEY:13A. That is fairly standard.14COMMISSIONER:15Q. He won't be operating the crane when a16helicopter comes along?17MR. PRITCHARD:18A. The crane needs to be actually in its cradle19by procedure, so consequent it's a good fit19by procedure, so consequent it's a good fit10that the crane driver becomes the helicopter21landing officer for that very reason.22COMMISSIONER:23Q. One other question that occurred to me this24morning, when tasks like that, fairly routine25morning, when tasks like that, fairly routine24morning, when tasks like that, fairly routine25tasks, but tasks nevertheless, have some		Page	158		Page 160
2netrocopter landing, and the crane operator2nave routine type of tasks and we have3would be the helicopter landing officer, and3critical tasks or critical procedures.4he would have a support team of firefighters3critical tasks or critical procedures.5and baggage handlers during helicopter5be a critical tasks, and that means there's a6operations. So they wouldn't be doing5be a critical task, and that means there's a7concurrent tasks while a helicopter is down.7knowledge required is different than if it was8That would be their primary task.9everybody that's on the helideck or supports10Q. Because I heard of that principle before that11landings or takeoff, they actually undergo11a crane operator11landings or takeoff, they actually undergo12Q. Mew on't be operating the crane when a13training course, for helicopter operations.14COMMISSIONER:14So they are trained specifically in how to15Q. He won't be operating the crane when a16COMMISSIONER:16D. He won't be operating the crane when a16COMMISSIONER:17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle1919by procedure, so consequent it's a good fit1020that the crane driver becomes the helicopter2021landing officer for that very reason.2122COMMISSIONER: <td></td> <td>Nova, we use our vessel crews as part of the</td> <td></td> <td></td> <td>installations, we have two types of tasks. we</td>		Nova, we use our vessel crews as part of the			installations, we have two types of tasks. we
3Would be the helicopter landing officer, and 43Crucial tasks of crucial tasks of crucial procedures.4he would have a support team of firefighters 53Something like refuelling a helicopter would5and baggage handlers during helicopter 65be a critical tasks of crucial tasks of that means there's a 66operations. So they wouldn't be doing 7so they endities tasks of that, and that means there's a 87operations. So they wouldn't be doing 8realize that adistic to that, 88a routine tasks. In addition to that, 9every three year training, which is a two day 1311a crane operator 111112every three year trained specifically in how to 1515Q. He won't be operating the crane when a 1	$ ^2$	nencopter landing, and the crane operator			nave routine type of tasks and we have
4Sometiming incerentering a helicopter would5and baggage handlers during helicopter6operations. So they wouldn't be doing7concurrent tasks while a helicopter is down.8That would be their primary task.9COMMISSIONER:10Q. Because I heard of that principle before that11a crane operator12MR. VOKEY:13A. That is fairly standard.14COMMISSIONER:15Q. He won't be operating the crane when a16helicopter comes along?17MR. PRITCHARD:18A. The crane needs to be actually in its cradle19by procedure, so consequent it's a good fit10that the crane driver becomes the helicopter21landing officer for that very reason.22COMMISSIONER:23Q. One other question that occurred to me this24morning, when tasks like that, fairly routine25morning, when tasks like that, fairly routine25means there's have some	3	would be the helicopter landing officer, and	3		Critical tasks of critical procedures.
5and baggage handlers during helicopter6operations. So they wouldn't be doing7concurrent tasks while a helicopter is down.8That would be their primary task.9COMMISSIONER:10Q. Because I heard of that principle before that11a crane operator12MR. VOKEY:13A. That is fairly standard.14COMMISSIONER:15Q. He won't be operating the crane when a16helicopter comes along?17MR. PRITCHARD:18A. The crane needs to be actually in its cradle19by procedure, so consequent it's a good fit10that the crane driver becomes the helicopter11landing officer for that very reason.22COMMISSIONER:23Q. One other question that occurred to me this24morning, when tasks like that, fairly routine25tasks, but tasks nevertheless, have some	4	ne would nave a support team of firefighters	4		Something like retuelling a helicopter would
6Operations. So they wouldn't be doing concurrent tasks while a helicopter is down.6Set sequence for doing everything. So the7concurrent tasks while a helicopter is down.7knowledge required is different than if it was8That would be their primary task.8a routine task. In addition to that,9COMMISSIONER:9everybody that's on the helideck or supports10Q. Because I heard of that principle before that10those type of operations during helicopter11a crane operator11landings or takeoff, they actually undergo12MR. VOKEY:12every three year training, which is a two day13A. That is fairly standard.13training course, for helicopter operations.14COMMISSIONER:14So they are trained specifically in how to15Q. He won't be operating the crane when a16COMMISSIONER:16helicopter comes along?17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle19between critical tasks and not so critical19by procedure, so consequent it's a good fit19between critical tasks.21landing officer for that very reason.21MR. VOKEY:23Q. One other question that occurred to me this24norning, when tasks like that, fairly routine24morning, when tasks like that, fairly routine25means there has to be a set sequence and	5	and baggage nandlers during neucopter	5		be a critical task, and that means there s a
 7 Concurrent tasks while a helicopter is down. 8 That would be their primary task. 9 COMMISSIONER: 10 Q. Because I heard of that principle before that 11 a crane operator 12 MR. VOKEY: 13 A. That is fairly standard. 14 COMMISSIONER: 15 Q. He won't be operating the crane when a 16 helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 10 those type of operations during helicopter 11 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 7 MR. VOKEY: 7 MR. PRITCHARD: 8 A. The crane needs to be actually in its cradle 19 between critical tasks, and by critical, it 20 Comming, when tasks like that, fairly routine 21 tasks, but tasks nevertheless, have some 22 Comming, when tasks like that, fairly routine 23 Comming, when tasks like that, fairly routine 24 consider critical tasks, and by critical, it 25 tasks, but tasks nevertheless, have some 	6	operations. So they wouldn't be doing	6		set sequence for doing everything. So the
8Inat would be their primary task.8a routhe task.In addition to that,9COMMISSIONER:9everybody that's on the helideck or supports10Q. Because I heard of that principle before that10those type of operations during helicopter11a crane operator11landings or takeoff, they actually undergo12MR. VOKEY:12every three year training, which is a two day13A. That is fairly standard.13training course, for helicopter operations.14COMMISSIONER:14So they are trained specifically in how to15Q. He won't be operating the crane when a15handle and refuel helicopters.16helicopter comes along?16COMMISSIONER:17MR. PRITCHARD:17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle19between critical tasks and not so critical19by procedure, so consequent it's a good fit19between critical tasks and not so critical20that the crane driver becomes the helicopter20tasks, routine tasks.21landing officer for that very reason.21MR. VOKEY:22Q. One other question that occurred to me this23Q. One other question that occurred to me this24morning, when tasks like that, fairly routine24consider critical tasks, and by critical, it25means there has to be a set sequence and25	7	concurrent tasks while a helicopter is down.	17		knowledge required is different than if it was
9COMMISSIONER:9everybody that s on the helideck of supports10Q. Because I heard of that principle before that10those type of operations during helicopter11a crane operator11landings or takeoff, they actually undergo12MR. VOKEY:12every three year training, which is a two day13A. That is fairly standard.13training course, for helicopter operations.14COMMISSIONER:14So they are trained specifically in how to15Q. He won't be operating the crane when a15handle and refuel helicopters.16helicopter comes along?16COMMISSIONER:17MR. PRITCHARD:17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle19between critical tasks and not so critical19by procedure, so consequent it's a good fit19between critical tasks and not so critical20that the crane driver becomes the helicopter20tasks, routine tasks.21landing officer for that very reason.21MR. VOKEY:22Q. One other question that occurred to me this23approximately one-third of all our tasks we24morning, when tasks like that, fairly routine24consider critical tasks, and by critical, it25means there has to be a set sequence and25	8	That would be their primary task.	8		a routine task. In addition to that,
10Q. Because I heard of that principle before that10those type of operations during hencopter11a crane operator11landings or takeoff, they actually undergo12MR. VOKEY:12every three year training, which is a two day13A. That is fairly standard.13training course, for helicopter operations.14COMMISSIONER:14So they are trained specifically in how to15Q. He won't be operating the crane when a15handle and refuel helicopters.16helicopter comes along?16COMMISSIONER:17MR. PRITCHARD:17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle19between critical tasks and not so critical19by procedure, so consequent it's a good fit19between critical tasks and not so critical20that the crane driver becomes the helicopter21Ianding officer for that very reason.21landing officer for that very reason.21MR. VOKEY:22Q. One other question that occurred to me this23approximately one-third of all our tasks we24morning, when tasks like that, fairly routine24consider critical tasks, and by critical, it25tasks, but tasks nevertheless, have some25means there has to be a set sequence and	9 00	OMMISSIONER:	9		everybody that s on the helideck of supports
11 <td></td> <td>Q. Because Theard of that principle before that</td> <td>10</td> <td></td> <td>landings on tale off, they actually undergo</td>		Q. Because Theard of that principle before that	10		landings on tale off, they actually undergo
 12 MR. VOKEY: 13 A. That is fairly standard. 14 COMMISSIONER: 15 Q. He won't be operating the crane when a helicopter comes along? 16 helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 12 between critical tasks, and by critical, it 26 consider critical tasks, and by critical, it 		a crane operator			landings of takeoii, they actually undergo
 13 A. That is fairly standard. 14 COMMISSIONER: 15 Q. He won't be operating the crane when a helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 13 Taihing course, for inercopier operations. 14 So they are trained specifically in how to 15 handle and refuel helicopters. 16 COMMISSIONER: 17 Q. Okay, thank you for that, because I didn't 18 realize that a distinction was made like that 19 between critical tasks and not so critical 20 tasks, routine tasks. 21 MR. VOKEY: 22 A. I can speak on behalf of Terra Nova, 23 approximately one-third of all our tasks we 24 consider critical tasks, and by critical, it 25 means there has to be a set sequence and 	12 M	IK. VOKET:	12		training course, for balicenter, operations
141415So they are trained specifically in now to15Q. He won't be operating the crane when a16handle and refuel helicopters.16helicopter comes along?16COMMISSIONER:17MR. PRITCHARD:17Q. Okay, thank you for that, because I didn't18A. The crane needs to be actually in its cradle17Q. Okay, thank you for that, because I didn't19by procedure, so consequent it's a good fit19between critical tasks and not so critical20that the crane driver becomes the helicopter20tasks, routine tasks.21landing officer for that very reason.21MR. VOKEY:22COMMISSIONER:22A. I can speak on behalf of Terra Nova,23Q. One other question that occurred to me this23approximately one-third of all our tasks we24morning, when tasks like that, fairly routine24consider critical tasks, and by critical, it25tasks, but tasks nevertheless, have some25means there has to be a set sequence and		A. That is failing standard.	13		So they are trained specifically in how to
 15 G. The won't be operating the craft when a helicopter comes along? 16 helicopter comes along? 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 16 COMMISSIONER: 16 COMMISSIONER: 17 Q. Okay, thank you for that, because I didn't 18 realize that a distinction was made like that 19 between critical tasks and not so critical 20 tasks, routine tasks. 21 MR. VOKEY: 22 A. I can speak on behalf of Terra Nova, 23 approximately one-third of all our tasks we 24 consider critical tasks, and by critical, it 25 means there has to be a set sequence and 	14 00	O He won't be operating the grane when a	14		handle and refuel belicopters
 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 10 CommissionEX. 17 MR. PRITCHARD: 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 19 between critical tasks and not so critical 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 10 CommissionEX. 17 Q. Okay, thank you for that, because I didn't 18 realize that a distinction was made like that 19 between critical tasks and not so critical 20 tasks, routine tasks. 21 MR. VOKEY: 22 A. I can speak on behalf of Terra Nova, 23 approximately one-third of all our tasks we 24 consider critical tasks, and by critical, it 25 means there has to be a set sequence and 	15	Q. The word to be operating the crane when a helicopter comes along?	15	COM	
 17 G. Okay, thank you for that, because F than t 18 A. The crane needs to be actually in its cradle 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 to kay, thank you for that, because F than t 27 that the crane needs to be actually in its cradle 28 that the crane driver becomes the helicopter 29 tasks, routine tasks. 20 tasks, routine tasks. 21 MR. VOKEY: 22 A. I can speak on behalf of Terra Nova, 23 approximately one-third of all our tasks we 24 consider critical tasks, and by critical, it 25 means there has to be a set sequence and 	17 M		17		Okay thank you for that because I didn't
 19 by procedure, so consequent it's a good fit 19 by procedure, so consequent it's a good fit 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 tasks in the transmission of transmission of transmission of the transmission of transmission of transmission of the transmission of transmissi	18	A The crane needs to be actually in its cradle	18	Q.	realize that a distinction was made like that
 20 that the crane driver becomes the helicopter 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 tasks but tasks nevertheless. 27 tasks but tasks nevertheless. 28 tasks but tasks nevertheless. 29 tasks but tasks nevertheless. 20 tasks, routine tasks and not so erntedinates and no	19	hy procedure so consequent it's a good fit	10		between critical tasks and not so critical
 21 landing officer for that very reason. 22 COMMISSIONER: 23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 tasks, routine tasks. 21 MR. VOKEY: 22 A. I can speak on behalf of Terra Nova, 23 approximately one-third of all our tasks we 24 consider critical tasks, and by critical, it 25 means there has to be a set sequence and 	20	that the crane driver becomes the beliconter	20		tasks routine tasks
22 COMMISSIONER:22A. I can speak on behalf of Terra Nova,23Q. One other question that occurred to me this23approximately one-third of all our tasks we24morning, when tasks like that, fairly routine24consider critical tasks, and by critical, it25tasks, but tasks nevertheless, have some25means there has to be a set sequence and	$\begin{bmatrix} 20\\ 21 \end{bmatrix}$	landing officer for that very reason	20	MR V	VOKEY.
23 Q. One other question that occurred to me this 24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 morning approximately one-third of all our tasks we 27 marticle approximately one-third of all our tasks we 28 morning approximately one-third of all our tasks we 29 morning approximately one-third of all our tasks we 20 morning approximately one-third of all our tasks we 20 morning approximately one-third of all our tasks we 21 means there has to be a set sequence and	22 0	OMMISSIONER.	21	Δ	I can speak on behalf of Terra Nova
24 morning, when tasks like that, fairly routine 25 tasks, but tasks nevertheless, have some 26 morning approximately one unit of an our tasks we 27 morning when tasks like that, fairly routine 28 morning approximately one unit of an our tasks we 29 morning approximately one unit of an our tasks we 20 morning approximately one unit of an our tasks we 20 morning approximately one unit of an our tasks we 21 morning approximately one unit of an our tasks we 22 morning approximately one unit of an our tasks we 23 morning approximately one unit of an our tasks we 24 morning approximately one unit of an our tasks we 25 morning approximately one unit of an our tasks we 26 morning approximately one unit of an our tasks we	$\begin{bmatrix} 22 \\ 23 \end{bmatrix}$	0. One other question that occurred to me this	23	11.	approximately one-third of all our tasks we
tasks, but tasks nevertheless, have some 25 means there has to be a set sequence and	24	morning, when tasks like that fairly routine	23		consider critical tasks and by critical it
	25	tasks, but tasks nevertheless, have some	25		means there has to be a set sequence and

January 11, 2010	Multi-Page	Offshore Helicopter Safety Inquiry	
F	Page 161	Page 163	
1 protocol of how that job is done.		not it was a valid right to refuse dangerous	
2 COMMISSIONER:	2	work or whether it wasn't. At that point, if	
3 Q. Does that apply to the your installations a	s 3	it's not satisfied to the satisfaction of the	
4 well?	4	individual or to any kind of the JOHS	
5 MR. PRITCHARD:	5	Committee, it would then get referred to the	
6 A. We have routine tasks, what's considered	low 6	Board. At that point in time, the Board would	
7 risk tasks, and some critical tasks.	7	normally involve two of their safety officers	
8 MR. SACUTA:	8	to do an investigation. As I mentioned, we	
9 A. We have what's considered integrity critic	al, 9	had three refusals shortly after Mr. Decker's	
10 and in those situations we have specific	10	testimony. Two were associated with the	
11 procedures that have to be followed step	by 11	auxiliary fuel tank, and one was associated	
12 step and initialled off as you complete eac	ch 12	with the cracks on the main gear box mounting	
13 step to identify the task is a critical task	13	feet, which was a known issue with Sikorsky.	
14 and you need to follow the procedures step	by 14	So the Board investigated all three of those	
15 step as part of that process.	15	and when the Board does their investigation,	
16 COMMISSIONER:	16	they interview the individual to make sure	
17 Q. Okay, thank you. Mr. Roil.	17	they understand exactly what the individual's	
18 ROIL, Q.C.:	18	issues are. They interview the JOHS Committee	
19 Q. Thank you. Thank you, Mr. Pritchard, and	l Mr . 19	to make sure that they understand the	
20 Sacuta, I believe that you had something y	you 20	discussion that was held of the JOHS Committee	
21 wanted to clarify from what you said thi	is 21	and the opinion of the JOHS Committee. They	
22 morning.	22	interview the individual's supervisor. They	
23 MR. SACUTA:	23	also gather information from an other sources.	
24 A. Yes. This morning we talked a little bit	24	In these three situations, they gathered	
about the three refusals that we had on	25	information from Cougar, they gathered	
F	Page 162	Page 164	
1 Hibernia, and although I intended to discu	iss 1	information from Sikorsky through Cougar on	
2 them tomorrow as part of the Helicopte	er 2	all of these issues. In all three cases, the	
3 Operation Task Force section, I thought it	was 3	Board issued a very detailed summary report	
4 appropriate that maybe I give a little summ	nary 4	with the process and the steps that they took	
5 on those three refusals.	5	to evaluate whether or not it was a valid	
6 COMMISSIONER:	6	right to refuse dangerous work, and in all	
7 Q. By all means, yes.	7	three cases they indicated that the auxiliary	
8 MR. SACUTA:	8	fuel tank and the main gear box feet cracks	
9 A. Yes, the process that you follow when any	body 9	did not result in an unacceptable increase in	
10 refuses work is that the individual raises	10	risk for helicopter transportation. That was	
11 concern with his immediate supervisor. T	hey 11	their opinion. After that ruling was provided	
12 have a discussion and they attempt to resol	lve 12	by the Board, all three individuals then	
13 the issue. If it's not resolved to the	13	subsequently transited offshore by helicopter	
14 employee's satisfaction, he then has the	e 14	and the item was considered closed based on	
15 option to take it to the JOHS Committee. He	e's 15	that fact. It went through the entire	
16 supposed to discuss it with the safety rep a	ind 16	process, the Board issued its opinion or its	
17 it would go to the JOHS Committee. At th	nat 17	decision, and then in this case the three	
18 point in the process, the offshore	18	individuals have now returned to their normal	
19 installation manager must contact the on d	uty 19	helicopter operations.	
20 board safety officer to let them know that	we 20 CO	MMISSIONER:	
21 have a right to refuse dangerous work bei	ing 21	Q. Okay, thank you.	
22 reviewed and it's going to the JOHS Comm	ittee. 22 RO	IL, Q.C.:	
23 The individual would then present his case	e to 23	Q. Unless there are any other follow up items	
24 the JOHS Committee and the JOHS Comm	nittee 24	from this morning, Mr. Vokey, I think you had	
25 would evaluate and try to determine wheth	er or 25	taken us to approximately 59, slide 59.	

January 11, 2010		i-Pag	e TM	^M Offshore Helicopter Safety Inquiry		
	Page 165			Page 167		
1 MR. VOKEY:	6	1		installations and vessels in the field.		
2 A. Yes.		2 R0	DIL,	Q.C.:		
3 ROIL, Q.C.:		3	Q.	Okay. Mr. Vokey, you slipped there from the		
4 Q. And we are now go	oing to focus a little more on	4	-	word "should" to the word "must", and that was		
5 helicopter operation	ns.	5		leading to my question, and I think we went on		
6 MR. VOKEY:		6		this this morning, but I just want to be		
7 A. That's correct, sir.	The next heading is	7		clear. These "shoulds", I take it become		
8 helicopter services	or helicopter operations.	8		"musts" by virtue of the fact that these are		
9 In this section of	the presentation, I'll	9		conditions for your work authorizations?		
10 speak in a little mo	re detail about helicopter	10 M	R. V	OKEY:		
11 operations, includin	ng the following; the oil	11	А.	That's correct. If it is included there, the		
12 and gas related reg	gulatory references, the	12		"shoulds" in essence for us would become a		
13 helicopter operation	ns, the general types of	13		"must". There are there are some isolated		
14 services provided,	the selection of Cougar as	14		instances, and I'll give you an example.		
15 the helicopter servi	ce provider by each of the	15		Where do I find it here; be equipped with		
16 operators, the activ	ity that Cougar led in the	16		externally mounted life rafts. There are		
17 selection of the S	-92 airframe, and the	17		helicopters that we have used here, the		
18 decision process by	each of the operators to	18		Sikorsky S-61, for example, does not have		
19 move to using the S	S-92 airframe for respective	19		externally mounted life rafts, but it does		
20 operations, and som	ne of the key features of	20		have internally mounted life rafts right at		
21 the S-92 which ma	de it an appropriate airframe	21		the main entrances and exits from the		
22 for use in our opera	ting environment. In his	22		helicopter. So in a case like that, if the		
23 testimony last fall	, Mr. Pike, the Chief	23		operator can demonstrate an equivalency, they		
24 Safety Officer wit	h the Board, provided	24		may be granted that equivalent standard, but		
25 excerpts from seve	eral places in the Board's	25		for the most part, where it says "should" it's		
	Page 166			Page 168		
1 guidelines where he	elicopter operations are	1		in essence a "must".		
2 referenced. The	guidelines respecting	2 R0	OIL,	Q.C.:		
3 drilling operations	contain specific	3	Q.	And you don't change from "must" to a		
4 expectations. All op	perators must meet these	4		"should", I take it, unless you have approval		
5 expectations in orde	er to hold a valid work	5		from the C-NLOPB to make that deviation?		
6 authorization. All	operators are in	6 M	R. V(OKEY:		
7 compliance with the	ese expectations. This	7	A.	Yeah. The other way is a lot more difficult		
8 slide and the next v	will provide a list of	8		to go.		
9 specific references f	or helicopter operations.	9 R(OIL,	Q.C.:		
10 I'm just taking a loo	k at some of them there.	10	Q.	Yeah, because I think the word "should" sounds		
Helicopter operation	is or helicopters must be	11		to most of those of us that use English in its		
12 certified by Transpo	rt Canada, the pilots must	12		ordinary parlance, it s, on, well, you know,		
13 be licensed by Trans	sport Canada, training and	13		you try to do it nine times out of ten, that		
14 experience of crev	vs and first response	14	D I <i>U</i>	sort of thing.		
15 tecnnicians are requ	nred, and provision of	15 M	K. V(UKEY: No for the most part, where it cave "should"		
16 Inght time for first	an to tally in general	10	A.	Ino, for the most part, where it says should ,		
1/ diffis. It also goes (should have. They should	10		demonstrate on equivalency		
10 about what all charts	a designs be capable of		יזר			
19 have multiple elign	r in at least moderate see	19 KU	JIL,	Q.C.: Okay		
20 idluing off the water	mai reast mouerate sea	20	ע. סיז ס	OKAY. OKEV.		
22 externally mounted	life rafts and be	$\begin{vmatrix} 2 & 1 & 1 \\ 2 & 2 & 2 \\ 2 & 2 & 2 \\ 2 & 2 & 2 \\ 2 & 2 &$	л. V Д	So the next slide I haliava is slide 62 in		
22 configured to allow	w emergency egress of	$\begin{vmatrix} 22\\ 23 \end{vmatrix}$	А.	the pack. It talks about some other things:		
24 passengers In addit	tion they must be able to	2.5		nassengers must receive helicopter underwater		
25 communicate with	shore-based and other	25		escape training suitably briefed prior to		

Page 165 - Page 168

January 11, 2010 Mult		Multi-P	Pa	age	Offshore Helicopter Safety Inquiry
		Page 169			Page 171
1	transport, don't normally carry freight a	nd 1	1		egress issue, or in the event of a helicopter
$ ^2$	passengers in the same aircraft or in the s	ame 2	2		impact, something becoming dislodged and
3	compartment, flying at night should be a	voided 3	3		becoming more of a projectile, but in recent
4	to the extent possible, reserve helicopte	r 4	4		times none of the operators, to my knowledge,
5	fuel to be kept in the field, consideration	1 5	5		and I ve gotten that from Cougar, with the new
6	given to providing goggles and appropri	iate e	6		airframes, have carried both cargo and
7	breathing devices to assist underwater es	cape,	7		passengers in the same.
8	and proven automated usage and moni	toring 8	8	ROIL,	
9	systems should be used where practicable	».	9	Q.	Does that mean they can't take a mail bag with
10	ROIL, Q.C.:		0		some mail in it or are we talking
	Q. Again I just want to slow us down a little		1	MR. V	OKEY:
12	and focus on some of these because ther	e are 12	2	А.	No, that would be in the cargo compartment.
13	some words like normally and should	d be 13	3		I in taiking specifically in the passenger
14	avoided, that I want to focus on. First (4	роц	compariment.
15	an, an passengers must receive HUE		5 6	ROIL,	Q.C.: Otrav so the sinfrome does have the conscitu
10	channing. I think we ve got a lot of eviden	ice le	6 7	Q.	to have a portion
$ _{10}^{1/}$	time on that Descenders must be suited		/		to have a portion
18	hrisfed prior to transport and wear appre	ne is	ð	MR. V	OKEY:
19	balicenter transportation suits We know		9	A.	Fol luggage.
$ _{21}^{20}$	about the transportation safety suite. Wh	Jw 20	1	KUIL,	Q.C.: For luggage
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	doos the briefing take place?		ו ר	Q.	TOT luggage.
$\begin{vmatrix} 22 \\ 22 \end{vmatrix}$	MB VOKEY:	22	2	MR. V	Ear luggage and that type of cargo, yes
23	MR. VORE I: A The briefing for baliconter transport we		э 1	A.	o C.
24	A. The bhering for hencopter transport wo	ulu 24	4 5	KUIL,	Q.C.: Consideration of weather and heliconter loads
25	take place at Cougar's facilities out adjac	D 170	5	Q.	consideration of weather and hencopter loads
		Page 1/0			Page 1/2
	to the airport.		1		when planning flights, what does that mean?
$\begin{bmatrix} 2 \\ 2 \end{bmatrix}$	ROIL, Q.C.:		2	MR. V	UKEY: I think when the question was asked this
	Q. Okay, and we in hear from them specification of the details of thet. Descengers and	iny as 3	3	A.	I think when the question was asked this
	freight should not normally be corried on	the 4	4		morning, Mr. Sacuta said, you know, our
3	asma aircraft. I don't know what "norm		э с		weather potential and the hersphase of it. In
	manna Doog that maan you can do it or		0		terms ofin terms of the weather and the
	weak once a month once a year? Expl	in to	/		lead limits any belicenter that's going
	the Commissioner and to those of us in	the for	0 0		offebore must carry arough fuel to not only
10	room what how that's interpreted and l	100	9 0		get to the offshore location, but it needs to
	that worked and applied over the past nu	iow s it	1		anticipate what the maximum head winds would
$ _{12}^{11}$	of years?		י ר		be for an example. They would also need to
12	MP VOKEV.	12	2 2		have enough fuel to hold a holding pattern
11	A Okay On occasion in the past and I'll sr	eak 1/	5 Л		and in the event that they weren't able to
15	in more recent years in a minute but in t	be 15	- -		land offshore that they would have enough
15	nast depending on the type of aircraft y	Ne 16	5		fuel to come back not only to St. John's but
17	did have latitude in terms of physical		7		its closest alternate and that could be Long
18	dimensions the size of doors and what	$\frac{1}{15}$	' 8		Pond it could be Gander it could be any
19	there was an ability to carry equipment	and 19	9		other known area where you're known to have
$ _{20}^{1}$	people concurrently. Since we've had the	$rac{1}{2}$	0		adequate weather
$\frac{20}{21}$	92s. I know of no instance where we've	carried 21	1	ROIL	0.C.:
22	both equipment and people in the sa	me 21	2	0	So the business of how the helicopter works on
23	compartment, and the reason the Roard	would 22	-3	ب	a daily basis has to take into consideration
$ _{24}^{-3}$	say it shouldn't normally be carried the	e. 22	4		the weather that is prevailing at that time?
25	because it does have the potential of bein	$\frac{1}{2}$ g an $\frac{1}{2}$	5	MR. V	OKEY:

Jar	nuary 11, 2010	Multi-P	'age ^{TT}	^M Offshore Helicopter Safety Inquiry
		Page 173		Page 175
1	A. That's correct, sir.			Gander doesn't have any limitations, your
2	ROIL, O.C.:	2	2	weather to and from offshore in terms of
3	O. Flying at night should be avoided to the	3	3	winds, sea states, is adequate, and you don't
4	extent possible. So there's two wiggle room	1 4	Ļ	reasonably have a good chance of mitigating
5	places there. What does that mean practically	v 5	5	your backlog over the next day or so because
6	in terms of how in the world of	6	ó	of weather forecast. So in cases like that,
7	Newfoundland we have now. I come to wor	k and 7	,	the operator will elect to fly at night.
8	it's fairly dark and I go home and it's fairly	8	ROIL	. O.C.:
9	dark. What's a night flight and when do you	. 9	0.	At this time of the year on the coast of
10	fly in the night, and again all of your	10)	Newfoundland, what number of flights or
11	answers here obviously can be ultimately ask	xed 11		percentage of flights, or how much flying
12	of Cougar, who are perhaps the better people	. 12	2	happens either in the early morning before
13	but I need to know what your understanding	is 13	;	it's light, or in the late afternoon, early
14	of what that caveat means to you?	14	L	evening, when it's dark?
15	MR. VOKEY:	15	MR.	VOKEY:
16	A. For us, a night flight is any night that	16	бА.	I don't have the exact numbers, but I know
17	either starts or terminates in darkness. So	17	,	statistically you're probably looking in the 5
18	it doesn't have to be fully in darkness, but	18	3	percent.
19	if a helicopter, say, leaves St. John's this	19	ROIL	, Q.C.:
20	time of the year at 3 o'clock, gets to the	20) Q.	As opposed to 25 or 30 percent?
21	offshore installation at 4:30, it's still	21	MR.	VOKEY:
22	daylight.	22	2 A.	Yes.
23	ROIL, Q.C.:	23	ROIL	2, Q.C.:
24	Q. This is in the afternoon we're talking?	24	4 Q.	Yes, okay.
25	MR. VOKEY:	25	MR.	VOKEY:
		Page 174		Page 176
1	A In the afternoon sorry but the return	1 uge 1 / 1	А	Five percent range and I wouldn't be off more
2	flight a portion or all of that flight would	1 2)	than a percent there
3	be in darkness. So that's one example	e. 3	ROII	
4	Another example could be the total, the y	whole 4		Okay, reserve helicopter fuel to be kept on
5	of the flight would be in darkness. So	it 5	· · ·	the facility, including rationale. I
6	includes either total darkness or partia	1ϵ	ő	understand the first part of that. What's the
7	darkness. I think it's fair to say from all	7	,	rationale about?
8	operators that are operating here, it is	8	MR.	VOKEY
9	certainly our business objectives an	d 9) А.	I think that goes they go back to some
10	operational objectives to complete all o	our 10)	outdated regulations, but
11	flying in daylight hours. The challenge	is 11	ROIL	. O.C.:
12	because of the weather, the inclement we	eather 12	2 0.	Outdated wording?
13	that we do experience from time to time,	we do 13	MR.	VOKEY:
14	have the potential for creating what we re	efer 14	A.	Outdated well, outdated wording, and the
15	to as a backlog of people to be transported	ed to 15	5	specific reference was the MODUS, which is
16	and from the offshore installations, and	in 16	5	Mobile Offshore Drilling Units, but what it
17	cases like that to clear it up, we are	17	,	takes into account is you need to be able to
18	required to get rid of the backlog by flyi	ng 18	3	justify the amount of fuel you keep on board
19	into the evening, and it's not an issue that	ıt's 19)	by the type of operations you're conducting.
20	taken lightly with the operators. We do l	have 20)	So if you're flying three helicopters a day,
21	a number of criteria or check sheets that	we 21		don't sit there with a tow tank of one cube on
22	go through, and as an example, you would	Id need 22	2	board because that wouldn't be adequate for
23	to verify that, you know, your own fir	st 23	;	the type of operation that you have.
24	response capability is not inhibited in a	ny 24	ROIL	<i>2</i> , Q.C.:
25	way, that the 103 Cormorant, 103 Squad	lron in 25	5 Q.	Okay, so it isn't to minimize the amount, it's

Page 177 Page 177 1 A Each of the operators have a contract with 2 MR. VOKEY: 3 A I's to ensure that you've got adequate in the 4 Congar Helicopters for the provision of 5 RCT, QC.: 6 O And these helicopters of orface 1 at the various 7 installations that they land at? 8 MR. VOKEY: 9 A Yes. 10 Q. At all of them? 11 Q. At all of them? 12 MR. VOKEY: 13 A Yes. 14 Q. Cansideration be given to goggles. I think 15 Q. Consideration be given to goggles. I think 16 Werk versively. Proven automated usage 17 breathing devices, we've dealt with that 18 faily versively. Proven automated usage 19 and consideration be given to goggles. I think 16 breathing devices, we've dealt with that 17 breathing weights. So in the 18 faily versively. Proven automated usage 19 andomonitoring systems. Irec	January 11, 2010	Multi-P	age [™]	Offshore Helicopter Safety Inquiry	
1 A. It's to ensure that you've got adequate in the 1 A. It's to ensure that you've got adequate in the 2 A. It's to ensure that you've got adequate in the 2 Cougar Helicopters for the provision of 4 event you need it. 5 Cougar Helicopters for the provision of 6 Q. And these helicopters do refuel at the various 6 maintenance of helicopters, specialized 7 maintenance of helicopters, specialized 7 presonnel. That would include; supply, operation, and 8 MR. VOKEY: 8 dispatch, mechanics, maintenance personnel, 1 9 A. Yes. 9 terminal services, things like baggage 1 10 ROIL, OC: 10 preceor Carego that truvels to the Offshore by 13 A. Tes, 11 maintenance of kin, their BST certification, their 16 we' ve come across goggles, and appropriate 16 Even down to individual set with so in their 14 maintenance of kin, their BST certification, their 16 adout ther? 20 12 metaring devices, we've dealt with that 18 pregoe goes strong h ascentry semanding and mointoring systems. Freacill with one of 10 adout their lat	Pa	age 177		Page 179	
2 MR. VORTY: 2 Cougar Helicopters for the provides of nervices 4 event you need it. 5 5 ROH, Q.C.: 6 mathematic of helicopters are previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous. The types of services is the provide include jiological previous include jiological previous is the provide include jiological previous include in a flight is the provide include jiological previous include in a flight is the provide include jiological previous include in a flight is the provide include jiological previous include in a flight is process then and individuals and the previous previous include in a flight is process then and individuals and propriate if previous previous include in a flight is process then and individuals and manifers include in a flight is flight weight. So in the previous previous include in a flight is process when an individuals and propriate if a mathematic prevent arcmange is previous include in a flight is flight include in a flight include in the singer and anonitoring systems. Frecall with one of the receive weight is flight include in a flight is process when an individual sheegit. So in the include in a flight is flight is flight include in a stand the prevention, and the is prevention, and the is system is prevention, and the is system is previse. The prevention and shout the weight is flight include in the intervise is an administrative is prevention. The prevention of the prevention is more deal previnde intere preventing deal preventing deaverses. The pr	1 to rationalize the amount?	1	A.	Each of the operators have a contract with	
3 A. It's to ensure that you've got adequate in the event you need it. 3 helicopter services. The types of services 4 event you need it. 5 MCLL QC: 5 6 Q. And these helicopters to refuel at the various installations that they land at? 5 would include; supply, operation, and 7 installations that they land at? 7 personnel. That would include; supply, operation, and 8 MR. VOKEY: 8 dispatch, mechanics, maintenance personnel, that would include; supply, operation. and 9 A. Yes. 10 handling, security, and screening. Passenger 11 Q. At all of them? 11 and earge administration. Every person and 12 MR. VOKEY: 12 piece of cargo that travels to the offshore by 13 A. Yes. 13 helicopter tars to be included in a llight 14 ROIL, QC: 13 helicopter set travels to the offshore by 13 helicopter set to be included in a llight maintenance of the individual feecks in, it talks 16 Were come across Soggles, and appropriate 16 Event and individual set set individual feecks in, it talks 16 administrative 12 administrative	2 MR. VOKEY:	2		Cougar Helicopters for the provision of	
4 went you need it. 4 that Cougar has been contracted to provide 6 Q. And these helicopters do refuel at the various. 6 maintenance of helicopters, specialized 7 installations that they land at? 7 personnel. That would include pilots, flight 8 MR. VOKEY: 8 dispatch, mechanics, maintenance personnel. 9 A Yes. 9 terminal services, things like bagsage 10 Q. At all of them? 11 and cargo administration. Every person and 12 MR. VOKEY: 12 piece of cargo that travels to the offshore by 13 A Yes. 12 piece of cargo that travels to the offshore by 14 ROIL, Q.C.: 14 mainfest, and like I say, both individuals and 15 Q. Consideration be given to goggles. I think 16 Even down to individual for kecks in, it talks 16 thereating winews some colourth exchange 19 about ther? 10 next of kin, their TST certification, heir 21 between be and I about HUMS, and what HUMS 21 medical certification, and like I say, tagh adiministrative 23 about there? 24 would be by supply boats, periodically smaller	3 A. It's to ensure that you've got adequate in th	ie 3		helicopter services. The types of services	
5 ROI, Q.C.: 5 would include; supply, operation, and 7 installations that they land at? 7 8 MR, VOKEY: 8 0 A Yes. 9 10 O.A tall of them? 10 11 O.A tall of them? 11 12 MR, VOKEY: 12 13 A. Yes. 10 14 ROI, Q.C.: 10 15 Q. Consideration be given to goggles. I think 15 16 we've come across goggles, and appropriate 16 17 between he and I about HUMS; and what HUMS 16 18 fairly extensively. Proven automated usage 18 19 about ther? 23 11 called. HUMS is Health Usage Monitoring 21 12 System. 32 13 he licopters, and alle is explobel individual's woight. Sing Hawait the software, and like 'starsport of cargo administrative is and and individual's woight. Sing Hawait the software, and like 'starsport of cargo administrative is and and individual's woight. Sing Hawait the software, and like 'starsport of cargo administrative is a sing Hawait the software, and like 'starsport of cargo administrative. The software, and like 'starsport of cargo administrative. Thea	4 event you need it.	4		that Cougar has been contracted to provide	
6 Q. And these helicopters do refuel at the various instruction of the copters, specialized instruction of the copters, specialized instruction. Every outdoin clude pilots, flight is NR, VOKEY: a maintenance of helicopters, maintenance personnel, dispatch, mechanics, maintenance personnel, a dispatch, mechanics, maintenance, dispatch, mechanics, dispatch, mechanich, dispatch, mechanics,	5 ROIL, Q.C.:	5		would include; supply, operation, and	
7 mstallations that they land at? 7 personnel. That would neulode pitots, llight 9 A. Yes. 9 thread at the second	6 Q. And these helicopters do refuel at the vario	us 6		maintenance of helicopters, specialized	
8 MR, VOREY: 8 displatch, mechanics, maintenance personnel, 9 A. Yes. 9 terminal services, things like baggage 10 ROIL, Q.C.: 10 handling, security, and screening. Passenger 11 Q. At all of them? 11 partial cargo administration. Every person and 12 MR, VOREY: 12 piece of cargo that travels to the offshore by 13 A. Yes. 13 handling, security, and screening. Passenger 14 ROIL, Q.C.: 14 mainfest, and like I say, both individuals and 15 Q. Consideration be given to goggles. I think 15 baggage goes through ascerrity screening. 16 we've come across goggles, and appropriate 16 Even down to individual's weight. So in the 17 breathing devices, we've cault with that 18 process when an individual checks in, it talks 19 the earlier witnesses some colouril exchange 20 net of kin, the'r BST certification, their 21 between he and I about HUMS, and what HUMS 21 medical certification of fier 22 about ther? 23 would be by supply boats, periodically smaller 23 about ther? 24 WR. VOREY: 24 24 Q. Right. 7 Tatal so provide first responseservices, and offshore facaliby or medical	7 installations that they land at?	7		personnel. That would include pilots, flight	
9 A Yes. 9 ferminal services, tings like baggage 11 Q. At all of them? 11 10 handling, security, and screening. Passenger 12 MR, VOKEY: 12 piece of cargo that travels to the offshore by 13 A Yes. 12 marifest, and like 1 say, both individuals and 14 ROIL, Q.C.: 14 marifest, and like 1 say, both individuals and 15 0. Consideration be given to goggles. In think 16 Even down to individual 's weight. So in the 17 breathing devices, we've dealt with that 17 case of passengers, there's an administrative 16 it and monitoring systems. I recall with one of 19 about their latest identification, their 10 hate it, that's what you're talking 21 methere? 23 while the primary means of transport of cargo 21 mether? 24 would be by supply boats, periodically smaller Page 178 23 A ROIL, Q.C.: 24 would be by supply boats, periodically smaller Page 180 1 in the same compartment as individuals. 2 Cougar also provides air ambulance services, and offshore Recaility for medical resontris. Page 180	8 MR. VOKEY:	8		dispatch, mechanics, maintenance personnel,	
10ROIL Q.C.10handling, security, and screening. PassengerPassenger110. At all of them?11and cargo administration. Every person and piece of cargo that travels to the offshore by 1313MR VOKEY:12piece of cargo that travels to the offshore by helicopter has to be individuals and baggage goes through a security screening. If the baggage goes through a security screening. See administrative16we've come across goggles. I think if breathing devices, we've dealt with that if the earlier winnesses some colourful exchange 201817breathing devices, we've dealt with that if the earlier winnesses some colourful exchange 211810meant. I take it, that's what you're talking 22 about there?2023meant. I take it, that's what you're talking 232124MR VOKEY:2425A. For the S-92, that's what the system is 252526A. For the S-92, that's what the system is 252521called. HUMS is Health Usage Monitoring 2223ROIL, Q.C.:3and that's to ensure the timely evacuation of anyone who has to be evacuated from an 3 offshore facility for medical reasons. They a so hat vo to provide first response services, a and that's to ensure the timely evacuation of a sone of the helicopter, and after every flight, 443A chait just monitors all the critical aspects 46also have to provide first response services, a and that's to ensure the side ariticer, 43A of t he helicopter, and after every flight, <br< td=""><td>9 A. Yes.</td><td>9</td><td></td><td>terminal services, things like baggage</td></br<>	9 A. Yes.	9		terminal services, things like baggage	
11 Q. At all of them? 11 and cargo administration. Every person and pice of cargo administratives to the offshore by 13 A. Yes. 12 pice of cargo administratives to the offshore by 14 ROLL, Q.C. 14 manifest, and like I say, both individuals and baggage goes through a security screening. 16 we've come across goggles, and appropriate 16 Even down to individual's weight. So in the 17 breathing devices, we've dealt with that 17 case of passengers, there's an administrative 18 parcess when an individual checks in, it talks about their latest identification of their 20 20 the earlier witnesses some colourful exchange 20 next of kin, their BST certification, their 21 between he and I about HUMS, and what HUMS 21 medical certification, and like I say, right 22 meant. I take it, that's what you're talking 22 down to their weight. Like I said earlier, 23 about there? 23 while the primary means of transport of cargo 24 Would be ys usply boats, periodically smaller 25 25 parts will be flown on a helicopter, but not 26 3 ROL, Q.C: 3 and that's to ens	10 ROIL, Q.C.:	10		handling, security, and screening. Passenger	
12 MR VOKEY: 12 piece of cargo that travels to the offshore by 13 A. Yes. 13 helicopter has to be included in a flight 14 ROIL, Q.C.: 13 helicopter has to be included in a flight 14 ROIL, Q.C.: 14 manifest, and like I say, both individuals and 15 O. Consideration be given to goggles. Ithink 16 Even down to individual's weight. So in the 17 breathing devices, we've dealt with that 17 case of passengers, there's an administrative 18 process when an individual checks in, it talks in down to their weight. Like I said, right 20 the earlier witnesses some colourful exchange 20 next of kin, their BST certification, their 21 between he and I about HUMS, and what HUMS 21 medical certification, and like I say, right 22 meant. I take it, that's what you're talking 22 down to their weight. Like I said earlier, wuild be by supply boats, periodically smaller 23 about ther? 23 while the primary means of transport of cargo 24 R.VOKEY: 24 Roung arcutally takes the information and a more on a helicopter shot has to be evacuated from an offshore facility for medical reasons. They 3 A. And it just monitors all the critical aspect	11 Q. At all of them?	11		and cargo administration. Every person and	
13 A. Yes. 13 hetcopter has to be included in a fight 14 ROIL, Q.C. 14 manifest, and like I say, both individuals and 15 Q. Consideration be given to goggles. It hink 15 baggage goes through a security screening. 16 we've come across goggles, and appropriate 16 Even down to individual's weight. So in the 18 fairly extensively. Proven automated usage 18 process when an individual's weight. So in the 19 and monitoring systems. I recall with one of 19 about their BST certification, their 21 between he and I about HUMS, and what HUMS 21 medical certification, and like I say, right 22 about ther? 24 would be by supply boats, periodically smaller 23 about ther? 24 would be by supply boats, periodically smaller 24 R. For the S-92, that's what the system is 25 parts will be flown on a helicopter, but not 25 system. 3 and that's to ensure the individuals. 2 Cougar also provides air ambulance services, 3 and that's to ensure the individuals. 2 System. 3 and tha's to ensure the individuals.	12 MR. VOKEY:	12		piece of cargo that travels to the offshore by	
14 MBL QC: 14 mantlest, and like 1 say, both individuals and baggag goes through a security screening. 16 we've come across goggles, and appropriate transmistrative we've come across goggles, and appropriate transmister. 16 Even down to individual's weight. So in the transmistrative process when an individual checks in, it talks about their latest identification of their transmistrative process when an individual checks in, it talks about their latest identification of their medical certification, their stay, right about there? 20 the earlier witnesses some colourful exchange about there? 20 next of kin, their BST certification, their medical certification, and like I say, right about there? 21 between he and I about HUMS, and what HUMS 21 medical certification, and like I say, right about there? 23 about there? 21 down to their weight. Like I said earlier, would be by supply boats, periodically smaller 23 xolff. Page 178 Page 180 1 called. HUMS is Health Usage Monitoring 2 2 cougar also provides air ambulance services, and tha's to ensure the timely evacuation of a anyone who has to be evacuated from an 5 5 3 ROIT, Q.C:: 3 and that's to ensure the timely evacuation of a anyone who has to their offware, and it gets 9 offshore facility for medical reasons. They also have to provide first response services, and anyone who has to be evacuated from an 5 5 <td>13 A. Yes.</td> <td>13</td> <td></td> <td>helicopter has to be included in a flight</td>	13 A. Yes.	13		helicopter has to be included in a flight	
15 0; Consideration be given to goggles. Ithink 15 baggage goes through a security screening. 16 we've come across goggles, and appropriate 16 Even down to individual's weight. So in the 17 breathing devices, we've dealt with that 17 case of passengers, there's an administrative 18 process when an individual's weight. So in the 18 process when an individual's weight. So in the 20 the earlier witnesses some colourful exchange 20 next of kin, their BST certification, their 21 between he and 1 about HUMS, and what HUMS 21 medical certification, and like I say, right 23 about there? 23 while the primary means of transport of cargo 24 MK VOKKY: 24 would be by supply boats, periodically smaller 25 system. 25 page 178 1 called. HUMS is Health Usage Monitoring 1 in the same compartment as individuals. 2 Cougar also provides air ambulance services, 3 and that's to ensure the timely evacuated from an 5 MK VOKKY: 3 and that'll be described later in the 8 6 A. And it just monitors all the critical aspects <t< td=""><td>14 ROIL, Q.C.:</td><td>14</td><td></td><td>manifest, and like I say, both individuals and</td></t<>	14 ROIL, Q.C.:	14		manifest, and like I say, both individuals and	
16We ve come across goggles, and appropriate16Even down to individual s weight. So in the17breathing devices, we've dealt with that17case of passengers, there's an administrative18fairly extensively. Proven automated usage18process when an individual s weight. So in the19and monitoring systems. I recall with one of19about their latest identification of their20the earlier witnesses some colourful exchange10next of kin, their BST certification, and like I say, right21between he and I about HDMS, and what HUMS21medical certification, and like I say, right23about there?23while the primary means of transport of cargo24MK.VOKEY:24would be by supply boats, periodically smaller25A. For the S-92, that's what the system is25parts will be flown on a helicopter, but not26A. MOKEY:24would be by supply boats, periodically smaller27values1in the same compartment as individuals.28System.2Cougar also provides air ambulance services, and that's to ensure the timely evacuation of3and that's to ensure the timely evacuation of44Q. Right.4anyone who has to be evacuated from an5MR.VOKEY:5also have to provide first response services, and they we could of may6A. And it just monitors all the critical aspects77of the helicopter, and after every flight, and that'll be described later in the8Cougar	15 Q. Consideration be given to goggles. I think	K 15		baggage goes through a security screening.	
17breatning devices, we ve deat with that17case of passengers, there s an administrative18fairly extensively. Proven automated usage17case of passengers, there s an administrative19and monitoring systems. I recall with one of19process when an individual checks in, it talks10the earlier witnesses some colourful exchange20next of kin, their BST certification, their21between he and I about HUMS, and what HUMS21medical certification, and like I say, right22about there?23while the primary means of transport of cargo24MK VOKEY:24would be by supply boats, periodically smaller25parts will beflown on a helicopter, but notPage 1781called. HUMS is Health Usage Monitoring1in the same compartment as individuals.2System.2and that's to ensure the timely evacuation of3ROIL, QC:3and that's to ensure the timely evacuation of4Q. Right.4anyone who has to be evacuated from an5off the helicopter, and after every flight,5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects6also have to provide first response services,7of the helicopter, and after every flight,7and that'l be described later in the8Cougar actually takes the information and9colleagues. Some other specialized services,10sent to Sinssky's headquarters.10in the past Terra Nova has done it a co	16 we've come across goggles, and appropri	ate 16		Even down to individual's weight. So in the	
18ratify extensively. Proven automated usage18process when an individual checks in, it taks19and monitoring systems. I recall with one ofabout their latest identification of their20the earlier witnesses some colourful exchange20about their latest identification of their21between he and I about HUMS, and what HUMS21medical certification, and like I say, right22about there?23down to their weight. Like I said earlier,23about there?23while the primary means of transport of cargo24MR, VOKEY:24would be by supply boats, periodically smaller25A. For the S-92, that's what the system is25parts will be flown on a helicopter, but not26Called. HUMS is Health Usage Monitoring1in the same compartment as individuals.2Cougar also provides air ambulance services,3and that's to ensure the timely evacuation of3ROL, Q.C:3and the's to ensure the timely evacuation of4Q. Right.5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects67of the helicopter, and after every flight,78Cougar actually takes the information and89colleagues. Some other specialized services,10sent to Skorsky's headquarters.1011ROIL, Q.C:1112Q. Do I take it that other airframes at different1213times may or may not have the same level of <t< td=""><td>breatning devices, we ve dealt with that</td><td>17</td><td></td><td>case of passengers, there s an administrative</td></t<>	breatning devices, we ve dealt with that	17		case of passengers, there s an administrative	
19 and monitoring systems. Frecari with one of particular dentification of their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one their parts identified of kin, their BST creation of one the same compartment as individuals. 20 meant. I take it, that's what you're talking about there? 20 next of kin, their BST creation of one their medical certification, and like I say, right down to their weight. Like I said earlier, while the primary means of transport of cargo and that's to ensure the timely evacuation of any one who has to be evacuated from an offshore facility for medical reasons. They also have to provide first response services, and that's to ensure the timely evacuation of may of the helicopter, and after every flight, and that'i be described later in the services, and that's heat definerent to Sikorsky's headquarters. 9 collagues. Some other specialized services, some of the second of the cargo or users of the second of the cargo or users of the second of the secores of the S-92 that makes it the there of the	18 fairly extensively. Proven automated usag	ge 18		process when an individual checks in, it talks	
20 the earnier Withesses some colournul exchange 20 next of kin, neith Si Certification, neith 21 between he and Labout HUMS, and what HUMS 21 medical certification, and like I say, right 22 meant. I take it, that's what you're talking 22 down to their weight. Like I said earlier, 23 about there? 23 while the primary means of transport of cargo 24 MR. VOKEY: 24 would be by supply boats, periodically smaller 25 A. For the S-92, that's what the system is 25 parts will be flown on a helicopter, but not 25 called. HUMS is Health Usage Monitoring 1 in the same compartment as individuals. 2 System. 2 Cougar also provides air ambulance services, 3 ROIL, Q.C.: 3 and that's to ensure the timely evacuation of 4 Q. Right. 5 offshore facility for medical reasons. They 6 A. And it just monitors all the critical aspects 6 also have to provide first response services, 7 off the helicopter, and after every flight, 7 and that'' to censchilzed services, 8 Cougar actually takes the information and 8 presentation in more detai	and monitoring systems. I recall with one of	DI 19		about their latest identification of their	
21 between ne and rabout HOMS, and What HOMS 21 medical certification, and like 1 say, right 22 meant. I take it, that's what you're talking 22 down to their weight. Like I said earlier, 23 about there? 23 while the primary means of transport of cargo 24 MR. VOKEY: 24 would be by supply boats, periodically smaller 25 A. For the S-92, that's what the system is 25 means of transport of cargo 24 MR. VOKEY: 24 would be by supply boats, periodically smaller 2 System. 25 means of transport of cargo 2 System. 26 Cougar also provides air ambulance services, and that's to ensure the timely evacuation of 3 ROIL, Q.C: 3 and that's to ensure the timely evacuation of 4 Q. Right. 5 offshore facility for medical reasons. They 6 A. And it just monitors all the critical aspects 6 also have to provide first response services, 7 of the helicopter, and after every flight, 7 and that'll be described later in the 8 Cougar attually takes the information and 8 presentation in more detail by one of my	20 the earlier witnesses some colourful exchan	ige 20		next of kin, their BST certification, their	
22 meant. I take It, that s what you re taiking 22 down to therer weight. Like I sad earlier, 23 about there? 23 while the primary means of transport of cargo 24 MR. VOKEY: 23 while the primary means of transport of cargo 24 MR. VOKEY: 24 would be by supply boats, periodically smaller 25 Page 178 Page 178 1 called. HUMS is Health Usage Monitoring 1 in the same compartment as individuals. 2 Cougar also provides air ambulance services, and that's to ensure the timely evacuation of and that's to ensure the timely evacuation of 3 ROIL, Q.C.: 3 and that's to ensure the timely evacuation of 4 Q. Right. 4 anyone who has to be evacuated from an 5 MR. VOKEY: 5 offshore facility for medical reasons. They 6 A. And it just monitors all the critical aspects 6 and that'll be described later in the 8 Cougar actually takes the information and 9 colleagues. Some other specialized services, 10 sent to Silorsky's headquarters. 10 helicopters use the helicopters or users of 11 MR. VOKEY: 10 <td>21 between he and I about HUMS, and what H</td> <td>IUMS 21</td> <td></td> <td>medical certification, and like I say, right</td>	21 between he and I about HUMS, and what H	IUMS 21		medical certification, and like I say, right	
23 about there? 23 while the primary means of transport of cargo 24 MR. VOKEY: 24 would be by supply boats, periodically smaller 25 A. For the S-92, that's what the system is 25 page 178 1 called. HUMS is Health Usage Monitoring 25 parts will be flown on a helicopter, but not 2 System. 22 cougar also provides air ambulance services, and that's to ensure the timely evacuation of 3 ROIL, Q.C.: 3 and that's to ensure the timely evacuation of 4 Q. Right. 5 offshore facility for medical reasons. They 6 A. And it just monitors all the critical aspects 7 and that'll be described later in the 7 of the helicopter, and after every flight, 7 and that'll be described later in the 8 Cougar actually takes the information and 8 presentation in more detail by one of my 9 downloads it into their software, and it gets 10 some of the helicopter operators or users of 11 ROIL, Q.C.: 11 helicopters use the helicopters for things 12 Q. Do I take it that other airframes at different 12 like changing out flare tips offshore. I know	22 meant. I take it, that's what you're talking	22		down to their weight. Like I said earlier,	
24 Mk. VOKEY: 24 would be by supply boats, periodically smaller 25 A. For the S-92, that's what the system is 25 parts wild be flown on a helicopter, but not 26 A. For the S-92, that's what the system is 25 parts wild be flown on a helicopter, but not 26 A. And it just monitors all the critical aspects 3 and that's to ensure the timely evacuation of 3 ROIL, Q.C: 3 and that's to ensure the timely evacuation of 4 Q. Right. 4 anyone who has to be evacuated from an 5 MR. VOKEY: 6 also have to provide first response services, 7 of the helicopter, and after every flight, 7 and that'll be described later in the 8 Cougar actually takes the information and 8 presentation in more detail by one of my 9 downloads it into their software, and it gets 9 colleagues. Some other specialized services, 10 sent to Sikorsky's headquarters. 10 some of the helicopters for things 12 Q. Do I take it that other airframes at different 11 helicopters use the helicopters for things 15 MR.VOKEY: 16 ROIL, Q.C: 16 A. That's correct. I mean, it is a later 17 Q. This is a heavy lift function? 19 attractive. It does have the ability to 19 Attractive. It does have the ability to 20 Mound be a heavy lift function of the 21 effective use of helicopters that Cougar	23 about there?	23		while the primary means of transport of cargo	
25 A. For the S-92, that is what the system is 25 parts will be flown on a helicopter, but not Page 178 Page 180 1 called. HUMS is Health Usage Monitoring in the same compartment as individuals. 2 System. Cougar also provides air ambulance services, 3 ROIL, Q.C: 3 and that's to ensure the timely evacuation of 4 Q. Right. 4 anyone who has to be evacuated from an 5 MR.VOKEY: 5 offshore facility for medical reasons. They 6 A. And it just monitors all the critical aspects 7 and that'll be described later in the 8 Cougar actually takes the information and 9 presentation in more detail by one of my 9 downloads it into their software, and it gets 9 colleagues. Some other specialized services, 10 sent to Sikorsky's headquarters. 10 some of the helicopters for things 12 Q. Do I take it that other airframes at different 12 like changing out flare tips offshore. I know 13 times may or may not have the same level of 13 in the past Terra Nova has done it a couple of 14 times, and I believe Hibernia and Sea Rose 15 </td <td>24 MR. VOKEY:</td> <td>24</td> <td></td> <td>would be by supply boats, periodically smaller</td>	24 MR. VOKEY:	24		would be by supply boats, periodically smaller	
Page 178Page 1801called. HUMS is Health Usage Monitoring1in the same compartment as individuals.2System.2Cougar also provides air ambulance services,3ROIL, Q.C.:3and that's to ensure the timely evacuation of4Q. Right.4anyone who has to be evacuated from an5MR.VOKEY:5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects6also have to provide first response services,7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and8presentation in more detail by one of my9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14times, and I believe Hibernia and Sea Rose15MR.VOKEY:15MR.VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of the ~-1 guess,17Q. This is a heavy lift function ?18one of the aspects of the	25 A. For the S-92, that's what the system is	25		parts will be flown on a hencopter, but not	
1called. HUMS is Health Usage Monitoring1in the same compartment as individuals.2System.2Cougar also provides air ambulance services,3 ROIL, Q.C.:3and that's to ensure the timely evacuation of4Q. Right.4anyone who has to be evacuated from an5MR. VOKEY:5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects6also have to provide first response services,7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and8presentation in more detail by one of my9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopters for things11ROIL, Q.C.:11helicopters use the helicopters for things12Like changing out flare tips offshore. I know13times may or may not have the same level of1314monitoring and usage information?1415MR. VOKEY:16A. That's correct. I mean, it is a later1617technology and that is one of theI guess,1818one of the aspects of the S-92 that makes it1819attractive. It does have the ability to1920monitor some of the critical aspects of the2021airframe.2122operaters, the operators pool th	Pa	age 178		Page 180	
2System.2Cougar also provides air ambulance services,3 ROIL, Q.C.:3and that's to ensure the timely evacuation of4Q. Right.3and that's to ensure the timely evacuation of5MR. VOKEY:4anyone who has to be evacuated from an6A. And it just monitors all the critical aspects5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects6also have to provide first response services,7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and9colleagues. Some other specialized services,9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different1213times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later17Q. This is a heavy lift function?18one of the aspects of the S-92 that makes it18MR. VOKEY:19attractive. It does have the ability to19A. This would be a heavy lif	1 called. HUMS is Health Usage Monitorin	ng 1		in the same compartment as individuals.	
3 ROL, Q.C.:3and that's to ensure the timely evacuation of4Q. Right.3and that's to ensure the timely evacuation of5MR.VOKEY:5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects6also have to provide first response services,7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of theI guess,18MR. VOKEY:19attractive. It does have the ability to19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.22<	2 System.	2		Cougar also provides air ambulance services,	
4Q. Right.4anyone who has to be evacuated from an5MR.VOKEY:5offshore facility for medical reasons. They6A. And it just monitors all the critical aspects5offshore facility for medical reasons. They7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and8presentation in more detail by one of my9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of the I guess,18MR. VOKEY:18one of the aspects of the S-92 that makes it19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.21effective use of helicopters that Cougar22QO.L.;22	3 ROIL, Q.C.:	3		and that's to ensure the timely evacuation of	
 5 MR. VOKEY: 6 A. And it just monitors all the critical aspects 7 of the helicopter, and after every flight, 8 Cougar actually takes the information and 9 downloads it into their software, and it gets 9 downloads it into their software, and it gets 9 attractive. It does have the ability to 10 attractive. It does have the ability to 11 attractive. It does have the ability to 12 ROIL, Q.C.: 13 attractive. It does have the ability to 14 attractive. It does have the ability to 15 attractive. It does have the ability to 16 A. That's correct. I mean, it is a later 17 technology and that is one of the I guess, 18 one of the aspects of the S-92 that makes it 19 attractive. It does have the ability to 10 monitor some of the critical aspects of the 21 airframe. 22 ROIL, Q.C.: 23 Q. Okay, I think that's all the questions I had 24 for this page. Thank you, we can move on. 25 MR. VOKEY: 34 for this page. Thank you, we can move on. 35 MR. VOKEY: 36 A. WOKEY: 37 A. That way, there's a specific if there's a 36 A. VOKEY: 37 A. That way, there's a specific and the appendiced for that. So it is an 	4 Q. Right.	4		anyone who has to be evacuated from an	
 A. And it just monitors all the critical aspects of the helicopter, and after every flight, Cougar actually takes the information and downloads it into their software, and it gets sent to Sikorsky's headquarters. sent to Sikorsky's headquarters. Q. Do I take it that other airframes at different times may or may not have the same level of monitoring and usage information? times may or may not have the same level of monitoring and usage information? M. VOKEY: A. That's correct. I mean, it is a later technology and that is one of theI guess, one of the aspects of the S-92 that makes it attractive. It does have the ability to attractive. It haves aspects of the S-92 that makes it attractive. It does have the ability to A. This would be a heavy lift function of the effective use of helicopters that Cougar colleagues. In order to ensure the most effective use of helicopters that Cougar colleagues. In order to ensure the most the helicopters. I mank you, we can move on. St MR, VOKEY: 	5 MR. VOKEY:	5		offshore facility for medical reasons. They	
7of the helicopter, and after every flight,7and that'll be described later in the8Cougar actually takes the information and9colleagues. Some other specialized services,9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopter sor things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later17Q. This is a heavy lift function?17technology and that is one of the I guess,18MR. VOKEY:19attractive. It does have the ability to19A. This would be a heavy lift function of the20monitor some of the critical aspects of the21effective use of helicopters that Cougar21airframe.22operators pool the helicopters.23Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25operators is penalized for that. So it is an	6 A. And it just monitors all the critical aspects	6		also have to provide first response services,	
8Cougar actually takes the information and 98presentation in more detail by one of my9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of the I guess,17Q. This is a heavy lift function?18one of the aspects of the S-92 that makes it19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.21effective use of helicopters that Cougar22ROIL, Q.C.:22operators pool the helicopters.23Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25operator is penalized for that. So it is an <td>7 of the helicopter, and after every flight,</td> <td>1 7</td> <td></td> <td>and that'll be described later in the</td>	7 of the helicopter, and after every flight,	1 7		and that'll be described later in the	
9downloads it into their software, and it gets9colleagues. Some other specialized services,10sent to Sikorsky's headquarters.10some of the helicopter operators or users of11ROIL, Q.C.:10belicopters use the helicopter operators or users of12Q. Do I take it that other airframes at different1113times may or may not have the same level of1114monitoring and usage information?1215MR. VOKEY:1316A. That's correct. I mean, it is a later1617technology and that is one of the I guess,1818one of the aspects of the S-92 that makes it1919attractive. It does have the ability to1920monitor some of the critical aspects of the2021airframe.2122ROIL, Q.C.:2223Q. Okay, I think that's all the questions I had2324for this page. Thank you, we can move on.2425MR. VOKEY:25	8 Cougar actually takes the information an	a 8		presentation in more detail by one of my	
10sent to Sikorsky's headquarters.10some of the helicopter operators of users of11ROIL, Q.C.:11helicopters use the helicopter operators of users of12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of the I guess,16ROIL, Q.C.:18one of the aspects of the S-92 that makes it19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.21effective use of helicopters that Cougar22Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25operator is penalized for that. So it is an	9 downloads it into their software, and it gets	9		colleagues. Some other specialized services,	
11ROIL, Q.C.:11helicopters use the helicopters for things12Q. Do I take it that other airframes at different12like changing out flare tips offshore. I know13times may or may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:15likewise.16A. That's correct. I mean, it is a later16ROIL, Q.C.:17technology and that is one of the I guess,16ROIL, Q.C.:18one of the aspects of the S-92 that makes it19A. This is a heavy lift function?19attractive. It does have the ability to19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.21effective use of helicopters that Cougar22ROIL, Q.C.:22operates, the operators pool the helicopters.23Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25operator is penalized for that. So it is an	10 sent to Sikorsky's headquarters.	10		some of the helicopter operators or users of	
12Q. Do I take it that other arritanes at different12Take changing out hare tips offshore. I know13times may or may not have the same level ofin the past Terra Nova has done it a couple of14monitoring and usage information?1415MR. VOKEY:1516A. That's correct. I mean, it is a later1517technology and that is one of the I guess,1618one of the aspects of the S-92 that makes it1719attractive. It does have the ability to1920monitor some of the critical aspects of the1921airframe.2122ROIL, Q.C.:2223Q. Okay, I think that's all the questions I had2424for this page. Thank you, we can move on.2425operator is penalized for that. So it is an	11 ROIL, Q.C.:			like shareing out flore ting offshare. I know	
13times may of may not have the same level of13in the past Terra Nova has done it a couple of14monitoring and usage information?14times, and I believe Hibernia and Sea Rose15MR. VOKEY:14times, and I believe Hibernia and Sea Rose16A. That's correct. I mean, it is a later15likewise.17technology and that is one of the I guess,16ROIL, Q.C.:18one of the aspects of the S-92 that makes it17Q. This is a heavy lift function?19attractive. It does have the ability to19A. This would be a heavy lift function of the20monitor some of the critical aspects of the20helicopters. In order to ensure the most21airframe.21effective use of helicopters that Cougar22Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25operator is penalized for that. So it is an	12 Q. Do I take it that other airframes at different	12		like changing out flare tips offshore. I know	
 14 Inflex, and Tbelleve Filbernia and Sea Rose 15 MR. VOKEY: 16 A. That's correct. I mean, it is a later 17 technology and that is one of the I guess, 18 one of the aspects of the S-92 that makes it 19 attractive. It does have the ability to 20 monitor some of the critical aspects of the 21 airframe. 22 ROIL, Q.C.: 23 Q. Okay, I think that's all the questions I had 24 for this page. Thank you, we can move on. 25 MR. VOKEY: 14 Iffree and Tbelleve Filbernia and Sea Rose 15 likewise. 16 ROIL, Q.C.: 17 Q. This is a heavy lift function? 18 MR. VOKEY: 19 A. This would be a heavy lift function of the 20 helicopters. In order to ensure the most 21 effective use of helicopters that Cougar 22 operates, the operators pool the helicopters. 23 That way, there's a specific if there's a 24 specific airframe out of service, that no one 25 MR. VOKEY: 	13 times may or may not nave the same level	01 13		times and Libelians Liberia and See Deer	
15 MR. VOKEY:15 Inkewise.16 A. That's correct. I mean, it is a later17 technology and that is one of the I guess,16 ROIL, Q.C.:17 technology and that is one of the S-92 that makes it19 attractive. It does have the ability to17 Q. This is a heavy lift function?19 attractive. It does have the ability to19 A. This would be a heavy lift function of the20 monitor some of the critical aspects of the20 helicopters. In order to ensure the most21 airframe.21 effective use of helicopters that Cougar22 ROIL, Q.C.:22 operates, the operators pool the helicopters.23 Q. Okay, I think that's all the questions I had23 That way, there's a specific if there's a24 for this page. Thank you, we can move on.24 specific airframe out of service, that no one25 MR. VOKEY:25 operator is penalized for that. So it is an	14 monitoring and usage information?	14		liberrise	
 A. That's correct. Threah, it is a fater technology and that is one of the I guess, one of the aspects of the S-92 that makes it attractive. It does have the ability to monitor some of the critical aspects of the airframe. ROIL, Q.C.: MR. VOKEY: A. This would be a heavy lift function of the helicopters. In order to ensure the most effective use of helicopters that Cougar operates, the operators pool the helicopters. That way, there's a specific if there's a specific airframe out of service, that no one operator is penalized for that. So it is an 	15 MR. VOKEY:	15	DOIL	likewise.	
 17 G. This is a heavy fit function? 18 one of the aspects of the S-92 that makes it 19 attractive. It does have the ability to 20 monitor some of the critical aspects of the 21 airframe. 22 ROIL, Q.C.: 23 Q. Okay, I think that's all the questions I had 24 for this page. Thank you, we can move on. 25 MR. VOKEY: 	A. That's confect. Thread, it is a fater	10	KUIL,	Q.C.: This is a heavy lift function?	
18 <td>17 technology and that is one of the</td> <td>+ 10</td> <td>Q.</td> <td>This is a heavy intrunction?</td>	17 technology and that is one of the	+ 10	Q.	This is a heavy intrunction?	
 autractive. It does have the ability to monitor some of the critical aspects of the airframe. airframe. ROIL, Q.C.: Q. Okay, I think that's all the questions I had for this page. Thank you, we can move on. M. This would be a heavy fit function of the helicopters. In order to ensure the most effective use of helicopters that Cougar operates, the operators pool the helicopters. That way, there's a specific if there's a specific airframe out of service, that no one operator is penalized for that. So it is an 	10 attractive It does have the ability to	10		This would be a heavy lift function of the	
20Informer some of the critical aspects of the21airframe.21airframe.22ROIL, Q.C.:23Q. Okay, I think that's all the questions I had24for this page. Thank you, we can move on.25MR. VOKEY:26hencopters. In order to ensure the most27effective use of helicopters that Cougar28operates, the operators pool the helicopters.29for this page. Thank you, we can move on.20operator is penalized for that. So it is an	monitor some of the critical espects of the	19	А.	halicoptors. In order to onsure the most	
21annume.21checkive use of hencopters that Cougar22ROIL, Q.C.:22operates, the operators pool the helicopters.23Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25operator is penalized for that. So it is an	21 airframe	20		effective use of heliconters that Cougar	
23Q. Okay, I think that's all the questions I had23That way, there's a specific if there's a24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25				operates the operators pool the belicopters	
25252511at way, there is a specific informed out of service, that no one24for this page. Thank you, we can move on.24specific airframe out of service, that no one25MR. VOKEY:25operator is penalized for that. So it is an	23 0 Okay I think that's all the questions I had	22		That way there's a specific if there's a	
25 MR. VOKEY: 25 operator is penalized for that. So it is an	for this page. Thank you we can move on	23		specific airframe out of service that no one	
	25 MR. VOKEY:	25		operator is penalized for that. So it is an	

Page 181	Page 183
	1 ugo 10./
1 agreement between the operators saving that 1 theirs and Statoil had a S-92 as well	
2 we'll each contribute a helicopter, and any 2 ROIL O.C.:	
day one is out for maintenance that we'll 3 0 Okay. So the pooling arrangement can i	nvolve
4 each share the remaining heliconters	
5 ROIL O.C.	
6 0 Okay before we go on there I'd like to 6 A If you elect to participate in the poolin	a
7 evolore a little bit how the system works 7 arrangement yes	8
 because ultimately, the individual contracts BOIL O.C. 	
8 because utilinately the multitudal contracts 8 Koll, Q.C.: 9 ore individual to the companies and we'll 9 O Ves okey. We have as an exhibit I d	on't
9 are individual to the companies, and we in 9 Q. Tes, okay. We have as an exhibit I d	on t
10 have to bring those contracts out in that 10 want to take it out and tak about it line is have of this inquiry, but we understand that 11 line, other, people may, but we do have	the
there is a pooling arrangement we understand that the holicenter pooling shorter. Mr. Valey, i	f the
12 there is a pooling arrangement, we understand 12 nencopter pooling charter. Mr. Vokey, 1	i you
13 that there are an frames. So let the just take 13 can just explain to us because againts in the behavior of the short of the shor	1
14 It in baby steps. Each of the three major 14 probably a lot of people don't understan	a, on
15 companies that are out there has an arritame 15 any given day, does do an three	. 6
16 that is assigned to it, that s it's contracted 16 helicopters fly out, is there always one for the first state of the s	eft
17 for? 17 behind? How does the flight thing work	k in a
18 MR. VOKEY: 18 practical sort of way?	
19 A. We each contract one helicopter. 19 MR. VOKEY:	
20 ROIL, Q.C.: 20 A. Well, these helicopters, I mean, like a	ny
21 Q. One helicopter, so that gives us three. We 21 piece of machinery, require maintenance	es, but
22 now have a new operator out there called 22 let's assume there's a day when none o	f the
23 ConocoPhillips. Do they share in that third 23 helicopters are US or unserviceable, the	re's
24 one or is there a fourth one? 24 always that one helicopter on the groun	d and
25 MR. VOKEY:25that would help satisfy the first respon	se
Page 182	Page 184
1 A. Not as such, but they're flying program takes 1 requirements for one hour wheels up. S	o the
2 them into somewhere around 1 o'clock in the 2 other two helicopters on a pre-set agree	ment
3 day, and when their helicopter when they're 3 between the operators would be flying of	out to
4 finished with their helicopter, they do make 4 each of the installations. So Hibernia w	ould
5 their helicopter available to us. 5 not necessarily fly the helicopter that th	ey
6 ROIL, Q.C.: 6 contracted, that helicopter might, in fact	, be
7 Q. So there are now out at the, call it Torbay 7 going to Terra Nova, but in the event	of
8 Airport, giving away my age, St. John's 8 backlog or weather, the pooling agreeme	ent sets
9 Airport, there are four airframes? 9 out the structure for when we'll resu	me
10 MR. VOKEY: 10 helicopter operations, how you get rid	of
11 A. There's four available, that's correct. 11 backlog, and what constitutes priority,	you
12 ROIL, Q.C.: 12 know, where technical emergencies wou	ıld fit in
13 Q. On March 11th and 12th, how many were there 13 there, where medivacs would fit in there	, and
14 then? I'm trying to establish the size of the 14 it's just it's not real tightly worded, I	
15 pool. Is it always three, has it been four, 15 think, if you read it, but it's the spirit of	
16 does it go down to two, does it go up to five? 16 there's three operators here, we have the	iree
17 Just give us some understanding of that. 17 aircrafts, let's use them to the mutua	1
18 MR. VOKEY: 18 advantage.	
19 A. I stand to be corrected on this, but I believe 19 MR. PRITCHARD:	
20 there was four. Maybe Mr. Sacuta can just 20 A. Just for clarity there, Mr. Roil, to sort o	f
21 comment. 21 make a point that the three aircraft do ro	tate
22 MR. SACUTA: 22 in operational duties on any one day.	
23 A. There was four. Statoil was drilling at the 23 ROIL, Q.C.:	
time. Statoil brought in an helicopter for 24 Q. Yes.	
25 their operation, so each of the operators had 25 MR. PRITCHARD:	

January 11, 2010	Multi-Pag	e TM Offshore Helicopter Safety Inquiry
Р	age 185	Page 187
1 A. So the three helicopters are operationally in	n 1	process is similar to that of the other
2 use, but any one time we always maintain a	since 2	operators, and included a pre-qualification of
3 March 12th, and back to service, one airfra	ime 3	the potential bidders, and that's a process
4 on the ground designated for SAR duties.	4	that ensures that each potential service
5 ROIL, Q.C.:	5	provider is qualified to perform the work.
6 Q. So three are able to fly. Two could be takin	ng 6	Cougar was one of three qualified bidders.
7 people back and forth?	7	The competitive bid process included a
8 MR. PRITCHARD:	8	detailed scope of work, the submission of a
9 A. Correct.	9	formal bid proposal from the three qualified
10 ROIL, Q.C.:	10	bidders, a detailed analysis of the bidders by
11 Q. The third one would be there.	11	a multidisciplinary team, and they assessed
12 MR. PRITCHARD:	12	the safety and environmental performance of
13 A. Yes.	13	the bidders, technical capabilities of the
14 ROIL, Q.C.:	14	bidders, the commercial proposal from each of
15 Q. That third one might be a different airfram	ne 15	the bidders, and to the extent to which in the
in the afternoon than it was in the morning	? 16	delivery of the service contract the bidder
17 MR. PRITCHARD:	17	met the Canada and Newfoundland and Labrador
A. In three hours time that airframe that's on	ı 18	benefits commitments. Cougar was awarded the
19 the ground may well be an operationa	1 19	contract for Hibernia's helicopter service
20 helicopter, just to make sure	20	provision in 1995. As a normal component of
21 MR. VOKEY:	21	any major contract service bid, the
22 A. Yeah. I guess no, for clarity, probably	22	recommendations of the operator are shared by
23 what I should have said, the one that's on t	he 23	the operator with the regulator, in this case
ground can't leave until one of the other tw	vo 24	the C-NLOPB, in advance of the contract award.
25 helicopters are back.	25	If the regulator has specific issues or
р	age 186	Page 188
1 ROIL. O.C.:	1	concerns with a contract recommendation at
2 0. Yeah. So it's not that there is a helicopter	2	this stage, they may direct the operator to do
3 marked number one that goes to Hibernia e	every 3	additional contract analysis. In this case,
4 day, and marked number two that goes to 7	Ferra 4	the recommendation was reviewed and validated
5 Nova, or three that goes to White Rose. Th	his 5	by the Board.
6 pooling and sharing arrangement happens	on a 6 R	OIL, Q.C.:
7 daily basis?	7	0. Now prior to 1995, I take it from all of your
8 MR. VOKEY:	8	evidence, or at least a couple of you, that
9 A. That's correct, it's an ongoing basis. The	9	you might have been here prior to that time,
10 next slide I'll talk a little bit about	10	was Cougar or was there another company or
selection of Cougar as the helicopter servic	ce 11	companies providing the service prior to that
12 provider.	12	time?
13 ROIL, Q.C.:	13 M	R. VOKEY:
14 Q. Before you go on, there was a line there ag	ain 14	A. There were other companies. I believe CHC was
that we didn't deal with. It says, "Operator	s 15	here at one point, a company called Sealand
16 conduct independent formal audits of Cous	gar". 16	Helicopters.
17 This is something that I gather we will dea	al 17 R	OIL, Q.C.:
18 with in more detail in the individual	18	Q. But from 1995, the only provider has been
19 presentations?	19 M	R. VOKEY:
20 MR. VOKEY:	20	A. That would have been the only because the
21 A. That's correct. On slide 64, as the Hiberni	a 21	drilling programs would have finished up in
22 Development was the first of the offshor	re 22	the mid to late 80s.
23 producing assets, HMDC led the first biddin	ng 23 R	OIL, Q.C.:
24 process for the selection of the helicopter	24	Q. Right.
25 service provider. HMDC's competitive b	id 25 M	R. VOKEY:

January 11, 2010	Multi-P	age	Offshore Helicopter Safety Inquiry
H	Page 189		Page 191
1 A. So during that period there wasn't any	/ 1	I A.	In preparation for the White Rose Project,
2 offshore activity.	2	2	Husky also initiated a competitive billing
3 ROIL, Q.C.:	3	3	process for helicopter services, and they too
4 Q. Right.	4	1	selected Cougar, and their contract was
5 MR. VOKEY:	5	5	established in 2003. Contracts between the
6 A. So other operator selection of Cougar	r 6	5	operators and Cougar are for specified periods
7 Helicopters. For Petro Canada, in preparat	ion 7	7	with extension options, and all operators
8 for the development of the Terra Nova Pro	oject, 8	3	revisit their helicopter contracts
9 Petro Canada also initiated a competitive b	oid 9)	periodically to determine if extensions and/or
10 process for helicopter services. The	10)	a rebid is required. I'm just going to talk a
11 requirements outlined in Petro Canada's se	cope 11	l	little bit about the selection of the S-92A
12 of work included those noted in the	12	2	now, the Sikorsky S-92A. At the time Cougar
13 description of helicopter services in the	13	3	was selected by each of the operators, they
14 first slide of this section, and also the full	14	1	were using Super Puma as their primary
15 time the requirement for a full deicing	15	5	airframe. As their base of operations
16 capabilities. The bid analysis process for	: 16	5	expanded, Cougar added the Sikorsky S-61 to
17 Petro Canada followed essentially the sa	me 17	7	the fleet, and they
18 process that HMDC conducted. Petro Car	nada 18	8 ROIL,	Q.C.:
also used a multidisciplinary team to revie	ew 19) Q.	I'm sorry, before you go on, people don't
20 its bid and contracted an aviation consulta	nt 20)	necessarily understand, is Eurocopter a piece
to participate in the bid review process.	21	l	of equipment manufactured by Sikorsky, or is
22 This was not a core skill for Petro Canada	at 22	2	that
the time. Cougar was recommended as a s	service 23	3 MR. V	OKEY:
24 provider and the contract between Terra N	Nova 24	4 A.	Eurocopter Super Puma is manufactured by
25 or Petro Canada and Cougar was establish	ed in 25	5	Eurocopter.
H	Page 190		Page 192
1 the fourth quarter of 1998.	1	ROIL,	Q.C.:
2 ROIL, Q.C.:	2	2 Q.	It's an entirely separate company?
3 Q. Can I just stop you there. Can I take it from	n 3	3 MR. V	OKEY:
4 that, that there was there may have bee	en 4	4 A.	It's a competitor, but
5 good commercial reasons to use Cougar,	but 5	5 ROIL,	Q.C.:
6 there was no requirement as a term of yo	our 6	5 Q.	Okay, I was trying to avoid that word, but,
7 licence that you use the same provider th	at 7	7	yes, exactly.
8 was being used by HMDC?	8	3 MR. V	OKEY:
9 MR. VOKEY:	9) A.	And the S-61, S-92 is Sikorsky.
10 A. There was absolutely no requirement. I gu	ess, 10) ROIL,	Q.C.:
11 one of the things when you are working i	na 11	l Q.	So Sikorsky have various airframes?
12 basin, though, as small as the Grand Ban	iks 12	2 MR. V	OKEY:
13 that we are operating in, there are econom	ies 13	3 A.	That's correct.
14 of scale. So the fact that Cougar was here	e, 14	A ROIL,	Q.C.:
15 was set up, had the facilities, had the space	e, 15	5 Q.	And they're all designated as "S" something.
16 whether or not you could argue they had	a 16	5 MR. V	OKEY:
17 competitive advantage or not, I mean, it's	up 17	7 A.	And Eurocopter in terms of Super Pumas, they
18 to the individual, but for some of those	18	3	have a couple of different models. We had the
19 companies when they're here first, it doe	es 19)	model 332L here. They do have a new
20 give them an advantage because the mone	ey has 20)	competitor now for the S-92, which is the
21 been invested and now it's just a matter of	it 21	l	Eurocopter 225.
being shared among the participants.	22	2 ROIL,	Q.C.:
23 ROIL, Q.C.:	23	3 Q.	Okay.
24 Q. Uh-hm. Okay, thank you.	24	4 MR. V	OKEY:
25 MR. VOKEY:	25	5 A.	So as I mentioned, as their base of operations

Januar	y 11, 2010 N	/Iulti-Page Th	M Offshore Helicopter Safety Inquiry
	Page	193	Page 195
1	expanded. Cougar added the Sikorsky S-61 to	0 1	certification process through the Federal
2	the fleet, and at that time they also began to	2	Aviation Authority, or as we know, the FAA,
3	assess new helicopter technology. The	3	and their production program for the 92, and
4	analysis was conducted over a number of year	s 4	also a detailed technical analysis of the
5	beginning back in 2000. Cougar's analysis	5	Eurocopter and Sikorsky airframes,
6	began to centre on two leading contenders; the	6	particularly in terms of their respective
7	Sikorsky S-92A, and the Eurocopter E-225.	7	size, fuel capacity, range, and availability
8	Cougar eliminated the Agusta Westland EH-10	01 8	of simulators for each of the respective
9	from a detailed consideration due to	9	airframes. This process led Cougar to
10	maintenance support limitations and weight	10	recommending the Sikorsky S-92 to Petro
11	considerations. The EH-101 is a significant	11	Canada. Petro Canada had been working jointly
12	heavier helicopter, and it may have been a	12	with Cougar through this analysis and
13	limitation for a number of vessels or rigs	13	ultimately agreed to proceed with the
14	operating on the Grand Banks if they weren't	14	selection of the S-92A. This recommendation
15	designed for those heavier loads for their	15	again was brought forward to the C-NLOPB for
16	helidecks. So that would have been one of the	16	endorsement in accordance with the guidelines
17	reasons.	17	that I referenced earlier in this section.
18 ROIL	. O.C.:	18	Petro Canada took delivery and put the first
19 Q.	If I can stop you there, you mentioned Agusta	19	of Cougar's S-92 in service in April of 2005.
20	Westland, which I take it is another	20	Ultimately, Cougar worked with HMDC and Husky,
21	manufacturer. Are there any other	21	who conducted independent reviews and
22	manufacturers in the world, and you may not h	be 22	ultimately adopted the S-92A. I just want to
23	the expert, so if you can't answer it clearly,	23	talk about some of the considerations. When
24	then answer it with, "I don't know", or "I'm	24	Cougar conducted its review of the airframe
25	not sure", but is it essentially a choice of	25	options, there were several key considerations
	Page	194	Page 196
1	three heavy lift helicopters, or are there 27	1	for the east coast offshore basin. One was
2	or 50 or 100? Is it like buying a car or is	2	the overall safety features of the S-92.
3	it like buying a particular vehicle for a	3	Another was passenger capacity. The S-92 had
4	particular purpose?	4	greater passenger capacity than that of the
5 MR.	VOKEY:	5	Super Pumas, and the S-61, that's previously
6 A.	You're right. I'm not the one to ask. The	6	been used by Cougar. Speed and range was a
7	only other one that I am familiar with that we	7	factor. Range is a critical consideration for
8	have used here in the past is the Bell	8	offshore producing fields because they are in
9	Helicopters.	9	excess of 300 kilometres offshore. Cargo
10 ROIL	, Q.C.:	10	capacity and flexibility was another
11 Q.	So there are other manufacturers. Whether	11	consideration, maintenance requirements and
12	they make anything that is appropriate, you	12	support capability, and the overall track
13	wouldn't be best to ask that?	13	record of the manufacturer. Ultimately,
14 MR.	VOKEY:	14	Cougar's recommendation to the operators was
15 A.	That's correct. Cougar would certainly be	15	that the S-92 was configured best for
16	able to shed some light, though.	16	Newfoundland offshore basin. To speak a
17 ROIL	,, Q.C.:	17	little bit about the transition from the 61
18 Q.	Okay, thank you.	18	Super Puma to a full fleet, as noted
19 MR.	VOKEY:	19	previously, Petro Canada took delivery of the
20 A.	Cougar determined that the S-92A and the	20	first S-92 in April of 2005. Cougar's pool of
21	Eurocopter 225 were the two airframes most	t 21	helicopters at that point expanded to include
22	suitable for east coast offshore operating	22	Super Pumas, S-61s, and then S-92As. From
23	environment. Cougar conducted a detailed	23	2005 to 2007, HMDC and Husky independently
24	analysis of the leading contenders, which	24	completed their assessment of the S- 92
25	included an analysis of Sikorsky's	25	airframe and executed their respective

Janua	ry 11, 2010 Mu	ulti-Page ¹	M Offshore Helicopter Safety Inquiry
	Page 1	97	Page 199
1	agreements with Cougar. By mid 2007, Cougar	1	pickup points far surpassed any of the older
2	completed its transition to a full S-92A fleet	2	vintage helicopters. So in the early days
3	as it is today. Some of the features of the	3	when we were improving on that technology and
4	S-92, and we did talk about the HUMS, but the	4	trying to perfect it, we did have some bogus
5	S-92, like the Eurocopter, the newer	5	or, you know, false chip lights, but that
6	helicopters do use newer technology. Anti-	6	didn't help our workforce and you know, in
7	deicing would be more of a standard, the	7	some ways, justifiably so. I mean, the chip
8	Health Usage Monitoring System, the S-92 has	8	lights were supposed to work and if you got a
9	bird strike protection, and some of the older	9	light indication, you know, there must be
10	aircraft didn't have those types of features.	10	something wrong. So Cougar's procedure was
11	Another reason, I guess, why Cougar back in	11	any time there was a chip light, you know, you
12	2000 took a look at their fleet, I mean, the	12	don't try and wait to get to an installation
13	Super Pumas have been around for in excess of	13	or analyze it when you get back to town.
14	25 years, the S-61s, they had been around	14	Their protocol was to return to town. So we
15	since the 60s, and given the maturity of the	15	did have more what we refer to as boomerang
16	field, given that Hibernia was just on stream,	16	flights due to early chip indications and I
17	Terra Nova was coming next, and White Rose was	17	think anybody that's flown offshore for any
18	next to become the third, Cougar took it upon	18	period of time, they've all got their
19	themselves to say, okay, you know, this basin	19	preferences on the types of airframes and it's
20	is going to be here for the next 15, 20, 25	20	no different than, you know, somebody likes a
21	years as a minimum, what do we have to do in	21	Ford or a Chev or you know, even down to a
22	order to secure airframes for the greater part	22	particular model. A lot of our people like
23	of the life of the field, and take advantage	23	the S61. Most of them have flown in Super
24	of some of the new technologies. So that was,	24	Pumas for a long time, but some of the motion
25	in essence, the driver for that, and I believe	25	or vibration characteristics of the S-92 which
	Page 1	98	Page 200
1	that concludes the section I don't know if	1	is inherent in the design it's not a
2	vou've got any questions.	2	substandard design, but it's just different.
3 ROI	. O.C.:	3	we did have feedback from some of our
4 0	. Before vou leave, let me just ask vou one	4	employees that, you know, they felt there was
5	question, and that is and this is really	5	more vibration than previous. But that was
6	for all three of you prior to the incident	6	the only indicators that I can recall right
7	on March the 12th, 2009, did any of your	7	now.
8	companies have any reason to believe that	8 ROII	. O.C.:
9	there were concerns with respect to the S-92.	9 0	Either of the other gentlemen have anything to
10	in terms of its safety and its performance	10	offer on that?
11	generally?	11 MR.	PRITCHARD:
12 MR.	VOKEY:	12 A	A. From Husky's point of view, we took our due
13 A	. I will just speak on behalf of Suncor first.	13	diligence with our aviation expert and we had
14	The answer to that is no. We did have, I	14	no safety concerns from that point of view
15	guess, indications from our workforce. The	15	with the S-92.
16	issue of chip lights came up more prevalent	16 MR.	SACUTA:
17	with the S-92.	17 A	A. Also from HMDC's perspective, we did do annual
18 ROI	L, Q.C.:	18	audits using an aviation specialist of
19 Q	. Chip lights, what are those?	19	ExxonMobil's corporate aviation group and no
20 MR.	VOKEY:	20	issues were raised at any point in time before
21 A	. Chip lights, that's where you pick up magnetic	21	the March 12th incident.
22	particles in the gearbox and it may or may not	22 ROII	L, Q.C.:
23	be indicative of a premature or an impending	23 Q	2. So I take it that by these chips, there was
24	failure. The S-92, in terms of what it	24	more information being given to the pilots
25	measures and the number of chip lights or chip	25	which enabled them to make decisions about

Page 201 Page 202 1 turning around, but that was interpreted by 1 latest Transport Canada standard, which is the 2 the consumers, by the users, by the workers, 3 as being that these machines were perhaps less 4 reliable? 3 reliable? 3 reliable? 5 AR VOKEY. 5 It was identified to be HUBAA capable. 5 6 A. Your chips are your early warning system. 7 HUEBA and they wanced to make sure that 8 Q. Okay. I think we're ready to move on now to 8 whatever suit we got had to be capable of 9 using the HUEBA. 10 Any replacement suit was also required to 11 IMR.SACUTA: 10 Any replacement suit was also required to 12 I transportation and it's specific to passenger 13 Transport Canada 13 protective equipment for helicopter 13 Transport Canada and the marine abandonment 14 tanumber of regulations that require or outline 13 0. Okay, and I think we as adopted by Transport Canada 15 safety. statadar bis morning, in my	January 11, 2010	Multi-P	age	Offshore Helicopter Safety Inquiry
1 turning around, but that was interpreted by 2 1 latest Transport Canada standard, which is the 2 2 the consumers, by the users, by the workers, 3 coss and PLBs. The 2 The 2 Coss standard, which also includes the requirement for life vess and PLBs. The standard, which states and PLBs. 1 The 2 5 A. Your chips are your early warning system. 7 KOLL, QC: 8 Use and the ware also required to be HUEBA. 1 The state state also be the 2 6 A. Your chips are your early warning system. 7 Huiba and the ware going to implement 4 Huiba and the ware going to implement 4 7 HULBA and they warned to make stare that 5 State ware state to make stare that 4 State ware state to make stare that 4 State ware state ware state to make state that 4 State ware state ware state 4 State ware state 4 State ware ware state 4 State ware ware state 4 State ware ware state 4 State ware ware ware state 4 State ware ware ware state 4 State ware ware ware ware state 4 State ware ware ware ware state 4 State ware ware ware ware ware ware ware war	Р	age 201	-	Page 203
2 the consumers, by the users, by the workers, a being that these machines were perhaps less 2 CGSB standard, which also includes the requirement for life vests and PLBA. capable. 5 MR. VOKLY: 5 If. was identified in 2006, when we started 6 A. Your chips are your early warning system. 6 If. was identified in 2006, when we started 7 ROIL, Q.C.: 7 If. was identified in 2006, when we started 7 OKay. I think we're ready to move on now to 0 6 N. Your chips are your early warning system. 10 take us through. 11 If. was identified in 2006, when we started 10 take us through. 11 If. MR.SACUTA: 12 A. The next section is to discuss personal 12 transportation suit stystem that was adopted by 13 and filescussed this morning, there are a 13 Transport Canada and the marine abandomment 14 the requirements for passenger transportation 13 Q. Okay, and I think the next slide deals with a 15 standard which was adopted by Transport Canada 14 Marine. 16 Ast I discussed this morning to go through them 14 Q. Okay, and I think the next slide deals with a 16 sta	1 turning around, but that was interpreted b	y 1	l	latest Transport Canada standard. which is the
3 as being that these machines were perhaps less requirement for life vests and PES. The suits were also required to B HUEA capable. Suits were also required to B HUEA capable. It was identified in 2006, when we started It was identified in 2006, when we started this process, that we were going to implement the next section, which Mr. Sacuta is going to the next section is to discuss personal transportation and it's specific to passenger transportation and it's specific to passenger transportation and it's specific to passenger transportation suit standard, which is also a CGB stately. As I discussed this morning, there are a number of regulations that require or outline the requirements for passenger transport canada the requirements for passenger transport and stated this morning in my regulatory overview section on that itum, unless you have any questions. the requirements for passenger transport and stated this morning in my regulatory overview section on this itum, the weight of the med to take it lime for all again, but they're all shown on this the subt we don't need to take it lime for as badded to 2007. It was recognized that the suits were continuous improvement the dial of 2007. It was recognized the med to late 2006, the morting the more as the disthe yellow suit. the suit was eve going to r	2 the consumers, by the users, by the worke	rs, 2	2	CGSB standard, which also includes the
4 sums were also required to be HUBBA capable. 5 MR VOREY: 5 6 A. Your chips are your early warning system. 5 7 ROIL, Q.C.: 5 8 Q. Okay. 1 think we're ready to move on now to the next section, which Mr. Sacuta is going to the next section is to discuss personal 7 10 take us through. 10 Any replacement suit was also required to the capable of 9 12 A. The next section is to discuss personal of protective equipment for helicopter 11 13 protective equipment for helicopter 12 14 transport Canda and the marine abnonment immersion suit standard, which was adopted by Transport Canada 16 As 1 discussed this morning, there are a 17 17 number of regulations that require or outline 18 20 stade and I'm not planning to go through them 19 21 stided this morning in my 22 regulatory overview section on that item, 23 a. Right. 24 ROIL, OC: 25 Q. No, that's fine. 1 think we've had some 2 we balout that and nice to refresh 2 that the suits were coming to the	3 as being that these machines were perhaps	less 3	3	requirement for life vests and PLBs. The
5 MR, VOREY: 5 It was identified in 2006, when we started 6 A. Your chips are your early warning system. 5 It was identified in 2006, when we started 6 A. Your chips are your early warning system. 6 6 this process, that we were going to implement 7 PROL, QC: 7 HUEBA and they warned to make sure that 8 Q. Okay. I think we're ready to move on now to 8 whatever suit we got had to be capable of 9 the next section is to discuss personal 10 Any replacement suit was also required to 11 MR. SACUTA: 11 mest section is to discuss personal 12 13 protective equipment for helicopter 13 Transport Canada and the marine abandonment. 14 transportation suit system that was adopted by Transport Canada 16 Marine. 17 number of regulations that require or outline 18 Q. Okay, and I think the next slide deals with a slas o a CSB is standard which was adopted by Transport Canada 18 the requirements for passenger 17 TROIT, QC: 2 regulatory overview section on that item, and ince to refresh a sub, and that is why was there a 18 2 Q. No, that's fini	4 reliable?	4	Ļ	suits were also required to be HUEBA capable.
6 A. Your chips are your carly warning system. 6 this process, that we were going to implement 7 ROL, Q.C.: 6 this process, that we were going to implement 8 Q. Okay. 1 think we're ready to move on now to 7 this process, that we were going to implement 9 the next section, which Mr. Sacuta is going to 8 whitever suit we go thad to be capable of 9 using the IUEBA. 10 Any replacement suit was also required to 11 mext section is to discuss personal 11 mext section and it's specific to passenger 12 A. The next section is to discuss personal 12 transportation suit standard, which is also a CCSB 13 As I discussed this morning, there are a 13 Transport Canda and the marine abnonment 16 As I discussed this morning in my 21 decision made to combine them? 22 regulatory overview section on that item, 21 decision made to compliant. 23 A. Right. 24 A. Right. 24 white we had some 25 O. Because it's hard to understand what the 25 Q. No, that's fine. I think we're had Mis2000 suit 11 A. A Lideestand the 2	5 MR. VOKEY:	5	5	It was identified in 2006, when we started
7 ROIL, Q.C.: 7 HUEBA and they wanted to make sure that 8 Q. Okay. I think we're ready to move on now to 6 7 HUEBA and they wanted to be capable of 9 Use of the next section, which Mr. Sacuta is going to 7 HUEBA and they wanted to make sure that 10 take us through. 11 MR. SACUTA: 10 12 A. The next section is to discuss personal 11 mest section, which Mr. Sacuta is going to 13 protective equipment for helicopter 14 mmesroin suit standard, which is also a CGSB 15 safety. 12 Transport Canada and the marine abandonment 16 As I discussed this morning, there are a 16 Marine. 17 number of regulations that require or outline 17 ROIL, QC:: 18 the requirements for passenger transportation 18 Q. Okay, and I think the next slide deals with a 19 guestion that I have and I'm sure other people 20 have had as well, and that is why was there a 21 slide. So it was talked this morning in my 22 SQ. REACUTA: 23 A. Right. 23 Question that I have and I'm sure other people 20 have had as well, and that. Page 204 1 rationale wa	6 A. Your chips are your early warning system.	. 6	5	this process, that we were going to implement
8 Q. Okay. I think we're ready to move on now to 8 whatever suit we got had to be capable of 9 the next section, which Mr. Sacuta is going to 10 Any replacement suit was also required to 11 MR. SACUTA: 10 Any replacement suit was also required to 12 A. The next section is to discuss personal 11 meet both the helicopter passenger 12 transportation suit system that was adopted by 13 safety. 11 meet both the helicopter passenger 14 transportation suit system that was adopted by Transport Canada and the marine bandonment 16 As I discussed this morning, there are a 11 Marine. 17 Rout, QC: 12 Ak and thich enxt slide deals with a 19 suits and I'm not planning to go through them 13 Q. Okay, and I think the next slide deals with a 19 regulatory overview section on that item, 21 decision made to combine them? 22 we don't need to take it line for 3 A. Right. 24 AK DI, QC: 24 A RU 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the <td>7 ROIL, Q.C.:</td> <td>7</td> <td>7</td> <td>HUEBA and they wanted to make sure that</td>	7 ROIL, Q.C.:	7	7	HUEBA and they wanted to make sure that
9 the next section, which Mr. Sacuta is going to take us through. 9 using the HUEBA. 10 take us through. 10 Any replacement suit was also required to the helicopter passenger 12 A. The next section is to discuss personal transportation and it's specific to passenger 11 Transport Canada and the marine abandonment 13 protective equipment for helicopter tarsportation and it's specific to passenger 13 Transport Canada and the marine abandonment 14 transport Canada 16 Marine. 17 15 safety. 0 Okay, and I think the next slide deals with a 19 19 16 matine. 17 ROIL, QC.: 18 0 Okay, and I think the next slide deals with a 19 21 slide. So it was talked this morning in my 22 regulatory overview section on that item, 23 20 No, that's fine. I think we've had some 21 A. Right. 23 Q. No, that's fine. I think we've had some 25 Q. Bocause it's hard to understand what the 26 Q. Decause it's hard to understand that. For operators undertook a joint bidding exercise. A. I understand that. For operators undertook a joint bidding exercise. 17 norta cost saving exercise. As a matter	8 O. Okay. I think we're ready to move on nov	w to 8	3	whatever suit we got had to be capable of
10 take us through. 11 MR SACUTA: 12 A. The basic scope of work was for the supply of 13 protective equipment for helicopter 14 transportation and it's specific to passenger 15 standard which was adopted by 16 As I discussed this morning, there are a 17 number of regulations that require or outline 18 As I discussed this morning in og ot brough them 20 all again, but they're all shown on this 21 stide. So it was talked this morning in my 22 regulatory overview section on that item, 23 all again, but they're all shown on this 24 ROIL, QC.: 25 Q. No, that's fine. I think we've had some 14 rationale was for that. 24 ROIL, QC.: 25 Q. No, that's fine. I think we've had some 14 Wastag fight commander suit and MS2000 suit 15 satadard whity, the operators used the 16 Mustag fight commander suit and MS2000 suit 18 Q. Okay. Now the Mustang suit, in terms of 19 operators undertook a joint bidding exervise	9 the next section, which Mr. Sacuta is going	g to 9)	using the HUEBA.
11 MR.SACUTA: 11 meet both the helicopter passenger 12 A. The next section is to discuss personal 11 meet both the helicopter passenger 12 A. The next section is to discuss personal 11 maport Canada and the marine abandonment 14 mapsortation and it's specific to passenger 11 maport Canada and the marine abandonment 14 mapsortation and it's specific to passenger 13 Transport Canada and the marine abandonment 16 As I discussed this morning, there are a 16 Marine. 17 number of regulations that require or outline 18 Q. Okay, and I think the next slide deals with a 18 using in put they're all shown on this 10 mares to mark the mast added this morning in my 21 regulatory overview section on that item, 21 Mc SACUTA: 23 unless you have any questions. 21 Mc SACUTA: 24 ROLL QC: 24 RA SACUTA: 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 25 us, but we don't need to take it line for 3 A. I understand that. For operational 4 MK SACUTA: 3 A. I understand that. For operational 7 until the fall of 2007. It was recognized 4 Mc SACUTA: 8 that the suits were coming to the end of their 5 suits to be compliant. 9 neckural flight commander suit and Ms2000 suit 10 more exis	10 take us through.	10)	Any replacement suit was also required to
12 A. The next section is to discuss personal protective equipment for helicopter transportation and it's specific to passenger transportation and it's specific to passenger transportation 12 transportation suit standard, which is also a CGSB 15 safety. 13 Transport Canada and the marine abandomment immersion suit standard, which is also a CGSB 16 As I discussed this morning, there are a 16 Marine. 17 number of regulations that require or outline 16 Marine. 18 the requirements for passenger transportation 19 question that the next slide deals with a 19 suits and I'm not planning to go through them 0 Okay, and I think the next slide deals with a 21 regulatory overview section on that item, 23 a. Right. 21 decision made to combine them? 22 Q. No, that's fine. I think we've had some 23 A. Right. 24 MCL QC: 25 Q. Because it's hard to understand what the 25 Q. Because it's hard to understand what the 2 a. Timitally, the operators used the 5 suits to be compliant. 3 A. I understand that. For operational 4 flexibility, the operators used the 5 suits to be compliant. 3 <	11 MR. SACUTA:	11		meet both the helicopter passenger
13 protective equipment for helicopter transportation and it's specific to passenger targets. 13 Transport Canada and the marine abandonment immersion suit standard, which is also a CGSB. 14 transport Canada and the marine abandonment immersion suit standard, which is also a CGSB. 15 safety. 15 16 As I discussed this morning, there are a number of regulations that require or outline. 16 18 the requirements for passenger transportation. 17 20 all again, but they're all shown on this 20 18 Q. Okay, and I think the next slide deals with a question that I have and I'm sure other people 20 21 regulatory overview section on that item, 22 Q. No, that's fine. I think we've had some 20 A. Right. 22 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 21 evidence before about that and rice to ferfesh 2 2. ME SACUTA: 3 A. I understand that. For operational 4 3 the suits were corning to the end of their 9 3 A. I understand that. For operational 4 4 flexibility, the operators used the end of their 9 3 A. I understand that. For operational 4 4 flexibility, the operators used took a joint bidding exercise i 13 5	12 A. The next section is to discuss personal	12	2	transportation suit system that was adopted by
14 transportation and it's specific to passenger safety. 14 immersion suit standard, which is also a CGSB 15 safety. 15 standard which was adopted by Transport Canada 16 As I discussed this morning, there are a 15 standard which was adopted by Transport Canada 16 As I discussed this morning in my 16 Marine. 17 number of regulations that require or outline 17 ROIL, Q.C. 20 all again, but they're all shown on this 19 question that I have and I'm sure other people 21 stick. So it was talked this morning in my 22 RR SACUTA: 23 A. Right. 24 ROIL, Q.C. 24 ROIL, Q.C. 24 ROIL, Q.C. 25 Q. Decause it's hard to understand what the 2 us, but we don't need to take it line for 3 A. I understand that. For operational 16 16 17 18 A. I understand that. For operational 6 abandonment immersion suit standard. This was 14 2 ME.SACUTA: 3 A. I understand that. For operational 16 16 17 18 A. I understand that. For operational 16 16 16 18 14 <td>13 protective equipment for helicopter</td> <td>13</td> <td>3</td> <td>Transport Canada and the marine abandonment</td>	13 protective equipment for helicopter	13	3	Transport Canada and the marine abandonment
15 safety. 15 standard which was adopted by Transport Canada 16 As I discussed this morning i, there are a 16 Marine. 17 number of regulations that require or outline 16 Marine. 18 the requirements for passenger transportation 16 Marine. 19 suits and I'm not planning to go through them 16 Marine. 20 all again, but they're all shown on this 19 question that I have and I'm sure other people 20 have had as well, and that is why was there a 20 have had as well, and that is why was there a 21 stidle. So it was talked this morning i my 21 decision made to combine them? 22 regulatory overview section on that item, 22 A. Right. 24 ROIL, Q.C: 24 ROIL, Q.C: 24 2 nust and nice to refresh 2 0. Because it's hard to understand what the 2 us, but we don't need to take it line for 3 A. I understand that. For operational 4 flexibility, the operators also specified the 5 suits to be compliant with the marine 6 Mustang flight commander suit and MS2	14 transportation and it's specific to passenge	r 14	Ļ	immersion suit standard, which is also a CGSB
16 As I discussed this morning, there are a 16 Marine. 17 number of regulations that require or outline 17 Rott, QC:: 18 the requirements for passenger transportation 18 Q. Okay, and I think the next slide deals with a 19 suits and I'm not planning to go through them 18 Q. Okay, and I think the next slide deals with a 20 all again, but they're all shown on this 18 Q. Okay, and I think the next slide deals with a 21 slide. So it was talked this morning in my 2 Meva had as well, and that is why was there a 21 evidence before about that and nice to refresh 2 M. R.SACUTA: 2 us, but we don't need to take it line for 3 A. I understand that. For operational 4 MR.SACUTA: 3 A. I understand that. For operational 4 flexibility, the operators also specified the 5 suits to be compliant with the marine 6 abandonment immersion suit standard. This was not a cost saving exercise. As a matter of 8 that the suits were coming to the end of their 9 standard? 10 initiative in the mid to late 2006, the 10 Roit, QC::	15 safety.	15	5	standard which was adopted by Transport Canada
17 number of regulations that require or outline 17 ROIL, Q.C.: 18 the requirements for passenger transportation 18 Q. Okay, and I think the next slide deals with a 19 suits and I'm not planning to go through them 19 question that I have and I'm sure other people 20 all again, but they're all shown on this 20 have had as well, and that is why was there a 21 slide. So it was talked this morning in my 22 MR. SACUTA: 23 unless you have any questions. 23 A. Right. 24 ROIL, Q.C.: 24 ROIL, Q.C.: 24 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 2 us, but we don't need to take it line for 3 A. I understand that. For operational 4 flexibility, the operators used the 5 suit sub er comigit on the end of their 9 useful life, so as a continuous improvement 10 standard? 14 MR.SACUTA: 11 Q. So the suit to be dual standard would have 12 operators undertook a joint bidding exercise 11 Q. So the suit to be dual standard would have <td< td=""><td>16 As I discussed this morning, there are a</td><td>16</td><td>ó</td><td>Marine.</td></td<>	16 As I discussed this morning, there are a	16	ó	Marine.
18 the requirements for passenger transportation 18 0. Okay, and I think the next slide deals with a 19 suits and I'm not planning to go through them 19 question that I have and I'm sure other people 20 all again, but they're all shown on this 19 have had as well, and that is why was there a 21 slide. So it was talked this morning in my 21 regulatory overview section on that item, 22 23 unless you have any questions. 23 A. Right. 24< ROIL, Q.C:	17 number of regulations that require or outlin	ne 17	ROIL.	0.C.:
19 suits and I'm not planning to go through them 20 all again, but they're all shown on this 21 slide. So it was talked this morning in my 22 regulatory overview section on that item, 23 unless you have any questions. 24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 26 evidence before about that and nice to refresh 2 us, but we don't need to take it line for 3 line, absolutely. Thank you. 4 MR.SACUTA: 3 A. Frior initially, the operators used the 6 Mustang flight commander suit and MS2000 suit 7 not a cost saving exercise. As a matter of 8 that the suits were coming to the end of their 9 useful life, so as a continuous improvement 10 initiative in the mid to late 2006, the 11 operators undertook a joint bidding exercise 12 Q. Okay. Now the Mustang suit, in terms of 13 ROIL, Q.C.: 14 Q. Okay. Now the Mustang suit, in terms of 15 A. That is the yellow suit? 17 MR. SACUTA:	18 the requirements for passenger transportati	ion 18	3 O.	Okay, and I think the next slide deals with a
20 all again, but they're all shown on this 21 slide. So it was talked this morning in my 22 regulatory overview section on that item, 23 unless you have any questions. 24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 26 Page 202 27 Q. No, that's fine. I think we've had some 28 Page 202 29 Q. Because it's hard to understand what the 29 Q. Because it's hard to understand what the 20 No, that's fine. I think we've had some 29 Q. Because it's hard to understand what the 20 No, that's fine. I think we've had some 21 evidence before about that and nice to refresh 2 us, but we don't need to take it line for 3 In. quertain that. For operational 4 flexibility, the operators also specified the 5 A. Prior initially, the operators used the 6 Mustang flight commander suit and Ms2000 suit 7 not a cost saving exercise. As a matter of 8 that the suits were coming to the end of their 9 standard compliant. <td>19 suits and I'm not planning to go through th</td> <td>iem 19</td> <td>)</td> <td>question that I have and I'm sure other people</td>	19 suits and I'm not planning to go through th	iem 19)	question that I have and I'm sure other people
21 slide. So it was talked this morning in my 21 decision made to combine them? 22 regulatory overview section on that item, 22 MR. SACUTA: 23 unless you have any questions. 23 A. Right. 24 ROIL, Q.C.: 23 A. Right. 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 2 evidence before about that and nice to refresh 2 MR. SACUTA: 3 A. I understand that. For operational 4 MR. SACUTA: 3 A. I understand that. For operational 4 4 MR. SACUTA: 3 A. I understand that. For operational 4 4 MR. SACUTA: 3 A. I understand that. For operational 4 6 Mustang flight commander suit and MS2000 suit 6 abandonment immersion suit standard. This was 7 not a cost saving exercise. As a matter of 8 fact, the suits were more expensive to be dual 9 useful life, so as a continuous improvement 9 standard compliant. 10 oreplace the existing Mustang suit. 10 ROIL, Q.C.: 13 ROIL	20 all again, but they're all shown on this	20)	have had as well, and that is why was there a
22 regulatory overview section on that item, 23 unless you have any questions. 24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 25 26 Q. No, that's fine. I think we've had some 25 27 us, but we don't need to take it line for 26 28 us, but we don't need to take it line for 26 3 line, absolutely. Thank you. 4 4 MR. SACUTA: 3 3 Initially, the operators used the 6 6 Mustang flight commander suit and MS2000 suit 3 7 until the fall of 2007. It was recognized 7 8 that the suits were coming to the end of their 9 9 useful life, so as a continuous improvement 10 10 initiative in the mid to late 2006, the 11 Q. So the suit to be dual standard compliant. 11 operators undertook a joint bidding exercise 11 Q. So the suit to be dual standard would have 12 cost you more per unit than simply to one 13 standard? 14 M. SACUTA: 14 M. SACUTA: 13	21 slide. So it was talked this morning in m	\mathbf{v} $\begin{vmatrix} \mathbf{z}_{0} \\ \mathbf{z}_{1} \end{vmatrix}$		decision made to combine them?
23 unless you have any questions. 23 A. Right. 24 ROIL, Q.C.: 23 A. Right. 25 Q. No, that's fine. I think we've had some 23 A. Right. 24 ROIL, Q.C.: 24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 24 ROIL, Q.C.: 24 ROIL, Q.C.: 25 Q. Because it's hard to understand what the 2 us, but we don't need to take it line for 3 A. I understand that. For operational 4 4 MR.SACUTA: 3 A. I understand that. For operational 4 6 Mustang flight commander suit and MS2000 suit 7 not a cost saving exercise. As a matter of 8 that the suits were coming to the end of their 9 standard compliant. 10 9 useful life, so as a continuous improvement 10 ROIL, Q.C.: 11 Q. So the suit to be dual standard would have 12 to replace the existing Mustang suit, in terms of 14 Q. So the suit to be dual standard would have 13 ROIL, Q.C.: 14 Q. Okay. Now the Mustang suit, in terms of <td>22 regulatory overview section on that item</td> <td>) 22</td> <td>2 MR S</td> <td>ACUTA:</td>	22 regulatory overview section on that item) 22	2 MR S	ACUTA:
24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 24 ROIL, Q.C.: 25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the Page 202 1 evidence before about that and nice to refresh 2 2 us, but we don't need to take it line for 3 3 line, absolutely. Thank you. 4 4 MR.SACUTA: 3 5 A. Prior initially, the operators used the 6 6 Mustang flight commander suit and MS2000 suit 7 7 until the fall of 2007. It was recognized 8 8 that the suits were coming to the end of their 9 9 useful life, so as a continuous improvement 10 10 initiative in the mid to late 2006, the 10 11 operators undertook a joint bidding exercise 11 12 to replace the existing Mustang suit, in terms of 13 13 ROIL, Q.C.: 13 14 Q. Okay. Now the Mustang suit? 14 14 Q. Okay. Now the Mustang suit? 14 17 MR.SACUTA: 15 1	23 unless you have any questions	23		Right
25 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the 225 Q. No, that's fine. I think we've had some 25 Q. Because it's hard to understand what the Page 202 1 evidence before about that and nice to refresh 2 Q. Because it's hard to understand what the 2 us, but we don't need to take it line for 3 A. I understand that. For operational 4 MR.SACUTA: 3 A. I understand that. For operational 4 MR.SACUTA: 3 A. I understand that. For operational 4 flexibility, the operators used the 5 suits to be compliant with the marine 6 Mustang flight commander suit and MS2000 suit 6 abandonment immersion suit standard. This was 7 not a cost saving exercise. As a matter of 8 fact, the suits were more expensive to be dual 9 useful life, so as a continuous improvement 10 ROIL, Q.C.: 11 Q. So the suit to be dual standard would have 12 to replace the existing Mustang suit, in terms of 14 MR.SACUTA: 13 X. That's so correct, and it was not the intent that 16 photographs that you see that are takena	24 ROIL OC:	24		
Page 202Page 202Page 204Page 2041colspan="2">colspan="2">Page 2041rational ewas for that.2us, but we don't need to take it line for3line, absolutely. Thank you.4MR. SACUTA:5A. Prior initially, the operators used the6Mustang flight commander suit and MS2000 suit7until the fall of 2007. It was recognized8that the suits were coming to the end of their9useful life, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, Q.C.:14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17TMR. SACUTA:18A. That is the yellow suit?17meck seal, a different design to the suit that19neck seal, a different design to the suit that20we've got today.21NR. SACUTA:22Q. Okay.23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of	25 0. No. that's fine. I think we've had some	25	5 O.	Because it's hard to understand what the
I age 2041evidence before about that and nice to refresh2us, but we don't need to take it line for3line, absolutely. Thank you.4MR.SACUTA:5A. Prior initially, the operators used the6Mustang flight commander suit and MS2000 suit7until the fall of 2007. It was recognized8that the suits were coming to the end of their9useful life, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, QC::14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR.SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, QC::21NR.SACUTA:22Q. Okay.23MR.SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of		2000 202	<u> </u>	Page 204
111	I avidance before about that and nice to refr	age 202		rationale was for that
2us, but we don't neer to take it line for2MR. SACUTA:3line, absolutely. Thank you.3A. I understand that. For operational4MR. SACUTA:3A. I understand that. For operational6Mustang flight commander suit and MS2000 suit4flexibility, the operators also specified the7until the fall of 2007. It was recognized5suits to be compliant with the marine8that the suits were coming to the end of their8fact, the suits were more expensive to be dual9useful life, so as a continuous improvement9standard compliant.10initiative in the mid to late 2006, the10ROIL, Q.C.:11operators undertook a joint bidding exercise11Q. So the suit to be dual standard would have12cost you more per unit than simply to one1313ROIL, Q.C.:13standard?14Q. Okay. Now the Mustang suit, in terms of15A. That's correct, and it was not the intent that16years ago, is that the yellow suit?14MR. SACUTA:17requirement we have offshore to maintain 2001818A. That is the yellow suit. It did also have a1919neck seal, a different design to the suit that1920we've got today.2121ROIL, Q.C.:2122Q. Okay.2223MR. SACUTA:2324A. The basic scope of work was for the supply of2425MR. SACUTA:23 <td>1 evidence before about that and fince to fello</td> <td></td> <td>MDC</td> <td></td>	1 evidence before about that and fince to fello		MDC	
3A. Prior initially, the operators used the4MR. SACUTA:5A. Prior initially, the operators used the6Mustang flight commander suit and MS2000 suit7until the fall of 2007. It was recognized8that the suits were coming to the end of their9useful life, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, Q.C.:14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR. SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, Q.C.:22Q. Okay.23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of	2 us, but we don't need to take it line for		2 MK. 5	L understand that For operational
4 MR. SACUTA:5 A. Prior initially, the operators used the6 Mustang flight commander suit and MS2000 suit7 until the fall of 2007. It was recognized8 that the suits were coming to the end of their9 useful life, so as a continuous improvement10 initiative in the mid to late 2006, the11 operators undertook a joint bidding exercise12 to replace the existing Mustang suits.13 ROIL, Q.C.:14 Q. Okay. Now the Mustang suit, in terms of15 photographs that you see that are taken a few16 years ago, is that the yellow suit?17 MR. SACUTA:18 A. That is the yellow suit. It did also have a19 neck seal, a different design to the suit that20 we've got today.21 ROIL, Q.C.:22 Q. Okay.23 MR. SACUTA:24 A. The basic scope of work was for the supply of	3 Inte, adsolutely. Thank you.	3	6 A.	flexibility the operators also specified the
5A. Prior initially, the operators used the5Suits to be compriant with the mattile6Mustang flight commander suit and MS2000 suitabandonment immersion suit standard. This was7until the fall of 2007. It was recognizednot a cost saving exercise. As a matter of8that the suits were coming to the end of theirfact, the suits were more expensive to be dual9useful life, so as a continuous improvementstandard compliant.10initiative in the mid to late 2006, theIn ROIL, Q.C.:11operators undertook a joint bidding exerciseIn Q. So the suit to be dual standard would have12to replace the existing Mustang suits.In Q. So the suit to be dual standard would have13ROIL, Q.C.:In Q. Okay. Now the Mustang suit, in terms ofIn MR. SACUTA:14Q. Okay. Now the Mustang suit, in terms ofIn MR. SACUTA:In A. That is the yellow suit?17MR. SACUTA:In this suit was ever going to replace the17MR. SACUTA:In requirement we have offshore to maintain 20018A. That is the yellow suit. It did also have aIn generation suits. There's a19neck seal, a different design to the suit thatIn generation suits. There's a20we've got today.20of 280, like we do at Hibernia, you have to21Now.21have -22Q. Okay.22ROIL, Q.C.:23MR. SACUTA:2324A. The basic scope of work was for the supply of2424A. The basic scope of work was	4 MR. SACUTA:	4	+	nexibility, the operators also specified the
6Mustang fight commander suit and MS2000 suit6abandomient miniersion suit standard. This was7until the fall of 2007. It was recognized7not a cost saving exercise. As a matter of8that the suits were coming to the end of their9not a cost saving exercise. As a matter of9useful life, so as a continuous improvement9standard compliant.10initiative in the mid to late 2006, the10ROIL, Q.C.:11operators undertook a joint bidding exercise11Q. So the suit to be dual standard would have12to replace the existing Mustang suits.12cost you more per unit than simply to one13ROIL, Q.C.:13standard?14Q. Okay. Now the Mustang suit, in terms of14MR. SACUTA:15photographs that you see that are taken a few16this suit was ever going to replace the17meck seal, a different design to the suit that19requirement we have offshore to maintain 20018A. That is the yellow suit. It did also have a19regulated requirement that if you've got a POB20we've got today.20of 280, like we do at Hibernia, you have to21ROIL, Q.C.:23Q. A POB, what's a POB?24A. The basic scope of work was for the supply of24MR. SACUTA:	5 A. Phot Initially, the operators used the) -	shandonment immension suit standard. This was
7Initial the fail of 2007. It was recognized8that the suits were coming to the end of their9useful life, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, Q.C.:14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR. SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, Q.C.:23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of	6 Wustang Hight commander suit and MS200)	abandonment miniersion suit standard. This was
8That the suits were conting to the end of them9useful life, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, Q.C.:14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR. SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, Q.C.:23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of	that the suite were coming to the end of the		/ >	foot the suite were more expensive to be duel
9userul me, so as a continuous improvement10initiative in the mid to late 2006, the11operators undertook a joint bidding exercise12to replace the existing Mustang suits.13ROIL, Q.C.:14Q. Okay. Now the Mustang suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR. SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, Q.C.:23MR. SACUTA:23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of	8 that the suits were coming to the end of the	311 8 mt 0	5	aton dord compliant
10Initiative in the find to fate 2006, the10 koll, Q.C.:11operators undertook a joint bidding exercise11Q. So the suit to be dual standard would have12to replace the existing Mustang suits.12cost you more per unit than simply to one13ROIL, Q.C.:13standard?14Q. Okay. Now the Mustang suit, in terms of14MR. SACUTA:15photographs that you see that are taken a few15A. That's correct, and it was not the intent that16years ago, is that the yellow suit?16this suit was ever going to replace the17MR. SACUTA:17requirement we have offshore to maintain 20018A. That is the yellow suit. It did also have a19regulated requirement that if you've got a POB20we've got today.20of 280, like we do at Hibernia, you have to21ROIL, Q.C.:23MR. SACUTA:23MR. SACUTA:23Q. A POB, what's a POB?24A. The basic scope of work was for the supply of24MR. SACUTA:	9 useful life, so as a continuous improvement			
11Operators undertook a joint bidding exercise11Q. So the suit to be dual standard would have12to replace the existing Mustang suits.12cost you more per unit than simply to one13ROIL, Q.C.:13standard?14Q. Okay. Now the Mustang suit, in terms of14MR. SACUTA:15photographs that you see that are taken a few15A. That's correct, and it was not the intent that16years ago, is that the yellow suit?15A. That's correct, and it was not the intent that17MR. SACUTA:16this suit was ever going to replace the18A. That is the yellow suit. It did also have a19percent abandonment suits. There's a19neck seal, a different design to the suit that19regulated requirement that if you've got a POB20we've got today.20of 280, like we do at Hibernia, you have to21ROIL, Q.C.:21have -22Q. Okay.22ROIL, Q.C.:23MR. SACUTA:23Q. A POB, what's a POB?24A. The basic scope of work was for the supply of24MR. SACUTA:	10 Initiative in the find to fate 2006, the	10) ROIL,	Q.C.:
1210 replace the existing Mustang suits.1112cost you more per unit than simply to one13ROIL, Q.C.:13standard?14Q. Okay. Now the Mustang suit, in terms of14MR. SACUTA:15photographs that you see that are taken a few15A. That's correct, and it was not the intent that16years ago, is that the yellow suit?15A. That's correct, and it was not the intent that17MR. SACUTA:16this suit was ever going to replace the18A. That is the yellow suit. It did also have a19percent abandonment suits. There's a19neck seal, a different design to the suit that19regulated requirement that if you've got a POB20we've got today.20of 280, like we do at Hibernia, you have to21ROIL, Q.C.:22Q. Okay.22Q. Okay.22ROIL, Q.C.:23MR. SACUTA:23Q. A POB, what's a POB?24A. The basic scope of work was for the supply of24MR. SACUTA:	to replace the existing Mustang suite	.50 11	Q.	So the suit to be dual standard would have
13 ROIL, Q.C.: 13 Standard? 14 Q. Okay. Now the Mustang suit, in terms of 14 MR. SACUTA: 14 MR. SACUTA: 16 years ago, is that the yellow suit? 15 A. That is the yellow suit. It did also have a 16 this suit was ever going to replace the 17 MR. SACUTA: 16 this suit was ever going to replace the 18 A. That is the yellow suit. It did also have a 19 percent abandonment suits. There's a 19 neck seal, a different design to the suit that 19 regulated requirement that if you've got a POB 20 we've got today. 20 of 280, like we do at Hibernia, you have to 21 ROIL, Q.C.: 21 have - 22 Q. Okay. 22 ROIL, Q.C.: 23 23 MR. SACUTA: 23 Q. A POB, what's a POB? 24 A. The basic scope of work was for the supply of 24 MR. SACUTA:	12 to replace the existing Mustang suits.	12	<u>,</u>	eter derd?
14Q. Okay. Now the Mustally suit, in terms of15photographs that you see that are taken a few16years ago, is that the yellow suit?17MR. SACUTA:18A. That is the yellow suit. It did also have a19neck seal, a different design to the suit that20we've got today.21ROIL, Q.C.:22Q. Okay.23MR. SACUTA:24A. The basic scope of work was for the supply of24A. The basic scope of work was for the supply of25We' Mathematic state stat	13 ROIL, Q.C.:	of 14		
 15 photographs that you see that are taken a few years ago, is that the yellow suit? 16 years ago, is that the yellow suit? 17 MR. SACUTA: 18 A. That is the yellow suit. It did also have a neck seal, a different design to the suit that 19 neck seal, a different design to the suit that 20 we've got today. 21 ROIL, Q.C.: 22 Q. Okay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 25 A. That's correct, and it was not the intent that 16 this suit was ever going to replace the 17 requirement we have offshore to maintain 200 18 percent abandonment suits. There's a 19 regulated requirement that if you've got a POB 20 of 280, like we do at Hibernia, you have to 21 have - 22 Q. Okay. 23 MR. SACUTA: 24 MR. SACUTA: 25 A. That's correct, and it was not the intent that 26 this suit was ever going to replace the 27 requirement we have offshore to maintain 200 28 OKay. 29 Not of 280, like we do at Hibernia, you have to 20 have - 22 Q. Okay. 23 MR. SACUTA: 24 MR. SACUTA: 25 A. The basic scope of work was for the supply of 24 MR. SACUTA: 	14 Q. Okay. Now the Mustaing suit, in terms	01 14	• MK. S	ACUIA:
16years ago, is that the yellow suit?16this suit was ever going to replace the17MR. SACUTA:17requirement we have offshore to maintain 20018A. That is the yellow suit. It did also have a18percent abandonment suits. There's a19neck seal, a different design to the suit that19regulated requirement that if you've got a POB20we've got today.20of 280, like we do at Hibernia, you have to21ROIL, Q.C.:21have -22Q. Okay.22ROIL, Q.C.:23MR. SACUTA:23Q. A POB, what's a POB?24A. The basic scope of work was for the supply of24MR. SACUTA:25MR. SACUTA:24MR. SACUTA:	15 photographs that you see that are taken a fo	ew 15	р А.	this suit was even going to replace the
 17 MR. SACUTA: 18 A. That is the yellow suit. It did also have a 19 neck seal, a different design to the suit that 20 we've got today. 21 ROIL, Q.C.: 22 Q. Okay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 25 A. The basic scope of work was for the supply of 26 MR. SACUTA: 27 A. The basic scope of work was for the supply of 28 MR. SACUTA: 29 A. The basic scope of work was for the supply of 20 A. The basic scope of work was for the supply of 21 MR. SACUTA: 22 A. The basic scope of work was for the supply of 24 MR. SACUTA: 25 A. The basic scope of work was for the supply of 24 MR. SACUTA: 	16 years ago, is that the yenow suit?	10)	uns suit was ever going to replace the
18 A. That is the yenow suit. It did also have a 18 percent abandonment suits. There's a 19 neck seal, a different design to the suit that 19 regulated requirement that if you've got a POB 20 we've got today. 20 of 280, like we do at Hibernia, you have to 21 NR. SACUTA: 23 Q. A POB, what's a POB? 24 A. The basic scope of work was for the supply of 24 MR. SACUTA:	17 MR. SACUTA:	1/		requirement we have on shore to maintain 200
 19 heck seal, a different design to the suit that 20 we've got today. 21 ROIL, Q.C.: 22 Q. Okay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 24 MR. SACUTA: 25 A. The basic scope of work was for the supply of 24 MR. SACUTA: 25 A. The basic scope of work was for the supply of 24 MR. SACUTA: 25 A. The basic scope of work was for the supply of 26 A. The basic scope of work was for the supply of 27 A. The basic scope of work was for the supply of 28 A. The basic scope of work was for the supply of 29 A. The basic scope of work was for the supply of 20 A. The basic scope of work was for the supply of 21 A. The basic scope of work was for the supply of 24 MR. SACUTA: 	A. That is the yellow suit. It did also have a	18	5	regulated requirement that if you've get a DOD
20 We ve got today. 21 ROIL, Q.C.: 22 Q. Okay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 24 A. The basic scope of work was for the supply of 24 MR. SACUTA: 25 Vert and the supply of 26 Vert and the supply of 27 Vert and the supply of 28 Vert and the supply of 29 Vert and the supply of 20 Vert and the supply of 21 Net and the supply of 24 Net and the supply of 26 Vert and the supply of 27 Vert and the supply of 28 Vert and the supply of 29 Vert and the supply of 20 Vert and the supply of 21 Vert and the supply of 24 Vert and the supply of 25 Vert and the supply of	19 neck seal, a different design to the suit that	19	,	of 280 like we do at Ukamia way have to
21 ROIL, Q.C.: 22 Q. Okay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 24 A. The basic scope of work was for the supply of 25 WR. SACUTA: 26 WR. SACUTA: 27 WR. SACUTA: 28 WR. SACUTA:	20 we ve got today.)	or 200, like we do at Hiderina, you have to
22 Q. OKay. 23 MR. SACUTA: 24 A. The basic scope of work was for the supply of 24 A. The basic scope of work was for the supply of 24 A. The basic scope of work was for the supply of 25 V. A. The basic scope of work was for the supply of 26 V. A. The basic scope of work was for the supply of 27 V. A. The basic scope of work was for the supply of 28 V. A. The basic scope of work was for the supply of 29 V. A. The basic scope of work was for the supply of 29 V. A. The basic scope of work was for the supply of 29 V. A. The basic scope of work was for the supply of 20 V. A. The basic scope of work was for the supply of 20 V. A. The basic scope of work was for the supply of 20 V. A. The basic scope of work was for the supply of 21 V. A. The basic scope of work was for the supply of 22 V. A. The basic scope of work was for the supply of 23 V. A. The basic scope of work was for the supply of 24 V. A. The basic scope of work was for the supply of 25 V. A. The basic scope of work was for the supply of 26 V. A. The	21 KUIL, Q.C.:			
25 WK. SACUTA: 23 Q. A POB, what s a POB? 24 A. The basic scope of work was for the supply of 24 MR. SACUTA:	22 Q. UKAY.	22	ι KUIL,	Q.U.: A pop what's a pops
A. The basic scope of work was for the suppry of 24 MK. SACUTA:	25 MK. SACUTA:	$V \text{ of } \begin{bmatrix} 23 \\ 24 \end{bmatrix}$	VQ.	A PUB, WHAT S & PUB?
125 heliconfer passenger suits compliant with the 125 A. It you've got personnel on board corry	A. The basic scope of work was for the supply heliconter passenger suits compliant with t	the $\frac{24}{25}$		If you've got personnel on board sorry

Ja	nuary 11, 2010 M	Iulti-F	age [™]	Offshore Helicopter Safety Inquiry
	Page	205		Page 207
1	that's 280, you need to have 200 percent of		MR. S	ACUTA:
2	those, of that number in abandonment suits.		2 A.	Flexibility.
3	ROIL, O.C.:		8 ROIL	0.C.:
4	0. So if you have 280 personnel working on board	d. 4	↓ 0.	Flexibility.
5	you have to have 560 of those suits?		5 MR. S	ACUTA:
6	MR. SACUTA:		б А.	Flexibility during crew changes.
7	A. Right.	-	ROIL	0.C.:
8	ROIL. O.C.:	8	3 O.	An expensive flexibility. I take it.
9	O. And so this was not meant to be part of that		MR. S	ACUTA:
10	complement?	10) A.	Yes.
11	MR. SACUTA:	11	ROIL.	0.C.:
12	A. This was not meant to eliminate the what we	2 12	2 0.	Okay.
13	currently in place on board the facilities.	13	MR. S	ACUTA:
14	ROIL, Q.C.:	14	A.	So as far as the process -
15	Q. Okay.	15	ROIL	Q.C.:
16	MR. SACUTA:	10	5 Q.	Sorry, just one other thing. There was some -
17	A. What it did recognize is that we would have	17	,	- not some, there's been a lot of talk about
18	the ability to use this suit during both	18	3	the thermal qualities and Mr. Decker, I think,
19	helicopter transportation and during	19)	in this evidence indicated that he would have
20	situations where we crew changed by vessel, if	20)	preferred to have the immersion suit because
21	there was weather that would not allow	2	l	he felt it had higher thermal standards. Do
22	helicopter transportation. Then we were able	22	2	you know anything about the thermal standard
23	to take the suit and use a suit that was	23	3	issue?
24	already certified for marine evacuation or	24	MR. S	ACUTA:
25	marine abandonment as well as part of your	25	5 A.	Both comply with the same standard, so they
	Page	206		Page 208
1	vovage on the supply vessel when you were			meet the same standard.
2	transiting to the facility.		2 ROIL	O.C.:
3	ROIL, O.C.:		3 O.	So if it's properly fitting -
4	O. Okay. So if I was going out by supply boat	4	MR. S	ACUTA:
5	but I might be coming back by helicopter, the		5 A.	Yes.
6	helicopter alone suit would not be permitted		5 ROIL.	Q.C.:
7	for me to transport on the vessel?		7 Q.	- they should both provide the same amount of
8	MR. SACUTA:	8	3	thermal capacity?
9	A. That's correct.	Ģ	MR. S	ACUTA:
10	ROIL, Q.C.:	10) A.	Yes, that's correct.
11	Q. If it was a single standard?	11	ROIL,	Q.C.:
12	MR. SACUTA:	12	2 Q.	To a user.
13	A. That's correct.	13	MR. S	ACUTA:
14	ROIL, Q.C.:	14	A.	That's correct. I think the key statement
15	Q. And then I'd have to have an aviation standard	1.	5	there is if the suit fits properly, yes,
16	one to fly back?	10	5	that's true, and we'll talk about that a
17	MR. SACUTA:	17	7	little bit further.
18	A. Correct.	18	8 ROIL,	Q.C.:
19	ROIL, Q.C.:	19) Q.	Yeah, okay, thank you.
20	Q. My marine one would not work?	20) MR. S	ACUTA:
21	MR. SACUTA:	21	A.	The selection of the new suit, we did issue an
22	A. Correct.	22	2	expression of interest in early 2006 and the
23	ROIL, Q.C.:	23	3	evaluation included that there would be
24	Q. Okay. So this was what, to just provide	24	ŀ	Transport Canada certification requirements,
25	additional -	25	5	i.e. the suit design, the weight, the need for

January 11, 2010 Multi-		Iulti-F	Page	^M Offshore Helicopter Safety Inquiry
	Page	209		Page 211
1	a lifeiacket, the location of that lifeiacket.		1	an opportunity for them to say that?
	the durability thermal protection et cetera		2 MR	PRITCHARD
3	We wanted to make sure the manufacturer had	a 3	2 ΜΠΟ. 3 Δ	Ves
	quality control processes in place and there		1 MR	SACUTA:
	was a requirement for the turnaround time to		τ IVIIC.	I'm sure there would have been an opportunity
	transition to the new suit for the suit		с п	but I'm not aware that that was raised at any
	implementation. There were four potential	-	7	point in this process
	suppliers who hid. There were four potential			point in this process.
	suppliers who bld. There was a detailed			Clear thank you
9	hidden and Hally. Hansen was awarded the			
	the E452 suit obtained the highest technical	• 10) MR.	After the contract was awarded, there was an
	the E432 suit obtained the highest technical	1		. After the contract was awarded, there was an
12	score and Helly Hansen was recommended as	the 12	2	implementation plan put in place. They were -
13	preferred supplier for our new nelicopter	1.	3	- Helly Hansen was awarded the contract in May
14	transportation suits.	14	1 -	of 2007 and they were to supply approximately
15	ROIL, Q.C.:	15	5	1400 suits to meet the needs of the existing
16	Q. Okay. Now often during bidding processes,	16	5	operators and any ad-hoc supply for short term
17	there are bidders meetings and there's an	17	7	programs. A transition plan was developed as
18	exchange of information or concerns and that	18	3	the new suits were manufactured. Helly Hansen
19	sort of thing. During the stage from the	19	Ð	worked in parallel with the Marine Institute's
20	issuing of the expression of interest until	20)	east coast training programs, in other words
21	when Helly Hansen was chosen to supply the	e 21	l	the basic survival training. There was a
22	E452, did anybody explain or say to either one	22	2	transition strategy developed to introduce the
23	of you, I guess, and who led this? Was	23	3	E452 suit to the offshore workforce and the
24	there one company that led or did -	24	1	new suits were introduced on November 1st,
25	MR. SACUTA:	25	5	2007. I'd also like to comment, Mr.
	Page	210		Page 212
1	A. It was a joint effort.	1	1	Commissioner, that although the Marine
2	ROIL, Q.C.:	2	2	Institute was not involved in the bidding and
3	Q. It was done by a committee, was it, of -	3	3	selection process, they were informed of the
4	MR. SACUTA:	4	1	transition plan well in advance of November
5	A. Yes.	4	5	1st, 2007.
6	ROIL, Q.C.:	e	5	I know that Helly Hansen was actually
7	Q. Okay. Was there any expression of concern		7	here and had a suit during testimony. I'm
8	about the fact that the two standards would	8	3	just going to highlight a couple of the key
9	create a problem or an issue or a concern for	q)	components of the E452 suit. It does have an
10	the manufacturer?	10)	integral hood with a zipper that creates a
11	MR. SACUTA:	11	1	face seal, as opposed to the old suit that had
12	A. I'm not aware of any concern that was raised.	12	2	a neck seal assembly. It has a PLB, a
13	I'm not sure if either of the other panel	13	3	personal locator beacon. You can see this
14	members were.	14	1	also the HUEBA, how the HUEBA sits onto the
15	MR PRITCHARD	14	5	suit Also included is a splash guard and an
16	A. I'm not aware of anything.	16	ń	integrated carbon dioxide inflated lifeiacket
17	MR VOKEY	17	7	is all part of the suit. I wasn't planning to
18	A No	18	2	go through it any further because I know that
19		10))	Helly Hansen did testify
$ _{20}^{1}$	O I'm not suggesting there is but I'm just	20	, BUI	
$ _{21}^{20}$	wondering you know it's often been said that			Yes indeed You have to wear or have had to
22	a camel is a horse designed by a committee		- y	wear one of these I presume Mr Sacuta?
23	and sort of if somebody felt that marrying the		- 3 MR	SACUTA.
$ _{24}^{23}$	two standards was not a good thing then I	2	1 A	Every time I fly offshore I wear one
25	suppose there was but would there have bee	n \int_{24}^{24}	5 RUI	
1-5	suppose there was but would there have been			-, -, -, -, -, -, -, -, -, -, -, -, -, -

Janua	ry 11, 2010	Multi-	Page TM	¹ Offshore Helicopter Safety Inquiry
	F	Page 213		Page 215
1 0	. Yes, exactly. You know, there's been a lo	t of	1	the face in the face area.
2	talk about it now in hindsight and a lot of		2 ROIL	, Q.C.:
3	criticism of it. When it was first introduced	b	3 Q.	Were you able to choose your own size? When
4	to the workforce, was there general accept	ance	4	you were fitted, do you recall whether you
5	of the concept of a new suit or was that		5	actually just went in and tried one on and
6	resistant right from the beginning?		6	said "no, I want a bigger or smaller one" or
7 MR.	SACUTA:		7	was there somebody that helped you choose a
8 A	. I don't think I'm not aware of any		8	size?
9	resistance. Of course, I didn't come bac	k	9 MR. S	SACUTA:
10	until October of 2007, but I know that we	had 1	0 A.	There was a Helly Hansen person there that
11	a communication plan which included dev	eloping 1	1	helped me, you know, made sure that the suit
12	I'm speaking for Hibernia now a	1	2	seemed to fit properly. On that first flight
13	PowerPoint package which was sent offshe	ore to 1	3	that I went offshore after I returned, there
14	be used for the JOHS committees. It was all	lso 1	4	was a Helly Hansen representative there to
15	used by each of the individual departments	s at 1	5	check the suit size for me. Although I can
16	their department safety meetings. The	e 1	6	say that it would have been easy for me, I
17	presentation was also posted on the offsho	ore 1	7	think, to say "this suit doesn't fit quite
18	bulletin board so that all personnel could	l 1	8	right. Can I have a different suit?" after
19	have access to the whole transition plan as	nd 1	9	this transition period, and I think that's one
20	the fact that we were introducing a new su	it. 2	20	of the issues that came up as part of our
21	I think most people understood that	2	1	follow up after March 12th.
22	transportation suits don't last forever, that	2	2 ROIL	, Q.C.:
23	the Mustang suits we'd had in place for	a 2	23 Q.	Okay. We'll perhaps get onto that when we get
24	number of years, since initial oil production	n 2	.4	to that slide.
25	for Hibernia. It was time for a new suit. S	o 2	5 MR. S	SACUTA:
	F	Page 214		Page 216
1	I think most people understand that proces	s.	1 A.	Personal locator beacons, all E452 suits are
2 ROII	L, Q.C.:		2	fitted with a personal locator beacon. It's a
3 Q	. And the whole issue of the fit, which we r	now	3	Sea Marshall is the manufacturer. It
4	know was a concern or has turned out to b	be a	4	automatically activates on contact with water.
5	concern, was there any flags went up to yo	ou to	5	Should you enter the water, it activates. The
6	say to you right from the beginning, you k	now,	6	signal facilitates rescue of personnel on the
7	there's problems with this suit because	;	7	water surface, but it's important to note that
8	they're not fitting properly?		8	these PLBs are not designed to operate at
9 MR.	SACUTA:		9	submerged water depths. They're designed to
10 A	. I think the issues with the suit fitting came	: 1	0	indicate the location of an individual who's
11	after March 12th. We did have, during th	nis 1	1	at the surface. Should you be submerged, you
12	transition, during the transition to the new	1	2	will not receive the signal.
13	suit, Helly Hansen personnel at Cougar to	make 1	3 ROIL	, Q.C.:
14	sure that everybody put on, had an appropri-	riate 1	4 Q.	Okay. If the vehicle is submerged and a
15	sized suit. There may not have been th	e 1	5	person is in it, how would the vehicle be
16	appropriate focus on the face seal of that	1	.6	found?
17	suit. I think we've discovered after the	1	7 MR. S	ACUTA:
18	fact, based on and we'll talk about this a	l 1	8 A.	The helicopter itself has also has locator
19	little bit later in the package that there	(1)	9	beacons. So you can locate the helicopter
20	were some concerns with making sure that	11 the 2	20	usen, and that was done on subsequent follow
21	and I think there were also serves the	suit, $ 2\rangle$		up to locate the hencopier after March 12th.
$ _{22}^{22}$	with personnal that were asso some other issu	2	Z KUIL	, Q.C.: Dight
23	different sizes of suits based on comfart		JUNE C	
24	opposed to paying attention to the seeling	of $\frac{2}{2}$	4 IVIK. S	There was a lot of talk early on in the days
145	opposed to paying attention to the sealing	01 2	. п.	There was a for or tark early on in the days

Jan	uary 11, 2010 N	Aulti-P	age™	¹ Offshore Helicopter Safety Inquiry
	Page	217		Page 219
1	after the crash that the PLBs did not work on	1		The overall process of evaluating an
2	the suits, but Cougar has verified and	2		appropriate breathing device and the eventual
3	confirmed that the PLBs functioned properly	3		selection of HUEBA was a very complex process.
4	for those that were at the surface.	4		This included engagement of a CAPP led
5 R		5		committee, complete with representatives from
6	O Okay There was also some evidence about th	ne 6		the industry The process also included
	locator beacon being a little different one			evaluating both the compressed air and a
	here than it is in Nova Scotia Do you have	8		rebreather device and making the
9	anything to offer on that? Different	9	,	recommendation on the style to be used in the
10	frequencies or something I'm not sure if	10		Newfoundland basin There were associated
11	much turns on it but -	11		risks that were reviewed against the
12 N	AR SACUTA:	12		following: the operational performance of the
13	A Different frequencies Yeah they do have a	13		units: the training requirements whether you
14	different PLB being in use in Nova Scotia	14		needed to train under one metre or over one
15	waters It is a satellite tracked PLB There	15		metre of depth: there were medical
16	are some differences. One of the differences	16		requirements to consider for both under one
17	it actually requires the individual to	17		metre and over one metre and a medical
18	activate it as opposed to automatic	18		screening that needed to be complete. Safety
19	activation when it enters the water. So that	19		was our main concern throughout this process.
20	does raise some concerns for. I think, the	20		In 2006, it was determined that the compressed
21	operators in the Newfoundland area that if a	21		air device, the HUEBA, was the best available
22	person is unconscious, he won't be able to	22		technology. It is my understanding that the
23	activate the beacon itself. So we've decided	23		Newfoundland basin operators are the first
24	to stay with the beacon that we have now. We	24		civilian use of the HUEBA device.
25	know that eventually we'll have to get another	r 25	ROIL	, Q.C.:
	Раде	218		Page 220
1	beacon as these beacons age and then we'll	1	0	In the what in the world or -
$\begin{vmatrix} 1 \\ 2 \end{vmatrix}$	look at the available technology the next time	2	MR S	SACUTA:
3	we go through the process of determining the	3	Δ	In the world, my understanding. The military
4	best PLB for our service	4		has used the device, but my understanding is
5 R		5		we are the first civilian use of the HUEBA
6	O. Right. Okay. that's. I think, all we need to	6		After the HUEBA was selected, the committee
7	have on personal beacons today.	7		had to identify and address the specific
8 N	IR. SACUTA:	8		medical concerns associated with completing
9	A. Okay. Next, helicopter underwater emergenc	y 9	1	training and needed to determine a mitigation
10	breathing apparatus or the HUEBA. The HUEBA	A 10)	plan to lower the risk to the workforce. We
11	provides the user with an additional supply of	11		did not want to introduce a new piece of
12	breathable air. Depending on the individual,	12		safety related equipment without fully
13	you could get 25 or 30 breaths. It depends on	13		determining the risks and identifying the
14	how excited you are. The process has been	14		required mitigation plan. Quite simply, we
15	under way since 2000. The implementation pla	an 15		wanted to do it once and we wanted to do it
16	was in the final stages as when the helicopter	16		right.
17	crashed on March 12th. As of October 1st, all	17		Based on the length of time it took to
18	personnel travelling to and from offshore via	18		implement the HUEBA, CAPP is developing the
19	helicopter are required to have both HUEBA	19		terms of reference for a review to identify
20	training and they're required to wear a HUEBA	20)	the lessons learned from the assessment,
21	device on their E452 suit.	21		decision making and implementation process.
22	As far as the implementation goes, the	22		It is anticipated this review will include a
23	operators acknowledge that the decision to	23		final report complete with recommendations for
24	implement the HUEBA did take a long period o	of 24		improvement and we expect the target date for
25	time. Mr. Commissioner, it took too long.	25		completion will be in the second quarter of

Ja	nuary 11, 2010 Mu	lti-P	ag	e TM	Offshore Helicopter Safety Inquiry
	Page 22	21			Page 223
1	2010.	1			unacceptable level of risk to complete the
2	ROIL OC	2)		training The main benefit of the training is
3	O Can I take it that you would undertake to	3	3		to provide the passenger with an understanding
4	provide that document to us once it is	4	Ĺ		of the steps, they must consider in order to
5	complete?	5	5		prepare for an escape situation At this
6	MR SACUTA:	6	ń		time it is not intended to provide aircraft
	A I'll take it with legal counsel but I would	7	, ,		specific training in all egress situations
	expect we would ves		2		which may occur
	ROLL OC:	0	,) P(л II	
	O Veah My objective here is not to criticize	10)	0	What does that mean?
11	the past, but to make sure, that we all learn	11	, M	ي. R ۲	
$ _{12}^{11}$	from what's gone on in the past so that we can	12)	К. 57 Л	That means that the training that we have in
12	avoid -	12	2	А.	Newfoundland through the Marine Institute is
11	MP SACUTA:	14	,		not aircraft specific. It hasn't been for
14	A And that is our desire as well As I've	14	r		vears whether it was the Super Puma or now
15	mentioned it took too long and we need to do	15	,		the S 02. The overall training was to provide
10	a lessons learned review of that to make sure	17	,		the individual with the basic steps, that he
10	that it doesn't happen again	10	, >		needs to take for a successful agrees. It was
10	POIL O.C.	10	, ,		not to train for the inclusion of the
20	O And again I think if we had access to that	20	,)		auxiliary fuel tenk for windows the same size
20	Q. And again, I think If we had access to that, it would be very helpful in our objectives and	20)		auxiliary fuel tank, for windows the same size
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	I'm sure that we can work with your counsel on	21)		a high level training program
$\begin{vmatrix} 22\\ 22 \end{vmatrix}$	This sure that we can work with your counsel on that	22	2 2 DC	211	
23	III.	23		JIL, I	Q.C.: So the objective I take it from the cil
24	MR. SACUTA:	24	+	Q.	companies' concern at least is not so much
23	A. So the next section is qualifications and	23)		companies concern at least, is not so inden
	Page 22	22			Page 224
1	training required to work offshore.	1			fidelity, which I think was a word that we
2	ROIL, Q.C.:	2	2		used before, but rather learning the processes
3	Q. Okay. Now I've been cautioned that we should	3	3		and steps that one should go through?
4	never run over time for various reasons,	4	I MI	R. SA	
5	including the commitment of our broadcasters,	5	5	A.	That's correct.
6	so we'll go on for a little bit longer and	6	5 R(JIL, (Q.C.:
7	then when we find a convenient place to stop	7	7	Q.	And so whether you had a Super Puma at the
8	for our afternoon break, we should.	8	3		airport or a Eurocopter or a not the
9	MR. SACUTA:	9)		Eurocopter, but if you had a Super Puma or a
10	A. Okay. You just let me know.	10)		61 or a 92 -
11	ROIL, Q.C.:	11	M	R. SA	ACUTA:
12	Q. Yean. I'm the policeman on that and I've been	12	2	Α.	Right.
13	told to watch the clock, so I will.	13	B RC	JIL, (Q.C.:
14	MR. SACUTA:	14	ŀ	Q.	- you were not focused on trying to make the
15	A. Okay. All the training which is completed as	15) - 1 -	D G	training -
16	part of the basic survival training program is	16	• MI	R. SA	ACUTA:
1/	tailored to a controlled differing situation.	1/		A.	That's correct.
18	It's purpose is to provide the individual with the basic requirements to aid in the	18	S RC	JIL, (Q.C.: Usering loomed what way?see loomed after the
19	the basic requirements to aid in the	19	,	Q.	maying learned what you ve learned after the
$ ^{20}_{21}$	successful egress from a difficient neitcopter.	$ ^{20}_{21}$)		time is there any intention to shares that?
$\begin{vmatrix} 21\\ 22 \end{vmatrix}$	speed impact or crash situation. During		։) Ն./Դ	ם פי	
$\begin{vmatrix} 22 \\ 22 \end{vmatrix}$	spece impact of clash shuallon. During training personnal are exposed to a level of		2 IVI. 2	к. SA л	We'll certainly discuss that later in the
23	rick. The desire is to provide an accontable	23	, I	А.	presentation. So if I could defar to a
$\begin{vmatrix} 24 \\ 25 \end{vmatrix}$	level of training without introducing an	24	r S		further a later section in the
120	iever of duming without mitoutoning un	140			

Ja	nuary 11, 2010	Multi-P	Pa	ige	Offshore Helicopter Safety Inquiry
	Pag	e 225			Page 227
1	presentation?	1	1		1st since March of 2001 and it was revised
2	ROIL O.C.:		2		in 2008.
3	0. Absolutely. Excuse me. I just wonder if the		3	ROIL	0.C.:
	vibrating thing in the back of the room could	4	4	0	We talked earlier about engagement of the
5	be encouraged to stop Somebody's got a	5	5	×۰	workforce. Is there any level or place where
6	Blackberry there that's just going and going	6	6		engagement of the workforce happens in the
	Okay go ahead		7		process of setting forth this training and
	MR SACUTA:	,	, 8		qualifications standard?
9	A All operators ensure that the training and		9 9	MR S	ACUTA:
10	qualifications of personnel in their	10	0	A	I mean representatives from the operators
11	operations comply with the requirements of t	he 11	1		would include users of the system.
12	CAPP training and qualifications guidelines	12	2	ROIL	0.C.:
13	and operators specific requirements for	13	3	0	Yes
14	offshore training The industry regulators	14	4	MR S.	ACUTA
15	and the training institutions continually	15	5	A	So workforce can be anyone who in any way has
16	review training requirements for offshore	16	6		to travel offshore. So there were people that
17	workers through the CAPP Training and	17	7		were people that would be required to
18	Qualifications Committee	18	, 8		complete the complaining (sic) that were part
19		10	9		of the committees. They might not necessarily
$ _{20}^{17}$	0 Is that the best place for this work to be	20) N		have been the offshore workforce but they
$ _{21}^{20}$	done in your view?	21	1		were people that would have had to have gone
$\begin{vmatrix} 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 \\ 2 $	MR SACUTA	21	1 7		through the training in order for them to
23	A In my opinion the Training and Qualification	ns 23	3		travel offshore like some of our safety
$\begin{vmatrix} 23 \\ 24 \end{vmatrix}$	Committee is an example of how working w	with 24	1		health and environmental professionals
25	CAPP works. They've put together a very	25	5	ROIL	0.C.:
\vdash		- 226			Dago 228
	rag	e 220	1	0	Page 220
	avample where CAPP has affectively facilitate		l n	Q.	know you don't get the opportunity to have
	an availant pieze of work		2		somebody who has actually travelled. So are
	an excellent piece of work.	2	3 1		some body who has actually travelled. So are
	KOIL, Q.C.:		4		there for at least some of these
	yoll in your view?) с		representatives to be people who have actually
	MD SACUTA.		7		travelled?
	MR. SACUTA: $1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 $		/ 0	MDC	
	POUL O.C.		0. n	MR. 57	Well I say a number of those representatives
10	NOIL, Q.C.:	10	9	А.	do travel offshore with a regular frequency
	MD SACUTA:	11	1	DOII	
12	A So the Standard Practice for Training and	12	י ר	NOIL,	Okay. This might be the place where we should
12	Qualifications of Personnel provides guidance	Pe 13	2	Q.	take our break for the afternoon Mr
11	on the appropriate level of training required		3 1		Commissioner
15	for the offshore workforce is a joint	15	+ 5	COM	
16	committee effort by the operators the CAODO	16	5	0	Okay
17	which is the Canadian Association of Oilwel	10 17	7	v	(BREAK)
18	Drilling Contractors CAPP the regulatory	18	, 8 .	ROIL	
19	bodies and the training institutions. It	19	9	0.	Mr. Sacuta when you're ready. I think we're at
20	outlines the required qualifications and	20	0	κ.	slide No. 84?
$ _{21}^{-3}$	certified safety training for east coast	21	1	MR. S.	ACUTA:
22	offshore workers, including offshore travel.	22	2	A.	That's correct. The offshore work site in
23	It outlines the approved training institutions	23	3		Newfoundland and Labrador area existed in a
24	for a course as deemed necessary by the	24	4		remote and potentially hostile environment.
25	committee. It has been in effect since March	25	5		Adverse weather may cause delays in medical

Page 229Page 231evacuation and convert a minor medical problem1this is a person who is on board a facility2into a major emergency. As such, medical2who needs to be evacuated because of their3assessments are conducted for all personnel3medical condition at that time?4prior to their travel to an offshore work4MR. SACUTA:5site. The objectives of a medical assessment5A. Yes, we do have a nurse on board our6are to ensure that designated offshore6facilities to provide medical services, they7personnel are medically fit to work safely at7are in direct consultation with AOMS on shore8an isolated location with job accommodation8and in that review, they may decide that it is9where possible, to anticipate and where9best suited to medivac the individual if the10possible prevent the avoidable occurrence of10medical service provide offshore isn't11ill health offshore which would place the11sufficient.12individual, his or her colleagues and the12ROIL, Q.C.:13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helico
1evacuation and convert a minor medical problem1this is a person who is on board a facility2into a major emergency. As such, medical3who needs to be evacuated because of their3assessments are conducted for all personnel3medical condition at that time?4prior to their travel to an offshore work3medical condition at that time?5site. The objectives of a medical assessment5A. Yes, we do have a nurse on board our6are to ensure that designated offshore6facilities to provide medical services, they7personnel are medically fit to work safely at8and in that review, they may decide that it is9where possible, to anticipate and where9best suited to medivac the individual if the10possible prevent the avoidable occurrence of10medical service provide offshore isn't11ill health offshore which would place the11sufficient.12individual, his or her colleagues and the12ROIL, Q.C.:13emergency rescue service at risk, to provide15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical20there, I don't know whether it would be a21<
2into a major emergency. As such, medical assessments are conducted for all personnel 42who needs to be evacuated because of their medical condition at that time?3assessments are conducted for all personnel 4prior to their travel to an offshore work site. The objectives of a medical assessment are to ensure that designated offshore f2who needs to be evacuated because of their medical condition at that time?4MR. SACUTA:5A. Yes, we do have a nurse on board our 66are to ensure that designated offshore an isolated location with job accommodation 98an isolated location with job accommodation 99where possible, to anticipate and where 109best suited to medivac the individual if the 910possible prevent the avoidable occurrence of 1111sufficient.12individual, his or her colleagues and the 1212RoIL, QC:13emergency rescue service at risk, to provide 140Q. Right. You've indicated that the primary purpose of AOMS is to determine whether the purpose of AOMS is to determine whether the person is fit medically to work offshore.16meet any regulatory requirements for hazard 1716Have you ever in, again the three of your to assessments and determine fitness to work 2019Services to complete work or medical 2119don't know whether it would be a there, I don't know whether it would be a 2122service provider who is familiar with the 2222but I'm just wondering have you ever had a situation where a person was able to work but 23
3assessments are conducted for all personnel3medical condition at that time?4prior to their travel to an offshore work5site. The objectives of a medical assessment3medical condition at that time?5site. The objectives of a medical assessment6are to ensure that designated offshore6A. Yes, we do have a nurse on board our6are to ensure that designated offshore6facilities to provide medical services, they7personnel are medically fit to work safely at7are in direct consultation with AOMS on shore8an isolated location with job accommodation8and in that review, they may decide that it is9where possible, to anticipate and where9best suited to medivac the individual if the10possible prevent the avoidable occurrence of10medical service provided offshore isn't11ill health offshore which would place the12ROIL, Q.C:12emergency rescue service at risk, to provide12Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know whether it would be a20assessments and determine fitness to work20there, I don't know whether it would be a <tr< td=""></tr<>
4prior to their travel to an offshore work4MR. SACUTA:5site. The objectives of a medical assessment5A. Yes, we do have a nurse on board our6are to ensure that designated offshore6facilities to provide medical services, they7personnel are medically fit to work safely at7are in direct consultation with AOMS on shore8an isolated location with job accommodation8and in that review, they may decide that it is9where possible, to anticipate and where9best suited to medivac the individual if the10possible prevent the avoidable occurrence of10medical service provided offshore isn't11ill health offshore which would place the11sufficient.12individual, his or her colleagues and the12ROIL, Q.C:13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17services to complete work or medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know whether it would be a21offshore. AOMS is a specialized medical20there, I don't know whether it would be a22services no workin
5site. The objectives of a medical assessment are to ensure that designated offshore are to ensure that designated offshore5A. Yes, we do have a nurse on board our facilities to provide medical services, they are in direct consultation with AOMS on shore a an isolated location with job accommodation 95A. Yes, we do have a nurse on board our facilities to provide medical services, they are in direct consultation with AOMS on shore 87personnel are medically fit to work safely at 87are in direct consultation with AOMS on shore 88an isolated location with job accommodation 98and in that review, they may decide that it is 99best suited to medivac the individual if the 10possible prevent the avoidable occurrence of 111011ill health offshore which would place the 1211sufficient.12individual, his or her colleagues and the 1412ROIL, Q.C.:13emergency rescue service at risk, to provide 1413Q. Right. You've indicated that the primary purpose of AOMS is to determine whether the purpose of AOMS is to determine whether the purpose of AOMS is to determine whether the to work offshore.16meet any regulatory requirements for hazard 1616Have you ever in, again the three of your experience, come across a person who was nadically unfit to travel by helicopter? I19Services to complete work or medical 2018medically unfit to travel by helicopter? I20assessments and determine fitness to work 2120there, I don't know whether it would be a 2322se
6are to ensure that designated offshore6facilities to provide medical services, they7personnel are medically fit to work safely at7are in direct consultation with AOMS on shore8an isolated location with job accommodation8and in that review, they may decide that it is9where possible, to anticipate and where9best suited to medivac the individual if the10possible prevent the avoidable occurrence of10medical service provided offshore isn't11ill health offshore which would place the11sufficient.12individual, his or her colleagues and the12ROIL, Q.C.:13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter fo
7personnel are medically fit to work safely at a n isolated location with job accommodation 97are in direct consultation with AOMS on shore and in that review, they may decide that it is 99where possible, to anticipate and where 109best suited to medivac the individual if the 1010possible prevent the avoidable occurrence of 1111ill health offshore which would place the 121012individual, his or her colleagues and the 1312ROIL, Q.C.:13emergency rescue service at risk, to provide 1413Q. Right. You've indicated that the primary 1414occupational health surveillance; that is 1513Q. Right. You've indicated that the primary 1416meet any regulatory requirements for hazard 1716Have you ever in, again the three of your 1717monitoring. Mr. Commissioner, the operators 1818medically unfit to travel by helicopter? I 1919Services to complete work or medical 20assessments and determine fitness to work 211921offshore. AOMS is a specialized medical 22service provider who is familiar with the 232223situation where a person was able to work but 2424not able to travel by helicopter for medical 2524offshore Medical Services, as in our medical 2525reasons?
8an isolated location with job accommodation99where possible, to anticipate and where910possible prevent the avoidable occurrence of11ill health offshore which would place the12individual, his or her colleagues and the13emergency rescue service at risk, to provide14occupational health surveillance; that is15monitoring for specific job demands and to16meet any regulatory requirements for hazard17monitoring. Mr. Commissioner, the operators18have contracted Atlantic Offshore Medical19Services to complete work or medical20assessments and determine fitness to work21offshore. AOMS is a specialized medical22service provider who is familiar with the23expectations of working in an offshore24environment and all three of us use Atlantic25Offshore Medical Services, as in our medical25Offshore Medical Services, as in our medical
9where possible, to anticipate and where possible prevent the avoidable occurrence of 119best suited to medivac the individual if the medical service provided offshore isn't11ill health offshore which would place the individual, his or her colleagues and the emergency rescue service at risk, to provide to occupational health surveillance; that is monitoring for specific job demands and to meet any regulatory requirements for hazard monitoring. Mr. Commissioner, the operators have contracted Atlantic Offshore Medical to services to complete work or medical complete work or medical9best suited to medivac the individual if the medical service provided offshore isn't sufficient.13emergency rescue service at risk, to provide monitoring for specific job demands and to monitoring. Mr. Commissioner, the operators have contracted Atlantic Offshore Medical to assessments and determine fitness to work complete work or medical to offshore. AOMS is a specialized medical complete work or securical to fishore. AOMS is a specialized medical complete medical to offshore10medically unfit to travel by helicopter? I to morit in an offshore to moritical to moritical to there, I don't know whether it would be a to a light or a lower or a different challenge, to a situation where a person was able to work but to a table to travel by helicopter for medical to work but the or able to travel by helicopter for medical complete to medical to a table to travel by helicopter for medical to a table to travel by helicopt
10possible prevent the avoidable occurrence of10medical service provided offshore isn't11ill health offshore which would place the10medical service provided offshore isn't12individual, his or her colleagues and the11sufficient.13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reas
11ill health offshore which would place the individual, his or her colleagues and the11sufficient.12individual, his or her colleagues and the11ROIL, Q.C.:13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
12individual, his or her colleagues and the12ROIL, Q.C.:13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
13emergency rescue service at risk, to provide13Q. Right. You've indicated that the primary14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
14occupational health surveillance; that is14purpose of AOMS is to determine whether the15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
15monitoring for specific job demands and to15person is fit medically to work offshore.16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
16meet any regulatory requirements for hazard16Have you ever in, again the three of your17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
17monitoring. Mr. Commissioner, the operators17experience, come across a person who was18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
18have contracted Atlantic Offshore Medical18medically unfit to travel by helicopter? I19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
19Services to complete work or medical19don't know if you see the subtle difference20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
20assessments and determine fitness to work20there, I don't know whether it would be a21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
21offshore. AOMS is a specialized medical21higher or a lower or a different challenge,22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
22service provider who is familiar with the22but I'm just wondering have you ever had a23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
23expectations of working in an offshore23situation where a person was able to work but24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
24environment and all three of us use Atlantic24not able to travel by helicopter for medical25Offshore Medical Services, as in our medical25reasons?
25 Offshore Medical Services, as in our medical 25 reasons?
Page 230 Page 23
1assessment process.1MR. SACUTA:
2 ROIL, Q.C.: 2 A. There are examples, for example if I was to
3Q. Mr. Sacuta, how manyyou mentioned earlier or3break my wrist and had a cast on, I certainly
4 one of the other panellists did, talked about 4 wouldn't be able to wear the helicopter
5 doing medical evacuations from the various 5 transportation suit, I wouldn't be able to
6 facilities, approximately how many of those 6 ensure a proper seal. So in those situations,
7 are done a year? Is it two or twenty or 200, 7 somebody would be determined medically unfit
8 we have no idea, so - 8 for offshore work and we would accommodate the
9 MR. SACUTA: 9 individual or his employer would make every
10 A. I would say somewhere between 2 and 20 in a 10 effort to accommodate as per the regulations
11 year, speaking for Hibernia.
12 ROIL, Q.C.: 12 frequency is dependent on the age of a worker.
13 Q. Each of you have approximately the same 13 As you get older, the medical requirements
14 experience? 14 become more frequent.
15 MR. PRITCHARD: 15 ROIL, Q.C.: 16 O. That's not on a financial formula in the first of the f
16 A. Yes, somewhat less for us, Hibernia some doing 16 Q. That's not good news for some of us.
17 It, they re out more, we have a maximum of 90 17 MR. SACUTA:
18 POB. 18 A. Tes, I all infinitely familiar with that. For
19 ROLL, Q.C.: 19 allybody less than 59 years old, it's every 20 by three years between 40 and 40, it's every two
20 Q. 165, 0kay. [20 unce years, between 40 and 49, it's every two
21 MP VOVEV.
21 MR. VOKEY: 21 years; and once you reach the big 50, it's 22 A. For Supcore, it would definitely be less than 22 overy year. First time, offshore workers are
21 MR. VOKEY:21years; and once you reach the big 50, it's22A. For Suncore, it would definitely be less than22every year. First time offshore workers are23one a month23required to complete the five day basic
21 MR. VOKEY:21years; and once you reach the big 50, it's22A. For Suncore, it would definitely be less than22every year. First time offshore workers are23one a month.23required to complete the five-day basic24ROIL OC:24survival training program. This must be

Ja	nuary 11, 2010	Multi	-Pa	age TM	Offshore Helicopter Safety Inquiry
	Page	e 233			Page 235
1 2 3 4 5 6 7	completed using a basic survival training recurrent, BSTR program which is a two-day course. In situations where the full BST is not justified, a one-day offshore survival introduction is allowed. This provides for six days offshore during a one-year period and I believe Mr. Commissioner, that's what you	d 1	1 2 3 4 5 6 7	ROIL, Q.	officer has detailed training to be completed, as well as anyone who works on the helideck teams and any of our offshore fire teams also have detailed training which must be completed to them fulfilling those roles. Q.C.: Okay, we've talked a bit about the helicopter
8	completed when you travelled offshore to the	e	8		landing officer before and about the offshore
9	Hibernia platform. It should be noted that		9		fire team, what does the helideck team do?
10	BST equivalent certificates issued by OPITO		10	MR. SA	ACUTA:
11	and OLF who are North Sea standards, will be	e	11	A.	Helideck teams would be the people that would
12	considered valid to travel to offshore		12		remove baggage from the helicopter baggage
13	facilities on the east coast, although they		13		compartment, put baggage in, put baggage out.
14	may be valid for four years in their own		14		Sometimes, depending on circumstances they may
15	three years in eastern Canada		15		accompany individuals to the edge of the
10			10		work on the belideck, mainly to get beggege
11/	And I presume that you're aware that there's	a	17		and cargo on and off the compartment itself
19	debate amongst workers as to whether or no	t l	19	ROIL	
20	this training should be done more often or	•	20	0.	So they're associated with the loading and
21	less often?		21	Č.	unloading in the offshore.
22	MR. SACUTA:		22	MR. SA	ACUTA:
23	A. Yes, I know that during my time working		23	A.	And they also are responsible for the
24	offshore as the offshore installation manager,		24		refuelling of the helicopter.
25	I was involved in numerous discussions, shou	ld	25	ROIL,	Q.C.:
	Page	e 234			Page 236
1	it be more frequent, should it be less		1	Q.	Okay.
2	frequent, should we have the sea day, should		2	MR. S	ACUTA:
3	we not have the sea day. There was a lot of		3	А.	So operators and their respective contractors
4	issues that were brought up over time and I		4		are responsible for maintaining employee
5	think that March 12th has refocussed everybo	dy	5		qualifications and certification records and
6	about the importance of the basic survival		6		compliance with the CAPP training and
7	training program.		7		qualification standard practice. The Board
8	ROIL, Q.C.:		8		does monitor operator compliance with those
9	Q. Okay, but at this point in time, to your		9		standards. As Cougar represents the final
10	knowledge there has been no change in these	e	10		gateway to offshore, operators have worked
	protocols or these timetrames?		11		tracking systems which maintain records of all
12	MR. SACUTA:		12		mandatory training and medical certifications
14	requirements related to heliconter operations		13		and their expiry dates to ensure that all
15	and safety. We talked earlier about		15		personnel travelling offshore are either
16	helicopter safety briefings which are		16		eligible to travel offshore or have the
17	completed both on shore and offshore prior to	,	17		appropriate exemption. And I'll talk about
18	getting on the helicopter for all personnel.		18		exemptions on my next slide.
19	There is regulatory awareness training that's		19	ROIL,	Q.C.:
20	completed; transportation of dangerous good	s	20	Q.	Okay, before we leave, electronic personnel
21	for any situation where we may be transportin	ng	21		tracking systems tracks the information about
22	dangerous goods in the cargo hold of the		22		these people, not the people themselves, I
23	helicopter; work place hazard materials		23		take it?
24	information systems or WHMIS. Anyone wh	10	24	MR. S	ACUTA:
25	fulfils the role of the helicopter landing		25	Α.	That's correct. Under Chapter 7 of the

Ja	nuary 11, 2010 M	ulti-I	Page	e TM	Offshore Helicopter Safety Inquiry
	Page	237			Page 239
1	Standard Practice for Training and		1 RC	DIL.	0.C.:
2	Oualifications of Personnel, which is the		2	0.	And how long can an exemption apply? Is it
3	exemption procedures, because of the		3		two days, two weeks, two years, two months?
4	intermittent nature of employment, course		4 M	R. S.	ACUTA:
5	scheduling and other factors, it may not		5	A.	Each time an individual travels offshore, you
6	always be possible for an individual to fulfil		6		would have to fill a new exemption, but I
7	all the qualification and training		7		think one of the expectations of the Board is
8	requirements set out in this document prior to		8		that you will close the need for that
9	travelling offshore. Exemptions may be		9		exemption as soon as you possibly can.
10	granted on a case by case basis with the	1	0 RC	DIL.	0.C.:
11	approval of the operator senior offshore	1	1	0.	So each time the person wants to travel, if
12	representative and the offshore installation	1	2		there's a reason for an exemption, it has to
13	manager. As identified in Mr. Decker's	1	3		be granted for that travel?
14	testimony, he did travel offshore on March	1	4 M	R. S.	ACUTA:
15	12th under one of these exemptions because he	1	5	А.	Each time they travel, that's correct.
16	had not completed the sea day component of his	1	6 R(DIL.	0.C.:
17	BSTR program. The Board monitors all	1	7	0.	Thank you.
18	exemptions and will notify the operator of any	1	8 M	R. S.	ACUTA:
19	questions or concerns and reserves the right	1	9	A.	Although I expect that Cougar may provide a
20	to deny any exemption or to issue an order to	2	0		more detailed summary of their check-in
21	an operator relating to exemptions if the	2	1		process, we've prepared a summary slide based
22	process is abused. Every exemption that we	2	2		on our knowledge and I'm just going to step
23	approve goes to the Board, the Board sees them	2	3		you though it because I know the Commissioner
24	all.	2	4		did have some questions around some of the
25	ROIL, Q.C.:	2	5		steps in the sequence offshore, so -
	Page	238			Page 240
1	0 Have you ever had a situation where the Board		1 RC	ЭП.	0.0.
2	has said we are not prepared to allow this		2	0.	Yeah, and I take it again from earlier
3	exemption to continue?		3	x .	evidence that a person who is ultimately going
4	MR. SACUTA:		4		to travel on a vessel would also go through
5	A. Speaking for Hibernia. I am not aware of any		5		the Cougar facility first?
6	situation where that's occurred.		6 M	R. S.	ACUTA:
7	MR. PRITCHARD:		7	A.	Absolutely.
8	A. No, nothing from Husky either.		8 R(DIL,	0.C.:
9	MR. VOKEY:		9	Q.	Yes.
10	A. None that I am aware of from Suncore.	1	0 M	R. S.	ACUTA:
11	MR. SACUTA:	1	1	A.	Cougar Helicopters has a standard check-in
12	A. We have been questioned quite often about the	1	2		process which is utilized by all the
13	specific exemptions.	1	3		operators. Individuals arrive at the heliport
14	ROIL, Q.C.:	1	4		at least one hour prior to their schedule
15	Q. What about the nature of the questions as to	1	5		departure. During their check-in process,
16	why it's gone on so long or -	1	6		identification and training is verified. Any
17	MR. SACUTA:	1	7		personnel who have gaps in the required
18	A. Circumstances around why the individual wasn	't 1	8		training are not checked in unless the
19	able to complete the training. In some cases	1	9		required exemptions and approvals are in
20	they've asked, you know, what are you plans	2	0		place. Individuals and baggage weights are
21	for follow up, which is normally included on	2	1		measured, medications are collected and sealed
22	the exemption form, there's a section around	2	2		for transport. They are delivered to the
23	making sure that you identify what the next	2	3		nurse offshore and he or she distributes them
24	steps will be so that the person doesn't have	2	4		to the individual when they get on board.
25	to continue to fly offshore on exemption.	2	5		It's a twofold purpose: we want the nurse to

Ja	nuary 11, 2010	Multi-F	Pa	ige [™]	Offshore Helicopter Safety Inquiry
		Page 241			Page 243
1	be aware of any medications that peopl	e are	1		the offshore check-in process utilized by the
2	taking when they get offshore in case he	e has	2		three operators. The summary on this slide is
3	to treat the individual and the individual	is	3		a typical departure process. Individuals
4	unconscious, he needs to know what me	dications	4		arrive at the Heli-Admin approximately one
5	they're on. Passengers then travel through	gh a s	5		hour prior to the scheduled departure. Each
6	security check which includes a baggage	search	6		facility has a designated helicopter admin
7	and an x-ray inspection, as well as al	1 7	7		area for departing passengers.
8	individuals pass through a metal detect	tor.	8	ROIL,	Q.C.:
9	Personnel are then issued a flight suit. T	'he g	9	Q.	And that's the expression "Heli-Admin"
10	flight suit includes the personal locato	r 10	0		referred to earlier, Helly Hansen referred to
11	beacon and a HUEBA. Their individual si	ze is	1		it, it doesn't have to do with Helly from
12	documented in the Cougar flight suit dat	abase. 12	2		Helly Hansen.
13	If an employee wishes to use another s	ize, 13	3	MR. S	ACUTA:
14	they must be refitted by a qualified Cou	ıgar 14	4	A.	No, it's helicopter administration.
15	personnel. Helly Hansen has trained C	ougar 15	5	ROIL,	Q.C.:
16	personnel in the proper suit fitting proces	ss, 16	6	Q.	Helicopter administration, okay.
17	so if an individual shows up and says,	you 17	7	MR. S	ACUTA:
18	know, I ate a lot of turkey over Christma	is and 18	8	A.	T cards, which are the cards that individuals
19	I gained a new pounds, I'd like a new su	it, he 19	9		use when they go offshore to say they're there
20	has to be refitted as part of that process.	20	0		and put them in their slot when they first
21	ROIL, Q.C.:	21	1		arrive and they're used during emergency
22	Q. So the individual cannot now change his	s suit 22	2		situations so we can account for all personnel
23	size.	23	3		are returned and individuals and baggage
24	MR. SACUTA:	24	4		weights are confirmed. Passengers normally
25	A. The individuals don't choose their suit si	ize. 25	5		bringing their flight suit from their room to
		Page 242			Page 244
1	All personnel are shown a video briefing.	Any	1		Heli-Admin and are issued a PLB and a HUEBA
2	other information which may be relevant	at the	2		which they fit to their suit. All personnel
3	time may also be discussed during th	is 3	3		are shown a video briefing. Any other
4	briefing situation. As an example, during	the 4	4		information which may be relevant at the time
5	period that we had the right to refuses bei	ng s	5		may also be discussed during that briefing.
6	analyzed by the Board, all the boardir	ng e	6		The flight suit is donned and all passengers
7	passengers were informed that there wa	sa 7	7		must once again demonstrate the ability to
8	right to refuse in process, just so that they	, 8	8		fully zip up their suit while in a seated
9	were aware that that was being follow	ved 9	9		position. The helicopter arrives at the
10	through through the C-NLOPB. The flight	suit 10	0		facility and the arriving passengers
11	is donned and all passengers must demon	strate 11	1		disembark. If required, the helicopter is
12	the ability to fully zip up their suit while	12	2		refuelled. The helicopter is not normally
13	in a seated position. Air muffs and air plu	lgs 13	3		refuelled with passengers in the passenger
14	are donned and personnel are escorted to	the 14	4		cabin. Air muffs and air plugs are donned and
15	helicopter for boarding. Individuals put	on 15	5		personnel are escorted to the helicopter for
16	their seatbelt and a Cougar representative	ve 16	6		boarding. An individual puts on his seatbelt
17	checks the seatbelt to ensure it is on	17	7		and a heli-deck team member checks the
18	properly, it does not interfere with the HU	EBA 18	8		seatbelt to ensure it is on properly and does
19	or any other component of the flight suit.	So 19	9		not interfere with the HUEBA or any other
20	each individual passenger will be checked	d by a 20	0		component of the flight suit. You will
21	Cougar person to make sure his seatbelt	is 21	1		notice, Mr. Commissioner, that there is no
22	positioned correctly and it is not interferin	ng 22	2		formal security check implemented offshore.
23	with any of the components of the flight s	suit 23	3		Each departing passenger signs a departure
24	itself. From an offshore check-in	. 24	4		sheet which clearly articulates expectations
25	perspective, there may be small difference	es in 25	5		tor luggage. Furthermore, the platform is

Ja	nuar	y 11, 2010 Mu	ılti-l	Page	TM Offshore Helicopter Safety Inquiry
		Page 24	45		Page 247
1		considered to be a secure environment, we		1	documents now, those will come out in the
2		would equate it to the interflight departure		2	individual presentations.
3		area at an airport where a security check is		- 3 MR	PRITCHARD
4		not required unless you leave the secure area		4	• We can certainly have that feedback type forms
5		that's why there's no security check offshore		5	from training on board
6	ROII			6 RO	
	OIL,	Now I've heard it said that some workers are		7 (T I was speaking about the actual emergency
	Q٠	concerned that by not going through a security		, , 8	response plans each companyit's an
		check then that there is less interest in or		0	individual plan is it?
		focus on their personal security on the flight	1	/ 0 MR	
		hack	1	1	A Ves
12	MDS		1	1 PO	
12	Δ	All Lean say is that the platform is a	1	2 KO	D. Ves that's right and we'll deal with the
11	л.	secured site and each individual does sign a	1	л (details of them in your original or your
14		form saving that you know they're not taking	1	+ 5	individual presentations
15		any inappropriate material on the flight	1	ј 6 мр	
17		returning flight to town, so there is some	1	7	. FRITCHARD. M. Well, this slide just shows a typical day of
18		nersonal accountability for your own security	1	v 1 Q	the resources available in terms of the
10		as well	1	0	operator's resources of vessels and
	POII) 0	heliconters. There may be other merchant
$ _{21}^{20}$	NOIL,	Okay thank you And I think we now move on		1	vessels coast guards or other aircraft in the
$\begin{bmatrix} 21\\ 22 \end{bmatrix}$	Q٠	to Mr Pritchard again now we move into		1 2	area Cougar's blue sky tracking system gives
22		operator emergency preparedness		2	details of the heliconter locations and the
$\begin{vmatrix} 2.3 \\ 2.4 \end{vmatrix}$	MD D			Л	AIS system which is an automated information
$ _{25}^{24}$	Δ	That's correct Operators have recognized a		 	system gives details of the vessels. This
-		Poge 2	16		Does 249
		Page 24	40	1	Page 248
		great importance of emergency preparedness and	1		gives us -
		surve to identify the critical scenarios and		2 KU	IL, Q.C.:
		prepare an necessary actions to be taken to		3 (2. So the vessels and the hencopters can be
		protect people, the environment and the		4 5 MD	protection a precise location at any time?
3		threat. The americancy response plan has an		5 MR	. PRIICHARD:
		understanding of the resources, evaluable and		0 1 7	A. So we can see them on the outer sky tracking
		notential escalation of events what		/	system and on the automated mormation
		mitigations are evaluable and what control		0 0 DO	
9		would control the situations. Tunically	1	9 KU	IL, Q.C.:
10		perhaps in an evecuation and the use of	1	1 MD	2. Who maintains mose systems:
		belicopters would be identified in that	1	1 MK	. PRIICHARD.
12		emergency response plan. The plans also	1	2 1	and L brought a little, bit of information on
13		reference organizational structure and	1	5 1	the automated information system so all
14		training recruitments. The emergency response	1	+ 5	merchant vessels over 300 tonnes have this
15		plan to continuously update it both at a high	1	5	system it gives information regarding the
17		level document in terms of the overall plan	1	7	vessel's identification number, its direction
18		but also from feedback from the training	1	, 8	and heading and speed. So that's a
19		exercises that occur offshore. So on a weekly		9	requirement of vessels over 300 tonnes some
20		hasis we have emergency response type training		- 0	of them actually use GPS and satellite to
21		exercises and feedback from that just		1	transfer the information other more coastal
22		continues to improve our courses and		2	vessels would use VHF.
23		procedures in the event of emergencies.	2	3 RO	IL, O.C.:
$ _{24}$	ROIL	. O.C.:	2	4 (). Right.
25	0	Now these again we don't have those actual		5 MR	PRITCHARD

Janu	uary 11, 2010	Multi-l	Page TM	Offshore Helicopter Safety Inquiry
	Р	age 249		Page 251
1	A. So the operators contract Cougar to supply	an	1	specialist SAR equipment and trained
2	operational service of transportation to and	1	2	personnel. Another little bit of detail on
3	from the offshore and also maintain the	,	3	that on the next slide. The SAR helicopter is
4	standby helicopter for SAR, the first response	se	4	available all year round, 24 hours a day, with
5	duties on a 24-hour 7-day-a-week basis. W	e've	5	a wheels up maximum time of one hour. The SAR
6	discussed a little bit earlier in the		6	S-92 is fully equipped with the necessary
7	testimony about the posture for the SAR du	ty	7	search and rescue equipment, such as hoists,
8	helicopter is maintained by ensuring one of	of	8	life rafts, emergency locator transmitters and
9	the three operational helicopters is		9	receivers. There is now a dedicated group of
10	physically on the ground in a serviceable	e 1	0	SAR technicians or SAR techs as they're
11	condition for that SAR service.	1	1	described and SAR pilots. So since this is
12 R	OIL, Q.C.:	1	2	another change since March the 12th. We've
13	Q. Okay. Do I take it that sometime ago, and	I 1	3	got a more dedicated group of SAR specialists
14	don't think this is tied or I don't know if	1	4	and that gives us the ability to train a more
15	it's tied to the March the 12th incident, but	1	5	concentrated group with the time allocated for
16	there was some additional flexibility that y	ou 1	6	the training side of it, of SAR duties.
17	were allowed to have with respect to tha	t 1	7 ROIL,	Q.C.:
18	third particular piece of equipment?	1	8 Q.	So the pilots that are responsible to conduct
19 M	R. PRITCHARD:	1	9	SAR missions or first response missions and
20	A. Yes, we utilized as the helicopters	2	20	the technicians that accompany them are now a
21	before March the 12th, or return to service	e 2	-	dedicated group rather than what, the entire
22	rather, the inbound helicopter, when it was	30 2 the	2	workforce?
23	minutes from St. John s, we were allowing	the 2	3 MR. P	RITCHARD:
24	outward bound to be dispatched, on the	e 2	4 A.	I am unsure if it was the entire workforce,
25	understanding that 50 minutes fullning this		.5	but certaining when you have a number of hours
	P	age 250		Page 252
1	the one helicopter, if a SAR requirement w	as	1	for training, spreading those hours too thin
$\begin{vmatrix} 2 \\ 2 \end{vmatrix}$	required, we could actually gear up within that final 20 minutes to be achieved and actually	n	2	over too many pilots would mean that the
3	that final 30 minutes to be wheels up with	in	3	pilots weren t getting pernaps the best
	one nour to get that aircraft outbound.		4	training and now we re looking for them to get
	OIL, Q.C.:	hit		note dedicated training in the SAR duties.
	Q. Okay. So the expectation was even mough was in the air, it would be able to get up in	111	o KOIL,	Actually, I suspect we should be asking Cougar
	the one hour?		/ Q.	for the particular details on that but your
			0	understanding is that there is a -
	A That's correct	1) 0 MR P	RITCHARD.
		1	1 A	That in essence is where we've -
12	O. Okay.	1	2 ROIL	O.C.:
13 M	R PRITCHARD	1	3 O.	- an increased focus.
14	A. So now we don't utilize that. We've chan	ged 1	4 MR. P	RITCHARD:
15	our scheduling such that we now operate	a 1	5 A.	- moved to, ves. So I've mentioned response
16	schedule one hour earlier in the daylight	1	6	efforts are also supported by the Blue Sky
17	time. So we start flying operations now a	t 1	7	Tracking System and the automated information
18	7:00 in the morning to gain daylight hours	and 1	8	system as previously described.
19	we're allowing ourselves now to have th	at 1	9 ROIL,	Q.C.:
20	helicopter dedicated on the ground, rathe	r 2	20 Q.	Okay. Now before you move on, there's a
21	than using that 30-minute running time. So	it 2	21	question that I have that arises from the
22	is a change since March the 12th.	2	2	safety planning guideline, which says that it
23	In alignment with the regulations and	2	3	is expected that a helicopter would be
24	described in detail in the safety plan	2	4	maintained on standby and dedicated to search
25	guidelines, the SAR helicopter has certain	2	.5	and rescue on a 24-hour per day basis.

Ja	nuary 11, 2010	Multi	-Pa	age ^{TI}	^M Offshore Helicopter Safety Inquiry
	P	age 253			Page 255
1	Nowhere in there or anywhere else that I can	0	1		wanted to make sure when the one hour was
2	see is there an indication that the one-hour		2		measured from. So it's not from the incident.
3	wheels up requirement. How did that get		3		It's from when Cougar are notified that there
4	established? By whom? And what does it mea	n?	4		has been -
5	MR. PRITCHARD:		5	MR.	PRITCHARD:
6	A. In our safety plan, we commit to a one-hour		6	A.	From the request.
7	wheels up, and that's how we contract Cougar,		7	ROIL	2, Q.C.:
8	and through the submission of our safety plan		8	Q.	Yes, okay. And the equipment, I know that
9	to the Board is how we established that one-		9		we've seen it in other places. There's a list
10	hour wheels up commitment.		10		of things like a hoist and, you know, a basket
11	ROIL, Q.C.:		11		and other things. Who determines what the
12	Q. I'm sorry, the last part? How was the one		12		equipment is? Is that dictated by the C-NLOPB
13	hour chosen?		13		or do the companies, in consultation with
14	MR. PRITCHARD:		14		somebody, determine what should be in the
15	A. It was chosen by the operators to put in the		15		equipment list?
16	safety plan.		16	MR.	PRITCHARD:
17	ROIL, Q.C.:		17	A.	We find that in the guidelines from the C-
18	Q. Yes.		18		NLOPB. There's a dedicated list of equipment.
19	MR. PRITCHARD:		19		I think slide 94, guidelines 71.2 starts to go
20	A. And the safety plan is acknowledged by the		20		into some of the actual requirements.
21	Board and that we give our one-hour commitm	ent	21	ROIL	-, Q.C.:
22	wheels up to the Board and that is accepted by		22	Q.	Okay. So, yeah, those are generic things.
23	the Board.		23		Not generic, but with a hoist. So there's no
24	ROIL, Q.C.:		24		particular specification on the hoist, but it
25	Q. And one-hour wheels up means what? From t	he	25		at least has to be capable of lifting -
	Р	age 254			Page 256
1	moment that an incident is reported? How	v is	1	MR.	PRITCHARD:
2	that time measured? That's what I'm curi	ous	2	A.	Well, it says that it's got to be capable of
3	as to where the hour is.		3		lifting one or multiple people, so a dual
4	MR. PRITCHARD:		4		tandem type of hoist.
5	A. The time would be measured from the red	quest	5	ROIL	2, Q.C.:
6	for a SAR requirement to the point of whee	els	6	Q.	Okay.
7	up and on the mission.		7	MR.	PRITCHARD:
8	ROIL, O.C.:		8	A.	Proceed?
9	0. Okay. So incident happens. At that point	in	9	ROIL	
10	time, the burden is on somebody to say	to	10	Q.	Yes. Thank you, sir.
11	Cougar, "we need to have a helicopter in a	SAR	11	MR.	PRITCHARD:
12	mode now" and the one hour is measured	from	12	A.	Canada is party to the International
13	then?		13		Convention on Marine Search and Rescue and
14	MR. PRITCHARD:		14		maintains responsibility for the preservation
15	A. Correct. It may be response to a Couga	ır	15		of search and rescue services based in its
16	helicopter. It could also be a response to		16		geographical profile. I have a map a little
17	the OIM offshore saying that he has a		17		bit later which will outline the area of
18	situation with people in the water.		18		responsibility.
19	ROIL, Q.C.:		19	ROIL	2, Q.C.:
20	Q. Yes, exactly.		20	Q.	Yes. Just so that we're clear now, I take it
21	MR. PRITCHARD:		21		that you're going to explain to us what the
22	A. So not just necessarily Cougar, but, you kr	now,	22		operators understand that SAR, the Government
23	any SAR mission.		23		SAR does in the offshore?
24	ROIL, Q.C.:		24	MR.	PRITCHARD:
25	Q. Yeah, okay. I think I understand. I just		25	A.	This is still the interface of DND and the

Jan	uar	y 11, 2010 Mu	lti-Pa	age™	^M Offshore Helicopter Safety Inquiry
		Page 25	57		Page 259
1		operators. I'll move on a little bit later in	1		in the next section. So we'll leave it at
2		the presentation here to what we perceive to	2		that.
3		be DND's responsibilities.	3		So DND has established joint coordination
	ROIL	OC ·			centres in various regions of Canada which
5	0.	Okay. So this is at the operator level.	5		are responsible for the coordination of
6	 MR∓	PRITCHARD	6		aeronautical and marine SAR operations within
	A	This is still the operator DND I'm moving			a prescribed region Upon the initial
	11.	into that interface	8		notification of any event occurring offshore
9	ROIL		9		the emergency response procedure calls for the
10	0.	Okay.	10		IRCC to be informed. This is on the basis
	MR. I	PRITCHARD:	11		that early notification is good rather than to
12	A.	So it's more than likely that the operators	12		gear up when it is realized that the actual
13		dedicated SAR helicopter will be the first	13		requirement is required. In other words, it's
14		response to an offshore emergency. However,	14		easy to stand down a resource than gear it up
15		it should be noted that DND has the authority	15		for when it's we know it's needed.
16		to command vessels or aircraft of opportunity	16		Whenever the standby helicopter is mobilized,
17		to help in a SAR mission, as was the case on	17		which may be for a medical evacuation or non-
18		March the 12th when the PAL flight was the	18		occupational, for instance, we've heard we
19		first flight over, and that was just by	19		used a helicopter for that, then JRCC are
20		opportune that that flight was going over.	20		notified. So they then understand that we are
21	ROIL	, Q.C.:	21		utilizing that SAR resource for a medical
22	Q.	Okay. So you're saying that that flight was	22		repatriation perhaps.
23		directed by DND?	23		If there's an emergency requiring
24 1	MR. I	PRITCHARD:	24		specific search and rescue, then JRCC will
25	A.	No, I'm not saying that. That just so	25		assume command of the situation, including
		Page 24	58		Page 260
1		happened I'm saving in the normal course of	1		deployment of the dedicated SAR first response
		events we have a SAR helicopter available for	2		helicopters as well as any of the other
3		response. As it happened on March the 12th.	3		resources, such as the other S-92 aircraft
4		the PAL flight was in the air and was first on	4		that are available in St. John's, and of
5		the scene.	5		course, any of the DND resources, such as the
6	ROIL	. O.C.:	6		Cormorants and the Hercules. So they would
7	0.	Okav.	7		have a vast array of resources available for
8	MR. I	PRITCHARD:	8		DND, as I say including the first response S-
9	A.	The practical response with the SAR technician	9		92 out of St. John's and thereafter, any other
10		was from the Cougar SAR helicopter. DND will	10		of the helicopters that would be either coming
11		control the Cougar helicopter during an	11		back or be on the ground at that time.
12		offshore SAR mission, or once launched, JRCC,	12		So I'll now move into what we consider as
13		Joint Rescue Coordination Centre, is informed	13		the responsibilities of DND, recognizing that
14		and would control the mission from thereafter.	14		the Department of National Defence will be
15		DND's assets, a single purpose full-time	15		coming and testifying as to what they actually
16		search and rescue specialists with significant	16		do. This is our understanding of what DND do
17		training. So DND have dedicated resources and	17		for us.
18		all the specialist equipment that facilitates	18	ROIL	, Q.C.:
19		successful search and rescue missions. They	19	Q.	Before you move there, Mr. Pritchard, let me
20		have a high level of training and have	20		just put to you a comment, I guess, that came
21		completed many successful missions. In the	21		from Mr. Decker that I'd like you to reflect
22		event of an event escalating, emergency DND	22		on. In his evidence, I recall him saying that
23		can call upon resources from Federal or	23		he had expected to see a search and rescue, an
24		others, and I'll detail in what we consider to	24		orange or a yellow or whatever colour
25		be the responsibilities and resources of DND	25		helicopter.

January 11, 2010	Multi-P	age	M Offshore Helicopter Safety Inquiry
	Page 261		Page 263
1 MR. PRITCHARD:			commitments are.
2 A. Yeah.	2	ROI	L, Q.C.:
3 ROIL, Q.C.:	3	s ç	Okay. Okay, thank you.
4 Q. And he saw a grey and blue Cougar one.	If the 4	MR.	PRITCHARD:
5 operator is normally first response, if th	e 5	i A	. So we know Canada is obliged by International
6 Cougar helicopter is normally first respo	nse, 6	ō	Convention to maintain the provision of SAR
7 how can that message get out to the work	cforce? 7	7	coverage in a geographical location and this
8 Because it appears that the expectation	was 8	8	comes with an agreement of the Minister of
9 something different than that.	9)	National Defence, supported by the National
10 MR. PRITCHARD:	10)	Search and Rescue Secretariat, of which there
11 A. I'm not sure. It's kind of speculation, he	ow 11		is a number of components there. So the
12 we can get that message out, because if w	ve do 12	2	Transport Canada, RCMP, Coast Guard, Parks
13 have repatriations or first response, it is	13	;	Canada, MET Services Canada and Environment
14 the Cougar helicopter that we use, in gen	eral 14	Ļ	Canada are all part of that National Search
15 terms.	15	i	and Rescue Secretariat as a support service
16 ROIL, Q.C.:	16	5	and group there.
17 Q. So you're saying that you can't be certain	n on 17	7	Marine Aeronautical SAR is carried out
18 every day and every moment it will be a	Cougar 18	8	principally by DND and the Coast Guard. DND
19 first response. It depends on what's	. 19)	has overall responsibility for the
20 happening with the other.	20)	coordination and operation of SAR missions.
21 MR. PRITCHARD:	21		DND has established Joint Rescue Coordination
A. I can't really speculate as to how we can	1 do 22	2	Centres in various regions of Canada, which
23 that, but you know, what circumstances	would 23	5	are responsible for the coordinations of a SAR
the yellow helicopter come along? They	would 24	ŀ	operation within that region, and for our
25 have to be almost in the vicinity, perhaps	on 25	5	region, Halifax is the JRCC.
	Page 262		Page 264
1 a training mission 60 miles off, when	an 1	ROII	, Q.C.:
2 incident occurs for them to be first on the	1e 2	2 Q). So a JRCC exists in Halifax and covers the
3 scene. We are the closest by nature in S	št. 3	3	region that we are in?
4 John's, rather than Gander.	4	MR.	PRITCHARD:
5 MR. SACUTA:	5	i A	A. Correct.
6 A. If I could add, Mr. Roil, I think one of th	ie 6	5 ROII	L, Q.C.:
7 things that all the operators have done is	to 7	Ç Ç	2. Yes, okay, and that's Joint Rescue -
8 make sure that we clarify with our work	force 8	3 MR.	PRITCHARD:
9 what our commitments are from a first re	sponse 9) A	. Coordination Centre.
10 search and rescue capability. So I thin	k 10) ROII	L, Q.C.:
11 we've recently all had an update provide	ed to 11	Ç) Coordination Centre, where comes the acronym
12 the workforce to make sure that there's	no 12	2	JRCC.
13 confusion about what we supply from a	first 13	MR.	PRITCHARD:
14 response search and rescue capability.	So 14	A	A. Yeah.
15 that was done just before Christmas.	15	ROII	, Q.C.:
16 ROIL, Q.C.:	16	5 Ç	0. Okay, thank you.
17 Q. So I take it that that might have been and	ther 17	MR.	PRITCHARD:
18 small lessons learned from that incident,	18 18	S A	A. Some regions also have marine rescue
19 that informing the workforce as to what	t0 19	,	subcentres, or MRSC, which is manned by the
20 expect is a part of being an informed wo	rker 20)	Coast Guard and the centre in St. John's is
21 and a more comfortable worker.			actually one of these marine subcentres, and
22 MR. SACUIA:	$ ^{22}$	5	part of our emergency response training
25 A. I mean, I units our workforce has been to	10 III 23)	in St. John's to give up on understanding of
25 an opportunity for us to remind them wh	at our 24	r í	the overall response and the management and

Januar	y 11, 2010	Multi	-Page ^T	M Offshore Helicopter Safety Inquiry
	Ι	Page 265		Page 267
1	coordinations of SAR events.	0	1	are Cormorants and at Greenwood in Halifax,
2	So a JRCC has direct tasking authority		2	there are four Cormorants and one Hercules.
3	over primary SAR resources, such as the DND		3	The Cougar resources, we have a one-hour
4	aircraft, the fixed wing and the helicopters		4	wheels up on a continuous basis 365 days a
5	and the Coast Guard vessels. The secondary		5	year and we have three S-92s currently
6	Federal resources, such as the Navy vessels,		6	available, one of course which is on permanent
7	can also be deployed. Under the Canadian		7	SAR duty. There are occasions when other
8	Shipping Act, JRCC and the Coast Guard have		8	airframes are available, such as the
9	the authority to direct all of the civilian		9	ConocoPhillips airframe, which is the other S-
10	vessels and aircraft to render assistance to a		10	92 that's in the picture now, so that gives us
11	nearby vessel or aircraft in distress. So		11	four. I prescribed three here from the three
12	there's a level of authority there that DND		12	operators, but now, I guess, we have four.
13	and the Coast Guard have over civilian		13 ROII	L, Q.C.:
14	facilities. Additionally, both JRCC and Coast		14 Q	And I take it that the supply and support
15	Guard can (unintelligible) aviation and marine		15	vessels that you operate would also be
16	civilian resources to assist with a SAR		16	resources that could be and would be called
17	operation, such as the Canadian Coast Guard		17	upon in the event of an emergency?
18	Auxiliary.		18 MR.	PRITCHARD:
19	During a mission, it is likely DND will		19 A	. Absolutely. All our resources and the DND
20	set up command and communications from a fi	xed	20	have the authority to command those other
21	wing plane, such as a Hercules. This allows		21	merchant vessels and other aircraft, civilian
22	distinct individual resources to concentrate		22	aircraft.
23	on a designated task and allow the overhead		23	On March the 12th, Cougar 491 was bound
24	aircraft to keep the full picture and keep		24	for Sea Rose and Hibernia Platform with 16
25	with the communications aspects.		25	passengers and two pilots when it crashed into
		Page 266		Page 268
1	So this slide now is the map of the	uge 200	1	the sea All operators mobilized their
2	IRCC's region So the IRCC in Halifax has t	he	2	respective emergency response teams and all
3	responsibility for the area outlined in red		3	assistance was offered between the operators
4	This is a vast area of some 4.7 million squa	are	4	in respect of both physical resources and
5	kilometres 29 000 kilometres of coast lin	e	5	human resources I very much appreciated the
6	and 80 percent of its responsibilities lies	•	6	response by DND and was only too pleased to
7	over water The Grand Banks is part of th	is	7	have Major Dennis McOuire taking an active
8	vast area and like any other user of the oce	an	, 8	role in St. John's while his coordination
9	within this geographical boundary we look	to	9	centre in Halifax worked its (unintelligible)
10	DND to utilize our resources of Cougar SA	AR	10	Operators will discuss in detail their
10	first response and any other Cougar		11	own emergency response in the next testimony
12	helicopters and our fleet of vessels, along		12	in the coming weeks. After this tragic
13	with the DND resources of helicopters and	đ	13	accident has happened and it was discovered
13	fixed wing aircraft. Coast Guard vessels. N	avv	13	there had been a ball failure in the oil
15	vessels et cetera and their authority to	u v y	15	filter housing all S-92s worldwide were
16	command other resources in the area in sur	port	16	grounded and that will lead us into the
17	of a well coordinated search and rescue	port	17	discussion by Mr. Vokey tomorrow on the
18	mission	, 	18	Helicopter Operations Task Force which was
19	So the next slide is the level of		19	raised as post March the 12th
20	resources and response times. So IRCC ha	ve	20 ROII	
21	two distinct response times. Monday to		21 O	Commissioner. I had indicated to the witnesses
22	Friday, 8 a.m. until 4 n.m. they are 30	-	X 22	iust after lunch that we would take the
23	minute wheels up as a maximum. After fl	hese	23	evidence to this point today and that I would
24	hours and at weekend, they are at two hours	irs	24	prefer to leave all the dealings with the
25	wheels up maximum. The resources at G	ander	25	Helicopter Operations Task Force for the

January	y 11, 2010	Multi-	Pag	ge TM Offshore Helicopter Safety	Inquiry
January 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17 Q. 21 Q. 23 24 25 Q. 1 2 3	y 11, 2010 morning. That's a separate and anothe document. It is a document that we have a some minor changes to in the past 24 hour I wanted to take the time now to have meeting with legal counsel for all of the parties to simply alert them to the change why they were made and then we would to on tomorrow. We're pretty much, in term progress, going at the rate that I had hope and expected and we will certainly although there's a significant amount of material there, I think we will have it easily covered by the end of the morning, and we could in into the examination by other counsel in the afternoon tomorrow. WISSIONER: All right then. Well, we'll adjourn now a you can go into this other process that you have described. Q.C.: Yeah, I'd simply ask other counsel, just the legal counsel, to remain and we'll have meeting here. MISSIONER: Yes. Will counsel remain for a discussion I with Mr. Roil after we disperse? Okay, that you. UPON CONCLUSION AT 4:15 P.M.	Multi- Page 269 er made rs and a ess, ake this ns of ed ough 1 ed move the a a on Page 270 nk	Pag 1 2 3 4 5 6 7 8 9 10 11 12 13 14	ge [™] Offshore Helicopter Safety I CERTIFICATE We, the undersigned, do hereby certify that the foregoing is a true and correct transcript of a hearing heard on the 11th day of January, 2010 at Tara Place, 31 Peet Street, Suite 213, St. John's Newfoundland and Labrador and was transcribed to the best of our ability by means of a sound apparatus. Dated at St. John's, NL this 11th day of January, 2010 Cindy Sooley Discoveries Unlimited Inc. Judy Moss Discoveries Unlimited Inc.	Inquiry Page 271 t by us

January 11, 2010

Multi-PageTM

& - adopted Offshore Helicopter Safety Inquiry

			Olishore men	opter safety inquiry
	1998 [2] 28:14 190:1	3 [2] 32:4 173:20	266:6	account [2] 176:17
-&-	1999 [1] 22:22	30 [8] 26:22 153:4 175:20	80s [1] 188:22	243:22
& [1] 22:6	1A [1] 2:15	218:13 249:22,25 250:3	84 [1] 228:20	accountability [2]
	1B [1] 8:11	30 -minuto (11, 250-21	85 [1] 44:5	94.10 243.18 accountable (2) 26.17
-'-	1st [6] 129:16,18 211:24	300 (a) 17.22 106.0		93·18
'90s [1] 42.12	212:5 218:17 227:1	248.15 19		achieve (1) 111.20
,		31 m 271:5	9 _[1] 32:18	achieved (1) 42:17
-1-	-2-	315 (1) 42·14	90 [3] 47:20 54:8 230:17	achieving [1] 78.22
1 m 7.05 9.10 11 21.00	2 [5] 8:1,9,10 31:24	31st [2] 68.21 69.23	92 [4] 195:3 224:10 260:9	acknowledge 11 218.23
1 [5] 7:25 8:10,11 31:20	230:10	32 [2] 45:18 153:5	267:10	acknowledged [1]
1 000 [1] 49·18	2.4 [1] 35:21	332I (1) 192.19	92s [1] 170:21	253:20
1 2 m 43·16	20 [3] 17:23 197:20	3400 [1] 40.6	94 [1] 255:19	acquisition [1] 69:5
1 3 (1) 44.9	230:10 200 m 204 17 205 1	36 (1) 21.15		acreage [2] 33:18,19
10 (1)	200 [3] 204:17 205:1	365 [1] 267.4	-A-	acronym [2] 31:9 264:11
10 [2] 27.22 32.20 100 [6] 24.4 20.20 35.16	2000 [6] 23.1 47.7 90.17	30 [1] 207.4	a.m [1] 266:22	acronyms [2] 31:4.14
47:20 139:7 194:2	193:5 197:12 218:15	37 [1] 232.19	abandonment 151	Act [7] 63:10 86:5 90:11
103 [2] 174:25.25	2001 [1] 227:1		203:13 204:6,18 205:2	123:24 124:3,13 265:8
11 [4] 1:1 17:21 32:23	2002 [4] 26:11.13 46:18	- -	205:25	actions [2] 39:18 246:3
59:5	90:17	4 [2] 32:6 266:22	Aberdeen [2] 23:2 24:9	activate [2] 217:18,23
117-100 [1] 17:18	2003 [2] 29:3 191:5	4. 7 [1] 266:4	ability [8] 6:24 170:19	activates [2] 216:4,5
117-207 [1] 17:19	2005 [5] 23:15 29:5	40 [2] 50:14 232:20	178:19 205:18 242:12	activation [1] 217:19
117-209 [1] 17:21	195:19 196:20,23	401 [1] 17:24	able 244.7 231.14 271.7	active [2] 65:11 268:7
11th [3] 182:13 271:4,10	2006 [6] 50:14 55:23	402 [1] 17:24	16.22 45.21 82.21 93.1	activities [16] 46:2 54:23
12 [6] 17:21 33:1 138:3	202:10 203:5 208:22	42 [1] 154:5	96:14 102:4 134:14	56:5,17,19 57:4,12 64:12
140:5 153:4,4	219:20	4414 [2] 70:11 90:21	166:24 172:14 176:17	65:17 75:17,21 76:21
120 [1] 50:16	2007 [10] 29:12,12 47:7 196:23 197:1 202:7	49 [1] 232:20	185:6 194:16 205:22	93:12 100:13,20 108:4
12:30 [1] 153:18	211:14,25 212:5 213:10	491 [2] 27:10 267:23	215:5 217:22 251:25,24 232:4 5 238:19 250:7	63.19 20 73.22 23 86.10
12th [26] 2:4 13:5 20:3	2008 [1] 227:2	4:15 [1] 270:3	above [2] 43:23 52:16	102:4 111:10 112:1
27:11 33:4 122:3,9	2009 [2] 68:21 198:7	4:30 [1] 173:21	ABS [1] 88.4	165:16 189:2
140:21 182:13 185:5	2010 [4] 1:1 221:1 271:4	4th [1] 90:17	absolutely (22) 31.8	actual [9] 19:12 22:1
215:21 216:21 218:17	271:10		34:20 36:21 37:14 52:9	42:22 159:3,15 246:25
234:5 237:15 249:15,21	2016 [1] 40:5	-5-	56:3 64:8 65:7 66:8 78:6	247:7 255:20 259:12
250:22 251:12 257:18	2017 [1] 40:5	5 [2] 32:7 175:17	78:13 83:9 84:20 90:5	au [2] 121:20 150:25
258:5 207:25 208:19 13 m 17:00 22:5	206 [1] 17:19	50 [2] 194:2 232:21	94:20 111:2 122:11	83.17 84.10 18 211.16
13 [2] 17:22 33:5	208 [1] 17:20	560 [1] 205:5	267:19	add 121 92:12 262:6
14 [2] 1/:22 31:17 1400	21 [3] 28:24,25 129:19	59 [2] 164:25,25	abused [1] 237:22	added [2] 52:12 202:0
1400 [1] 211:15	21-day-off [1] 50:19		accept [2] 78:3 91:6	193:1
14 (1) 129:21 15 cm 17 02 47 05 70 05	21-day-on [1] 50:19	-6-	acceptable [3] 77:5	addition [4] 1:9 7:16
15 [5] 17:23 47:25 72:25	210 [1] 17:21	6 [2] 32:11 55:22	97:13 222:24	160:8 166:24
150 (1) / (2)	2121 [1] 5:3	60 [1] 262:1	acceptance [1] 213:4	additional [8] 49:22
15th m 7.11	213 [1] 271:5	60-storey [1] 43:14	accepted [1] 253:22	50:14 122:22 188:3
16 [1] 7.11 16 [2] 17.23 267.24	21st [1] 129:17	60s [1] 197:15	access [7] 3:3 14:21,23	206:25 218:11 234:13
10 [2] 17.23 207.24 17 [2] 17.23 35.10	220 [1] 46:7	61 [2] 196:17 224:10	65:21 83:20 213:19	Additionally 11 265.14
18 [1] 42.20	224 [1] 43:13	62 [1] 168:22	221:20	address (2) 0.11 220.7
18 (1) 42.20	225 [2] 192:21 194:21	64 [1] 186:21	accessible [1] 14:13	adaguate (4) 172:20
18th (a) 2.15 7.12	22nd [1] 90:17		accident [4] 20:7 24:15	175:3 176:22 177:3
100 (1) 25.17	23rd [1] 69:22	-7-	accommodate (2) 232.8	adequately [2] 8:18.19
1073 (1) 21.16	24 [3] 35:18 251:4 269:3	7 (2). 22.14 226.25	232:10.11	adhere (1) 139:18
1973 [1] 21:10 1070 (1) 42:16	24-hour [2] 249:5 252:25	7 [2] 52.14 250.25	accommodation [3]	adjacent m 169:25
1081 (a) 25.4 26.22	25 [6] 27:24 51:7 175:20	7-uay-a-week [1] 249.5	52:13,16 229:8	adjourn [1] 269:17
1901 [2] 25.4 20.25 108 $I_{(2)}$ 25.11 46.16 51.9	197:14,20 218:13	700 [1] 40.8	accommodations [4]	admin [1] 243:6
1904 [3] 25:11 40:10 51:8	250 [1] 46:7	71.2 [1] 233:19 75 (2) 27.7 25.19 50.17	43:21,23 50:12 86:7	administration [3]
1008 (1) 25.15	27 [1] 194:1	7.00 (1) 250.10	accompany [2] 235:15	179:11 243:14,16
100 [1] 23:13	280 [4] 43:21 204:20	7.00 [1] 230:16	251:20	administrative [1]
1003 [1] 22:15	205:1,4	/ 11 [1] 129:20	Accord [3] 63:9 86:5	179:17
1005 [1] 22:10	29 [1] 24:24		124:20 accordance 12, 00:17	admit [1] 17:15
1975 [3] 18/:20 188:/,18 1006 (a) 20:14 16 25 19	29,000 [1] 266:5	-0-	89:3.7 195.16	admitted [2] 9:17,18
1770 [3] 22:14,16 25:18		8 [2] 32:16 266:22	accordingly 11 101.5	adopted [4] 119:24
1777 [2] 42:17 46:17	-3-	80 [5] 42:15 43:5,7 50:13		195:22 203:12,15

Discoveries Unlimited Inc., Ph: (709)437-5028

Index Page 1
advance [2] 187:24	266:14 267:21,22	annual [2] 83:13 200:17
212:4	aircrafts [1] 184:17	answer [8] 19:10,11 66:6
advantage [4] 184:18	airframe [14] 165:17,19	104:20 145:19 193:23,24
190:17,20 197:23	165:21 171:16 178:21	198:14
Adverse [1] 228:25	180:24 181:15 185:3,15	answers [1] 173:11
advise [1] 5:9	185:18 191:15 195:24	Anti [1] 197:6
advised [2] 2:6 8:13	190:25 207:9	anticipate [2] 172:11
advisor [2] 159:10,23	airirames [11] 1/1:6	229:9
aeronautical [2] 259:6	178.12 181.13 182.9	anticipated [2] 2:8
263:17	197:22 199:19 267:8	220:22
Aerosafe [2] 11:6 110:10	airport 151 170.1 182.8	AOMS [3] 229:21 231:7
affairs [1] 12:16	182:9 224:8 245:3	231:14
affected [3] 20:9.12	AIS [1] 247:24	apparatus [2] 218:10
153:14	akin [2] 60:9.13	271:8
affirmed [3] 18:24 19:1	Alberta [4] 25.13 18	appear [1] 3:12
19:2	28:1.5	applicable [3] 67:16
Africa [5] 23:5 25:9 29:7	alert 111 269:6	
29:10,16	alerts m 92.11	application [2] 82:22
afternoon [7] 173:24	aligned (1) 130-15	
174:1 175:13 185:16	alignment (1) 250.22	applied [9] 96:16 102:14
222:8 228:13 269:15	allocated (1) 251.15	124:10 126:24 170:11
again [33] 40:14 41:22	allocated [1] 251:15	apply (11) 12:18 67:2
45:19,22 48:4 60:18 66:1		99:14 100:24 110:4 111:3
107.9 109.13 113.11	allow [7] 106:11 112:25	124:14,15,18 161:3 239:2
116:8 125:17 128:15	238.2 265.23	appointed [2] 23:19
137:1 142:19 146:4	allowed (4) 83.4 113.3	59:21
169:11 173:10 183:13	233:5 249:17	appraisal [1] 55:5
186:14 195:15 201:20	allowing [2] 249.23	appreciated [1] 268:5
221:18,20 231:16 240:2	250:19	approach [5] 6:19,22
against 1121 66:11 84:2	allows [6] 16:21 21:8	111:19 112:24 119:24
84:4 85:2 97:8 102:20	92:8 112:24 153:12	appropriate [16] 3:19
105:15,17 109:2 136:3	265:21	4:11 18:24 83:18 96:15
156:3 219:11	almost [1] 261:25	104:21 162:4 165:21
age [3] 182:8 218:1	alone [2] 106:15 206:6	214.14 16 219.2 226.14
232:12	along [8] 54:24 67:6	236:17
agency [1] 15:21	134:16 137:14 155:5	approval 151 77.20 84.24
agenda [2] 132:11,16	158:16 261:24 266:12	84:25 168:4 237:11
ago [4] 26:7 145:9 202:16	alongside [1] 153:12	approvals [2] 62:6
249:13	alternate [1] 172:17	240:19
agree [1] 78:9	always [14] 7:8 8:1 22:2	approve [2] 77:3 237:23
agreed [3] 7:19 155:11	30:9 114:13 147:19	approved [4] 77:7 89:7
195:13	151:12,14,19 182:15	169:19 226:23
agreement [8] 34:5	ambulance (1) 180.2	April [3] 90:17 195:19
59:24 80:15 145:10 181:1	amoutance [1] 180.2	196:20
184.2,8 203.8	A morizon (4) 88-6 8 10	area [53] 3:16 14:7 20:1
80.11 13 15 20 90.8	American [4] 88.0,8,10 88.11	32:15 33:3,17 35:14
150:9 197:1	A methyst (1) 51.12	36:23,24 40:1 41:8 42:5
Agusta [2] 193:8.19	Amethyst [1] 51.12	61:3 63:14 65:17 70:7
ahead 161 19:21 113:3	among [1] 190.22	79:14 86:11 96:3,8,23
133:16 153:25 156:18	amongst [1] 255:19	98:19 104:11 128:2
225:7	177.1 208.7 269.11	137:12 141:6,7 154:12
aid [7] 80:11,13,15	analogy (5) 08.15 00.18	154:12,13,23 155:3,13
136:24 150:9 152:24	103·21 114·12 115·2	215.1 217.21 228.23
222:19	analysis 1161 105.8 10	243:7 245:3,4 247:22
air [12] 35:19 157:10	105:23 106:5.5.14 154:13	256:17 266:3,4,8,16
180:2 218:12 219:7,21	187:10 188:3 189:16	areas [17] 11:4 36:10
2+2.13,13 244.14,14 250:7 258·4	193:4,5 194:24,25 195:4	37:2 38:2 59:10,13 70:16
aircraft [201 38.18	195:12	92:3,7 93:23 94:4 108:12
166:18 169:2 170:6.16	analyze [1] 199:13	124:0,21 141:7 142:2
184:21 197:10 223:6,14	analyzed [1] 242:6	arone (1) 100-16
247:21 250:4 257:16	anchor [1] 150:4	arise (1) 7.6
260:3 265:4,10,11,24	Andrews [1] 77:25	

advance - aviation **Offshore Helicopter Safety Inquiry** arises [1] 252:21 Atlantic [6] 63:9 71:11 arising [3] 129:10 130:15

Multi-PageTM

29:6

ATL [2] 149:22,24

86:5 124:20 229:18,24 136:17 attached [1] 159:1 arrangement [6] 58:14 attaches [1] 55:11 156:2 181:12 183:3,7 attempt [2] 8:17 162:12 186:6 attend [1] 123:3 arrangements [3] 56:4 attendance [1] 151:15 101:4 156:18 attended [2] 25:1 119:11 array [1] 260:7 **attending** [1] 130:10 arrival (1) 54:11 attention [1] 214:25 arrive [3] 240:13 243:4 attitude [4] 24:12 97:16 243:21 97:20 118:7 arrives [1] 244:9 attitudes [1] 94:11 arriving [2] 131:11 attractive [1] 178:19 244:10 audit [12] 83:13 84:2,6 arrow [1] 114:21 84:20 107:21.23 140:10 art [1] 92:18 140:13 142:17 143:14,15 articulates [1] 244:24 143:20aspect [4] 22:12 37:22 audited [2] 108:1 140:8 117:9 123:1 auditing [2] 142:22 aspects [13] 22:21 26:17 143:21 36:11 90:2 94:17 95:12 audits [11] 13:18 83:23 118:13 133:8 141:5 178:6 86:17.17 107:25 140:4 178:18,20 265:25 140:15,19 143:7 186:16 assembly [1] 212:12 200:18 assess [3] 82:8 96:5 193:3 August [1] 23:18 assessed [2] 156:13 authorities [4] 23:13 187:11 79:20 86:3 87:24 assesses [1] 77:8 authority [27] 59:23 assessment [7] 79:21 71:2 85:7,22,23 86:9,12 106:16 156:6 196:24 86:16.22 87:4.8.16 88:13 220:20 229:5 230:1 88:16,18 89:1,12 90:18 104:11 154:23 195:2 assessments [3] 112:7 257:15 265:2,9,12 266:15 229:3.20 267:20 asset [6] 18:21 24:22 Authority's [1] 70:15 26:16,19 54:3,13 authorization [23] assets [7] 34:2 54:25 63:16,18,22 67:6 68:3,5 57:15 92:21 151:14 72:7.7.9.13.18 73:13.20 186:23 258:15 73:23 74:12,13,23 75:22 assign [1] 135:21 75:25 76:2 77:3 84:24 **assigned** [2] 133:6 166:6 181:16 authorizations [11] assignment [2] 28:16 64:10.10 67:24 69:11 73:7 74:14.19 75:18 80:6 assist [6] 21:6 54:4 129:7 124:9 167:9 152:16 169:7 265:16 authorize [1] 86:10 assistance [3] 81:5 authorized [1] 73:10 265:10 268:3 automated for 169:8 associate [1] 105:21 177:18 247:24 248:7,14 associated [24] 51:19 252.17 55:4,7 56:4 75:24 81:20 automatic [1] 217:18 100:3,6,21 101:6,22 automatically [1] 216:4 102:19 104:14 106:21 auxiliary [4] 163:11 107:1 109:8 133:1 150:2 164:7 223:20 265:18 154:19 163:10,11 219:10 **availability** [4] 115:24 220:8 235:20 116:2,4 195:7 Association [5] 2:1 11:9 23:9 71:15 226:17 **available** [17] 14:22 18:2 45:16 81:7 182:5.11 assume [2] 183:22 218:2 219:21 246:7,9 259:25 247:18 251:4 258:2 260:4 **assurance** [1] 88:22 260:7 267:6,8 assure [1] 27:14 average [1] 46:6 ate [1] 241:18 aviation [12] 70:15,18

Discoveries Unlimited Inc., Ph: (709)437-5028

71:2,3 110:14 189:20

Multi-PageTM

avoid - case Offshore Helicopter Safety Inquiry

			Olishore Helic	copter Safety Inquir
195:2 200:13,18,19	basin-wide [1] 119:24	187:10,13,14,15 209:10	boundary [1] 266:9	calls [1] 259:9
206:15 265:15	basis [24] 15:8 20:24 21:7	209:17	bow [1] 53:5	camel [1] 210:22
avoid [3] 31:5 192:6	69:7 77:16 79:18 84:18	bidding [4] 186:23	box [3] 76:2 163:12 164:8	Canada [57] 11:4 15:22
221:15	135.24 115.10 110.0	202:11 209:10 212:2	BP _[2] 21:17,21	25:5 28:3,6 29:14 42:9
	153:9 172:23 186:7.9	Dig [3] 38:23 54:1 232:21	brand [1] 115:11	51:7 01:8,21 02:5,0,15
173·3	237:10 246:20 249:5	Digger [2] 106:19 215:6	breach [1] 64:19	89:25 90:7.10.14.20 91:3
awaiting (2) 116.22	252:25 259:10 267:4	Diggest [1] 137:12	break [11] 59:4 71:17	91:5 124:14 142:13 145:9
156.5	basket [1] 255:10	billing [1] 191:2	72:22 73:1 112:11 117:5	146:2 148:15 166:12,13
award 11 187.24	beacon [7] 212:13 216:2	bio [1] 141:24	153:17 222:8 228:13,17	187:17 189:7,9,17,18,22
awarded (4) 187.18	217:7,23,24 218:1 241:11	bird [1] 197:9	252.5	196.19 203.1 13 15
209:10 211:11,13	beacons [4] 216:1,19	bit [42] 2:11 6:19 7:22,24	breathable [1] 218:12	208:24 233:16 256:12
aware [13] 11:14 36:2,4	218:1,7	14:4 30:13 39:5,9 40:13	177.17 218.10 219.2	259:4 263:5,12,13,13,14
132:25 210:12,16 211:6	Decame [6] 21:20 22:19 22:22 23:1 122:20 123:12	76:4 90:5 92:24 93:15	hreaths (1) 218.13	263:22
213:8 233:18 238:5,10	become (12) 5.11 9.5	94:21 99:2 115:12,17	brief (2) 21:24 22:2	Canada's [3] 70:9 152:6
241:1 242:9	45:16 48:4 72:15.17	116:14,22 118:11 135:13	brief [2] 51.24 52.5	189:11
awareness [1] 234:19	124:10 132:21 167:7,12	149:6 151:23 161:24	169:19	Canadian [7] 1:25 71:14
away [7] 22:13 23:17	197:18 232:14	191:11 196:17 208:17	briefing (6) 169:22.24	265:7.17
182.8	becomes [2] 126:16	214:19 222:6 235:7	242:1,4 244:3,5	cannons [1] 38.21
awful m 5·4	158:20	248:13 249:6 251:2	briefings [1] 234:16	cannot (5) 106.15 133.13
avis 11 110.7	becoming [4] 22:8 109:7	256:17 257:1	bring [12] 1:11 5:6 22:24	136:18,20 241:22
		Blackberry [1] 225:6	46:16 57:6 73:16,21	CAODC [1] 226:16
	Deds [4] 50:13,15,16	blacked [1] 15:3	110:23 111:21 145:11,14	$CAP_{[1]}$ 70:18
	heforehand (1) 104-15	blue [7] 22:17 23:2	181:10	CAP437 [3] 70:18.21.24
B [2] 130:2,4	begon (2) 25.4 102.2 6	247:22 248:6,12 252:16	bringing [2] 93:3 243:25	capabilities [7] 32:25
baby [1] 181:14	begin [3] 25:4 195:2,0	201:4	broadcasters [1] 222:5	55:13 95:17 152:12 153:1
Bachelor [2] 24:25 25:2	28.15 42.4	22.9 22 40.12 46.4 48.12	broken [2] 16:10 31:16	187:13 189:16
backdrop [1] 5:14	heginning 131 193.5	54:7 56:1 60:9 63:17,21	brought [8] 11:24 12:10	capability [7] 38:4
backed [1] 42:13	213:6 214:6	65:12 66:18,23 68:21,24	102:25 145:14 182:24	152:25 153:3 174:24
background [2] 21:11	behalf [5] 19:23 60:6	72:8,12 73:8,20 74:14	BST [3] 179.20 233.3 10	196:12 262:10,14
24:19	157:25 160:22 198:13	//:5,6,20 /8:1 /9:22	BSTR [2] 233.2,10	203:8 255:25 256:2
Dackiog [5] 1/4:15,18	behavioural [3] 108:6	83:20.21 84:6.9.13.13	building (1) 43.14	canacity (a) 2.0 44.10
bag (1) 171.0	108:10,17	86:10 100:10 107:18	built (1) 116.4	50:16 171:16 195:7 196:3
hag $[1] 1/1.9$	behaviours [2] 93:9	121:16,18,19,21 123:18	bulk (5) 05.24 06.3 08.16	196:4,10 208:8
179:15 235:12.12.13.13	94:11	124:7,20 126:9 128:7	98:17 150:13	CAPP [27] 4:3,10,13
235:17 240:20 241:6	Denind [1] 183:17	133:12.13 136:10 137:14	bulletin [1] 213:18	11:9 70:25 71:14,14
243:23	Beliast [1] 22:5	139:19 147:19 148:24	bullets (1) 122:13	118:16,17,18 119:6,9,11
ball [1] 268:14	Bell [1] 194:8	150:19 152:25 157:20,23	burden (1) 254:10	121:12.21 219:4 220:18
Banks [11] 20:1 25:8	below [1] 47:25	159:10 162:20 163:6,6	Bureau [4] 88:6.8.10.12	225:12,17,25 226:2,18
27:9 35:15 38:10 57:17	benefit [1] 223:2	165:24 168:17 170:23	burning [2] 131.15	236:6
92:22 143:3 190:12	benefits [1] 187:18	176:18,22 188:5 204:25	137:18	car [1] 194:2
Barnes [3] 2:2 9:14	beside [1] 29:1	205:4,13 213:18 231:1,5	business [15] 20:18	carbon [1] 212:16
118:24	best [12] 12:10 93:3	236:7 237:17,23,23 238:1	21:15 25:16,20 27:25	card [1] 104:10
barrels [2] 44:9 50:10	113:20 140:12 194:13	259:7 240:24 242:6 247:5	86:13 139:3,8,11 141:5	cards [2] 243:18,18
barrier [4] 114:20,20,22	225:20 231:9 252:3 271:7	Board's 151 65.22 77.2	149:15,25 151:10 172:22	career [11] 24:3 25:4,7
118:5	better [4] 12:11 19:11	84:20 88:17 165:25	huving (2) 194.2 3	25:12,21 26:12 28:2,21
barriers [9] 92:5 93:24	50:2 173:12	boarding [3] 242:6,15		29:13,20 30:8
113:10,16,17 114:2,3	between [34] 10:7 15:9	244:16	-C-	careers [1] 26:25
11/:5 118:15	25:13 37:23 38:13 40:5	boat [1] 206:4	$\frac{1}{C_{12}}$	171.6 12 23 179.11 12
191.15 192.25	40:0 45:10,24 40:7 61:10	boats [3] 149:15 151:24	C[4] 17:11 75:7 151:5 255.17	179:23 196:9 234:22
hased [15] 20.18 33.17	80:16 124:12 135:4 138:7	179:24	C-118 (1) 17.25	235:18
34:1 39:19 40:8 44:4	142:7 145:10,21 147:3	bodies [2] 131:1 226:19	C-NLOPB [26] 2.22 4.3	carried [6] 98:11 170:5
69:8 89:25 149:7 164:14	155:13,19 160:19 177:21	bogus [1] 199:4	4:20 10:23 12:16,19 61:1	170:21,24 171:6 263:17
214:18,24 220:17 239:21	181:1 184:3 189:24 191:5	books [1] 156:4	61:13 62:7,19 63:16,17	carriers [1] 21:19
200:10	hevond (1) 116.13	boomerang [1] 199:15	70:1 77:24 123:18 129:1	carry [5] 63:10 98:20
Dasic [8] 202:24 211:21 222:16 19 223:17 232:23	hid [10] 143.3 186.25	born [3] 24:24 25:14	139:22 142:13,17 143:13	109:11/0:191/2:9
233:1 234:6	187:7,9,21 189:9.16.20	28:18	195:15 242:10 255:12	carrying [2] 153:1,3
basin [12] 38:11 41:4	189:21 209:8	Dottom [3] 42:23 43:11	cabin [1] 244:14	cascade [2] 98:25 99:6
46:15 49:15 56:14 145:12	bidder [1] 187:16	hound 121 240.24 267.22	cadet 11 21:17	cascades [1] 98:9
190:12 196:1,16 197:19	bidders [9] 187:3,6,10	houndaries 11 107.00	California III 29:15	cascading [1] 102:5
219:10,23		550110a1105[1] 127:22		case [22] 7:6 15:23 47:3

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

cases - concern Offshore Helicopter Safety Inquiry

			Olishore Hend	opter Safety Inquiry
47:13,19 49:3 60:13	234:10 241:22 250:22	clearance [1] 155:2	Commissioner [63] 1:2	188:10,15 192:2 209:24
113:3 140:5 144:23 145:4	251:12	clearly [6] 6:16 15:20,24	1:7,13,18,21,22 3:8 4:9	247:8
149:21 162:23 164:17	changed [7] 18:3 20:8	78:17 193:23 244:24	6:2,8 9:1 10:11,16 18:10	company's [1] 134:17
167:22 179:17 187:23	27:11 133:9 156:9 205:20	clock [1] 222:13	18:14 19:4,23 21:23	compartment [7] 169:3
188:3 237:10,10 241:2	250:14	close [7] 9:23 40:18 41:21	24:14,21 27:18,21 30:14	170:23 171:12,14 180:1
	changes [7] 17:8 18:8	95:25 117:3 156:16 239:8	46.13 68.19 72.23 24	235:13,18
Cases [5] 164:2,7 174:17	70:22 77:17 207:6 269:3	closed (1) 164.14	73.3 153.19 22 25 156.20	compelling [1] 11:8
175:0 238:19	269:6	closest (4) 36.24 37.17	156:25 157:4.16.25 158:9	competence [2] 97:2.17
cast [1] 232:3	changing [2] 93:14	172.17 262.3	158:14,22 159:14,19	competency (1) 141.6
categories [3] 94:1	180:12	closet (1) 5:15	160:16 161:2,16 162:6	competent (1) 06:25
109:13 113:16	Chapter [1] 236:25	closing the 22.0	164:20 170:9 212:1	competent [1] 50.25
category [1] 113:15	characteristics [1]		218:25 228:14,15 229:17	
catering [4] 144:6,16	199:25	CO [5] 33:22 35:2 59:18	233:7 239:23 244:21	competitive [5] 186:25
146:5,7	charter [4] 16:24 17:6	60:2,15	268:21 269:16,24	18/:/ 189:9 190:1/ 191:2
caution [1] 17:8	17:25 183:12	co-chair [1] 124:25	commissioning [3]	competitor [2] 192:4,20
cautioned [1] 222:3	CHC [1] 188:14	co-venturer [1] 34:2	22:24 25:25 26:10	complaining [1] 227:18
caveat [1] 173:14	check [9] 85:22 142:23	co-venturer's [1] 33:24	commit [1] 253:6	complaint [1] 152:10
Celsius [2] 35.20.21	174:21 215:15 241:6	co-venturers [6] 34:4	commitment [7] 4:16	complement [1] 205:10
centre [0] 47:1 51:21	244:22 245:3,5,9	34:13,25 42:8 59:21 60:6	33:18 93:6,6 222:5	complete [14] 65:16
193.6 258.13 264.9 11	check-in [5] 239:20	co-workers [2] 27:1,14	253:10,21	83:15 133:9 161:12
264:20.23 268:9	240:11,15 242:24 243:1	coast [24] 25:9 49:15 51:6	commitments [8] 84:3	174:10 219:5,18 220:23
centres [2] 259.4 263.22	checked [3] 116:6 240:18	118:18,21 144:2 175:9	84:8,23 85:3 187:18	221:5 223:1 227:18
CEO (2) 01:22 00:11 16	242:20	194:22 196:1 211:20	262:9,24 263:1	229:19 232:23 238:19
CEO [3] 91.23 99.11,10	checking [1] 117:12	226:21 233:13 247:21	committed [2] 20:23	completed [17] 73:11
CEP [3] 1:10 2:20 5:3	checklist [3] 100.23	263:12,18 264:20,23	27:15	100:9,25 154:15 155:24
certain [12] 36:11 103:16	101:3.18	265:5,8,13,14,17 266:5	committee [40] 60:1,8	196:24 197:2 209:9
103:19,23 112:1 115:8	checks [3] 179.18 242.17	266:14	60:14 92:24 118:23 119:5	222:15 233:1,8 234:17
115:22 124:15 150:0	244:17	coastal [1] 248:21	121:7,8 125:5 124:24	254:20 255:1,4 257:10
157.9 250.25 201.17	cheese [18] 93.15 113.12	Code [3] 124:1,4,14	125.7,20 120.10,24	250.21
10.7 26.11 50.12 60.25	113:14 114:5.7.12.13.14	cold [2] 35:19 37:4	129:2.25 130:5 133:12	
85.1 106.14 110.13	114:18 115:14,17,21	collaborate [1] 30:5	136:13 162:15,17,22,24	completion [3] 25:24
115:20 135:23 137:1	116:10,14,22,25 117:2	collaborative [4] 6:19	162:24 163:5,18,20,21	
174:9 194:15 224:23	117:20	6:22 140:3 145:15	210:3,22 219:5 220:6	complex [2] 106:4 219:3
232:3 247:4 248:12	Chemical [1] 28:2	collaboratively [1]	225:18,24 226:16,25	complexity [1] 106:2
251:25 269:10	Chev [1] 199:21	139:10	committees [15] 118:25	compliance [19] 60:5
certificate [7] 65:13 86:8	Chevron [2] 42:9 46:23	colleagues [3] 27:10	119:9,20,25 120:4,9,10	63:23 64:17 84:7,22 85:1
86:23 87:5 88:23,25	chief [5] 21.20 22.8 77.2	180:9 229:12	120:17 121:15 123:23	88:20,24 97:6,8 108:1,5
271:1	98:8 165:23	collected [1] 240:21	124:25 125:4 129:15	118:8 142:11 143:0 152:0
certificates [2] 90:15	children [2] 25.14 93.7	collectively 121 40.6	213.14227.19	100.7 250.0,8
233:10	chin [2] 23.14 95.7	126:5		204:5 0
certification [9] 66:17	198.25 199.5 7 11 16	colour (1) 260.24	communicate [3] 130:1	204.5,9
88:16,21 90:10 179:20	ching (2) 200-23 201-6	colourful (1) 177.20	134:15 166:25	
179:21 195:1 208:24	chips [2] 200.23 201.0	condition in 26.9	communication [6]	complies [1] 79:6
230:5		COMDITIATION [4] 30:8	120:8,12 121:16 135:3	comply [5] 67:4 68:2
certifications [1] 236:13	choose [4] 99:17 215:3,7	30:11 38:3,24	155:21 215:11	99:7 207:25 225:11
certified [6] 87:2 88:11	241:25		COMMUNICATIONS [8]	component [6] 44:4
148:16 166:12 205:24	CHOSEH [5] 11:18 12:9	compinea [1] 106:/	121:15 131:13 134.13	1/:13 18/:20 23/:16
220:21	209:21 235:15,15	comfort [1] 214:24	265:20.25	242:19 244:20
certifies [3] 86:16 87:10	CHEISUMAS [2] 241:18	comfortable [3] 107:17	community 121 137.20	components [5] 43:17
09:12	202:13	107:18 262:21	137:21	79:15 212:9 242:25 263:11
certify [2] 87:25 271:2	Cindy [1] 2/1:11	coming [9] 54:14 84:12	companies [24] 12:7.13	205.11
certifying [18] 23:13	circle [2] 44:1,3	127:1 197:17 202:8 206:5	12:17.25 13:10.12 14:1	08.10 139.4 226.1
85:7,21,23 86:3,9,12,16	circumstances [3]	260:10,15 268:12	14:3 15:9 21:13 33:20	20.10 139.4 220.1
88.18 80.1 11 00.18	235:14 238:18 261:23	command [7] 79:20,25	33:21 92:25 99:12 126:5	219:7
actors 121 200.2 266.15	civil [5] 25:3 70:15,18	257:16 259:25 265:20	140:2 149:20 181:9,15	219.20
CCSP 202.2.14	71:2,3	200:10 207:20	188:11,14 190:19 198:8	119.25
CGSB [2] 203:2,14	civilian [6] 219:24 220:5	commander [1] 202:6	255:13	comprised (1) 124-24
chair [1] 125:1	265:9,13,16 267:21	commencement [1]	companies ² [1] 223:25	
challenge [2] 174:11	clarify [2] 161:21 262:8	/4:15	company [36] 11:15,25	
231:21	clarity [2] 184:20 185:22	comment [4] 9:3 182:21	12:4 14:17 15:7 22:17	concentrated [1] 251:15
challenges [2] 37:19	class [1] 40:1	211:25 260:20	24:17 25:25 20:5 27:23	concept [7] 31:21 32:7
82:3	classed [1] 49:23	comments [11] 1:8 6:3	60.10 66.5 71.25 98.8	33:15 95:3 97:22 114:19
chance [1] 175:4		0:5,18,24 8:24 10:12,15	50.10 00.5 /1.25 /0.0	213:5
	cleaned [2] 48:13.14	11.13 10.21 /1.1/	126:8 129:4.4.5 133:2	0.1.1.1.0
change [9] 103:25 104:3	cleaned [2] 48:13,14 clear [3] 167:7 174·17	11:13 19:21 41:14	126:8 129:4,4,5 133:2 134:16 139:6 141:8 142:1	concern [10] 8:16 162:11
change [9] 103:25 104:3 129:24 168:3 224:21	cleaned [2] 48:13,14 clear [3] 167:7 174:17 256:20	11:13 19:21 41:14 commercial [2] 187:14 190:5	126:8 129:4,4,5 133:2 134:16 139:6 141:8 142:1 143:9 144:19 149:11	concern [10] 8:16 162:11 210:7,9,12 214:4,5 219:19 223:25 228:1

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

concerned - deck ffshore Helicopter Safety Inquiry

			Offshore Helio	copter Safety Inquiry
concerned [3] 5:16	4:17 9:21 109:6 169:5	126:9,12 139:2,10,12,17 140:4 142:11 143:21	cost [2] 204:7,12	crew [3] 132:13 205:20 207:6
concerning 11 155.1	193:9 196:7,11	226:18 236:3	62.4 63.2 2 146.17	Crews [2] 158.1 166.14
concerns [10] 6:21 15:15	considerations [4]	contracts [9] 13:15	147:14 163:25 164:1	criteria (1) 174.21
188:1 198:9 200:14	119:23 193:11 195:23,25	139:7 142:7 146:15,19	165:14,16 171:5 173:12	critical [16] 30.12 141.17
209:18 214:20 217:20	considered [9] 38:14	181:8,10 191:5,8	178:8 179:2,4 180:2,21	160:3.3.5.19.19.24.24
220:8 237:19	64:17 133:6 155:13 161:6	contribute [1] 181:2	186:11,16 187:6,18	161:7,9,13 178:6,20
conclude [1] 102:1	161:9 164:14 233:12	contributes [1] 94:11	188:10 189:6,23,25 190:5	196:7 246:2
concludes [1] 198:1	245:1	control [9] 54:4 89:9	193:1.8 194:15.20.23	criticism [1] 213:3
conclusion [3] 109:19	considers [1] 133:5	96:9,9 209:4 246:9,10	195:9,12,20,24 196:6	criticize [2] 13:21 221:10
136:23 270:3	consistence [1] 115:12	258:11,14	197:1,1,11,18 214:13	Cross [1] 85:22
concrete [1] 44:4	consistent [1] 112:24	controlled [4] 97:12	217:2 236:9,11 239:19	crude [3] 21:18 22:3 49:1
concurrent [1] 158:7	constantly [3] 8:12	101:3 134:20 222:17	240:3,11 241:12,14,13	cube [1] 176:21
concurrently [1] 170:20	114:7 135:3	109·1	252:7 253:7 254:11,15	culture [4] 21:5 133:3
condition [3] 67:20	126:23 127:2 128:10	controls (2) 77.8 100.2	254:22 255:3 258:10,11	134:18 139:12
231:3 249:11	constitutes (1) 184-11	convenient (1) 222.7	261:4,6,14,18 266:10,11	curious [1] 254:2
conditions [10] 36:9	constructed (a) 29.12	Convention m 256.12	267:3,23	current [9] 3:13 26:16
37:4,10 73:12 84:25 85:1	89·3	263:6	Cougar's [7] 169:25	29:13,21 33:5 50:6 53:25
100:17 147:25 155:7	construction (4) 76.14	conversation [2] 103.4	193:5 195:19 196:14,20	69:11 153:14
conduct (51, 50:22, 73:10	126:19,21 128:7	103:5	counsel (18) 2.5 6 6.14	
85:21 186:16 251:18	consultant [2] 11:6	converse [2] 130:6 131:9	7:13.17 9:8.12.13.22	-D-
conducted [7] 98:4	189:20	Conversely [1] 134:25	10:8 14:21 221:7,22	D'Arc [2] 38:11 46:15
189:18 193:4 194:23	consultation [2] 231:7	converses [1] 130:3	269:5,14,21,22,25	daily [6] 135:24 138:4,6
195:21,24 229:3	255:13	convert [1] 229.1	counterparts [1] 49:8	138:10 172:23 186:7
conducting [2] 19:24	consumers [1] 201:2	coordinate (1) 120.23	country [1] 28:7	dangerous [10] 123:21
176:19	contact [3] 81:5 162:19	coordinated (1) 266.17	couple [6] 19:20 145:9	133:4,7,8,15 162:21
conducts [1] 83:13	216:4	coordination (0) 140.0	180:13 188:8 192:18	163:1 164:6 234:20,22
conference [1] 138:14	contain [1] 166:3	258:13 259:3.5 263:20	212:8	dark [3] 173:8,9 175:14
conferred [1] 59:23	contains [3] 44:5,7 61:5	263:21 264:9,11 268:8	Course [27] 11:6 13:25	darkness [6] 173:17,18
confidential [11] 14:5	contemplate [2] 5:24	coordinations [2]	26:21 56:14 57:25 96:4	data m 60.5 149.10 10
14:10,15,19 16:22 17:3	6:1	263:23 265:1	96:19 97:4 101:21 107:20	databasa m 16:0 241:12
17:0,12,20,22,25	contenders [2] 193:6	copies [1] 69:25	108:5 116:24 144:12	data m 22:22 155:24
196.15	194:24	core [6] 51:10 141:6,7,11	155:9 156:1 160:13 213:9	uale [3] 22:23 155:24
confined (1) 100.16	CONTEXT [6] 3:17 5:10 6:17 10:12 21:12 95:22	151:10 189:22	226:24 233:3 237:4 258:1	dated (2) 60.22 271.0
confirmed (2) 217-3	contingency (2) 80.1 8	Cormorant [1] 174:25	Courses (2) 116:17	dates 11 236.14
243:24	continuelly (1) 225.15	Cormorants [3] 260:6	246:22	dav_to 11 07.23
conflict [1] 124:12	continuary [1] 225.15	267:1,2	cover [1] 90:22	day = to [1] 97.25
conform [1] 90:20	29:17 30:11 59:3 238:3	corporate [1] 200:19	coverage [1] 263:7	106:20
confusion (1) 262.13	238:25	Corporation [2] 18:18	covered (1) 269.12	davlight [4] 173.22
congested [2] 154.13	continues [4] 27:8 91:6	42:10	covering [2] 90.8 123.20	174:11 250:16,18
155:3	108:19 246:22	45.8 13 53.8 54.21 57.25	covers (1) 264.2	days [10] 10:8 12:20
conjunction [1] 147:22	continuous [8] 20:24	60:22 61:18 62:17,21	cracks [3] 83.7 163.12	31:20 104:17 145:16
connected [1] 47:2	21:7 30:3 92:13 108:22	67:22 71:2 74:4,8 75:13	164:8	199:2 216:25 233:6 239:3
Conoco [3] 12:4 40:10	134:21 202:9 267:4	76:12,17 78:11 79:8 86:5	cradle [1] 158:18	267:4
40:23	Continuousiy [3] 77:14	87:6,16,21 88:15 89:15	craft [2] 57:9 152:13	deal [11] 1:14,16,17,19 10:7 37:7 106:13 144:14
ConocoPhillips [4]	contract [25] 12:16 22:12	97:15 104:25 106:10	crafts (1) 152:14	186:15.17 247:13
40:21 56:13 181:23 267:9	140.7 142.5 9 144.3	110:12 117:24 125:10	crane [12] 154.22.23	dealings (1) 268.24
consecutive [1] 132:3	145:3,25 149:11,20 150:7	128:17 130:24 131:21	155:4,5,6 156:1,2 158:2	deals (2) 11:16 203:18
consent [1] 7:19	179:1 181:19 187:16,19	132:9 137:24 140:17	158:11,15,18,20	dealt [5] 2.25 7.14 9.7
consequence [9] 109:8	187:21,24 188:1,3 189:24	143:2,18,23 147:18 148:5	crash [2] 217:1 222:22	154:2 177:17
109:10 110:7 111:7,12	191:4 211:11,13 249:1	159:18 165:7 167:11	crashed [2] 218:17	debate [2] 6:11 233:19
113:6 155:20,23 156:17	contracted [14] 32.12	173:1 178:16 182:11	267:25	December [5] 2:4 68:21
consequences [2] 65:4	138:20 139:4 142:4 145:6	185:9 186:9,21 192:13	create [12] 82:2 92:5	69:22,23 90:16
consequent in 159-10	145:10 147:21 148:3	194:15 204:15 206:9,13	93:24 96:1 113:10 114:2	decide [1] 231:8
conservation 12 62.12	150:15 179:4 181:16	224:5.17 228:22 234:13	127:1 128:9 210:9	decided [3] 9:20 95:11
65:15	184:6 189:20 229:18	236:25 239:15 245:25	created [3] 10:24 26:21	217:23
consider [12] 5:9 63.19	contracting [2] 143:16	248:8 250:10 254:15	156:12	decision [7] 95:9 119:6
95:12,23 111:6 115:3	eontractor (4) 140.7	264:5 271:3	creates [3] 26:20 96:6	164:17 165:18 203:21
116:21 160:24 219:16	142:15.21 144·21	corrected [1] 182:19	212:10	decisions (1) 200.25
223:4 258:24 260:12	contractors [12] 125.5	correctly [2] 97:12	creating [3] 118:13	dools up 567
consideration [11] 4:2		242:22	148:7 174:14	ucck [1] 50:/

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

Decker - dynamic Offshore Helicopter Safety Inquiry

			Olishore Hend	opter Safety Inqui
Decker [4] 11:7 147:12	deployed [1] 265:7	deviation [1] 168:5	discussing [2] 57:11	down [17] 7:11 48:5
207:18 260:21	deployment [1] 260:1	device [6] 218:21 219:2	92:3	56:13 99:2,6 100:13
Decker's [3] 134:11	depth [5] 42:15 43:4	219:8,21,24 220:4	aiscussion [7] 18:5	158:7 169:11 179:16.22
decoration (11, 5·21	47:22 48:3 219:15	devices [2] 169:7 177:17	163:20 268:17 269:25	182:16 199:21 259:14
dedicated [14] 48:21	depuns [1] 216:9	diagram [3] 41:3 42:23	discussions [1] 233:25	download [1] 16:8
58:1,3 151:13 250:20	descending [1] 99:4	dial (1) 138.14	disembark [1] 244:11	downloads [1] 178:9
251:9,13,21 252:5,24	describe [6] 32:4,7 56:10 100:18 119:1 138:25	dictated (1) 255.12	dislodged [1] 171:2	DP [1] 153:11
255:18 257:13 258:17	described [12] 4.20	difference [2] 93.21	dispatch [1] 179:8	draft [11] 65:20 66:4,10
200:1 doom w 04.14	51:24 98:18 109:13	231:19	dispatched [1] 249:24	66:12 67:15 69:2,3,4,5,6
deemi [1] 84:14	118:24 124:7 138:23	differences [6] 52:2,5	dispensed [1] 116:5	81:14
226:24	180:7 250:24 251:11	52:13 217:16,16 242:25	disperse [1] 270:1	dramatic [1] 111:14
Defence [4] 32.23.24	252:18 209:19	different [47] 15:5,10	distance [2] 37:23 38:3	101.21 102.6 8 154.17
260:14 263:9	123.24	37:11 39:18 53:19 55:9	distinct [3] 130:4 265:22	dream [2] 26.23 27.2
defences [3] 113:21,21	describing 11 124.8	98·1 104·21 111·6 17 18	266:21	dried (1) 48.13
113:22	description [4] 16.12	111:19 112:3,9 125:8,19	distinction [1] 160:18	drill [11] 45.19 22 51.21
defer [1] 224:24	32:2,4 189:13	125:24,24,25 126:13,14	distress [1] 265:11	54:19 56:11 64:3 74:2
deferred [1] 118:21	design [17] 38:15 40:4	127:8,13 130:9,13 139:7	distributes [1] 240:23	87:23 88:1 145:2 150:12
define [1] 78:19	45:13 79:19 81:10,13,25	155:25 100:7 178:12	ditched [1] 222:20	drilled [4] 45:19 47:3,6
definitely [1] 230:22	94:6 96:22 97:18 109:11	200:2 202:19 214:24	ditching [1] 222:17	47:14
deflection [1] 38:20	202.19 208.25	215:18 217:7,9,13,14	diverse [2] 102:24 120:5	drilling [64] 12:3,5 19:25
degree [2] 28:1 159:1	designate 11 16.22	231:21 261:9	divided [2] 19:6,13	40.11 41.20 43.18 44.6
degrees [2] 35:20,21	designated [7] 14.15	differently [2] 39:7	diving [8] 65:16,18 75:3	44:7 45:2,2,10,14,15,18
deicing [2] 189:15 197:7	85:19 185:4 192:15 229:6	/3:1/	75:7,9,9 76:9 86:7	45:24,25 47:4,12 54:17
delay [1] 10:5	243:6 265:23	dig (m. 142.4	DND [23] 256:25 257:7	54:23 55:1,3 56:3,9,16
delays [1] 228:25	designed [12] 39:11,13	dig [1] 143:4	257:15,23 258:10,17,22	50:18 57:4,18 61:4 63:12 64:1 65:14 68:14 23 69:2
delegated [6] 85:7,19	44:10 50:13 70:12 89:2	dimensionary 40.17	260:16 263:18,18,21	73:21,22 74:1,22 81:16
88:13 90:6,14,16	210.22 216.8 9	114.25 170.18	265:3,12,19 266:10,13	81:16 86:6 127:3 144:6
delegates [1] 91:5	designing [2] 94.18 18	dioxide (1) 212.16	267:19 268:6	144:16,18,20,20,23 145:3
delegation [4] 85:25	designs [2] 95.14 166.19	direct (9) 58.7 18 78.15	DND'S [2] 257:3 258:15	145:24 140:1,2 152:4
89:25 90:9,22	desire [4] 5.19 30.10	78:24 142:16 188:2 231:7	DNV [4] 86:15 87:14	188:21 226:18
delineation [1] 55:5	221:15 222:24	265:2,9	90:13,17	drills [1] 166:17
deliver [1] 159:2	detail [12] 35:6 59:2	directed [2] 58:9 257:23	16.18 23 24 25 17.5 14	driver [3] 154:22 158:20
delivered [1] 240:22	119:1 127:6 144:8 165:10	direction [9] 40:23 41:10	17:21 92:10 221:4 226:1	197:25
195:18 196:19	180:8 186:18 250:24	41:24 59:25 62:14 69:17	237:8 246:17 269:2,2	drop [1] 48:4
demands [1] 229:15	dotailed (10) 164.2 197.9	directly (7) 11.1 40.4	documentation [5] 2:22	dropped [3] 155:15,15
demonstrate [5] 77:6	187:10 193:9 194:23	141:9 142:7 145:3	4:4 13:14 97:10 120:22	155:18
167:23 168:18 242:11	195:4 209:8 235:1,4	directors [1] 60:9	documented [2] 112:25	dropping [1] 155:22
244:7	239:20	discharge [2] 115:8.11	documents (201 2.18 20	dual [3] 204:8,11 256:3
Dennis [1] 268:7	details [6] 60:19 170:4	discharged [2] 48:15	3:2.15.24 4:1 5:10 7:18	due [4] 17:13 193:9
deny [1] 237:20	247:14,23,25 252:8	115:5	9:4 13:23 14:1,5,14,15	dunlicity (1) 82.2
departing [2] 243:7	detector [1] 241:8	discipline [1] 107:14	14:16,16,20,24 15:24	durability (1) 62.5
244:23	determine [10] 83:17 108:1 109:15 141:4	disconnectible [2] 49:16	16:4,10,11,16 17:10,11	during [41] 22.6 23.6
department [5] 25:6	162:25 191:9 220:9	53:15	247:1	25:24 29:9,20 44:22
denartmental [12]	229:20 231:14 255:14	discovered [5] 42:16	doesn't [13] 5:24,25 37:6	50:13,15 69:8,12 80:17
126:17 128:5 132:23	determined [4] 156:16	46:16 51:8 214:17 268:13	47:22 114:21 128:1 156:7	83:19 88:23 93:24 95:16
136:4,6,8,10,12,14,17	194:20 219:20 232:7	271.14	173:18 175:1 215:17	107:5,20 112:11 121:18
136:19 138:8	determines [1] 255:11	discovery 121 25.11	221:18 238:24 243:11 Deba w 29:11	189:1 205:18,19 207:6
departments [2] 128:23	determining [3] 109:24	42:19	Dolla [1] 28:11 dome.rep. (-02, 12-7, 54-17	209:16,19 212:7 214:11
215:15 doporturo (5) 240.15	devestating (1) 20.7	discuss [17] 18:7 21:4	108.15 109.8 128.2 131.6	214:12 222:22 233:6,23
243:3.5 244:23 245:2	develop (a 5,19,22,10	31:19,25 32:14,16 33:5	137:9 138:11 140:15	240.15 242:5,4 245:21
depend [1] 128:1	46:17.21 108:25 109:4	33:14 76:3 95:1 102:25	142:6 143:2 146:5,7	duties [7] 123:21 157:23
dependant [1] 232:12	developed [5] 7:10 46:24	224:23 268:10	161:1 169:17 180:13	184:22 185:4 249:5
depending [10] 46.7	98:14 211:17,22	discussed [12] 42:6	230:7 233:20 262:7 15	251:16 252:5
50:6,22 58:17 104:22	developing [4] 8:14	59:18 92:24 101:14	donned [4] 242:11.14	duty [6] 107:7 151:23
126:24 157:22 170:16	92:16 213:11 220:18	108:25 130:16 135:6,13	244:6,14	162:19 232:11 249:7
218:12 235:14	development [11] 18:18	201:16 242:3 244:5 249:6	door [1] 235:16	dynamic m 114.15
aepiction [1] 50:1	26:3 27:23 34:11 42:7	aiscusses [5] 32:11,20	doors [1] 170:18	153:11
aepiete [1] 45:21	186.22 189.8	52.25 55.1 155.11	dots [1] 40:17	

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

E-225 - expression Offshore Helicopter Safety Inquiry

				optil survij
	205:12	entire [7] 20:11 29:25	evacuated [2] 180:4	executed [3] 55:2 104:18
- <u>E</u> -	eliminated [2] 15:2	251:21,24	251:2 evacuation (5) 180:3	190:25 executive (5) 60:14
E-225 [1] 193:7	eliminating [2] 95:18	entirely [3] 83:24 112:3	205:24 229:1 246:11	71:24 78:20 98:8 119:5
e-mail [1] 7:16	109:1	192:2	259:17	executives [2] 91:23
E452 [7] 122:23 209:11 209:22 211:23 212:9	emergencies [2] 184:12	entities [1] 12:13	evacuations [1] 230:5	98:9
216:1 218:21	246:23	entitled [1] 19:9	evaluate [3] 83:18	exemption [11] 236:17
Earle [11] 1:9,10,20,23	emergency [27] 9:10	entity [2] 14:17 15:21	evaluated (1) 109.5	237.5,20,22 258.5,22,25
1:24 3:10 6:13,21 7:1 9:5	166:23 218:9 229:2,13	entrances [1] 107:21	evaluating [2] 219:1.7	exemptions [7] 236:18
earliest (1) 9.22	243:21 245:23 246:1,5,6	100:16,21,25	evaluation [3] 88:19	237:9,15,18,21 238:13
early [13] 25:12 26:11	240:13,15,20 247:7 251:8	environment [21] 20:19	208:23 209:9	240:19
40:4 42:12 144:25 175:12	264:22 267:17 268:2,11	32:1,5 33:20 36:4,6,15	evening [2] 174:19	exercises [2] 202.11 204.7
175:13 199:2,16 201:6	employ [1] 40:6	37:17 38:25 89:6 119:3	1/3:14	exhibit [8] 3:7 16:21 17:6
earth (1) 48.19	employee [4] 123:21	172:5 194:23 228:24	53:17 105:18,20 113:5,6	17:16,17 18:1,4 183:9
easily (1) 269.12	132:20 236:4 241:13	229:24 245:1 246:4	117:15 150:18 152:22	exhibits [11] 14:6,10,11
east [11] 25:9 49:15 51:6	employees [4] 20.15	environmental (11)	171:1 172:14 177:4 184:7	14:12,24 16:4 17:9,16
118:18,20 144:2 194:22	125:24 132:20 200:4	32:1 36:8 37:4 68:16,18	267:17	existed [1] 228.23
196:1 211:20 226:21	employer [4] 123:20	69:4 80:2 85:3 120:7	events [7] 20:2 33:4	existing [2] 202:12
255.15 eastern (2) 49.7 233.16	127:9,10 232:9	187:12 227:24	113:11 116:17 246:8	211:15
easy [2] 215.16 259.14	employers [7] 125:6,8	98.4	258:2 265:1	exists [1] 264:2
economies [1] 190:13	employment (1) 237.4	environments [1] 35:22	eventual[1] 219.2	exits [1] 167:21
edge [3] 38:11 116:11	enabled [1] 200:25	equate [1] 245:2	everybody [7] 5:16 7:19	expand [1] 112:22
235:15	encompassing [1] 159:8	Equatorial [1] 29:7	11:13 127:24 160:9	expanded [3] 191:16
education [1] 25:2	encouraged [2] 135:3	equipment [39] 32:17	214:14 234:5	expect [5] 63:2 220:24
effect [6] 98:25 102:21	225:5	51:18 79:22 92:4 93:22	everywhere [2] 31:14	221:8 239:19 262:20
effective [5] 38:22 68:20	end [8] 6:10 53:14 54:1	97:9,14,18 109:11 113:9	evidence [27] 1:12 3:6	expectation [2] 250:6
102:21 108:20 180:21	269:13	113:13,20 117:10,21	4:5,7 5:12 7:7 10:13,23	261:8
effectively [3] 77:8	endorsement [1] 195:16	118:6,9 150:21,24 152:16	10:25 11:8 12:9,19,25	72:8 115:13 166:4.5.7
143:20 226:2	Energy [14] 11:23 18:20	201:13 220:12 249:18	77:25 140:13 169:16	229:23 239:7 244:24
efficiently (1) 145.21	18:22 23:15,20 24:16	251:1,7 255:8,12,15,18 258:18	188:8 202:1 207:19 217:6	expected [9] 9:11 40:4
effort [3] 210:1 226:16	51:13 57:14 58:5	equipped [4] 61:16	240:3 260:22 268:23	64:11 66:8 115:19 139:13 252:23 260:23 269:10
232:10	enforceable [1] 66:22	153:10 167:15 251:6	evolves[1] 77.15	expensive [2] 204:8
efforts [2] 5:6 252:16	enforces [1] 72:12	equivalency [2] 167:23	exactly 6 99:10 113:24	207:8
egress [5] 166:23 171:1	engaged [1] 120:18	168:18	163:17 192:7 213:1	experience [10] 27:24
FH_101 [2] 103.8 11	engagement [4] 107:11	49:19 167:24 233:10	254:20	93:1 166:14 174:13
eight [3] 25.17 49.22	engine (1) 166·19	escalating [1] 258:22	examination [2] 2:17	230:14 231:17
68:25	engineer 161 21:20 22:9	escalation [1] 246:8	example [36] 7:23 15:6	expert [2] 193:23 200:13
either [22] 4:3 7:19 15:2	28:4,4 95:15 111:18	escape [3] 168:25 169:7	39:10 60:16 65:2 74:24	expertise [4] 141:6,7
36:1,17 52:21 59:18	engineering [12] 21:17	223:5	75:7 80:3,18 81:25 82:1	evnerts (5) 100.3 102.24
119:8 133:9,19 173:17	22:12 25:3,4 26:15,19 28:2 29:4 95:14,17 96:22	244:15	103:12 112:6,12 113:20	120:1 121:4 156:15
174:6 175:12 200:9	135:16	essence [4] 167:12 168:1	139:8 141:12,14 145:2	expiry [1] 236:14
209:22 210:13 236:15 238:8 260:10	English [1] 168:11	197:25 252:11	154:4 156:11 157:9 167:14.18 172:12 174:3	explain [6] 3:22 139:1
elect [3] 128:10 175:7	enjoy [2] 24:11,13	essential [1] 43:19	174:4,22 225:24 226:2	170:8 183:13 209:22
183:6	ensure [24] 30:5 86:22	essentially [3] 152:13	232:2 242:4	explanation [11 17:12
elected [4] 126:22 128:21	101:1 106:8 108:4 109:21	establish 131 109-19	examples [5] 11:18,24 12:12 114:9 232:2	explicit [1] 98:10
132:22 130:15	116:2 118:8 134:15,19	118:12 182:14	excavations [2] 47:24	exploration [2] 55:4
electronic (3) 16.9	139:21 177:3 180:3,20	established [8] 120:8	51:16	63:11
236:11,20	242:17 244:18	120:11 189:25 191:5	excellent [1] 226:3	explore [2] 33:18 181:7
electronically [1]	ensures [1] 187:4	estimate 121 24.2 29.19	excerpts [1] 165:25	explored [2] 8:18 36:5
138:11	ensuring [5] 90:18 98:3	et [2] 209:2 266:15	excess [4] 27:7 35:16	105:13 222:23
elements [2] 96:20 111:8	102:18 142:10 249:8	ethics [1] 24:13	exchange [2] 177.20	express [1] 3:15
elevation 11 132-17	101:11 216:5	Eurocopter [11] 191:20	209:18	expression [6] 71:19
eligible (1) 236.16	entered [2] 3:7 142:9	191:24,25 192:17,21	excited [1] 218:14	137:6 208:22 209:20
eliminate [2] 96:3	enters [1] 217:19	195:7 194:21 195:5 197:5 224:8,9	Excuse [1] 225:3	210:7 245:9
and the second		,	1	1

Discoveries Unlimited Inc., Ph: (709)437-5028

extension [1] 191:7	false
extensions [1] 191:9	fami
extensive [2] 32:13	125:
122:21	fomi
extensively [1] 177:18	27:1
extent [4] 9:15 169:4	fami
externally 131 166:22	28:1
167:16,19	fami
extra [1] 150:19	far [5
extracted [1] 36:5	207:
extracts [1] 15:24	foct
extremely [1] 108:14	foot
ExxonMobil ^[2] 42:8	50:8
46:22 EvyonMobillaru	197:
200:19	Febi
	fede
-F-	258
FAA [1] 195·2	feed
face 15 1 212:11 214:16	feed
214:21 215:1,1	92:1
faced [1] 3:11	121:
facilitate [1] 137:13	24/:
facilitated [1] 226:2	foot
facilitates [2] 216:6	49:2
258:18	felt [4
23:21 25:18 28:4.10 29:9	207:
31:23 32:3,14 39:4,13	few [
40:2,9,19 43:20 50:11	12:2
57:22.25 70:10 79:22.24	202:
95:16 137:22 139:17	fidel
147:16 148:23,25 149:8	field
205:13 230:6 231:6	46:1
233:13 265:14	197:
facility [31] 22:11 26:6	field
28:14 42:22 45:6,11	fifte
51:25 54:6 55:16,25 58:2	file [1
73:16 83:21 86:17,21	fill [1
87:14,14 126:10,11	fille
231:1 240:5 243:6 244:10	filte
fact [26] 2:11 5:17 6:20	tinal
12:20 26:5 66:2,3,9,11	final
68:1 81:24 94:16 100:25 103:16 104:15 125:4 18	final
129:23 164:15 167:8	find
184:6 190:14 204:8 210:8	find
213:20 214:18	fine
factors [1] 190.7	201:
237:5	fined
Fagan [2] 6:15 8:12	100.
fail [2] 95:21 96:20	fire
failure [2] 198:24 268:14	116:
fair [3] 139:5 150:14	firef
1/4:/ foirly (5):12:04	firef
173:8.8 177:18	152:
fall [3] 105:19 165:23	10.2

[1] 199:5 i**liar** [6] 107:16 12 137:7 194:7 22 232:18 ilies [3] 2:21 20:7 **ily** [7] 20:3 27:8 28:6 1,15 29:10 30:9 ily's [1] 28:19 51 20:6 145:23 199:1 14 218:22 her [2] 41:24,25 [1] 152:13 ures [8] 44:21 49:23 115:23 165:20 196:2 3.10ruary [2] 2:15 7:11 ral [8] 42:10,11 19 124:1,5 195:1 23 265:6 [1] 106:6 back [12] 69:9 92:8 4 93:2 94:8 108:16 3,6 200:3 246:18,21 [1] 5:5 6 35:19 47:25 49:18 0 163:13 164:8 41 26:24 200:4 21 210:23 11] 1:8 10:13 12:20 1 13:23 23:16 73:4 17 138:25 144:4 15 ity [1] 224:1 [14] 23:18 40:3 42:4 5,20,24 47:2,18 51:8 16 167:1 169:5 16,23 **S** [2] 49:19 196:8 en [1] 145:2 ı] 107:9] 239:6 **d** [1] 83:7 r [1] 268:15 [5] 17:2 218:16 23 236:9 250:3 **lly** [2] 16:23 17:24 ncial [2] 28:14 33:17 ing [1] 55:4 **S**[1] 55:6 [3] 50:24 112:14 25 **1** [1] 65:1 hed [3] 128:14 182:4 219] 115:4,6,7,7,22,23 3 235:3.9 **ighters** [1] 158:4 ighting [2] 96:24 24 [61] 3:20 6:17,23 10:22 12:22 14:10 20:5 21:22,25 26:7,10,13 42:4

42:17,19 46:18 49:14 130:10,15,20,22 131:10 131:10,18 152:24 166:14 166:16 169:14 174:23 176:6 180:6 183:25 186:22,23 189:14 190:19 195:18 196:20 198:13 213:3 215:12 219:23 220:5 232:22 240:5 243:20 249:4 251:19 257:13,19 258:4 260:1,8 261:5,6,13,19 262:2,9 262:13 266:11 **fishing** [1] 24:11 fit [15] 84:14 88:23 89:4 89:6 96:17 97:9 158:19 184:12,13 214:3 215:12 215:17 229:7 231:15 244:2 fit-for-purpose [1] 96:12 fitness [7] 65:13 86:8,23 87:5 88:20 89:1 229:20 fits [3] 78:18 82:1 208:15 fitted [2] 215:4 216:2 fitting [4] 208:3 214:8 214:10 241:16 five [9] 28:8 51:14 112:20 112:20,20 126:9,11 176:1 182:16 five-day [1] 232:23 fixed [5] 55:12 56:15 265:4,20 266:14 flags [1] 214:5 flare [1] 180:12 **fleet** [6] 191:17 193:2 196:18 197:2,12 266:12 flexibility [9] 156:7,10 196:10 204:4 207:2,4,6 207:8 249:16 flight [34] 41:11 80:10 121:24 122:14,20 166:16 173:9,16 174:2,2,5 178:7 179:7,13 183:17 202:6 215:12 241:9,10,12 242:10,19,23 243:25 244:6,20 245:10,16,17 257:18,19,20,22 258:4 flights [5] 29:20 172:1 175:10,11 199:16 **floating** [2] 21:23 46:24 floor [4] 42:24 43:2 50:5 50:8 flow [3] 48:11,25 54:15 flown [6] 27:5,7 29:14 179:25 199:17,23 fly [10] 29:18 150:18 173:10 175:7 183:16 184:5 185:6 206:16 212:24 238:25 flying [9] 169:3 173:3 174:11,18 175:11 176:20 182:1 184:3 250:17 foam [8] 115:4,4,20,20 116:5,8,8 117:13 focus [10] 13:20 90:3 93:5 118:1 165:4 169:12 frequencies [2] 217:10

Multi-PageTM

extension - general **Offshore Helicopter Safety Inquiry**

169:14 214:16 245:10 217:13 252:13 focused [2] 83:24 224:14 fog [1] 147:8 follow [10] 65:18 83:19 99:3,18 161:14 162:9 164:23 215:21 216:20 238:21 **follow-up** [1] 33:6 followed [5] 4:14 24:6 161:11 189:17 242:9 **following** [10] 13:5 14:2 68:24 79:17 80:7,10 123:20 152:11 165:11 219:12 follows [2] 18:15 133:7 football [1] 49:19 footsteps [1] 24:6 force [9] 33:2,8 118:23 118:25 128:2,6 162:3 268:18.25 forces [3] 119:4,20,25 Ford [1] 199:21 forecast [2] 148:10 175:6 forecasting [4] 144:7 146:10,22 147:2 forecasts [1] 148:8 **foregoing** [1] 271:3 forever [1] 213:22 form [5] 3:5 33:20 109:11 238:22 245:15 formal [5] 25:1 90:8 186:16 187:9 244:22 formally [1] 46:17 formed [1] 34:17 forms [6] 101:8,9,13,20 102:9 247:4 forth [7] 29:10 45:15,23 126:13 151:18 185:7 227:7 forward [9] 5:6 6:10 11:24 13:24 21:8 41:3 49:24 106:12 195:15 found [3] 65:22 112:1 216:16 four [22] 12:21 24:8 28:22 33:12 51:15,18 55:9 84:12 123:24 124:6 124:21 182:9,11,15,20 182:23 183:4 209:7 233:14 267:2,11,12 fourth [3] 11:25 181:24 190:1 **FPSO** [23] 21:22 22:23 23:4,11 26:9 39:4 47:1,1 47:16 48:8.10 49:15.16 50:8,18 51:13 53:20 87:25 95:6,10,10 151:16 154:10 FPSOs [4] 21:22 22:18 23:3 90:12 frankly [1] 5:2 **freely** [2] 14:13 48:25 freight [2] 169:1 170:5

frequency [3] 135:20 228:10 232:12 frequent [5] 111:11 121:22 232:14 234:1,2 **frequently** [2] 33:20 77:17 **Friday** [1] 266:22 friend [1] 6:6 fro [1] 57:8 front [2] 95:9 97:22 **fuel** [18] 95:24 96:3,13 98:16,17 104:1,2 105:12 163:11 164:8 169:5 172:9 172:13,16 176:4,18 195:7 223:20 **fuelling** [1] 95:2 fulfil [1] 237:6 **fulfilling** [1] 235:5 fulfils [1] 234:25 **full** [13] 14:21 31:8 83:20 102:25 103:11 104:5 106:9 189:14,15 196:18 197:2 233:3 265:24 **full-time** [1] 258:15 **fully** [8] 45:18 49:16 156:13 173:18 220:12 242:12 244:8 251:6 **function** [4] 129:4 143:15 180:17,19 **functioned** [1] 217:3 **functions** [3] 78:22 90:10 157:17 **funding** [1] 96:23 **Furthermore** [1] 244:25 **future** [6] 11:21 13:22 30:8 91:25 93:21 122:24 -Ggain [1] 250:18 gained [2] 69:9 241:19

Gander [4] 172:18 175:1 262:4 266:25 gaps [1] 240:17 Gary [3] 18:21,25 24:21 gas [16] 20:18 24:23 25:18 27:24 28:10,14 43:20 48:10,13,18,21 51:20,21 55:6,7 165:12 gateway [1] 236:10 gather [3] 110:22 163:23 186:17 gathered [2] 163:24,25 **GBS** [6] 39:4,11 43:12 44:15 87:25 95:6 gear [5] 163:12 164:8 250:2 259:12,14 gearbox [1] 198:22 general [13] 18:19 23:1 23:19 102:12 106:12 121:25 126:16 156:8 159:6 165:13 166:17 213:4 261:14

Discoveries Unlimited Inc., Ph: (709)437-5028

202:7

Multi-PageTM

generally - hole Offshore Helicopter Safety Inquiry

				copier Sarety Inqui
generally [13] 14:16 46:3	gravity [3] 43:1 44:4	handle [1] 160:15	helicopter [178] 5:8 11:2	help [5] 48:22 139:2
48.9 54.8 50.18,25 58.14	07:1 groot (2) 29:21 112:5	handlers [1] 158:5	16:17 20:17 22:16.21	holped (2) 27.2 215.7 11
159:25 198:11	246:1	handling [3] 43:20 150:4	23:6,23 29:20,23 30:2,6	helpful (1) 221.2215.7,11
generate [1] 43:19	greater [2] 196:4 197:22	hondover (1) 129.5	32:14,17 33:2,7 49:10	hence [1] 56.6
generation [1] 50:9	greatly [1] 20:2	handovers [1] 138.3	52:21 56:19 61:6,6,15	Honry (4) 47.6 13 57.17
generic [6] 100:12,13,16	green [4] 44:1.3 110:24	Hangon [14] 11.10	79:14 80:5.9 81:14.17	145:11
101:17 255:22,23	111:21	209:10,12,21 211:13,18	81:21,24 82:13,24 83:24	Hercules [3] 260:6
gentlemen [6] 1:3 18:14	Greenwood [1] 267:1	212:6,19 214:13 215:10	96:2 121:23,25 122:6,13	265:21 267:2
30:15,21 30:17 200:9	grey [1] 261:4	215:14 241:15 243:10,12	122:20 134:2,7,21 135:2	hereby [1] 271:2
263:7 266:9	ground [8] 48:3 183:24	happening [1] 261:20	151:5 157:21 158:2,3,5	Hibernia [51] 11:22
geological III 68:17	185:4,19,24 249:10	Harbour [1] 149:17	158:7,16,20 160:4,10,13	18:17 26:2,4 27:22 28:16
geophysical [3] 65:21	arounded (11 268.16	hard [2] 96:1 203:25	162:2 164:10,13,19 165:5	34:9.10.13 39:11 40:2
68:17 81:11	groun [42] 102.24 103.3	Harland [1] 22:6	166:9,11 167:22 168:24	42:4,7,9 43:11 44:8,21
geotechnical [1] 68:18	106:6,15 107:2 120:5	harsher [1] 36:6	169:4,20,24 171:1,25	47:18,23 49:2,8 56:14
given [12] 2:18 3:21 6:20	121:4 126:18,18,19,21	harshest [1] 35:22	172:8,22 173:19 176:4	60:14 86:14 87:1 145:23 145:24 149:22 150:20 23
40:9 120:20 135:1 169:6	126:25 127:3 128:6,7	172:6	178:7179:3,13,23180:10	151:14 162:1 180:14
200:24	130:20.20.21.22 131:5	Harvey [2] 149:11.16	182:24 183:12,24 184:5	184:4 186:3,21 197:16
giving [1] 182:8	131:10,18,24,25 132:6,6	hazard [14] 38:9 79:20	184:6,10 185:20 186:2	204:20 213:12,25 230:11
glad [1] 30:23	135:15,16 137:18 152:7	96:1,6 100:7 102:15,16	186:11,24 187:19 189:10	267:24
Glomar [1] 145:5	200:19 251:9,13,15,21	102:21 103:10 105:19	201:13 202:25 203:11	Hibernia's [2] 60:13
glory [3] 51:15,16 54:13	grouning [1] 40.16	234.23	205:19,22 206:5,6 209:13	187:19
glossary [1] 31:1	groups [3] 23:12 129:18	hazardous (4) 105:17	216:18,19,21 218:9,16	hierarchial [1] 101:15
goal [2] 69:13 98:2	130:4	106:5 113:5,11	232:4 234:14.16.18.23	high [14] 30:10 35:18
goal-based [1] 68:22	grow [1] 30:11	hazards [25] 92:17 95:18	234:25 235:7,12,16,24	40:12 43:13 44:5 98:6
goal-oriented [1] 68:22	grows [1] 115:17	95:19,19 99:25 102:19	242:15 243:6,14,16 244:9	222:21 223:22 246:16
goals [1] 78:23	GSF [1] 57:16	105:1,0 105:9,11,14,10	244:11,12,15 247:25	258:20
goes [10] 49:4 115:15	guard [11] 212:15 263:12	109:2,4,7,20 132:25	251:3 252:23 254:11,16	higher [4] 72:2,5 207:21
166:17 176:9 179:15	263:18 264:20,23 265:5	134:20 136:3 159:7	257:13 258:2,10,11	231:21
100.5,4,5 210.22 257.25	203.0,13,13,17 200.14	ho'd m 1.11	259:16.19.260:25.261:6	nignest [3] 43:12 99:5
GOOGLES [3] 169.6 177.15	ouards (1) 247.21		261.14.24.268.18.25	209.11
goggles [3] 169:6 177:15 177:16	guards [1] 247:21 guess [14] 52:21 69:18	head [2] 49:5 172:11	261:14,24 268:18,25 helicopters [59] 4:25 5:1	209:11 highlight m 212:8
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9	head [2] 49:5 172:11 heading [3] 54:4 165:7	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6	209:11 highlight [1] 212:8 highlighted [1] 76:3
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1]	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 23:211 25:10 5:01 (0:24)	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4 17	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4 16	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3	head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3 13 23 125:15 129:2	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 birgs [1] 72:1
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Coodrigh (*) 47:6 14	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 highlight [1] 213:2
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [15] 61:4 67:7	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 100:2 240:11 246:12	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 bigtorical [1] 125:8
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20.22	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2.12.13.14.17.18 69:6	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 bit [1] 48:5
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17 20 110:21 23	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5:8 41:13 125:3	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 UMDC 2 m 186 25
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 12:25 271:4	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMIDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMIDC's [2] 186:25 200:17
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 CPS = 240.20	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Guideline 20 16 122	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:45 116:15 16	209:11 highlight [1] 212:8 highlightd [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [21, 25:2]	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17 19	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9	209:11 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H-	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governace [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H-	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12 15 00:10 102:16	209:11 highlight [1] 212:8 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Holifay IG	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12,15 90:19 193:16 heliopert in 124:22	209:11 highlight [1] 212:8 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8 27:9 35:15 38:10 57:17	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Halifax [6] 122:23 263:25 264:2 266:2 267:1	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19 held [4] 22:6 128:24	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12,15 90:19 193:16 heliport [2] 134:23 240:13	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8 27:9 35:15 38:10 57:17 92:22 145:5 190:12 193:14 266:7	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Halifax [6] 122:23 263:25 264:2 266:2 267:1 268:9	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19 held [4] 22:6 128:24 156:22 163:20	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 261:12 52:6 61:11 25:6 49:12 52:6 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 157:10,11 160:15 166:11 167:17 167:17 176:20 179:2,6 180:11,11,20,21 180:22 181:4 183:20,23 184:2 185:25 188:16 189:7 196:21 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 240:2 246:12 266:13 helideck [35] helideck [35] 43:22 49:24 52:16 53:1 55:25 81:10,13,25 </td <td>209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22 holder [1] 46:19</td>	209:11 highlight [1] 212:8 highlighted [1] 76:3 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22 holder [1] 46:19
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8 27:9 35:15 38:10 57:17 92:22 145:5 190:12 193:14 266:7 grandchildren [1] 93:8	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Halifax [6] 122:23 263:25 264:2 266:2 267:1 268:9 hall [7] 135:9 136:25	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19 held [4] 22:6 128:24 156:22 163:20 Heli-Admin [3] 243:4,9 244:1	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 85:15 89:19 95:1,4,7,11,16,23,25 96:8,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12,15 90:19 193:16 heliport [2] 134:23 240:13 Helly [15] 11:10 209:10 209:12,21 211:13,18	209:11 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22 holder [1] 46:19 holding [2] 42:9 172:13
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governace [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8 27:9 35:15 38:10 57:17 92:22 145:5 190:12 193:14 266:7 grandchildren [1] 93:8 granted [5] 77:4 87:5	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidelines [1] 71:25 guidelines [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Halifax [6] 122:23 263:25 264:2 266:2 267:1 268:9 hall [7] 135:9 136:25 137:3,6,9,10,15	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19 held [4] 22:6 128:24 156:22 163:20 Heli-Admin [3] 243:4,9 244:1 heli-deck [1] 244:17	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 98:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12,15 90:19 193:16 heliport [2] 134:23 240:13 Helly [15] 11:10 209:10 209:12,21 211:13,18 212:6,19 214:13 215:10 215:14 241:15 242:10 11	209:11 highlight [1] 212:8 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22 holder [1] 46:19 holding [2] 42:9 172:13 holds [1] 34:2
goggles [3] 169:6 177:15 177:16 golfing [1] 24:14 gone [3] 221:12 227:21 238:16 good [21] 1:3,3,21,23 33:11 35:10 50:1 60:24 64:14 72:21 75:7 115:12 116:9 120:11 153:16 158:19 175:4 190:5 210:24 232:16 259:11 Goodrich [4] 47:6,14 57:17 145:12 goods [2] 234:20,22 Gorilla [1] 55:22 governance [1] 139:24 governed [1] 34:5 Government [3] 42:11 42:11 256:22 governs [1] 84:6 GPS [1] 248:20 graduated [2] 25:2 26:22 graduating [1] 27:25 Grand [11] 20:1 25:8 27:9 35:15 38:10 57:17 92:22 145:5 190:12 193:14 266:7 grandchildren [1] 93:8 granted [5] 77:4 87:5 167:24 237:10 239:13	guards [1] 247:21 guess [14] 52:21 69:18 144:17 149:19 151:9 152:2 178:17 185:22 190:10 197:11 198:15 209:23 260:20 267:12 guidance [5] 70:4,17 71:19 151:4 226:13 guidances [1] 71:25 guideline [10] 67:16 69:3 69:3,4,6 70:22 72:12 121:6,7 252:22 guidelines [35] 61:4 67:7 68:2,12,13,14,17,18 69:6 69:12,14 70:8,9 71:19 72:9,11 78:16,16,25 80:8 81:17,20 119:21,23 120:21 121:2,14 152:3 166:1,2 195:16 225:12 250:25 255:17,19 guinea [2] 29:7 123:12 Gulf [2] 29:16 42:12 -H- half [2] 24:8 28:22 Halifax [6] 122:23 263:25 264:2 266:2 267:1 268:9 hall [7] 135:9 136:25 137:3,6,9,10,15 hand [3] 44:23 138:19 150:11	he u [1] 1.11 head [2] 49:5 172:11 heading [3] 54:4 165:7 248:18 headquarters [1] 178:10 heads [1] 6:21 health [23] 60:3 65:19 66:2 77:9 90:23 120:6 120:10,17 123:17,22,24 124:3,13,23 125:15 129:2 129:5,13 178:1 197:8 227:24 229:11,14 hear [7] 11:7 23:14 60:18 91:19 122:24 151:3 170:3 heard [12] 5:21 10:22 11:3,5,8 41:13 125:3 129:11 158:10 245:7 259:18 271:4 hearing [4] 9:19,20 13:25 271:4 heavier [2] 193:12,15 heavy [3] 180:17,19 194:1 Hebron [1] 40:3 height [3] 105:14,18,19 held [4] 22:6 128:24 156:22 163:20 Heli-Admin [3] 243:4,9 244:1 heli-deck [1] 244:17 heli-refuelling [1] 96:10	261:14,24 268:18,25 helicopters [59] 4:25 5:1 5:5 24:5 27:6 29:14 32:6 49:12 52:6 56:1,23 58:25 61:11 62:4 82:18 95:2 138:24 146:17 147:14 157:10,11 160:15 166:11 167:17 176:20 177:6 179:2,6 180:11,11,20,21 180:22 181:4 183:4,16 183:20,23 184:2 185:1 185:25 188:16 189:7 194:1,9 196:21 197:6 199:2 240:11 246:12 247:20 248:3 249:9,20 260:2,10 265:4 266:12 266:13 helideck [35] 43:22 49:21 49:24 52:16 53:1 55:25 81:10,13,25 88:16,19 104:2 115:4,5 116:15,16 116:19 117:14 160:9 235:2,9,11,17 helidecks [6] 56:4 57:7 70:12,15 90:19 193:16 heliport [2] 134:23 240:13 Helly [15] 11:10 209:10 209:12,21 211:13,18 212:6,19 214:13 215:10 215:14 241:15 243:10,11 243:12	209:11 highlight [1] 212:8 highlight [1] 212:8 highlights [3] 25:21 26:12 79:12 highly [3] 15:6 59:8 141:17 hindsight [1] 213:2 hints [1] 72:1 hire [1] 141:8 hired [1] 142:21 historical [1] 135:8 hit [1] 48:5 HMDC [11] 11:22 29:24 34:11,16 59:20 145:23 186:23 189:18 190:8 195:20 196:23 HMDC's [2] 186:25 200:17 hoc [2] 121:20 150:25 hoist [4] 255:10,23,24 256:4 hoists [1] 251:7 hold [7] 63:14,16 82:20 153:13 166:5 172:13 234:22 holder [1] 46:19 holding [2] 42:9 172:13 holds [1] 34:2 hole [8] 114:22 115:14

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

	holes - introduction
Offshore Helic	copter Safety Inquiry
9 25 242 15 242 2	ingtallationary 72.10

				· · · · · · · · · · · · · · · · · · ·
116:24	66:12 90:13 100:1 101:4	included [17] 7:1 25:7	241:8,25 242:15 243:3	installations [15] 73:10
holes [9] 51:15,16 54:13	105:15 106:9,10,23 107:4	25:12,22 45:13 167:11	243:18,23	84:15 86:12 87:11 139:19
114:4,6,13,18,24 117:2	109:5 113:4 136:4 154:25	179:13 187:2,7 189:12	Indonesia [1] 28:8	147:20,23 153:7,13 160:1
home [2] 30:9 173:8	203:5 237:13 246:12	194:25 208:25 212:15	induction [1] 134:24	101:3 10/:1 1/4:10 1//:/ 18/:/
homecoming [1] 28:19		includes (14) 20.14 23.23	industry [32] 4:23 8:5	installed 11 100.21
hood [1] 212:10	Identify [13] 33:15	38:17.19.20 42:13 43:17	11:12,15 12:12 13:13	instancu [1] 109.21
hook [1] 26:10	100:20 101:1 102:19	46:21 69:2 79:14 174:6	16:17 20:16,23 21:4,9	102.9 170.21 259.18
hooked [1] 54:14	161:13 220:7.19 238:23	203:2 241:6,10	22:18 24:10,11,24 26:21	instances (1) 167.14
hope [2] 27:13 77:23	246:2	including [12] 20:9	35:13 71:12 92:16 110:14	
honed (11, 269.9	identifying [4] 7:25	77:10 142:12 148:23	110:14,17 118:17 155:17	223.13
horse (1) 210.22	69:22 105:9 220:13	152:1,12 165:11 176:5	155:18 219:6 225:14	Institute's (1) 211.19
hostile (1) 228.24	II [2] 124:1,5	222:5 220:22 259:25	226:5	institutions [3] 225.15
hour 141 35.17 138.3	ill [1] 229:11	inclusion (1) 223.10	inflated [1] 212:16	226:19.23
184:1 240:14 243:5 250:4	illustrates [1] 4:18	incorporated (2) 24:12	influence [2] 95:8 118:4	instruction [2] 100:22
250:8,16 251:5 253:10	immediate [1] 162:11	59·20	inform [1] 68:20	135:1
253:13 254:3,12 255:1	immediately [1] 33:3	increase [1] 164.9	information [24] 21:6	instructions [4] 100:17
hourly [1] 15:8	immersion [4] 81:10	increased (4) 122.2.3.6	101:8 147:5,6,13 148:7	101:18 108:3 134:24
hours [9] 174:11 185:18	203:14 204:6 207:20	252:13	163:23,25 164:1 178:8	integral [2] 77:13 212:10
250:18 251:4,25 252:1	impact [8] 39:12,14	indeed [17] 20:8.11 27:12	234:24 236:21 242:2	integrate [1] 139:2
266:24,24 269:3	44:11 66:4,10 146:23	55:22 57:3 91:18 107:14	244:4 247:24 248:7,13	integrated [1] 212:16
housing [1] 268:15	171:2 222:22	108:13 117:13 120:3	248:14,16,21 252:17	integration [2] 94:22
HSC [1] 120:1	impacted [1] 39:6	121:4,19 124:19 126:19	informed [5] 212:3	97:18
HUEBA [23] 4:16 7:23	impacts [1] 129:12	135:10 157:19 212:21	242:7 258:13 259:10	integrity [2] 134:15
121:24 203:4,7,9 212:14	impending [1] 198:23	independent [5] 84:10	262:20	161:9
212:14 218:10,10,19,20	implement [4] 203:6	88:19 140:3 186:16	informing [1] 262:19	intended [5] 89:4 139:3
220:6.18 241:11 242:18	218:24 220:18 236:11	independently 120.0	ingredient [1] 97:3	162:1 222:21 223:6
244:1,19	implementation [7]	145.14 196.23	ingress [3] 122:14,20	intent [10] 3:15 88:21
HUET [1] 169:15	63:10 109:18 209:7	indicate (4) 11.1 15.1	123:4	91:22 98:7,12 99:14,15
hulls [1] 153:9		15:25 216:10	inherent [1] 200:1	101:16 111:2 204:15
human [1] 268:5	implemented [1] 244:22	indicated [19] 1:9 37:18	inhibited [1] 174:24	Intention [1] 224:21
HUMS [4] 177.21 21	Importance [3] 32:11	45:1 49:11 56:16 58:2	initial [4] 42:19 55:4	Interact [1] 62:23
178:1 197:4	254:0 240:1	69:16 140:9 141:23	213:24 259:7	interchangeable [1]
Husky [26] 11:23 18:20	1111portant [19] 3:23 4:17	147:12 150:1,6,10 152:1	initialled [1] 161:12	33:24
23:15,20 24:16 35:3	32:10 37:22 52:7 63:20	153:2 164:7 207:19	initiated [2] 189:9 191:2	Interconnected [1]
46:22 51:6,9 57:14 58:5	64:16 90:2,21 102:17	indicating (1) 152.7	initiative [2] 23:11	interest (a) 14.19 20.25
59:19 118:21 135:10	117:9 118:13,15 134:13	indicating [1] 152.7	202:10	29.23 30.1 33.25 46.19
130:0 137:1 145:5,8,15	216:7	199.9 253.2	initiatives [1] 23:8	208:22 209:20 245:9
191:2 195:20 196:23	improve [3] 21:9 92:11	indications (2) 198-15	injected [2] 48:18,24	interesting [1] 144:17
238:8	246:22	199:16	injection [3] 51:20 55:6	interface [5] 32:22 61:9
Husky's [1] 200:12	improvement [9] 13:22	indicative (1) 198:23	55:6	70:19 256:25 257:8
Hydro [2] 12:2 25:22	108.13 22 202.9 220.24	indicators [1] 200.6	input [4] 93:2 94:8 106:7	interfaces [1] 23:12
	improvements [3] 92.6	individual [57] 12:12	119:21 inquiniog (n) 5:17	interfere [2] 242:18
-I-	92:9 110:5	14:1.3 15:13 16:11 19:7		244:19
iem 208.25	improving [2] 77:14	34:8 39:3 44:7,22 60:2	111quiry [23] 2:8,13,14 3:3 6 16 18 22 23 4:6 18	interfering [1] 242:22
ice [26] 36.13 37.5 7 28.0	199:3	60:19,22 79:3 82:8 100:6	4:22 5:19 6:14 11:25	interflight [1] 245:2
38:12.12.14.16.24 39:7	inadequate [1] 155:7	102:2 108:7,9 127:6	12:15 21:3,6 27:13 30:4	intermittent [1] 237:4
39:11,16,19 44:11 47:23	inappropriate [1]	162.10 23 163.4 16	61:9 93:19 181:11	internally [1] 167:20
49:23 55:14 144:6,18	245:16	179:18 181:8,9 186:18	inspection [8] 70:14	international [3] 13:13
146:9,21 147:11,22 150:5	inbound [1] 249:22	190:18 213:15 216:10	85:20 89:8 90:6,9 98:20	256:12 263:5
152:3 153:9	Inc [2] 271:12,14	217:17 218:12 222:18	100:18 241:7	interpret [1] 12:18
iceberg [3] 37:7,19 55:14	incident [20] 13:5 83:16	223:17 229:12 231:9	Inspections [5] 13:18	interpreted [2] 170:10
icebergs [10] 36:14 38:13	85:4 114:4,21 115:6	252.9 257.6 258.18 259.5	85:14,15 84:21 80:17	201:1
58:14,19,25 59:12,20 44·12 48·2 53·17	117:6,16 122:3,9,19	242:20 244:16 245:14	inspects [1] 80:10	interview [3] 163:16,18
idea (11, 230.8	198:6 200:21 224:20	247:2,9,15 265:22	install [1] 128:8	163:22
ideal (1) 230.0	262:2.18	individual's [3] 163:17	Installation [33] 22:16 22:10 26:8 28:24 53:1	intimately [1] 232:18
identical (1) 02:10	incidents 121 68.15 92.5	163:22 179:16	65:14 76:15 77:11 79:23	introduce [3] 31:22
identification (0) 70-21	inclement [1] 174.12	individually [4] 13:10	81:8 82:11,17,23 86:7	211:22 220:11
100.7 102.16 17 22	include (11) 40.10 65.12	146:7,9 149:20	88:22 89:2,13,18 90:19	introduced [5] 22:1
103:10 179:19 240:16	68:13 70:7 79:16 80:7	individuals [18] 21:12	90:24,25 115:3 137:2	08:21 95:22 211:24 213:3
248:17	179:5,7 196:21 220:22	151:1 14/:25 152:17	141:20 147:6 151:6	222.25
identified [17] 33:7	227:11	180:1 235:15 240.13 20	199:12 233:24 237.12	introduction (21.20
				min ouucion [3] 51:20

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

invention - lunch Offshore Helicopter Safety Inquiry

			Olishore nend	copter Safety Inquiry
33:14 233:5	106:8,14,25 108:14,15	kinds [1] 135:18	leave [6] 185:24 198:4	likes [1] 199:20
invention (1) 110.16	111:4,5 113:2 128:14	Kingdom (11, 23.9	236:20 245:4 259:1	likewise [2] 150.20
inventoriogue 50.17	154:14,24,25 156:19	Imotary 25.16	268:24	180.15
inventories [1] 58:17	159:4 161:1 229:8,15	KIIOUS [1] 35:16	leaves (1) 173.19	limitation (4) 102.12
invested [1] 190:21	iobs 131, 157-13, 159-6, 10	knowing [2] 5:14 26:25	lod (a) 165-16 196-22	
investigated [1] 163:14	John's 1201 24.9 12 25.15	knowledge [8] 21:5	105.0 200.23 24 210.4	limitations [3] 2:8 175:1
investigation [4] 68:15	301111 S [20] 24.6,12 23.13 28.15 18 20.11 147.4	106:18 133:2 146:12	195.9 209.23,24 219.4	193:10
129:6 163:8,15	149.16 172.16 173.19	160:7 171:4 234:10	left [8] 12:3 18:16 20:12	limited [4] 3:16 6:24
investigations 121 91.2	182.8 249.23 260.4 9	239:22	22:13 52:17 55:10 76:22	55:13,15
129:7	262:4 264:20.24 268:8	known [4] 133:20 163:13		limits [1] 172:8
invitation (1) 143.4	271:5,9	172:19,19	left-hand [1] 75:20	line [11] 4:14 78:22 91:20
	JOHS [30] 92.15 120.8	knows [2] 4:15 132:6	leg [1] 49:6	91:21 117:20 183:10,11
	121.3 7 15 123.3 125.14	Knutson (11, 58-11	legal [3] 221:7 269:5,22	186:14 202:2,3 266:5
invited [1] 137:14	126:10.24 127:10 128:11		legislation [2] 95:15	lined [2] 114:19.23
involve [3] 82:18 163:7	129:25 131:16 132:3.12	T	123:20	lines (2) 48:11 54:15
183:3	133:12 136:13,16,21,22	-L'-	legislative 121 60.5	linkod (2) 50.47
involved [22] 4:15 9:13	162:15,17,22,24,24 163:4	Labour [3] 123:25 124:4	142.12	IIIIKeu [2] 50:4,7
20:21 21:20 22:2 23:7	163:18,20,21 213:14	124:14	legitimate (1) 10.4	list [9] 16:3,6,7 18:1
42:12 51:6 86:13 90:18	join [1] 31:21	Labrador [9] 25:9 30:11	logg vie 55.11	139:4 166:8 255:9,15,18
92:23 94:17 100:4,5	ioined (4) 21.16 22.4 16	35:14 39:25 59:8 61:3	legs [1] 55:11	listen [1] 137:15
103:7 105:25 117:21	23:15	187:17 228:23 271:6	length [2] 42:18 220:17	living [2] 24:12 43:21
133:24 135:16 156:14	ioining (1) 130.17	ladies n 1:3	less [10] 115:18 130:22	Llovd's [3] 86:14 87:2
212:2 233:25	joining[1] 150.17	land (6) 38.4 52.22.24	201:3 230:16,22,25	90:15
involvement [1] 102:23	Joint [24] 13:24 16:12	55.13 172.15 177.7	232:19 233:21 234:1	Llovds (1) 90.13
involves [1] 133:11	51.0 50.22 120.0 16	landing (11) 56.1 70.16	245:9	
Island [2] 28:7 37:3	122.10 125.14 127.12 12	16101119[11] 50.1 /0.10	lesson [1] 13:6	
Islands (1) 27.16	125.19 125.14 127.12,15	90.1 151.5 157.21 150.2	lessons [3] 220:20 221:17	loading [2] 44:13 235:20
isianus [1] 57.10	226.15 258.13 259.3	235.8	262:18	loads [2] 171:25 193:15
Isolated [2] 167:13 229:8	263.21 264.7	Londing (4) 1(0.11	letter 121 69:21 152:5	local [3] 106:17,17 149:9
issue [21] 1:11 8:5 11:14	iointly (1) 105.11			locate [2] 216:19.21
15:25,25 86:20 90:15	JOINTY [1] 195.11	large [10] 16:7 29:1 47:24		located (5) 38.11 /1.25
	JKCC [12] 258:12 259:10	53:13 93:25,25 106:8	IEVEL [36] 28:13 72:2,5	42.14 47.1 95.25
12/:/ 102:13 103:13	259:19,24 205:25 204:2	128:0 145:15 154:11	/8:21 92:14,24 97:14	location (10) 22:25 05:12
1/1:1 1/4:19 198:10	266:20	larger [1] 15:18	96.7,9,21 99.11,12	10 Cation [10] 22:23 93:13
207.23 208.21 210.9		largest [3] 28:9 46:19	118.8 1/ 121.8 8 135.1	216.10 229.8 248.4 263.7
214.3 237.20	JRCC'S[1] 266:2	49:16	136.14 18 21 178.13	210.10 229.8 248.4 203.7
ISSUEU [18] 63:15,17 69:7	Judge [1] 8:23	last [7] 23:18 26:22 69:23	222.23 25 223.1 22	locations [2] 28:5 247:23
/2:0,10 /3:/ /4:14 80:0	Judy [1] 271:13	133:25 165:23 213:22	226:14 227:5 246:17	locator [7] 212:13 216:1
154.20 164.3 16 233.10	jump [3] 31:9 45:15.23	253:12	257:5 258:20 265:12	216:2,18 217:7 241:10
241.9 244.1	jumning [2] 123.6.9	lastly [4] 28:23 33:8	266:19	251:8
icsus (201, 7,14, 25, 9,2)	junctures (1) 117.7	44:13 71:11	levels [5] 90:24 104:22	logistical [1] 57:9
8.18 20 37.7 06.16 07.8		late 151 28:14 133:25	105:24 118:17.23	logistics [1] 23:22
121.12 21 130.15 131.15	June [3] 23:15 55:15	175:13 188:22 202:10	liabilities (1) 34.2	longer [3] 5:19 73:17
132.14 136.16 23 25	152:3	latest (3) 2.13 179.19	liconco ra 100.7	222:6
137:17 138:15 163:18	Jurisdiction [8] 15:20	203:1		look [30] 8:2.8.9 13:21
164:2 187:25 200:20	61:2,7 65:12 124:2,4,5	latitude nu 170.17	IICENSE [5] 63:15,22 64:3	14:19 18:8 35:5 52:19
214:10,22 215:20 234:4	233:15	lounabed in 152.14	64:4 66:23	70:21 79:2 82:16 92:6
issuing [3] 63:18 88:25	jurisdictions [1] 27:6	259:12	licensed [1] 166:13	92:13 97:7,8 98:1 103:7
209:20	justifiably [1] 199:7	230.12	licenses [2] 64:6 67:20	105:11 108:4,12 109:25
item [3] 65:25 164:14	justified [1] 233:4	Laurentian [2] 41:4	lies [1] 266:6	110:4 113:24 127:5
201:22	iustify (1) 176:18	50:14	life [12] 5:18 20:8 137:9	140:19 147:3 166:10
items (6) 2.4 79.17	J	lead [3] 104:12 107:19	141:16 166:22 167:16,19	197:12 218:2 266:9
121.24 132.11 154.2		268:16	167:20 197:23 202:9	looked [4] 5:23 102:1,3
164:23	<u></u>	leader [1] 107:6	203:3 251:8	113:25
itself [14] 3.18 11.12 16	keep [7] 5:3 48:22,24	leaders [1] 107:13	lifejacket [3] 209:1,1	looking [11] 2:12 79:5
55.11 91.7 105.5 6 106.3	73:5 176:18 265:24,24	leadership [2] 94:10	212:16	82:15 83:6 98:21 113:7
155:4 216:18 20 217:23	kept [3] 8:13 169:5 176:4	118:7	lift 161 51:20 55:7 155:5	117:10 155:12,25 175:17
235:18 242:24	kev [5] 92:3 165:20	leading [3] 167:5 193:6	180:17,19 194:1	252:4
	195:25 208:14 212:8	194:24	lifting [2] 255.25 256.3	loss [1] 27:10
	kilometres (61 35:17	leads (11 107:14	light (4) 175.12 104.16	low [4] 30:6 35:21 108:23
- J -	38:2 42:15 196:9 266:5	loarn [2] 27.13 30.5	100.011	161:6
jack-up [3] 55:10,19,19	266:5	221.11	lightly 174.00	lower [2] 220:10 231:21
January [4] 1:1 129:16	kin [1] 179:20	learned (a) 6:6 220:20	ngnuy [1] 1/4:20	luggage [4] 171:19.21
271:4,10	kind [13] 38.3 04.20	221.17 224.10 10 262.10	ugnts [6] 198:16,19,21	171:23 244:25
Jeanne [2] 38:10 46:15	102.4 113.14 120.5	221.17 224.17,19 202.10	198:25 199:5,8	LUK [2] 28.9 13
iet [2] 104:1 105:12	133:16 135:21 141:11 20	icarining[1] 224:2	likelihood [8] 109:6,10	lunch m 110.11 152.17
IOA [1] 24.6	155:13 156:5 163:4	least [7] 2:7 166:20 188:8	110:3,7 111:7,11 113:4	268·22
ich reg 105 7 10 22	261:11	223:25 228:5 240:14	156:17	200.22
JUD [20] 105:7,10,23		255:25	likely [2] 257:12 265:19	

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

machinery - moves Offshore Helicopter Safety Inquiry

			Offshore men	opter Barety Inqui
	115:13	252:2 253:4 262:23	193:19 221:16 230:3	132:25 136:3 159:7 246:9
-M-	manufacturers [2]	means [26] 6:11 16:9	252:15	mixture [1] 48:9
machinery [1] 183:21	193:22 194:11	49:12 56:24 92:7 96:19	merchant [4] 21:16	mobile [4] 47:12 48:4
machines III 201.3	map [2] 256:16 266:1	96:21,22 97:1 108:9	247:20 248:15 267:21	56:9 176:16
Maersk [2] 140.23 24	March [30] 13:5 20:2	109.5 111.20 117.2	mesh [1] 139:22	mobilized [2] 259:16
Waci SK [2] 149.23,24	27:11 33:4 38:13 122:3	152.24 155.0 152.19	message [2] 261:7,12	268:1
magnetic [1] 198:21	122:9 134:8 140:21 152:2	170:7 173:14 179:23	met [2] 187:17 263:13	mode [1] 254:12
mail [2] 171:9,10	182:13 185:3 198:7	223:12 253:25 271:7	metal [1] 241:8	model [5] 93:16 113:12
main [6] 43:17 163:12	200:21 214:11 215:21	meant [3] 177.22 205.9	meteorologist [1]	113:25 192:19 199:22
164:8 167:21 219:19	210:21 218:17 220:25	205:12	148:18	models [1] 192:18
223:2	249:15.21 250:22 251:12	measure [4] 84:2.4 95:20	meteorologists	moderate 11 166:20
maintain [7] 79:23 185:2	257:18 258:3 267:23	115:9	148:16	modifications (1) 4.24
204:17 230:12 248:12	268:19	measured of 66:11	meteorology [1] 147:24	MODI 151 47:6 0 10
249.3 203.0	marginal [1] 38:12	240:21 254:2,5,12 255:2	method (11, 98.3	56.10 88.1
Inamianeu [4] 89:7	marine [26] 22:11 23:22	measurement [2]	motro (4) 210-14 15 17	modulo (1) 42-24
91:3 249:8 232:24	37:17 90:22 126:18 144:5	101:10,12	219.17	MODUS 176 15
maintaining [1] 236:4	144:14 149:12 150:1,8	measures [4] 98:3 103:2	motros (7) 25.19 42.16	MODUS [1] 1/6:15
maintains [2] 248:10	152:6 203:13,16 204:5	109:16 198:25	A3.5 7 13 AA.5 A7.20	moment [4] 10:14
256:14	205:24,25 206:20 211:19	measuring [2] 97:4.5	Movico III 20.16	144:15 254:1 261:18
maintenance [21] 48:23	212:1 223:13 256:13	mechanic (1) 104·12	wiekico [1] 29:10	Monday [1] 266:21
50:14 89:8 92:9 96:16	259:6 263:17 264:18,21	mechanical (2) 25:24	mid [3] 188:22 197:1	money [1] 190:20
115.9 25 116.1 117.11	203.13	26·9	202.10	monitor [7] 84:21 85:2
126:18 146:1 179:6 8	Marked [4] 18:12 41:4	machanias ra 170.9	mignt [25] 7:5 8:20 9:13	107:24 115:9 159:10
181:3 193:10 196:11			15:9 04:4 72:21 87:25	178:20 236:8
maintenances	market [5] 44:15 58:7,9	mechanism [1] 48:6	104.11 12 21 105.18	monitoring [15] 80:3,3
183:21	58:11,19	media [1] 30:25	129:19.20 184:6 185:15	85:4,21 89:8 97:5,5
maior 161 50:13 92:22	markets [1] 49:7	medical [25] 152:24	188:9 206:5 227:19	150:5 169:8 177:19 178:1
181:14 187:21 229:2	marrying [1] 210:23	179:21 180:5 219:15,17	228:12 262:17	1/8:14 197:8 229:15,17
268:7	Marshall [1] 216:3	220:8 228:25 229:1,2,5	miles [1] 262:1	monitors [2] 178:6
majority [2] 47:5 58:19	material [2] 245:16	229:18,19,21,25,25 250:5	military [1] 220:3	
makers 11 119.6	269:11	236.13 259.17 21	million [4] 43.16 44.9	MONUM [7] 14:4 79:4,4
makes [2] 38.25 178.18	materials [1] 234:23	medically (4) 229.7	50:10 266:4	155:20 170:8 250:25,25
manege [2] 50.25 170.10	matrix [9] 97:7 109:23	231.15 18 232.7	mind [3] 37.2 126.3	monuns [5] 23:17 75:10
11121129 [4] 55:18 94:2	110:5,6,9 112:20,22	medications [3] 240.21	157:5	75:11 147:21 239:3
management [49] 19.17	154:4 155:12	241:1.4	minds m 5:7	1.22 10:6 72:22 157:7
26.2 27.23 32.8 34.10	matter [15] 3:1,18,19	medivac (11, 231.9	mingles (1) 130.3	1:25 10:0 72:22 157:7
38:17 42:7 59:25 60:8	4:17 9:8,18 68:1 95:11	medivace [1] 251.9	minimize (1) 176.25	167:6 172:4 175:12
65:5 77:7,14 78:18,19	102:23 120:1 121:4 126:6	mont up 0.12 19.7 (1.24	minimum (a) 77.19	185:16 201:16,21 250:18
78:21 79:12,18 85:5	150:15 190:21 204:7	111001 [15] 9:12 18:7 01:24	107·21	269:1,13
91:15,20,21 92:17,17,19	matters [6] 2:3,10 4:9	72.11 152.8 166.4 203.11	177.21 Ministor (1) 2(2.9	mornings [1] 6:9
92:23 97:4,25 98:6 99:5	4:11 5:2,22	208:1 211:15 229:16		Mosbacher [1] 46:23
102:2 104:8 107:10	mature [1] 4:21	meeting [25] 2.5 10.7	minor [2] 229:1 269:3	Moss (1) 271.13
108.20,20,23 124.23	maturity [1] 197:15	84:5 130:8,8,11,16,18	minus [2] 35:19,21	most [20] 13.8 25 36.10
136:24.25 139:14.15	maximum [6] 54:7	131:8,10,12,19 132:12	minute [2] 170:15 266:23	49:4 59:8.11.12.14 117:9
140:7 141:2 152:3 264:25	172:11 230:17 251:5	132:13,15,16 135:5	minutes [10] 72:25	133:24 142:6 149:23
manager [17] 18:20.22	200:23,25	136:13,16,17,20 137:2,9	128:25 131:8,18,23,24	167:25 168:11,16 180:20
22:19,23 23:1,16,19	may [56] 1:19 9:9 10:1	269:5,23	132:3 249:23,25 250:3	194:21 199:23 213:21
24:22 26:14,15,16 28:24	38.13 63.21 74.22 83.17	meetings [22] 119:11,13	mission [8] 254:7,23	214:1
137:2 141:20 162:19	83.17 84.25 91.19 97.25	121:17,18,20 128:23,25	257:17 258:12,14 262:1	motion [1] 199:24
233:24 237:13	106:4,6 108:7,24 109:16	131:14 132:4,24 133:9	203:19 200:18	motivated [1] 97:15
managers [1] 26:8	113:3 129:7 137:17 142:5	138:6.8.10 209:17 213:16	258.10.21.262.20	mounted [4] 166:22
managing [1] 34:17	153:4 157:23 167:24	member 151 27.8 126.23	238:19,21 205:20	167:16,19,20
mandatory [1] 236:13	178:13,13 183:11 185:19	128:10 136:15 244:17	milgale [6] 96:14	mounting [1] 163:12
manifest [1] 179:14	188:2 190:4 193:12,22	members [10] 1:4 5:8	156:3	move [25] 6:10 11:11
manned [1] 264:19	198:22,22 211:15 214:15	20:3 93:8 123:3 128:21	mitigated (1) 134.20	39:16 53:18,24 69:17
manner [5] 3.1 5.25 67.2	223.8 228.25 251.8	129:3,25 132:22 210:14	mitigating an 05-10	91:13 93:14 96:3,7 98:18
98:5 143:11	237:5.9 239:19 242:2.3	membership [1] 118:22	175.4	111:14 114:23 118:10
manning [1] 90.24	242:25 244:4,5 247:20	Memorial [2] 11.10 25.1	mitigation (126:13 165:19 178:24
manufactured	254:15 259:17	mention [2] /1.12 6/.16	103·2 107·4 110·1 125·7	201:8 245:21,22 252:20
191:21.24 211·18	McQuire [1] 268:7	124:6	220:9.14	moved (a) 25.15 19 20 4
manufacturer	mean [19] 7:3 39:9 46:3	mentioned up 21.16	mitigations rise 05.21	28.11 15 20.3 5 252.15
193:21 196:13 209:3	53:22 168:17 170:7 171:9	34:9 38:9.24 44:10 54:25	96:14 100:8.24 102:16	movement 121.04.5
210:10 216:3	172:1 173:5 178:16	84:10,21 90:7 122:13	103:7 106:11.24 109:20	117.4 147.9
manufacturer's m	183:20 190:17 197:12	134:13 163:8 192:25	109:25 110:4 111:3,5,20	$m_{0} = 11, -1 = 11, -2$
	199:7 223:10 227:10			110105[1] 23.13

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

moving - oil Offshore Helicopter Safety Inquiry

				copier survey inqui
moving [7] 14:6 91:11	156:13 158:18 172:10	244:12 261:5,6	201:17 205:2 213:24	offense [1] 64:17
93:13 114:7,25 129:17	211:15 223:18 231:2	Norsk [1] 25:22	228:9 248:17 251:25	offer [3] 19:9 200:10
257:7	241:4	North [11] 23:4,7 24:7	263:11	217:9
MRSC [1] 264:19	nets [1] 38:20	29:16 37:5,6,8,11 51:11	numbers [3] 105:5	offered [1] 268:3
Ms [2] 6:15 8:12	never [3] 20:6 83:4 222:4	70:20 233:11	126:20 175:16	office [1] 43:14
MS2000 [1] 202:6	nevertheless [1] 158:25	northern [2] 37:11 51:21	numerous [3] 27:5 65:11	officer [8] 77:2 157:21
muffs [2] 242:13 244:14	new [40] 12:4 51:12,12	Norway [1] 25:24	233:25	158:3,21 162:20 165:24
multidisciplinary [2]	51:14 68:22,22 69:1,12	note [5] 2:13 63:20 90:21	nurse [3] 231:5 240:23	235:1,8
187:11 189:19	69:13 73:21 109:16	147:18 216:7	240.23	officers [2] 98:8 163:7
multiple [6] 80:4 116:3	110:10 115:11 110:25	noted [6] 57:13 114:6		offload [4] 53:12,12,14
157:13,17 166:19 256:3	122.24 120.21,25 120.8	189:12 196:18 233:9	-0-	57:9
Murphy [2] 42:13 46:22	137:16 171:5 181:22	257:15	o'clock [3] 153:20	offloaded [2] 49:1,2
must [28] 77:6 78:17 89:1	192:19 193:3 197:24	nothing [3] 20:13 70:24	173:20 182:2	offloading [2] 21:24
93:10 99:7 139:14,17	208:21 209:6,13 211:18	238:8	object [1] 155:15	46:25
151:4 162:19 166:4,11	211:24 213:5,20,25	notice [1] 244:21	objective [4] 8:6 110:22	offshore [191] 11:17,20
168.3 24 169.15 18 172.9	241.12 220.11 239.0	notification [2] 259:8	221:10 223:24	11:20 12:15 20:1,10,16
199:9 223:4 232:24 235:4	newer [2] 197.5 6	239.11 notified to 255.2 250.20	objectives [5] 78:23	20:20,22 21:2,10 22:8,8
241:14 242:11 244:7	Newfoundland [24]	notified [2] 255:3 259:20	1/4:9,10 221:21 229:5	24:4.7 25:7.8.10.19.25
Mustang [4] 202:6,12	20:1 24:25 25:13 30:10	notify [1] 237:18	objects [1] 155:19	26:8 27:1,4,7,16 28:22
202:14 213:23	31:23 32:15 33:17 35:13	Notwithstanding [1]	obligations [1] 34:4	28:24 29:2,8,15,18,22
musts [1] 167:8	39:25 40:11 41:1 42:5	143:13 North and 10.22.24.22	obliged [1] 263:5	31:23 32:2,20 33:19
mutual [5] 80:11,13,15	44:16 55:13,20 58:20	INOVA [39] 18:22 24:23	observation [6] 108:6	35:12 36:24 39:25 40:7
150:9 184:17	59:7 61:3 63:9 70:6	29:17 34:6.24 39:12 40:2	108:17 144:7,18 146:10	40.9,11 42.3,13 44.13
myriad [1] 117:1	173.7 175.10 187.17	46:11,14,14 47:4,5,20	140.21	59:17 61:3 63:14 68:22
	196:16 217:21 219:10,23	49:4,14 50:12,18 51:23	observations [1] 108:10	70:12,16 71:11 78:2
-N-	223:13 228:23 271:6	53:16 54:3,13 86:14	147.11 19	79:14 83:16 84:12 86:11
Nalcor [2] 51:10.13	Newfoundlander [1]	151:16 158:1 160:22	obtain (2) 33.18 64.11	86:11 8/:11 91:18 106:6
name [2] 24:21 27:21	26:20	189:24 197:17 217:8.14	obtained (2) 35.18 04.11	110:16.17 120:2.3.9.11
named (1) 58.11	Newfoundlanders [1]	November [4] 42:17	obtaining (1) 75.25	121:3,5 123:23 124:3,23
names (1) 15.14	27:3	55:16 211:24 212:4	obtaining [1] 75.25	124:24 125:5 127:1,11
narrowed (1) 100.13	newly [1] 28:13	now [100] 1:4,14,16,19	29.17 100.11 113.15	127:24 131:11 133:21
national (7) 32.23 24	news [2] 137:15 232:16	2:12,19,25 4:5,6 9:3	173:11	134:25 137:1 138:7
118:18 260:14 263:9.9	next [48] 8:10 9:25 10:8	10:15 11:11 12:2,5,19	occasion [2] 106:19	147:4.16 150:3.21.24
263:14	10:13 12:20,21 18:18	14:6 15:18 17:11 21:11	170:14	151:25 153:6 154:20
nature [8] 60:11 61:23	44.23 46.10 54.21 56.8	25.23 35.1 12 41.16	occasional [2] 155:14	156:14 162:18 164:13
100:18 141:3 149:10	56:10 59:3,14 69:12 73:4	44:23 45:23 46:6,18 47:9	155:19	172:9,10,15 173:21
237:4 238:15 262:3	76:4 81:8 86:2 102:15	51:10 69:19,21 83:23	occasionally [1] 58:18	1/4:10 1/5:2 1/0:10
Navy [3] 21:16 265:6	108:19 109:22 131:25	85:9 89:11 91:11 93:22	occasions [3] 141:25	189:2 194:22 196:1.8.9
266:14	165:7 166:8 168:22 175:5	99:13 100:12,15,21	155:17 267:7	196:16 199:17 204:17
nearby [1] 265:11	197:20 201:9.12 203:18	113:13 116:16 118:10	occupational [12] 65:19	211:23 212:24 213:13,17
necessarily [10] 7:4 58:7	218:2,9 221:25 236:18	120:4 130:8 138:19	66:2 90:23 120:9,17	215:13 218:18 222:1
100.3 102.9 120.12	238:23 251:3 259:1	141:23 145:9 150:10	123:17 124:13 123:13	225:14,10 220:15,22,22
227:19 254:22	266:19 268:11	153:18 154:24 164:18	OCCUP 171 86:10 96:2	229:4.6.11.18.21.23.25
necessary [7] 7:20 69:8	nice [1] 202:1	165:4 1/3:/ 181:22 182:/	138:4.7 155:16 223:8	231:10,15 232:8,22 233:4
159:20,22 226:24 246:3	night [7] 169:3 173:3,9	192:20 200:7 201:8	246:19	233:6,8,12,24,24 234:17
251:6	173:10,16,16 175:7	202:14 209:16 213:2,12	occurred [4] 86:20	235:3,8,21 236:10,15,16
neck [2] 202:19 212:12	nimbleness [1] 4:23	214:3 217:24 222:3	134:10 158:23 238:6	237:9,11,12,14 238:25
need [52] 6:13 7:15,15	nine [1] 168:13	223:15 241:22 245:7,21	occurrence [1] 229:10	242:24 243:1,19 244:22
10:5 55:7 56:19 68:5	NL [1] 271:9	245:22 246:25 247:1	occurring [2] 113:5	245:5 246:19 249:3
76:9,15 80:9,9 82:24 83:7 03:17 05:3 6 11 12	NLOPB [3] 73:8 151:4	251:20 252:4.20 254:12	259:8	254:17 256:23 257:14
95:17.22.23 96:5 97:21	255:18	256:20 260:12 266:1	occurs [4] 61:10 114:4	258:12 259:8
101:1,2,20 105:20 109:4	Noble [1] 146:1	267:10,12 269:4,17	138:5 262:2	OITEN [11] 33:22 70:21
109:15,17,19 115:6,24	non [3] 63:22 64:16	nowhere [2] 2:15 253:1	ocean [3] 42:24 43:2	209.16 210.21 233.20 21
115:25 134:14 141:15	259:17	NTL [1] 58:8	200:8	238:12
150:5 152:11,25 161:14	HOHE [4] 145:12 171:4 183:22 238:10	number [37] 2:3 10:11	213.10 218.17	oil [44] 10:25 11:9.12
176:17 177:4 202:2 205:1	nonetheless 11 06.7	15:4 16:7,19 23:7 25:12	odd (1) 26.22	13:13 14:3 16:17 19:4
208:25 218:6 221:16	normal 141 50.15 164.10	27:4 28:4 29:1 33:12	off [11] 27.0 28.25 27.16	20:17 21:14,18 22:1,4
239:8 254:11	187:20 258:1	81:19 116:3 135.5 138.23	52:17 53:5 144:4 151.6	22:11,18 24:10,23 25:17
needed [5] 219:14,18	normally 1141 74.16 85.6	144:2 150:17 154:5	161:12 176:1 235:18	42:14.17 19 43·20 44·14
220:9 235:16 259:15	107:13 163:7 169:1,13	170:11 174:21 175:10	262:1	45:10 46:18 48:9,12,15
		106.2 / 102.4 12 100.25	99	

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

oilfield - people Offshore Helicopter Safety Inquiry

				copier barety inquir
53:15 55:5 71:24,25 91:1	onto [6] 16:8 39:22 48:10	77:15 78:2 80:11,16 81:4	256:17	participate [4] 132:21
165:11 213:24 223:24	59:1 212:14 215:23	91:19 121:19 125:18	outlined [2] 189:11	132:23 183:6 189:21
oilfield (1) 40.2	open [3] 116:25 117:3	140.8 141.14,15 142.4	266:3	participates [1] 118:17
	155.5	145:13 150:7,14 152:4,8	outlines [3] 61:1 226:20	participating [1] 33:21
Oilineius [1] 40:16	41:14 116:12	165:16,18 166:4,6 171:4	220:25	participation [2] 32:9
Oliwell [1] 226:17	opens (1) 116-14	174:8,20 179:1 180:10	OULWAFG [1] 249:24	
OINI [2] 135:11 254:17	operate (12) 11:22 12:12	180:22 181:1 182:25	Overall [8] 78:18 196:2	particles [1] 198:22
old [2] 212:11 232:19	32.5 36.9 40.19 45.14	191.6 7 13 196.14 202.5	246.17 263.19 264.25	particular [26] 11:15
older [3] 197:9 199:1	51:18 79:23 153:6,8	202:11 204:4 211:16	overboard [1] 48.16	20.8 22.9 34.3 30.0
252.15 OLE #1 222.11	216:8 250:15 267:15	217:21 218:23 219:23	overhang (1) 53.1	98:24 100:25 101:7 102:3
	operated [7] 34:10,24	225:9,13 226:16 227:10	overhead (1) 265:23	104:3 121:6 123:1 144:25
Office [26] 48:11 95:11 100-8 101-1 103-19 104-1	35:2 42:6 89:5 97:15	229:17 236:3,10 240:13	overnight (1) 18:0	146:23 150:16 157:8
109:12 112:20 113:11	151:14	253:15 256:22 257:1,12	oversight (1) 50.25	194:3,4 199:22 249:18
114:14 116:8 128:24,24	operates [3] 51:24 57:14	262:7 267:12 268:1,3,10	overview (2) 31.24 35.12	narticularly 1/2 52.5 6
135:19,19,19,20,25 170:7	100.22	opinion [5] 30:22 163:21	201:22	77:25 144:17 147:1 195:6
170:8,8 220:15 221:4	34:7 35:22 36:14 38:15	164:11,16 225:23	own [23] 5:18 13:14.15	parties [9] 2:9.17 3:3 7:1
232:21 244:7 238:12	38:25 59:24 63:15,21	OPITO [1] 233:10	13:15 36:1 50:9 85:2	8:13 14:20 17:9 18:6
15 :11 16:1 9 19:13 20:12	70:6 92:25 158:15 165:22	opportune [1] 257:20	112:23 127:9,10 128:9	269:6
21:21,24 23:4,10 25:22	172:5 174:8 190:13	opportunities [1] 13:22	131:14 139:13 143:10	partner [2] 51:11 93:7
26:1,7,12 28:9 31:12	193:14 194:22	opportunity [21] 3:13	144:20 146:15,19 148:18	partners [4] 33:23 34:14
34:24 35:2,21 36:1,6,9	operation [14] 60:10	3:21,22 5:13 6:13 7:11	245.18 268.11	46:20 51:10
36:17 37:8 40:18 44:6	129.12 150.11 162.3	9:23 15:23 18:7 21:3	owned (1) 34:13	parts [2] 16:11 179:25
45:11,14,17 40:4 52:15	176:23 179:5 182:25	28:21 58:10 08:20 84:9	ownershin [1] 33.25	party [3] 5:20 88:19
76:19 81:3,4 82:1 99:16	263:20,24 265:17	257:16 262:25	owns [1] 145.24	256:12
100:10 102:2 105:10	operational [10] 82:3	opposed [6] 38:23 53:4	0 with [1] 143.24	pass [3] 24:18 114:21
107:14 112:13 117:18	95:12 97:23 174:10	175:20 212:11 214:25		241:8
122:13 124:14 125:6,20	184:22 185:19 204:3	217:18		passenger [18] /0:/
120.1,23 127.1,12,17,23	219.12 249.2,9	opposite [1] 41:10	P [1] 17:10	166:21 171:13 179:10
137:7,8 139:8 144:8	127:17 185:1	option [1] 162:15	P-116 [1] 17:16	196:3,4 201:14,18 202:25
145:5 147:12 151:19	operations [74] 16:18	options [2] 191:7 195:25	P-II 7 [1] 17:17	203:11 223:3 242:20
154:2 158:23 163:11	17:1 18:20 19:25 20:17	orange [1] 260:24	p.m [2] 266:22 270:3	244:13,23
174:5 176:21 177:19	23:16,19,22,23 26:14,15	order [9] 45:19 65:16	pack [1] 168:23	passengers [17] 166:24
181:19,21,24,24 183:16	29:8 30:12 32:15 33:2,7	95:15 166:5 180:20	package [2] 213:13	108:24 109:2,15,18 1/0:4
183:24 184:1,22 185:2,3	35:15 50:15 54:24 57:4 57:18 59:22 61:2 6 7	197:22 223:4 227:22	214:19	242:11 243:7.24 244:6
185:11,15,23,24 186:3	62:15 64:9 65:21 66:6	ordinary (1) 168.12	page [1] 178:24	244:10,13 267:25
187:6 188:15 190:11	70:5 72:6 74:12,13,16	organization (a. 4.13	pages [2] 18:3 73:5	passwords [1] 3:4
193.10 194.0,7 190.1	79:13,15,18 80:5 83:24	70:25 104:22 118:19	PAL [2] 257:18 258:4	past [10] 13:21 170:11,14
207:16 209:22,24 212:22	93:2,4 95:1 97:20 118:21	129:3 133:4	panel [9] 1:4 12:23 13:24	170:16 180:13 194:8
212:24 215:5,6,19 217:7	149:8 150:1 156:8 158:6	organizational [2]	10.13,23 17.10 44.22	221:11,12 262:24 269:3
217:16 219:14,14,16,17	159:11 160:10,13 164:19	79:19 246:14	nanellists [2] 154.1	pat [1] 108:15
239.7 240.14 243.4 249.8	165:5,8,11,13,20 166:1	organizations [1] 90:12	230:4	path [1] 41:11
250:1,4,8,16 251:5 253:9	166:3,9,11 176:19 184:10	orientate [1] 54:5	panels [2] 14:2 60:22	pattern [1] 172:13
253:12 254:12 255:1	234.14 250.17 259.6	orientated [1] 52:15	parallel [1] 211:19	Paul [4] 18:15 19:2 27:21
256:3 261:4 262:6 264:21	268:18,25	orientation [2] 134:21	Parks [1] 263:12	118:24 DOM: 115-7
20/:2,0	operator [34] 16:12	135:2	parlance [1] 168:12	pay [1] 15:7
one-day [1] 233:4	38:16 46:20 59:17 71:24	orientations [1] 134:17	part [52] 3:5 4:5 5:11 8:9	paying [1] 214:25
one-nour [5] 253:2,6,21	74:15 77:4,6 139:16,18	oriented [2] 69:14 118:1	8:9,11 16:16 25:12 37:12	Peet [1] 271:5
one-third (1) 160.22	141:4,8 142:8,10 145:1	original [2] 51:7 247:14	43:22 45:6 49:5,24 50:3	penalized [1] 180:25
0110^{-1111} 11^{-1} 100.23	175.7 180.25 181.22	originally [2] 17:10	50:6 85:24 100:10 124:1	people [91] 5:4 10:3 20:9
ones (5) 21:25 28:22	187:22,23 188:2 189:6	50:12	124.5 125.19 151.12	56.5 6 18 90.3 92.4 93.5
117·11 123·6 9	236:8 237:11,18,21	OSH [2] 81:14 123:23	139:12 141:5 144:12	93:10,17,23 94:3,5,6,7
ongoing [3] 143.5 150.15	245:23 257:5,7 261:5	otherwise [1] 147:8	149:24 154:14,16,20	94:15,17 95:7 96:15,25
186:9	operator's [8] 32:21	ourselves [6] 3:24 21:12	156:11 158:1 161:15	97:3,6,16,21 104:13
onshore [18] 26:13 27:1	/8:20 83:19 97:25 108:8	27:14 40:15 70:20 250:19	102:2 107:23 108:10	109:12 112:21,25 113:9
28:9 29:3 40:7 106:5,15	onerators (02) 4.7.12	outage [1] 50:14	212:17 215:20 222:16	117:8,14,19.20 118:1.12
106:20 120:2,4 127:9	Perutor 5 [33] 4.7,12	outbound [1] 250:4	227.18 241.20 253.12	118:15 120:3.5 121:5
1 134:25 135:15 136:24	10:25 11:9,18,21.23	antdatad - 1751010	227.10 241.20 255.12	110110 12010,0 12110
138.7 150.22 154.15	10:25 11:9,18,21,23 12:17 19:5,24 23:9 31:22	outdated [4] 176:10,12	262:20 263:14 264:22	126:14 129:17,24 130:10
138:7 150:22 154:15 156:12	10:25 11:9,18,21,23 12:17 19:5,24 23:9 31:22 32:8,12 33:3,16 38:10	outdated [4] 176:10,12 176:14,14 outdoors (1) 24:13	262:20 263:14 264:22 266:7	126:14 129:17,24 130:10 130:13,21 131:9,11,16 134:14 127:13 21 141:0
138:7 150:22 154:15 156:12 onshore/offshore (1)	10:25 11:9,18,21,23 12:17 19:5,24 23:9 31:22 32:8,12 33:3,16 38:10 38:16 57:16 59:7 61:11 61:24 63:5 69:25 70:5	outdated [4] 176:10,12 176:14,14 outdoors [1] 24:13 outline [3] 30:17 201:17	227.18 241.20 255.12 262:20 263:14 264:22 266:7 partial [1] 174:6	126:14 129:17,24 130:10 130:13,21 131:9,11,16 134:14 137:13,21 141:9 141:11 142:1 152:2

Discoveries Unlimited Inc., Ph: (709)437-5028

92:11 94:9

232:10 252:25

205:1 266:6

perfect [1] 199:4

175:11

219:12

107:23

242:5

179:24 191:9

135:25,25 154:21

240:3 242:21

32:13 40:12 43:21 50:21

119:8

170:22 173:12 174:15 57:8 60:4 71:13 78:21 183:11,14 185:7 191:19 83:21 152:15 179:7,8 199:22 203:19 213:21 204:25 205:4 213:18 214:1 227:16,17,21 228:6 214:13,23 216:6 218:18 235:11 236:22.22 241:1 222:23 225:10 226:13 246:4 254:18 256:3 229:3,7 234:18 236:15 236:20 237:2 240:17 people's [4] 15:13 24:12 241:9.15.16 242:1.14 243:22 244:2.15 251:2 per [8] 24:1 29:22 35:17 persons [4] 46:4 105:24 83:14 113:14 204:12 105:24 147:13 **perspective** [7] 28:19 **perceive** [1] 257:2 28:21 43:15 82:6 140:1 percent [9] 50:17 51:14 200:17 242:25 175:18,20 176:1,2 204:18 pertain [1] 4:25 **pertaining** [1] 73:12 percentage [2] 34:1 Petro [13] 25:5 145:8 189:7,9,11,17,18,22,25 195:10,11,18 196:19 perform [8] 90:14 PetroCanada [2] 35:1 101:19 106:4 107:2.7 46:18 141:9 154:10 187:5 petroleum [14] 2:1 31:25 performance [5] 98:2 35:13 61:2 63:13 65:14 115:3 187:12 198:10 68:23 71:12,15 81:8,11 82:11.16 87:11 performed [2] 98:14 phase [9] 2:15 7:25 8:1 8:10,11,19 40:4 143:3 perhaps [32] 7:24 9:23 181:11 15:17 20:4 31:2 33:11 **Phillips** [1] 12:5 39:22 90:3 92:15 95:2 **philosophy** [1] 108:24 102:5 105:13 107:21 108:6,12 112:11 115:12 phonetic [1] 28:10 115:14,16 116:12,18,24 photograph [2] 52:19 117:8 127:5 155:7 173:12 52:20 201:3 215:23 246:11 photographs [1] 202:15 252:3 259:22 261:25 phrase [1] 91:20 **period** [13] 7:12 12:8 55:15 69:8 73:11 74:20 **physical** [8] 68:16 80:2 114:8 115:1 155:6,21 74:24 189:1 199:18 215:19 218:24 233:6 170:17 268:4 **physically** [6] 105:6 periodically [3] 85:20 114:23 155:4.15 159:13 249:10 periods [2] 50:21 191:6 pick [1] 198:21 picks [1] 124:8 permanent [1] 267:6 permit [18] 82:15,16 pickup [1] 199:1 98:23 99:25 100:14 104:4 picture [4] 56:8,11 104:7,13,16 106:22,24 265:24 267:10 107:21 108:3 135:12,23 piece [9] 114:14 147:2 156:6 179:12 183:21 permit-to-work [4] 191:20 220:11 226:3 99:17,23 107:24 108:2 249:18 permitted [2] 73:9 206:6 **pieces** [1] 118:9 persistent [1] 108:21 **pigs** [1] 123:12 person [20] 15:16 19:13 Pike [1] 165:23 54:7 104:21 133:6 141:21 pilots [9] 147:7 166:12 159:2,16 179:11 215:10 179:7 200:24 251:11.18 216:15 217:22 231:1,15 252:2,3 267:25 231:17,23 238:24 239:11 **pipe** [4] 96:11 155:14,21 156:6 personal [13] 29:23 place [36] 4:19 15:4.18 32:16 65:4 150:17 201:12 27:16 32:12 33:3 38:17 212:13 216:1,2 218:7 45:11 64:6 69:23 72:21 236:11 241:10 245:10,18 100:2 106:11,23 114:22 personally [3] 36:2 91:6 117:22,22 122:15,16,17 153:16 169:22,25 205:13 **personnel** [39] 29:1 209:4 211:12 213:23

228:12 229:11 234:23 240:20 264:23 271:5 places [7] 27:5 56:25 72:8 111:3 165:25 173:5 255:9 plan [49] 68:13 69:3,4 76:2,5,10,15,23 77:3,4,7 77:11,18,19 78:15,16,17 78:24 79:12,15 80:7 84:8 85:2.3 91:1 93:20 104:15 109:18,20 111:25 154:18 211:12,17 212:4 213:11 213:19 218:15 220:10,14 246:6,13,16,17 247:9 250:24 253:6,8,16,20 plane [1] 265:21 planned [3] 104:14 124:19 141:4 planning [7] 80:1,8 98:2 172:1 201:19 212:17 252.22 plans [6] 13:16 79:2 84:3 238:20 246:13 247:8 platform [18] 26:1,4 28:16,20 29:19 42:14 43:11,12,15,23 44:8,22 145:25 151:20 233:9 244:25 245:13 267:24 play [4] 80:25 81:2 92:2 94:3 players [1] 92:23 PLB [5] 212:12 217:14 217:15 218:4 244:1 PLBs [4] 203:3 216:8 217:1,3 pleased [2] 25:10 268:6 plot [1] 110:22 **plotted** [1] 248:4 plugs [2] 242:13 244:14 **POB** [6] 46:6 54:7 204:19 204:23,23 230:18 point [28] 3:25 5:23 8:3 43:12 71:18 91:13 94:5 94:9 115:18 138:6 147:18 148:10 156:21 162:18 163:2,6 184:21 188:15 196:21 200:12,14,20 211:7 224:20 234:9 254:6 254:9 268:23 points [2] 31:10 199:1 **policeman** [1] 222:12 policies [5] 79:16 99:6,7 139:18 142:24 **policy** [9] 98:7,12,21,22 99:12,14,24 101:16 123:22 polluting [1] 89:5 pollution [1] 91:1 **Pond** [1] 172:18 **pool** [3] 180:22 182:15 196:20 **pooling** [9] 16:24 17:5 17:25 181:12 183:3,6,12 184:8 186:6 port [3] 52:17 53:2,6 portion [5] 89:19 149:13

Multi-PageTM

people's - Pritchard **Offshore Helicopter Safety Inquiry**

149:25 171:17 174:2 present [2] 5:10 162:23 posed [1] 77:9 presentation [20] 16:13 16:14,25 17:17 19:5,12 position [8] 19:11 21:8 30:17 31:2,17 110:10 29:4,13 141:18 153:13 137:4.6 138:22 144:13 242:13 244:9 165:9 180:8 213:17 **positioned** [1] 242:22 224:24 225:1 257:2 **positioning** [1] 153:11 presentations [9] 4:7 positions [2] 22:7 28:13 12:22 44:23 60:20 79:3 possibility [1] 65:1 127:6 186:19 247:2,15 possible [10] 26:24 31:5 presenters [1] 19:7 65:4 88:2 110:25 169:4 presents [1] 111:11 173:4 229:9.10 237:6 preservation [1] 256:14 possibly [2] 9:17 239:9 President [3] 18:17 post [1] 268:19 27:22 29:24 posted [2] 68:24 213:17 pressure [2] 48:22,25 **posture** [1] 249:7 presume [4] 2:7 157:13 potential [11] 96:1 103:1 212:22 233:18 113:10 114:2 170:25 pretty [3] 106:2 110:12 172:6 174:14 187:3,4 269:8 209:7 246:8 **prevailing** [3] 39:19 potentially [7] 92:12 155:8 172:24 103:20 104:17 112:9 prevalent [2] 38:13 120:2 135:15 228:24 198:16 pounds [1] 241:19 prevent [3] 92:5 156:18 power [2] 43:19 50:9 229:10 **PowerPoint** [2] 16:14 preview [1] 16:3 213:13 previous [5] 113:8 practicable [2] 30:7 132:12,15 135:14 200:5 169:9 **previously** [10] 22:2 practical [4] 108:24 34:9 59:18 61:8 93:16 109:15 183:18 258:9 98:17 153:2 196:5,19 practically [1] 173:5 252:18 practice [7] 71:12 100:12 price [3] 15:7,11,11 100:13 166:16 226:12 pride [1] 26:20 236:7 237:1 **primarily** [4] 78:1 practices [4] 93:3 101:17 142:25 149:22 150:25 108:2 142:24 primary [6] 49:12 158:8 practises [1] 119:22 179:23 191:14 231:13 pre-qualification [2] 265:3 143:3 187:2 principally [2] 15:5 pre-set [1] 184:2 263:18 precise [1] 248:4 **principle**^[1] 158:10 predictions [1] 148:8 principles [2] 92:20 102:13 **predrilled** [1] 54:11 priority [2] 126:4 184:11 predrills [1] 54:13 Pritchard [159] 18:19 prefer [1] 268:24 19:1,18,20,22 37:21 38:1 preferences [1] 199:19 40:25 49:11 51:2,5 52:8 preferred [3] 56:24 52:12,23 53:7,11,23 207:20 209:13 54:20 55:21 56:2 57:2 preliminary [1] 1:11 57:23 58:13,21 66:15,16 66:21 67:1,5,21 78:12 premature [1] 198:23 82:7 91:16,17 94:19,25 preparation [6] 19:4 99:9,22 102:7 103:18 22:25 121:2 154:14 189:7 104:9,24 105:3 106:1 191:1 107:12 110:11.19 111:1 prepare [3] 119:22 223:5 111:16 112:4,15,19 246:3 113:23 114:17 117:23 prepared [3] 6:10 238:2 118:3 119:10,15,19 120:19 121:1,11 122:5 239:21 122:10,18 123:16 124:17 preparedness [3] 32:21 125:9,13,21 126:2,15 245:23 246:1 127:14,20 128:4,16,20 **prescribe** [1] 124:21 129:22 130:12,23 131:2 prescribed [2] 259:7 131:7,20 132:2,8,19 267:11

Discoveries Unlimited Inc., Ph: (709)437-5028

222:7 225:20 227:5

Index Page 15

134:12 135:22 136:11

Multi-PageTM

privacy - re-submitted Offshore Helicopter Safety Inquiry

			01101101 0 110110	copier burety inquir,
137:11.23 138:2.13.18	product [2] 21:19 48:8	222:18.24 223:3.6.16	43:25 44:17.25 45:5.9	249:12 250:5.11 251:17
143.12.22.150.10.154.3	production (2) 21.14	229.13 231.6 239.19	46.1 9 47.8 48.17 50.23	252.6 12 19 253.11 17
154.9 156.22 157.2 14	production [32] 21:14	• • • • • • • • • • • • • • • • • • • •	51.2 52.4 10 18 52.2 0	252.0,12,19 255.11,17
157 10 150 17 150 5 17	21:23 22:2,11 23:21	provided [15] 2:3,5,19	52 21 54 16 55 19 24	255.24 254.8,19,24 255.7
157:18 158:17 159:5,17	25:16,20 26:1,3,6,18	2:23 3:2,13 9:4,6,8 59:24	53:21 54:16 55:18,24	255:21 256:5,9,19 257:4
159:21 161:5,19 184:19	28:23 42:5 45:10 47:2	164:11 165:14,24 231:10	56:21 57:20 58:10,24	257:9,21 258:6 260:18
184:25 185:8,12,17	47:17 51:20 54:15,24	262:11	60:7,17,23 61:12,19 62:1	261:3,16 262:16 263:2
200:11 210:15 211:2	57:3 63:12 64:2 65:15	provider [15] 11.2	62:8,13,18,22 63:6,24	264:1,6,10,15 267:13
230:15 238:7 245:22,24	68.23 69.2 74.12 86.6	1 40.20 142.8 145.4	64:13,18,23 65:3,8,24	268:20 269:20
247:3,10,16 248:5,11,25	141.14 15 156.21 195.3	140:20 142:8 143:4	66:14,19,24 67:3,8,12	Oatar 11 28.12
249.19 250.9 13 251.23	212.24	146:13,18 148:3 165:15	67.18 23 68.4 9 69.15	
252.10 14 253.5 14 19	215:24	186:12,25 187:5 188:18	69.24 70.23 71.4 8 16	qualification [4] 119:23
252.10,14 255.5,14,17	products [2] 22:4 48:12	189:24 190:7 229:22	71.23 72.14 20 73.2 14	121:13 236:7 237:7
254.4,14,21 255.5,10	professional 121 30.1	providers (1) 148.14	71.25 72.14,20 75.2,14	qualifications [13]
250.1,7,11,24 257.0,11	36.3		75:24 74:5,9,18,25 75:4	32.10 71.13 70.25 221.25
257:24 258:8 260:19	50.5	provides [7] 32:2 70:17	75:8,14 76:8,13,18,24	225.10 12 18 22 226.12
261:1,10,21 263:4 264:4	professionals [2] 120:1	88:19 180:2 218:11	77:21 78:8 79:1,9 80:12	225:10,12,18,25 226:15
264:8,13,17 267:18	227:24	226:13 233:5	80:19,23 81:23 82:10	226:20 227:8 236:5 237:2
nrivacy [1] 15.14	profile (1) 256.16	providing [3] 63.4 169.6	83:1,5,10,22 84:16 85:8	qualified [5] 147:24
		100.11	85:12.16 86:1.25 87:7	187:5.6.9 241:14
proactive [1] 135:7	program [24] 12:3,6	100.11	87.12 18 22 88.9 89.10	qualities (1) 207.19
probe [1] 6:25	41:21 55:1 56:16 68:14	province [1] 20:11	80.16 22 00.1 01.4 10	quanties [1] 207:18
problem $(n = 10.1 \pm 210.0)$	74:22 75:3 76:9,15 81:17	provincial (2), 123-19	01.14.04.12.22.00.1.20	quality [5] 116:5,7,9
problem [4] 10:1,1 210:9	85:20 90:7 108:18 123:22	124.3	91:14 94:13,23 99:1,20	117:13 209:4
229:1	144.13 182.1 195.3	124.5	101:25 103:14 104:6,19	augrtor (2) 50.4 100.1
problems [2] 125:17	222.16 223.22 232.24	provision [8] 146:23	105:1,22 10/:8 110:8,15	qualter [3] 59.4 190.1
214:7	222.10 223.22 232.24	147:15 149:9 150:8	110:21 111:9,23 112:10	220:25
procedure to 09.24.25	233.2 234.7 237.17	166:15 179:2 187:20	112:17 113:19 114:11	quarterly [3] 83:14
procedure [9] 98:24,25	programs [12] 55:3 61:4	263:6	117:17.25 119:7.12.17	86:18 121:18
99:13,17,23,25 158:19	68:16 79:16 89:9 117:11	public [12] 10:10 14:12	120:13.24 121:9 122:1 8	questioned 121 7.5
199:10 259:9	129:5 140:10 152:4	Public [12] 10:19 14:12	120.13,21 121.9 122.1,0	
procedures [13] 79.17	188.21 211.17 20	14:23 16:23,25 17:2,11	122.12 123.3,11 124.11	238:12
97.15 98.14 23 99.7		17:19,21,24 21:1 30:25	125.2,11,10,25 120.7	questioning [2] 4:14
101.16 108.2 100.12	progress [3] 92:8 93:12	publication [4] 70:11	12/:4,16,25 128:12,18	9:15
101.10 100.2 109.12	269:9	70.19 71.3 90.21	129:9 130:7,19,25 131:4	questions (m. 2.0.2.14
100:5 101:11,14 257:5	project [25] 21:21 22:4,7		131:17,22 132:5,17	questions [17] 2:9 5:14
246:23	22:9.13.14.23 24:23 34:3	puii [2] 112:8 157:9	133:18 134:1,6 135:17	4:12 5:13 7:5 8:6 9:9
proceed [4] 2:14 133:16	34.8 18 24 35.2 5 42.13	Puma [6] 191:14,24	136:9 137:5,19,25 138:9	10:3 19:8 51:1 157:5
195:13 256:8	57.8 50.17 22 03.25 04.1	196:18 223:15 224:7.9	138:16 139:20 140:11.18	178:23 198:2 201:23
proceeding to 2.1 4.6	05.2 129.9 145.1 190.9	Dumos (5) 102.17 106.5	140.24 141.10 19 142.14	237:19 238:15 239:24
	95.5 120.0 145.1 109.0	1 uillas [5] 192:17 190:5	140.24 141.10,19 142.14	auite (14) 3.17 5.2 5
5:25	191:1	196:22 197:13 199:24	140:24 141:10,19 142:14 142:20 143:8,19,24 144:11 145:18 146:2 8	quite [14] 3:17 5:2,5
5:25 proceedings [2] 9:24	95.5 128.8 145.1 189.8 191:1 project-related [2] 57:5	1 umas [5] 192:17 196:3 196:22 197:13 199:24 pump [8] 104:3 105:5,6	142:20 143:8,19,24 144:11 145:18 146:3,8	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6
5:25 proceedings [2] 9:24 10:6	project-related [2] 57:5 57:6	1 umas [5] 192:17 196:5 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11.16.19.25	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13
5:25 proceedings [2] 9:24 10:6	95.5 128.8 145.1 189.8 191:1 project-related [2] 57:5 57:6	1 umas [5] 192:17 196:5 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8	projectile [1] 171:3	1 unias [5] 192.17 196.5 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [8] 104:1 115:7,7	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13	project: [1] 171:3 projects [2] 33:16 108:25	Tumps [5] 192.17 196.5 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [8] 104:1 115:7,7 115:10,14,22,24 116:3	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15.24
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25	project: [1] 171:3 project: [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21	1 unitas [5] 192:17 196:3 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [8] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13	project: [1] 171:3 project: [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21	1 unitas [5] 192:17 196:3 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [8] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purps [12] 13:20 34:17	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10	project: project: prop project: prop project: prop project: prope: prope: project: prope: project: prope: project: prope: project: project:	1 unitas [5] 192:17 196:3 196:22 197:13 199:24 pump [8] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [8] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 28:21 24 20:4 06:17	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20 24 172:21 173:2	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R-
proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16	umas [s] 192.17 196.5 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97.10 104:4 222.18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8 19 23 176:3	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R-
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3.15	umas [s] 192.17 196.23 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11 24 177:5 10 14	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25
proceedings [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1 37 189:10	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3	umas [s] 192.17 196.23 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3 11 22 180:16 181:5	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16 18 21 191:3 195:1	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 240:18 244:18	umas [s] 192.17 196.23 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8
proceedings [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18	umas [s] 192.17 196.23 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise (4) 5:1 104:11 13
5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:2 214:1 278:2 14	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5	umas [s] 192:17 196:3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20
proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 210:1 3 6 10 220:21	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14	umas [s] 192:17 196:3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purposes [4] 16:21 89:17 151:25 156:10	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20
proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 203:7 20:0 1 207:22	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6.18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12
proceedings [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 prosper [1] 30:11	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19
proceeding [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 prosper [1] 30:11 protect [n] 48:1 115:22	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17.24 204:10	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19
proceedings [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18	95.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3 8 14 206:3	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10
processing [3] 5.14.0 5:25 process [61] 4:15 14:8 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:47 15 18	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 prosper [1] 30:11 protect [3] 48:1 115:23 246:4	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10 14,19 23 207:3 7	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1
proceedings [3] 3.14.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11 10 108:22 116:1	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:7	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 rank (11 107:9
proceedings [3] 3.14.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protection [5] 47:22 51:17 69:4 197:9 209:2	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 rank [1] 107:9
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protection [5] 47:22 51:17 69:4 197:9 209:2	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2	$\begin{array}{r} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \textbf{raises} [2] \ 104:7 \ 162:10 \\ \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \textbf{ranks} [1] \ 107:9 \\ \textbf{ranks} [1] \ 21:19 \end{array}$
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protection [5] 47:22 51:17 69:4 197:9 209:2 protective [2] 32:16	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 puts [1] 244:16	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 ranks [1] 107:9 ranks [1] 21:19 rate [2] 18:11 269:9
processing [3] 5.14.0 5:25 process [61] 4:15 14:8 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 processing [1] 63:13	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protecting [1] 48:6 protecting [2] 32:16 201:13	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 ranks [1] 21:19 rate [2] 18:11 269:9 raise [2] 10.13:21
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produce [3] 64:4 94:7 106:15	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 propostion [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protecting [2] 32:16 201:13 protectole [3] 38:17 161:1	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6	$\begin{array}{c} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \hline \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \hline \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \hline \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \hline \textbf{raises} [2] \ 104:7 \ 162:10 \\ \hline \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \hline \textbf{rank} [1] \ 107:9 \\ \hline \textbf{ranks} [1] \ 21:19 \\ \hline \textbf{rate} [2] \ 18:11 \ 269:9 \\ \hline \textbf{rather} [12] \ 8:9,10 \ 13:21 \\ 20:22 \ 522 \ 105:16 \ 224 \ 22 \\ \hline \end{array}$
proceeding [3] 3.14.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produce [3] 64:4 94:7 106:15	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 propostion [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protecting [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 ranks [1] 21:19 rather [12] 8:9,10 13:21 2:23 22: 20:51:6 224:2 2:10:13:21
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produce [3] 64:4 94:7 106:15 produced [5] 2:10 4:1	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4	$\begin{array}{r} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \textbf{raises} [2] \ 104:7 \ 162:10 \\ \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \textbf{rank} [1] \ 107:9 \\ \textbf{ranks} [1] \ 21:19 \\ \textbf{rate} [2] \ 18:11 \ 269:9 \\ \textbf{rather} [12] \ 8:9,10 \ 13:21 \\ 20:23 \ 52:2 \ 105:16 \ 224:2 \\ 249:22 \ 250:20 \ 251:21 \\ \end{array}$
proceedings [3] 5.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produced [5] 2:10 4:1 48:14 49:1 71:14	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protection [5] 47:22 51:17 69:4 197:9 209:2 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2 12 19 24	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 ranks [1] 107:9 ranks [1] 21:19 rate [2] 18:11 269:9 rather [12] 8:9,10 13:21 20:23 52:2 105:16 224:2 249:22 250:20 251:21 259:11 262:4
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produced [5] 2:10 4:1 48:14 49:1 71:14 Producers [21 2:1 71:15	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 proprietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11 proven [2] 169:8 177:18	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 putting [2] 117:21,22 -Q- Q.C [415] 1:6,15,20,24 3:10 6:4,12 10:10,18	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 range [6] 153:4,5 176:1 195:7 196:6,7 ranks [1] 107:9 ranks [1] 21:19 rate [2] 18:11 269:9 rather [12] 8:9,10 13:21 20:23 52:2 105:16 224:2 249:22 250:20 251:21 259:11 262:4 rationale [3] 176:5,7
proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produced [5] 2:10 4:1 48:14 49:1 71:14 Producers [2] 2:1 71:15	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposition [1] 3:11 properietary [1] 14:17 prosper [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protecting [1] 48:6 protect [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11 proven [2] 169:8 177:18	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24 231:12 232:15 233:17 234:8 235:6 10 25 226:10	$\begin{array}{c} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \textbf{raises} [2] \ 104:7 \ 162:10 \\ \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \textbf{rank} [1] \ 107:9 \\ \textbf{ranks} [1] \ 21:19 \\ \textbf{rate} [2] \ 18:11 \ 269:9 \\ \textbf{rather} [12] \ 8:9,10 \ 13:21 \\ 20:23 \ 52:2 \ 105:16 \ 224:2 \\ 249:22 \ 250:20 \ 251:21 \\ 259:11 \ 262:4 \\ \textbf{rationale} [3] \ 176:5,7 \\ 204:1 \\ \end{array}$
proceedings [3] 3.1 4.0 5:25 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produce [3] 64:4 94:7 106:15 produced [5] 2:10 4:1 48:14 49:1 71:14 Producers [2] 2:1 71:15 producing [13] 19:25	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 property [1] 246:5 proposal [2] 187:9,14 proposition [1] 3:11 property [1] 246:5 proposition [1] 3:11 property [1] 246:5 proposition [1] 3:11 property [1] 246:5 proposition [1] 3:11 property [1] 48:6 protecting [1] 48:6 protecting [1] 48:6 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 proven [2] 169:8 177:18 provide [27] 4:2 22:18 21:20 24 28:10 22:18	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22 -Q- Q.C [415] 1:6,15,20,24 3:10 6:4,12 10:10,18 18:13 19:3,19 30:19 33:10 34:15 21 35:4 9	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24 231:12 232:15 233:17 234:8 235:6,19,25 236:19	$\begin{array}{c} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \textbf{raises} [2] \ 104:7 \ 162:10 \\ \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \textbf{rank} [1] \ 107:9 \\ \textbf{ranks} [1] \ 21:19 \\ \textbf{rater} [12] \ 8:9,10 \ 13:21 \\ 20:23 \ 52:2 \ 105:16 \ 224:2 \\ 249:22 \ 250:20 \ 251:21 \\ 259:11 \ 262:4 \\ \textbf{rationale} [3] \ 176:5,7 \\ 204:1 \\ \textbf{rationale} [3] \ 176:5,7 \\ 204:1 \\ \end{array}$
proceeding [3] 3.1 4.0 5:25 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 106:5 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 produce [3] 64:4 94:7 106:15 produced [5] 2:10 4:1 48:14 49:1 71:14 Producers [2] 2:1 71:15 producing [13] 19:25 28:10 29:8 31:23 32:3	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 properly [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 propostion [1] 30:11 protect [3] 48:1 115:23 246:4 protecting [1] 48:6 protecting [1] 48:6 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11 provide [27] 4:2 22:18 31:20,24 88:16,22 109:3 31:20,24 88:16,22 109:3	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22 -Q- Q.C [415] 1:6,15,20,24 3:10 6:4,12 10:10,18 18:13 19:3,19 30:19 33:10 34:15,21 35:4,9 35:24 36:16 20 38:6 39:2	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24 231:12 232:15 233:17 234:8 235:6,19,25 236:19 237:25 238:14 239:1,10	$\begin{array}{c} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \hline \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \hline \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \hline \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \hline \textbf{raises} [2] \ 104:7 \ 162:10 \\ \hline \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \hline \textbf{rank} [1] \ 107:9 \\ \hline \textbf{ranks} [1] \ 21:19 \\ \hline \textbf{rate} [2] \ 18:11 \ 269:9 \\ \hline \textbf{rather} [12] \ 8:9,10 \ 13:21 \\ 20:23 \ 52:2 \ 105:16 \ 224:2 \\ 249:22 \ 250:20 \ 251:21 \\ 259:11 \ 262:4 \\ \hline \textbf{rationale} [3] \ 176:5,7 \\ 204:1 \\ \hline \textbf{rationalize} [1] \ 177:1 \\ \hline \end{array}$
proceeding [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 processing [1] 63:13 produce [3] 64:4 94:7 106:15 produced [5] 2:10 4:1 48:14 49:1 71:14 Producers [2] 2:1 71:15 producing [13] 19:25 28:10 29:8 31:23 32:3 40:2 46:15 48:6 54:25	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 proposal [2] 187:9,14 propostion [1] 30:11 protecting [1] 48:6 protecting [1] 48:6 protecting [1] 48:6 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11 provide [27] 4:2 22:18 31:20,24 88:16,22 109:3 139:4 147:5,13 149:12	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22 -Q- Q.C [415] 1:6,15,20,24 3:10 6:4,12 10:10,18 18:13 19:3,19 30:19 33:10 34:15,21 35:4,9 35:24 36:16,20 38:6 39:2 30:15 24 36:16,20 38:6 39:2	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24 231:12 232:15 233:17 234:8 235:6,19,25 236:19 237:25 238:14 239:1,10 239:16 240:1,8 241:21	$\begin{array}{r} \textbf{quite} [14] \ 3:17 \ 5:2,5 \\ 57:14 \ 70:19,21 \ 72:6 \\ 116:9 \ 143:2 \ 154:13 \\ 155:22 \ 215:17 \ 220:14 \\ 238:12 \\ \textbf{quote} [2] \ 78:15,24 \\ \hline \textbf{-R-} \\ \textbf{radio} [2] \ 90:24,25 \\ \textbf{rafts} [5] \ 166:22 \ 167:16 \\ 167:19,20 \ 251:8 \\ \textbf{raise} [4] \ 5:1 \ 104:11,13 \\ 217:20 \\ \textbf{raised} [10] \ 24:25 \ 28:18 \\ 104:4 \ 132:14 \ 136:23 \\ 154:21 \ 200:20 \ 210:12 \\ 211:6 \ 268:19 \\ \textbf{raises} [2] \ 104:7 \ 162:10 \\ \textbf{range} [6] \ 153:4,5 \ 176:1 \\ 195:7 \ 196:6,7 \\ \textbf{rank} [1] \ 107:9 \\ \textbf{ranks} [1] \ 21:19 \\ \textbf{rate} [2] \ 18:11 \ 269:9 \\ \textbf{rather} [12] \ 8:9,10 \ 13:21 \\ 20:23 \ 52:2 \ 105:16 \ 224:2 \\ 249:22 \ 250:20 \ 251:21 \\ 259:11 \ 262:4 \\ \textbf{rationale} [3] \ 176:5,7 \\ 204:1 \\ \textbf{rationalize} [1] \ 177:1 \\ \textbf{RCMP} [1] \ 263:12 \\ \end{array}$
proceedings [3] 3.1 4.0 5:25 proceedings [2] 9:24 10:6 process [61] 4:15 14:8 14:24 15:3 40:21 69:13 72:13 85:5 92:10 105:25 10:65 109:23 113:9,13 113:21 117:12 133:7,10 133:14 161:15 162:9,18 164:4,16 165:18 179:18 186:24 187:1,3,7 189:10 189:16,18,21 191:3 195:1 195:9 203:6 207:14 211:7 212:3 214:1 218:3,14 219:1,3,6,19 220:21 227:7 230:1 237:22 239:21 240:12,15 241:16 241:20 242:8 243:1,3 269:18 processes [18] 79:13 92:4 93:23 94:4,7,15,18 97:11,19 108:23 116:1 117:22 118:6,10 134:19 209:4,16 224:2 produce [3] 64:4 94:7 106:15 produced [5] 2:10 4:1 48:14 49:1 71:14 Produces [2] 2:1 71:15 produces [2] 2:1 71:15 produces [3] 19:25 28:10 29:8 31:23 32:3 40:2 46:15 48:6 54:25 92:21 141:16 186:23	9.5.3 128.8 143.1 189.8 191:1 project-related [2] 57:5 57:6 projectile [1] 171:3 projects [2] 33:16 108:25 prop [1] 38:21 proper [4] 9:11 214:21 232:6 241:16 properly [7] 208:3,15 214:8 215:12 217:3 242:18 244:18 property [1] 246:5 proposal [2] 187:9,14 proposal [2] 187:9,14 proposition [1] 3:11 property [1] 30:11 protecting [1] 48:6 protecting [1] 48:6 protecting [1] 48:6 protection [5] 47:22 51:17 69:4 197:9 209:2 protective [2] 32:16 201:13 protocol [3] 38:17 161:1 199:14 protocols [1] 234:11 provide [27] 4:2 22:18 31:20,24 88:16,22 109:3 139:4 147:5,13 149:12 151:7 152:5 166:8 179:4	umas [s] 192.17 196.3 196:22 197:13 199:24 pump [s] 104:3 105:5,6 105:12 115:11,16,19,25 pumps [s] 104:1 115:7,7 115:10,14,22,24 116:3 purely [1] 102:11 purpose [12] 13:20 34:17 88:21,24 89:4 96:17 97:10 194:4 222:18 231:14 240:25 258:15 purpose-built [3] 49:3 49:14 56:18 purposes [4] 16:21 89:17 151:25 156:10 pursue [1] 120:15 put [22] 4:19 6:17 7:6,18 7:20 10:12 33:2 43:15 84:25 106:23 114:22 122:17 195:18 211:12 214:14 225:25 235:13,13 242:15 243:20 253:15 260:20 puts [1] 244:16 putting [2] 117:21,22 -Q- Q.C [415] 1:6,15,20,24 3:10 6:4,12 10:10,18 18:13 19:3,19 30:19 3:10 34:15,21 35:4,9 35:24 36:16,20 38:6 39:2 39:15,21 40:13 41:2,7 41:12 10:40:14 21:40:2 8	142:20 143:8,19,24 144:11 145:18 146:3,8 146:14,20 147:10 148:1 148:6,11,17,22 149:3,14 151:2,17 152:18 153:15 153:24 161:18 164:22 165:3 167:2 168:2,9,19 169:10 170:2 171:8,15 171:20,24 172:21 173:2 173:23 175:8,19,23 176:3 176:11,24 177:5,10,14 178:3,11,22 180:16 181:5 181:20 182:6,12 183:2,8 184:23 185:5,10,14 186:1 186:13 188:6,17,23 189:3 190:2,23 191:18 192:1,5 192:10,14,22 193:18 194:10,17 198:3,18 200:8 200:22 201:7,24 202:13 202:21 203:17,24 204:10 204:22 205:3,8,14 206:3 206:10,14,19,23 207:3,7 207:11,15 208:2,6,11,18 209:15 210:2,6,19 211:8 212:20,25 214:2 215:2 215:22 216:13,22 217:5 218:5 219:25 221:2,9,19 222:2,11 223:9,23 224:6 224:13,18 225:2,19 226:4 226:9 227:3,12,25 228:11 228:18 230:2,12,19,24 231:12 232:15 233:17 234:8 235:6,19,25 236:19 237:25 238:14 239:1,10 239:16 240:1,8 241:21 243:8,15 245:6,20 246:24	quite [14] 3:17 5:2,5 57:14 70:19,21 72:6 116:9 143:2 154:13 155:22 215:17 220:14 238:12 quote [2] 78:15,24 -R- radio [2] 90:24,25 rafts [5] 166:22 167:16 167:19,20 251:8 raise [4] 5:1 104:11,13 217:20 raised [10] 24:25 28:18 104:4 132:14 136:23 154:21 200:20 210:12 211:6 268:19 raises [2] 104:7 162:10 ranke [1] 107:9 ranks [1] 21:19 rate [2] 18:11 269:9 rather [12] 8:9,10 13:21 20:23 52:2 105:16 224:2 249:22 250:20 251:21 259:11 262:4 rationale [3] 176:5,7 colspan="2" rationale [3] 176:5,7 20:4:1 rationale [3] 176:5,7 20:4:1 rationale [3] 176:5,7 20:4:1 rationale [3] 176:5,7 20:4:1

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

reach - responsible ffshore Helicopter Safety Inquiry

			Offshore Helic	copter Safety Inquiry
reach [1] 232:21	recovery [8] 95:20 96:19	Registrar [1] 18:1	228:24	115:19 123:25 124:9
reaches [1] 98:10	96:21,23 97:1 109:3	regular [8] 23:12,24 27:9	removal [2] 154:11,19	142:12 152:11 184:1
reactions [1] 93:9	152:15,19	50:19 115:10 116:6	remove [2] 155:2 235:12	189:11 196:11 201:18
read [3] 31:13 144:4	recruitments [1] 246:15	119:13 228:10	render [1] 265:10	208.24 219.13,10 222.19
184:15	recurrent [1] 233:2	regulate [1] 11:2	renewal [2] 69:12 232:25	232:13 234:14 237:8
ready [4] 1:5 154:7 201:8 228:19	red [3] 40:17 110:24 266:3	regulated [6] 59:9,11 59:12,14 62:4 204:19	renewed [2] 74:16 232:25	255:20 requires [6] 74:23 86:6
real [5] 66:4 114:24 131:15 147:5 184:14	redacted [3] 15:13,14 17:4	regulates [2] 10:25 12:16	rental [1] 15:8	86:21 102:22 108:21 217:17
reality [1] 109:7	redaction [4] 14:8,25	regulation [2] 67:16	rep [1] 162:16	requiring [1] 259:23
realize [1] 160:18	14:25 15:4	124:22	repatriation [1] 259:22	rescue [21] 32:25 152:13
realized [1] 259:12	redactions [1] 15:19	regulations [38] 12:18	repatriations [1] 261:13	216:6 229:13 251:7
really [9] 9:8 12:21 38:22	reduce [4] 94:2 97:13	63:23 65:11,13,14,15,16	204:16	252:25 256:13,15 258:13
72:2 79:5 93:18 118:6	108:23 109:9	68:23 71:20 81:9,12,14	renlaced (1) 156.9	258:16,19 259:24 260:23
198:5 261:22	113·2	81:16,20 82:2,12,17,21	replacement (1) 203.10	264:7,18 266:17
rear [1] 53:4	reducing [2] 20.23 109.9	82:23 84:4,5,7,22 88:18	replete (1) 31.4	reserve [2] 169:4 176:4
reason [5] 158:21 170:23	reduction [1] 94:12	88:25 89:4 119:21 125:18	report 181 16:18 17:2	reserves [1] 237:19
197.11 196.8 259.12	redundancy (1) 116:4	250:23	33:8 119:4 147:25 148:2	reservoir [2] 26:18 48:9
109:17	reel [1] 53:14	regulator [2] 187:23,25	164:3 220:23	resistance [1] 213:9
reasonably [1] 175:4	refer [6] 50:3.20 60:8	regulators [2] 23:14	reported [2] 155:9 254:1	resistant [1] 213:6
reasons [6] 15:5 180:5	153:11 174:14 199:15	225:14	reporting [2] 68:14 69:5	resolve [1] 162:12
190:5 193:17 222:4	reference [6] 70:21	regulatory [15] 32:5	reports [1] 138:15	resolved [2] 133:13
231:25	82:12 121:23 176:15	59:16 60:5 61:10,23 63:1	represent [2] 21:13	162:13
reassess [1] 112:8	220:19 246:14	165:12 201:22 226:18	24:17	resolving [1] 136:24
rebid [1] 191:10	referenced [3] 81:15	229:16 234:19	representation [1]	resource [2] 259:14,21
rebreather [1] 219:8	references (8) 61.5 80.4	reinjected [2] 48:13,21	renresentative 61	resources [29] 57:10,15
receive [6] 62:5 84:24	80:8 81:9,12,19 165:12	reinjection [1] 51:22	24:16 126:8,12 215:14	80:17 81:6 145:21 148:24
129:3 108:24 109:15	166:9	reissue [1] 77:19	237:12 242:16	258:23,25 260:3,5,7
received [3] 4:4 16:6	referred [7] 6:14,14,16	relate [1] 52:6	representatives [10]	265:3,6,16,22 266:10,13
69:21	33:22 163:5 243:10,10	related [15] 4:24 32:17	11:3 12:24 60:2,15 65:5	266:16,20,25 267:3,16
receivers [1] 251:9	refitted [2] 241:14,20	44:21 61:5 63:11 68:15	228:9	207.19 200.4,5
receiving [1] 72:18	reflect [3] 108:11 134:17	126:17 134:8 165:12	represented [4] 126:1	53:19 122:23 198:9
recent [3] 121:21 170:15	reflection (1) 112.12	220:12 234:14	128:22,23 139:6	249:17 268:4
171:3	refocussed [1] 234.5	relates [1] 149:25	represents [4] 1:10 5:4	respecting [4] 61:4 70:9
recently [3] 7:17 133:24	refresh (1) 202.1	relating [1] 237:21	125:7 236:9	152:4 166:2
202:11 DECESS m 152:21	refresher [3] 116.17 20	relation [3] 30:23 40:19	request [3] 81:5 254:5	respective [10] 33:25
RECESS [1] 155:21	116:24	134:2	200:0	80:15 159:10 140:9 165:19 195:6 8 196:25
recognise (1) 20:25	refuel [3] 95:23 160:15	relationships [1] 78:20	72:19 136:5 145:16	236:3 268:2
recognize [1] 20.23	177:6	relative [3] 114:7,25	requesting [1] 214:23	respond [1] 6:6
25:23 100:14 111:4 117:7	refuelled [2] 244:12,13	relatively 151 40.18	require [11] 20:16 56:5	responds [1] 11:16
143:14 159:7 205:17	refuelling [5] 157:10,11	47:19 115:15 116:11,23	56:6,17 75:18,21 76:22	response [39] 7:10 8:24
recognized [7] 86:9	refuence (2) 161.25 162.5	relevance [1] 15:12	81:3 152:4 183:21 201:17	83:19 152:12 166:14,16
90:11 105:18 118:19	163:9	relevant [5] 10:4 15:15	required [33] 9:15 23:5	246:6.13.15.20 247:8
recognizing (1) 260.13	refuse [7] 123:21 133:4	15:16 242:2 244:4	80:17 83:15,18 86:18	249:4 251:19 252:15
recommendation [5]	133:10 162:21 163:1	reliable [2] 20:15 201:4	121:20 129:8 152:23	254:15,16 257:14 258:3
188:1,4 195:14 196:14	164:6 242:8	relied [1] 147:14	160:7 166:15 174:18	258:9 259:9 260:1,8
219:9	refused [1] 133:20	remain [5] 89:6 92:20	218.20 220.14 222.1	264:22,25 266:11,20,21
recommendations [4]	refuses [2] 162:10 242:5	95.11 209.22,25	226:14,20 227:17 232:23	268:2,6,11
33:6 92:15 187:22 220:23	regarding [5] 51:17	remains [2] 12:7 86:23	240:17,19 244:11 245:4	responsibilities [8]
recommended [2]	248:16	88:24	250:2 259:13	23:20 32:25 36:3 93:10
recommending [1]	Regardless [1] 30:7	remarks [1] 33:9	requirement [19] 72:15	266:6
195:10	regime [2] 10:24 129:11	remember [2] 6:18 7:24	150:2,5 189:15 190:6,10	responsibility [16] 11:5
record [1] 196:13	regimes [1] 61:10	Remembering [1] 13:20	203:3 204:17,19 209:5	31:18 78:4 91:7,19,21
recording [1] 101:9	region [5] 259:7 263:24	remind [3] 10:20 40:15	248:19 250:1 253:3 254:6	91:22 99:4 142:16,22,25
recordings [2] 101:20	263:25 264:3 266:2	262:25	requirements and 50.14	266:3
102:10	regions [4] 59:9 259:4	reminded [1] 154:9	61:23,25 63:1.3 69:1	responsible [20] 22:10
records [2] 236:5,12	203:22 204:18	reminding [1] 5:3	70:14 75:24 81:11,22	22:20 23:3 29:24 60:3
recovered [1] 152:14	90:16	remote [3] 155:14,20	84:23 90:25 100:19,20	60:13 73:4 78:1,6 89:19

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

responsiveness - scheduling Offshore Helicopter Safety Inquiry

			Olishore men	opter Safety Inquiry
98:5 119:5 139:21,23	154:4 155:12 159:1 161:7	194:10,17 198:3,18 200:8		100:12,16 105:7,10,23
142:10 235:23 236:4	164:10 220:10 222:24	200:22 201:7,24 202:13	S	106:14 133:9
231.10 239.3 203.25	rieks (7) 20.21 100.2	202.21 205.17,24 204.10	S [4] 170:20 192:15 260:8	sately [6] 21:11 89:5
rest (2) 50.20 155.8	110:23.23 133:14 219:11	206:10,14,19,23 207:3,7	267:9	90.14,22 99.13 229.7
restraints (1) 166.21	220:13	207:11,15 208:2,6,11,18	S-61 [5] 167:18 191:16	11:16 12:14 13:16 15:22
restrict (1) 156.7	risky [1] 112:2	209:15 210:2,6,19 211:8	192:9 193:1 196:5	16:2 20:13,19 21:2,4,5,9
$result_{11}$ 164.0	road [1] 149:10	212.20,25 214.2 215.2	S-61S [2] 196:22 197:14	22:21 23:10 27:16 29:23
$resume_{11} 184.0$	Robbie [1] 134:10	218:5 219:25 221:2,9,19	5-92 [27] 165:17,19,21	29:25 30:2 32:7,9 44:21
retrieving (1) 152.17	robust [1] 114:3	222:2,11 223:9,23 224:6	192:9.20 195:10.19 196:2	68:13.15 69:3 76:2.5.10
refurn [2] 17/1 100.1/	Roil [424] 1:4,6,15 6:4	224:13,18 225:2,19 226:4	196:3,15,20,24 197:4,5	76:15,22 77:2,3,4,7,10
249:21	6:11,12 10:10,18 18:13	228:18 230:2.12.19.24	197:8 198:9,17,24 199:25	77:11,14,18,19 78:2,6
returned [4] 29:12	19:3,19 24:21 30:18,19	231:12 232:15 233:17	200:15 223:16,21 251:6	/8:15,16,1/,1/,22,24
164:18 215:13 243:23	35:4.9.24 36:16.20 38:6	234:8 235:6,19,25 236:19	$S-92\Delta$ [7] 101.11 12	84:8 85:2 90:23 91:15
returning [2] 4:3 245:17	39:2,15,21 40:13 41:2,7	237:25 238:14 239:1,10	193:7 194:20 195:14.22	91:18,21 92:2,10 93:5
revalidated [1] 77:12	41:12,19 42:1,21 43:3,8	243:8,15 245:6,20 246:24	197:2	93:21 97:3 98:6 102:2
review [20] 32:18 100:7	43:25 44:17,25 45:5,9	247:6,12 248:2,9,23	S-92As [1] 196:22	102:18 108:22 118:11
100:8,9 107:3 132:12,14	51:3 52:4.10.18 53:3.9	249:12 250:5,11 251:17	S-92s [2] 267:5 268:15	123:22.24 124:3.13.23
138:15 154:24,25 156:2	53:21 54:16 55:18,24	252:6,12,19 253:11,17	S61 [1] 199:23	125:15 126:3 127:21,23
220:19.22 221:17 225:16	56:21 57:20 58:10,24	255:21 256:5.9.19 257:4	Sable [1] 29:17	129:2,5,13 132:23 133:2
231:8	60:7,17,23 61:12,19 62:1	257:9,21 258:6 260:18	Sacuta [224] 18:15,16	133:3 134:17,18 135:6,8
reviewed [7] 100:9	64:13.18.23 65:3.8.24	261:3,16 262:6,16 263:2	19:2,15,17 26:5 27:20	138:5 139:12.13.15.22
105:14 155:11 156:12	66:14,19,24 67:3,8,12	264:1,6,10,15 267:13	27:21 30:16,24 31:7,15	141:17 142:24,24 152:7
162:22 188:4 219:11	67:18,23 68:4,9 69:15	role (11) 26:3 14 14 16	36:7 37:1.13.18 38:7.8	159:2,4,9,16,23 162:16
reviews [5] 92:10,10	69:24 70:23 71:4,8,16	29:21 88:15 92:2 94:4	39:8,17,24 40:22 41:5,9	162:20 163:7 165:24
97.11 120.20 195.21	73:24 74:5.9.18.25 75:4	120:4 234:25 268:8	41:17,23 42:3,25 43:6	198:10 200:14 201:15
227:1	75:8,14 76:8,13,18,24	roles [2] 78:19 235:5	43:10 44:2,19 45:3,7,12	213:16 219:18 220:12
revisit (1) 191:8	77:21 78:8 79:1,9 80:12	roll [2] 136:12,13	60:12.21.25 61:17.22	226:21 227:23 234:15,16
revoke (1) 63:21	80:19,23 81:23 82:10	room [9] 10:20 14:20	62:3,11,16,20,25 63:8	250:24 252:22 253:6,8
revolve (1) 54:2	85:12.16 86:1.25 87:7	16:6 17:9 120:15 170:10	64:7,15,21,25 65:6,10	255.10,20
rid [2] 174:18 184:10	87:12,18,22 88:9 89:10	173:4 225:4 243:25	66:7 67:25 68:7,11 69:20	sail [2] 23.17 34.2
rig [8] 40:10 45:15.17.20	89:16,22 90:1 91:4,10	ropes [1] 38:19	72:16 73:3.6.18 74:3.7	sancu [1] 22.15
45:22 55:10 73:21 145:4	91:14 94:13,23 99:1,20	ROS [2] 90:12,13	74:11,21 75:2,6,12,16	SAR [20] 185:4 240:4 7
rig-share [1] 145:10	105:1.22 107:8 110:8.15	KOSE [26] 23:17,18 34:6	76:11,16,20 77:1,22 78:5	249:11 250:1.25 251:1.3
Rigging [2] 120:21	110:21 111:9,23 112:10	51:8.10.25 53:19 54:12	78:14 79:7,11 80:14,21	251:5,10,10,11,13,16,19
121:13	112:17 113:19 114:11	54:14 86:15 87:14 145:6	84:1.19 85:10.14.18 86:4	252:5 254:6,11,23 256:22
right [46] 17:23 18:21	117:17,25 119:7,12,17	145:7 150:20 151:16	87:3,9,15,20 88:3,7,14	256:23 257:13,17 258:2
41:13,18 45:23 46:6 50:7	120:13,24 121.9 122.1,8	154:10 180:14 186:5	89:11,14,20,24 90:4 91:8	260:1 263:6.17.20.23
113:24 123:21 128:9	125:2,11,16,23 126:7	rotate (2) 50.5 184.21	91:12 123:2,8,14 132:10	265:1,3,16 266:10 267:7
132:18,21,24 133:4,10	127:4,16,25 128:12,18	rotating 11 20.0	161:8.20.23 162:8 172:4	sat [2] 119:8,13
150:10 162:21 163:1	129:9 130:7,19,25 131:4	rotation [11] 23.25 27.9	182:20,22 183:5 200:16	satellite [2] 217:15
164:6 167:20 178:4	133:18 134:1.6 135:17	28:25 50:19 54:9 128:25	201:9,11 202:4,17,23	248:20
194:6 200:6 203:23 205:7	136:9 137:5,19,25 138:9	129:15,20,24 130:14	203:22 204:2,14,24 205:6	satisfaction [2] 162:14
213:6 214:6 215:18	138:16 139:20 140:11,18	138:4	206:21 207:1,5,9,13,24	163:3
216:23 218:6 220:16	140:24 141:10,19 142:14	rotational [1] 29:6	208:4,9,13,20 209:25	Saustieu [5] 89:2 127:18
224:12 231:13 237:19 242:5 8 247:13 248:24	144:11 145:18 146:3,8	round [1] 251:4	210:4,11 211:4,10 212:22	satisfy (1) 183.25
269:17	146:14,20 147:10 148:1	route [2] 120:12 147:7	212:23 213:7 214:9 215:9	saving [1] 204.7
right-hand [2] 75:22	148:6,11,17,22 149:3,14	routine [5] 158:24 160:2	218:8 220:2 221:6,14,24	Saw $[2]$ 110.9 261.4
101:23	151:2,1/152:18 153:15	160:8,20 161:6	222:9,14 223:11 224:4	Savs [7] 7.7 167.25
rights [2] 34:3 132:20	161:18 164:22 165:3	Power 41, 55, 00	224:11,16,22 225:8,22	168:16 186:15 241:17
rigs [14] 43:18 44:7 45:2	167:2 168:2,9,19 169:10	KOWAII [1] 55:22	226:7,11 227:9,14 228:8	252:22 256:2
45:2,14,16,17,25 55:19	170:2 171:8,15,20,24	$\frac{1000}{10} \frac{11}{10} \frac{12}{10} $	232:1,17 233:22 234:12	scale [1] 190:14
50:22 145:24 140:1 150:12 193:13	172:21 173:2,23 175:8	ruling (1) 164.11	235:10,22 236:2,24 238:4	scenarios [1] 246:2
rise 11 136:21	177:5,10,14 178:3,11,22	run (2) 20.17 111.25	238:11,17 239:4,14,18	scene [2] 258:5 262:3
risk [34] 20:24 30:5 77:9	180:16 181:5,20 182:6	222:4	240.0,10 241.24 243:13	schedule [9] 2:13,16,24
79:21 92:17,18 94:2,2	182:12 183:2,8 184:20	running [3] 154:6 249:25	saddened [11 20:2	3:13 4:1 5:24 7:9 240:14
94:12 97:13 102:19	186:13 188:6.17.23 189:3	250:21	safe [17] 20:14 22:24	scheduled 121 06.10
106:16 108:19,20,23	190:2,23 191:18 192:1,5	Russia [1] 37:4	27:16 30:12 77:10 79:13	243:5
111:6 112:7.20.22 113:1	192:10,14,22 193:18		79:18 90:23 98:4 99:24	scheduling [2] 237:5
				B

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

scope - specialists **Offshore Helicopter Safety Inquiry**

245:4 250:15 scope [29] 50:22 59:23 84:20 100:1,11 101:11 102:12,25 103:6,24 104:5 104:12.17 105:4.8 106:3 106:8,9,12,19 107:17,19 107:21 145:13 154:9 155:24 187:8 189:11 202:24 score [1] 209:12 Scotia [4] 25:8 29:17 217:8,14 screening [3] 179:10,15 219:18 sea [38] 21:18 23:4,7,17 24:7 29:16 35:20 37:6,6 37:8,12 38:2,5,12 39:11 39:13 44:11 50:4.7 51:25 53:19 54:11,14 70:20 145:6 147:8 151:15 154:10 166:20 175:3 180:14 216:3 233:11 234:2,3 237:16 267:24 268:1 seabed [1] 55:11 Seaboard [1] 49:7 seal [6] 202:19 212:11,12 214:16,21 232:6 Sealand [1] 188:15 sealed [1] 240:21 sealing [1] 214:25 54:18 search [15] 32:24 241:6 251:7 252:24 256:13,15 258:16,19 259:24 260:23 262:10,14 263:10,14 266:17 seas [1] 36:12 seatbelt [5] 242:16,17 213:13 242:21 244:16,18 seated [6] 18:14,16,18 18:21 242:13 244:8 seats [1] 223:21 269:1 seawater [1] 48:24 second [10] 7:7 44:4 94:15 46:14 49:5 130:10,18,20 131:5,12 220:25 secondary [1] 265:5 seconded [2] 120:2 141:25 239:25 secondment [2] 25:22 26:2 Secretariat [2] 263:10 263:15 section [28] 31:20,24 32:4,11,14,15,18,20,23 33:1,5 54:21 58:23 59:15 138:24 139:3 162:3 165:9 189:14 195:17 198:1 201:9,12,22 221:25 224:25 238:22 259:1 sections [5] 19:6,8 31:17 31:19 32:6 sector [1] 29:15 sectors [1] 111:22 Secunda [1] 149:22 144:3,5,9 146:24 149:7 secure [3] 197:22 245:1

secured [1] 245:14 security [10] 179:10,15 241:6 244:22 245:3,5,8 245:10,18 246:5 see [47] 15:18 17:3 26:20 30:10 37:10 40:15 42:18 43:22 46:21 52:20,20,25 53:13 55:8,10 58:3,5 61:14 62:9 75:19 79:5 81:18 97:17 101:15,23 102:4.13 109:12 110:5 113:11.20 116:8 117:1 129:25 134:24 149:15 151:3,18,22 156:2 159:15 202:15 212:13 231:19 248:6 253:2 260:23 seem [1] 8:3 sees [1] 237:23 segregated [1] 117:19 select [1] 95:3 selected [4] 159:22 191:4 191:13 220:6 **selection** [11] 97:22 165:14,17 186:11,24 189:6 191:11 195:14 208:21 212:3 219:3 semisubmersible [2] 47:13 56:9 semisubmersibles [1] send [1] 58:18 senior [5] 24:15 65:5 71:24 135:9 237:11 sense [1] 102:13 sensitive [1] 15:6 sent [3] 7:16 178:10 separate [11] 13:14,15 13:16 16:20 47:4,15 57:21,24 142:7 192:2 **separated** [2] 48:12 separately [3] 13:17 34:12 59:20 **separation** [1] 12:22 sequence [3] 160:6,25 series [2] 16:15 48:11 served [1] 226:5 service [31] 142:5,8,9 144:5 145:4 146:13,18 146:22 147:15 148:14 151:7 165:15 180:24 185:3 186:11,25 187:4 187:16,19,21 188:11 189:23 195:19 218:4 229:13,22 231:10 249:2 249:11,21 263:15 serviceable [1] 249:10 services [37] 13:17 22:17 23:2 32:12 43:19 63:4 128:13 138:20,23,25 139:5 140:8 142:3 143:16

149:12 150:8 165:8,14 179:3,3,9 180:2,6,9 189:10,13 191:3 229:19 229:25 231:6 256:15 263:13 session [2] 6:9 144:9 set [10] 17:10 43:2 53:5 76:4 127:7 160:6,25 190:15 237:8 265:20 sets [2] 98:7 184:8 setting [3] 98:2 99:5 227:7 seven [3] 28:5 112:21,21 several [3] 27:6 165:25 195:25 shafts [1] 44:6 shall [3] 63:17 72:10 77:2 shallow [2] 47:19,21 share [8] 13:2 15:11 80:17 144:19 145:20 150:16 181:4,23 shared [3] 59:1 187:22 190:22 shares [1] 34:12 sharing [2] 57:12 186:6 shed [1] 194:16 sheet [1] 244:24 sheets [1] 174:21 Shetland [1] 37:16 shift [3] 138:3,3,5 **ship** [7] 53:4 56:11 74:1 74:2 87:23 88:1,10 shipboard [1] 91:1 **shipment** [1] 44:15 **Shipping** [8] 21:17 88:6 88:8,10,12 90:11,16 265:8 ships [3] 22:3 54:19 70:10 **Shipyard** [1] 22:5 shore [3] 149:6 231:7 234:17 **shore-based** [4] 144:5 144:14 149:8 166:25 **short** [4] 35:12 74:22 145:19 211:16 shorter [2] 74:20,24 **shortly** [2] 99:19 163:9 **shoulds** [2] 167:7,12 **show** [6] 12:10 42:22 50:2 78:17 92:1 135:5 shown [5] 56:11 113:14 201:20 242:1 244:3 **shows** [10] 41:4 49:9,10 56:8 57:10 75:17 92:1 103:9 241:17 247:17 **shuttle** [3] 58:4,6,8 **sic** [1] 227:18 side [15] 25:16,20 45:20 52:17,17 53:2,6 75:20 75:22 76:22 116:16 117:20 118:16 120:7 251:16 sign [1] 245:14

signal [2] 216:6,12 signed [1] 90:8 significant [9] 38:9 94:3 139:11 149:13,24 155:23 193:11 258:16 269:11 signs [1] 244:23 **Sikorsky** [12] 163:13 164:1 167:18 191:12,16 191:21 192:9,11 193:1,7 195:5.10 Sikorsky's [2] 178:10 194:25 similar [13] 24:9 37:4 47:18,23 49:1 52:1,3 53:16 54:12 110:9 124:18 153:7 187:1 **similarly** [1] 2:23 **simply** [13] 7:1,20 8:6 15:1 54:2 57:15 71:25 82:14 132:23 204:12 220:14 269:6,21 simulators [1] 195:8 single [4] 45:24 60:10 206:11 258:15 sit [5] 9:25 26:25 128:10 130:18 176:21 site [8] 22:12 54:12 91:24 106:14 155:11 228:22 229:5 245:14 sites [3] 159:12,12,23 sits [2] 42:23 212:14 situation [17] 3:25 55:14 80:18 84:11 86:19 111:25 133:20 222:17.22 223:5 231:23 234:21 238:1.6 242:4 254:18 259:25 situations [15] 81:3 103:17,19,23 109:3,5 124:16 161:10 163:24 205:20 223:7 232:6 233:3 243:22 246:10 six [4] 24:1 42:8 131:24 233:6 size [19] 16:19 49:19 114:8 115:1 126:25 127:2 128:1,5 170:18 182:14 195:7 215:3,8,15 223:20 241:11,13,23,25 sized [1] 214:15 sizes [1] 214:24 skill [1] 189:22 sky [4] 247:22 248:6,12 252:16 **slice** [2] 93:16 113:12 slices [2] 113:14 114:5 **slide** [42] 33:12 36:13 40:15 56:7 57:10 61:1 62:10 64:8 75:17,23 76:22 86:2 94:9 101:23 101:24 102:15 103:9 108:19 109:22 114:1 135:5 140:12 154:5 164:25 166:8 168:22,22 186:10,21 189:14 201:21 203:18 215:24 228:20 236:18 239:21 243:2

247:17 251:3 255:19

266:1.19 **slides** [9] 39:10,23 55:8 59:15 76:4 91:25 113:8 117:19 122:25 slightly [4] 52:16 53:5 98:1 116:25 **Slinging** [2] 120:21 121:14 **slipped** [1] 167:3 **slot** [1] 243:20 **slots** [3] 45:16,18,18 slow [2] 99:2 169:11 small [7] 16:10 93:25 115:15 116:11 190:12 242:25 262:18 smaller [4] 38:22 116:22 179:24 215:6 **SNOW** [1] 37:5 **software** [1] 178:9 **Sokland** [1] 37:3 solve [2] 8:2 10:2 **someone** [2] 108:11 134:22 **sometime** [1] 249:13 sometimes [6] 77:16 100:4 140:3 145:19,20 235:14 somewhat [3] 24:9 37:11 230:16 **somewhere** [5] 41:22 96:4.6 182:2 230:10 **son** [1] 24:5 Sooley [1] 271:11 **soon** [2] 154:6 239:9 **sorry** [10] 8:10 17:20 47:12 66:20 85:9 174:1 191:19 204:25 207:16 253:12 sort [8] 16:19 88:1 135:20 168:14 183:18 184:20 209:19 210:23 sought [1] 4:10 sound [1] 271:7 sounds [1] 168:10 sources [2] 148:19 163:23 **South** [1] 23:5 southern [3] 37:12 38:11 40:24 southwest [2] 41:1 43:22 **space** [3] 100:16 150:23 190:15 speak [7] 19:14 154:4 160:22 165:10 170:14 196:16 198:13 **speaking** [7] 2:7 58:14 118:20 213:12 230:11 238:5 247:7 **special** [4] 86:18,22 97:3 101:4**specialist** [3] 200:18 251:1 258:18 **specialists** [2] 251:13

Discoveries Unlimited Inc., Ph: (709)437-5028

Index Page 19

258:16

$\boldsymbol{Multi-Page}^{^{\mathrm{TM}}}$

specialized - tasks Offshore Helicopter Safety Inquiry

			Olishore Heik	opter barety inqui
specialized [3] 179:6	152:10,15,16 249:4	structure [13] 43:1 44:5	suited [1] 231:9	survival [6] 211:21
180:9 229:21	252:24 259:16	55:12 56:15 79:19,20,25	suits [30] 81:10,13,15,18	222:16 232:24 233:1,4
specializing [1] 25:3	standing [2] 3:3 52:21	87.1 91.25 98.5 101.10 184.9 246.14	81:22,25 82:25 121:24	234:0
specific [35] 32:19 34:8	stands [2] 70:11,18	structures (1) 55·3	202:8 12 25 203:4 204:5	survivore (1) 11.7
81.9 82.12 90.11 100.17	start [8] 25:25 40:5 54:15	study (1) 157.1	204:8,18 205:2,5 209:14	Survivors[1] 2:21
100:18,19 101:18,22	250.17	style (2) 56.12 219.9	211:15,18,24 213:22,23	suspect [1] 252:7
118:20 126:25 129:4,6	started 151 11.25 22.1	styles (1) 54.23	214:24 216:1 217:2	Suspend [1] 63:21
133:2 159:23 161:10	28:2 31:13 203:5	stynes[1] 54.25	Sumatra [1] 28:7	SWISS [4] 93:15 113:12
166:3,9 176:15 180:23	starting [1] 99:4	sub-committees [3]	summary [5] 162:4	Sworn [2] 18.23.25
220:7 223:7,14 225:13	starts [5] 17:18 98:6	119:3,14,16	164:3 239:20,21 243:2	system [30] 21.24 44.14
229:15 238:13 259:24	106:25 173:17 255:19	subcentres [2] 264:19	Summer [1] 29:3	53:16 70:8 77:7,15 78:18
specifically [5] 32:17	state [2] 78:16 92:18	264:21	Suncor [18] 11:22 18:22	92:19 95:24 96:10 97:25
81:15 160:14 170:3	statement [6] 98:13	subject [7] 102:23 120:1	46:19 50:2 51:9 59:19	98:6,16,17 100:3 104:2
1/1:13	99:11,13,15,24 208:14	121:4 140:12,15 156:15	67:11,13 140:1,5 145:9	107:22,24 115:4,23
specification [1] 255:24	statements [1] 61:14	15/:6	145:16 149:21 198:13	177:25 178:2 181:7 197:8
specified [2] 191:6 204:4	states [3] 147:8 166:21	Subjects [1] 19:14	Suncor's [1] 140:1	201:6 203:12 227:11
specify [1] 73:8	1/5:3	Submerged [3] 216:9,11	Suncore [2] 230:22	247:22,24,25 248:7,8,14
speculate [1] 261:22	stating [1] 98:22	submission (2) 187.8	238:10 Summer 101 14 24	248:16 252:17,18
speculation [1] 261:11	stationed [1] 151:19	253:8	Super [11] 191:14,24	systematic [1] 98:3
Speed [4] 147:8 196:6	statistically [1] 1/5:1/	submit [1] 76:1	197:13 199:23 223:15	Systems [21] 32:8 43:18 79:13 96:16 24 102:3
speeds 111 25:16	Statistics [2] 135:11	subsea [4] 47:3.21 48:7	224:7,9	107:17 133:3 134:16
specus [1] 55.10	Statoil (11) 12:1 2 25:23	51:18	supervise [1] 142:16	139:14,14,15,22 140:7
169:17	41:15 42:14 46:22 145:8	subsequent [3] 39:10	supervising [1] 29:7	169:9 177:19 234:24
spent [4] 25:16 27:3 28:8	145:17 182:23,24 183:1	130:14 216:20	supervision [7] 100:10	236:12,21 248:10,12
28:22	StatoilHydro [1] 41:14	subsequently [1] 164:13	107:18 133:11 135:15	
spirit [2] 30:2 184:15	status [2] 18:2 33:6	substandard [1] 200:2	136:19 141:25 155:10	-1-
splash [1] 212:15	statutory [3] 85:19 90:6	Substantially [1] 122:6	supervisor [4] 26:4	T [1] 243:18
splitting [1] 31:18	90:9	subtle [2] 52:2 231:19	Supervisors (1) 26.6	table [1] 118:22
spoke [1] 51:17	stay [2] 12:6 217:24	success [1] 30:13	supervisory [3] 28.12	tabs [2] 16:20,20
spool [3] 154:11,19 155:2	stayed [1] 29:10	successful [4] 222:20	29:4 141:23	tag [1] 105:5
spot [1] 7:8	step [12] 75:19 94:20	223:18 258:19,21	supplier [3] 140:9 142:6	tailored [1] 222:17
spouse [1] 93:7	102:17 103:20 105:8	SUCH [23] 18:12 53:24 59:19 20 80:20 103:2	209:13	takeoff [1] 160:11
spreading [1] 252:1	161:15 239:22	109:7 121:12.24 130:1	suppliers [2] 140:4	takes [8] 15:4 30:8 91:22
Squadron [1] 174:25	steps [8] 110:23 164:4	139:11 147:24 182:1	209:8	114:1 1/6:1/ 1/8:8 182:1
square [2] 112:21 266:4	181:14 223:4,17 224:3	229:2 250:15 251:7 260:3	supply [18] 49:10 138:23	taking (6) 151.5 166.10
St [20] 24:8,12 25:15	238:24 239:25	260:5 265:3,6,17,21	149:15,21 151:11,11	185:6 241:2 245:15 268:7
28:15,18 29:10 147:4	stern [3] 52:14 53:13	sufficient (3) 126.20	209:21 211:14,16 218:11	talks [4] 64:9 86:2 168:23
149:10 1/2:10 1/3:19	95:10 still top 22:25 (C.9 72:4	127:2 231:11	249:1 262:13 267:14	179:18
262:3 264:20,24 268:8	82·14 86·23 91·2 98·19	suggest [2] 9:22 10:7	support [24] 30:4 57:3,9	tandem [1] 256:4
271:5,9	111:4 118:22 173:21	suggesting [1] 210:20	57:21,24 81:4 99:23	tank [11] 100:11,13,15
staff [1] 78:22	256:25 257:7	suggestions [1] 72:1	149:12 150:4.8 151:11	100:19,21,25 101:12
stage [2] 188:2 209:19	stimulates [1] 103:4	suggests [1] 128:6	151:12 152:25 158:4	223:20
stages [2] 145:1 218:16	stood [1] 28:25	suit [67] 70:8 82:13	159:15 193:10 196:12	tanker (1) 52:15
stake [1] 51:14	stop [9] 35:25 63:25 85:9	122:14,20,23,24 202:6,6	263:15 266:16 267:14	tankers [10] 21:18 44:15
stakeholders [1] 21:1	112:8 120:14 190:3	202:14,16,18,19 203:8	supported [3] 106:19	49:3,6 53:12 57:10 58:5
stand [6] 2:2 4:4 106:15	stong (2) 155.26	203.10,12,14 204.0,11	supports 11 160.0	58:6,9,18
156:19 182:19 259:14	stops[2] 155.5,0	206:6 207:20 208:15,21	supports[1] 100.9	Tara [1] 271:5
standard [26] 71:12	46:25 95:24 96:3.5	208:25 209:6,6,11 211:23	suppose [1] 210.23	target [1] 220:24
158:13 167:24 197:7	store 111 50:9	212:7,9,11,15,17 213:5	162:16 199:8	task [22] 33:2,8 100:4,5
203:1,2,14,15 204:6,9	storevs [1] 49:22	213.20,23 214.7,10,13	surface [4] 47:25 216:7	107:3,25 118:23,25 119:4
204:11,13 206:11,15	strategies [1] 38:15	215:17,18 218:21 232:5	216:11 217:4	160:5.8 161:13.13 162:3
207:22,25 208:1 226:12	strategy [1] 211:22	241:9,10,12,16,19,22,25	surpassed [1] 199:1	265:23 268:18,25
standards 110 70.17	stream [1] 197:16	242:10,12,19,23 243:25	surprise [1] 4:8	Taskforce [2] 16:18 17:1
82:20 95:14 96:11 121:13	Street 11 271:5	suitable 111.2 160.19	surprised [1] 8:15	tasking [1] 265:2
207:21 210:8,24 233:11	strengths III 153:9	194:22	surprising [1] 3:17	tasks [19] 91:24 107:23
236:9	strike [1] 197:9	suitably [1] 168:25	surveillance [2] 38:18	133:1 135:24 158:7,24
standby [19] 49:9 58:1,3	strive [1] 246:2	suite [3] 57:9 103:11	229:14	158:25,25 160:1,2,3,19
58:4 80:9 150:2 151:4,5	stroking 11 223-21	271:5	surveys [5] 85:6,13,15	161:7
131.13,13,22,24 132.9			85:21 90:15	

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

team - two Offshore Helicopter Safety Inquiry

team [19] 93:8,11 107:6	268:11	through [60] 2:14 16:13	top [4] 91:23 98:9 105:9	transport [42] 11:3
107:13 108:15 116:15.16	testing [5] 122.14 15 22	17:4.18.18.23 21:19 26:9	126:4	15:22 56:24 61:8.21 62:5
116:19.20.23.133:6	122.4.12	27:13 45:24 48:11 52:1	topics (2) 121.17 127.16	62:6.15 63:3 70:9.11
157.22 158.4 159.13	125:4,12	59.2 62.6 7 72.13 75.19		77.10 85.6 13 15 89.25
187.11 189.19 235.9 9	Texaco [1] 46:23	91.23 25 93.12 94.21	topsides [1] 43:17	90.7 14 20 20 91.3 5
244.17	thank [38] 1:7 6:3 10:9	99.18 103.21 105.8 108.5	Torbay [1] 182:7	95.7 142.13 148.15
244.17	10:11 27:17.19 30:15	111.20 114.4 21 24	torso (1) 166.21	140.10 150.19 152.1 6
teams [4] 235:3,3,11	33.11 38.7 50.24 52.11	111.20 114.4,21,24		149.10 150.18 152.1,0
268:2	63.7 70.1 73.3 76.25		total [2] 174:4,6	100:12,13 109:1,19,24
technical [7] 141:2.17	83.11 128.10 132.18	138:20 140:9 141:15	totality [1] 89:18	1/9:23 203:1,13,15 206:7
184:12 187:13 195:4	142.25 140.4 152.25	142:6 143:4,6 147:6	totally 111.10	208:24 240:22 263:12
209:9.11	143.23 147.4 133.23	164:1,15 174:22 179:15		transportation [43] 5:1
toohnigion (1) 25%0	164.21 179.24 100.24	195:1,12 201:10,19	toucn [1] 8:13	15:21 16:2 20:15 21:15
	104.21 178.24 190.24	212:18 218:3 223:13	tow [1] 176:21	22:3,22 23:6 30:6 32:18
technicians [3] 166:15	194:18 202:3 208:19	224:3 225:17 227:22	towards 121 103.8 105.11	49:13 56:6,20 63:13 70:8
251:10,20	211:9 239:17 245:21	240:4 241:5,8 242:10,10	110.5	76:7 81:13.15.18.21.22
techniques (11, 129:6	256:10 263:3 264:16	245:8 253:8		81:24 82:13.24 121:23
technologies (1) 107-24	270:1	throughout [4] 22:14	towing [2] 38:19 150:4	122:7 134:3.7.22 135:2
technologies [1] 197:24	that'll [1] 180:7	24.2 31.19 219.19	town [10] 135:9 136:25	164.10 169.20 21 201.14
technology [7] 4:21	theirs (1) 183-1	thmustoms w 54.2	137:3,6,9,10,14 199:13	201.18 203.12 205.19 22
178:17 193:3 197:6 199:3	thems [1] 105.1	Infusiers [1] 54:3	199:14 245:17	200.14 213.22 232.5
218:2 219:22	tnemselves [9] 18:8 54:5	tieback [1] 51:13	TP _[1] 70.10	209.14 215.22 252.5
techs (11 251·10	107:15 135:10 136:22	tied (3) 47.16 249.14 15		234.20 249.2
tolonhono 42, 120, 14	137:18 159:9 197:19	tichtly and 104 14	TP-7920 [1] 152:9	transported [1] 174:15
telephone [1] 138:14	236:22	ugnuy [1] 184:14	track [2] 109:17 196:12	transporting [11 234:21]
television [1] 20:4	thereafter [3] 28:11	timeframes [1] 234:11	trooked in 217.15	Transchinmont
telling [3] 69·18 140·14	258.14.260.9	timeliness 121 8.4 7		11 anssiipment [2]
143.9	themefore in 56 (10	timely 42, 190.2	tracking [5] 236:12,21	44:10 58:20
145.7	therefore [6] 56:6,19	timely [1] 180:3	247:22 248:6 252:17	travel [22] 20:21 21:10
temperatures [3] 35:19	97:1116:6126:22136:16	times [15] 24:1,4 27:5,7	tracks [1] 236:21	24:1 29:21 37:23 38:3
35:20 36:12	thermal [5] 207:18,21	29:22 84:14 121:22	traditional (1) 144.24	56:22 226:22 227:16,23
ten [3] 26:7 145:2 168:13	207:22 208:8 209:2	126:14 150:6 168:13		228:10 229:4 231:18,24
tend (1) 1/4.10	they've [2] 100.18	171:4 178:13 180:14	tragedy [1] 27:12	233:12 236:16 237:14
	225.25.228.20	266:20.21	tragic [1] 268:12	239:11.13.15 240:4 241:5
tender [1] 143:4	223.23 238.20	ting (1) 180.12	train (2) 210.14 222.10	trovollod (5) 22.15 24.2
tentative [1] 7:9	thickness [2] 101:10,12	ups[1] 180:12	11 alli [3] 219:14 225:19	229 2 7 222 9
term (9) 31.8 33.73 77.17	thin [1] 252:1	to-day [1] 106:13	251:14	228:3,7 233:8
74.22 89.22 129.22 100.6	third (11) 15.17 16.22	to-work [2] 98:24 107:22	trained [8] 96:15,24	travelling [3] 218:18
74:22 88:25 128:22 190:0	17.5 40.25 89.10 120.21	today (27) 0.24 12.22	116:16 117:14 157:17	236:15 237:9
211:16	17:5 49:25 88:19 130:21	100ay [27] 9:24 12:23	160:14 241:15 251:1	travels (3) 134.22 179.12
terminal [4] 44:16 58:8	181:23 185:11,15 197:18	13:25 16:4,14 17:7,17	trainer m 11.11	230.5
58:20 179:9	249:18	18:5,15 19:23 20:4,10		239.3
terminates (11, 173-17	thought [4] 3:23 6:22	21:13 24:17 26:25 29:2	training [76] 32:19 71:13	treat [2] 66:10 241:3
4 ample and 10 10 24 7	7:13 162:3	29:18 30:3 31:17 33:14	79:24 92:13 97:2,6,17	tremendous [1] 20:25
19:12 34:7	thoughts (1) 20.6	49:17 93:19 156:19 197:3	119:22 121:12 122:21,22	trends 11 135.8
36:25 37:17 49:17 50:11		202:20 218:7 268:23	122:22 129:3 160:12,13	
61:20 62:14 64:19 66:6	Inreal [2] 39:19 246:6	today's 11 17:14	166:13 168:25 169:16	1 revor [2] 18:19 19:1
89:17 96:9,13 97:19	three [109] 10:8 11:18,21	togothor (15) 12:1 2 4	211:20,21 218:20 219:13	trial [1] 69:7
98:10 99:5 109:1 114:8	11:23 12:7,11,24 13:9	12.11 24.11 07.21 102.25	220:9 222:1,15,16,23,25	tried (1) 215.5
114:24 117:3 118:20	15:9 16:11 18:14 19:7.9	13:11 24:11 97:21 102:25	223:2,2,7,12,16,22	trice [1] 213.5
122:6 126:16 135:7 147:2	19:24 20:10 23:3 25:14	116:21 117:4 125:14	224:15 225:9.12.14.15	trip [1] 137:2
147:14 155:23 170:17	29:8 31:22 33:16 38:15	126:4 127:12,23 131:16	225:16.17.23 226:12.14	troubling [1] 7:24
172:7,7 173:6 175:2	40.1 18 20 43.16 49.19	225:25	226.19.21.23.227.7.22	true (4) 3.9 27.3 208.16
192:17 195:6 198:10,24	51.19 54.9 25 57.21 24	tolerance [3] 116:7,12	232.24 233.1 20 234.7	271.3
202:14 220:19 232:11	58.12 16 25 64.1 66.5	116:13	232.24 233.1,20 234.7	271.5 Arms 20.21.21.5.0
246:17 247:18 261:15	74.17 24 75.10 10 77.12	tomorrow 181 0.24 17.15	236.13 237.1 7 238.10	LFY [8] 29:21 31:5,8
269:8	77.15 18 23 23 78.0	18.6 03.20 162.2 268.17	230.13 237.1,7 230.19	145:20 154:18 162:25
Torro (25) 19:22 24:22	20.16 22.12 26.11 02.2	16.0 93.20 102.2 208.17	240.10,18 240.13,18,20	168:13 199:12
1011 <i>a</i> [35] 18:22 24:22	00.10 05:15 00:11 92:5	209:0,15	247.5 251:10 252:1,4,5	trying [5] 154:10 182:14
25:11 26:8,11,16,19 34:6	92:7,21 94:1 109:13	ton [1] 155:21	258:17,20 262:1 264:22	192:6 199:4 224:14
34:24 39:12 40:2 46:11	128:24 129:15 130:9,14	tonnes [2] 248:15.19	transcend [2] 125:5	TSB (11, 11).4
46:14,14 47:4,5,20 49:3	131:25 133:25 134:5,7	tons (1) 42.16	127:22	
49:14 50:12,18 51:23	134:10 136:7,7 137:3	tons [1] 43:16	transcribed III 271.6	turkey [1] 241:18
53:16 54:3,12 86:14	144:16 145:8,11 148:24	too [12] 24:7 112:2 119:1		turn [2] 51:1 53:25
151:16 157:25 160:22	150:14 155:21 160:12	140:2 146:4 156:16 191:3	transcript [1] 2/1:3	turnaround (1) 200.5
180:13 184:7 186:4 189:8	161:25 162:5 163:9,14	218:25 221:16 252:1,2	transfer [3] 44:14 104:2	
189:24 197:17	163:24 164:2,7,12,17	268:6	248:21	turnea [1] 214:4
tested (1) 4·21	176:20 181:14,21 182:15	took [11] 42.18 122.15	transit [2] 11.17 12.14	turning [1] 201:1
togtified (1.0	183:4,15 184:16.16.21	164.4 195.18 196.10	thomasto J 164.10	turns (1) 217.11
testifieu [1] 61:9	185:1,6,18 186:5 187:6	197.12 18 200.12 218.25	transited [1] 164:13	turnet 50 0 50 15
testify [1] 212:19	187:9 194:1 198:6 229:24	220.17 221.16	transiting [1] 206:2	LUFFEL [4] 50:3 53:15
testifving n1 260.15	231:16 232:20 25 233.16	220.17 221.10	transition [10] 196.17	154:12,12
tostimony rate 02.17	243:2.249:9.267:5.11.11	tool [1] 112:22	197:2 209:6 211:17 22	twenty [1] 230:7
122.25 124.11 125 14	thron and 43 54 9	toolbox [5] 106:25 107:1	212.4 213.19 214.12 12	twice [2] 121.17 138.4
122:25 154:11 155:14	unee-anu [1] 54:8	107:5 135:12 136:2	212.7 213.17 217.12,12	
141:24 163:10 165:23	three-yearly [1] 77:16	tools 181 103.0 11 12 15	4	LWU [49] 6:3 10:8 14:11
212:/ 23/:14 249:/	thrilled [1] 28:17	103.22 22 25 124.10	transmitters [1] 251:8	25:14 31:19 37:2 43:18
		103.22,22,23 134:19		

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-PageTM

two-day - WHMIS Offshore Helicopter Safety Inquiry

			Ousliere neur	copter Safety Inquiry
44:6 45:2,13,24 46:2	undertaking [5] 2:3,21	users [3] 180:10 201:2	VHF [1] 248:22	152:23 166:20 216:4,5,7
57:17 58:6,11 61:10 64:5	3:19 7:3 23:11	227:11	via [2] 92.14 218.18	216:9 217:19 254:18
113:14 128:22 130:4	undertakings (2.10	11606 [2] 86.15 02.18		266:7
144:23 150:12 151:9	2.18 A.10 5.11 8.21	140.23	vibrating [1] 225:4	waters (1) 217.15
153:20 154:1.17 160:1	2.18 4.10 3.11 8.21	•	vibration [2] 199:25	
160:12 163:7.10 173:4		using [15] 31:6 38:19,20	200:5	wave [2] 37:18 153:14
182:16 184:2 185:6.24	undertook [1] 202:11	38:21 46:24 56:13 58:8	vicinity [1] 261:25	waves [1] 35:18
186:4 193:6 194:21 210:8	underwater [3] 168:24	103:24 127:12 165:19	video (2) 124-24 242-1	ways [6] 24:10 108:9
210:24 230:7 232:20	169:7 218:9	191:14 200:18 203:9	244.2	117:2 144:23 155:25
239:3.3.3.3.266:21.24	underway [2] 46:8 57:18	233:1 250:21	•	199:7
267:25	unfit (a) 221.18 222.7	usually [7] 29:21 34:4	VIEW [8] 27:11 94:6,9	W00r (c) 160.10 212.21
two down 222.2		60:1 95:18 107:6,15	200:12,14 225:21 226:6	weal [6] 109:19 212:21
	unintelligible [2] 265:15	136:15	226:8	212.22,24 218.20 232.4
twoiold [1] 240:25	268:9	utility (1) 43:18	vigilant [2] 93:11 114:10	wears [1] 115:16
type [19] 37:17 55:16	unique [6] 31:25 36:10	utilizo (n. 22.0 146.12	vintage (1) 199.2	weather [33] 37:5 38:25
73:8 94:1 95:5 108:16	36:14 38:5 39:1 172:5	250.14 266.10	virtuo (2, 67,10,110,6	53:20,22,24 80:3 144:7
112:22 135:7 145:7	unit [5] 47.4 12 56.10	230:14 200:10	120.22 120.12 126.1	144:18 146:9,21 147:1,3
150:11 160:2,10 170:16	88.17 204.12	utilized [3] 240:12 243:1	129:25 150:15 150:1	147:5,5,11,19,25 148:3
171:23 176:19,23 246:20	United (1) 22.9	249:20	107.8	148:8,10 155:7 171:25
247:4 256:4		utilizes [1] 58:6	visit [2] 83:17 86:22	172:6,7,20,24 174:12,12
types [16] 14:11 23 21:18	units [2] 176:16 219:13	utilizing (1) 259.21	visited [1] 155:10	175:2,6 184:8 205:21
39.4 54.23 55.9 56.7	University [3] 11:11		visits [3] 83.20 84.13	228:25
75.25 103.12 135.5 142.3	25:1 28:1		86.19	website [4] 3.2 14.13 22
160:1 165:13 179:3	unless 181 3.6 9.19 63.14	-V-	Vokowstan 19-21-25	65.23
197.10 199.19	164.23 168.4 201.23	valid 171 86.8 24 163.1	V UKEY [113] 18:21,25	
t	240.18 245.4	164:5 166:5 233:12 14	24:18,20,22 37:15 44:24	WEEK [7] 13:8 44:23 79:4
typical [5] 103:24 109:22	Unlimited as 071 10 14	volidated w 100.4	46:11,12 47:11 48:20	129:15,24 135:19 170:8
121:14 243:3 247:17	Ummiled [2] 2/1:12,14	vanuateu [1] 188:4	50:25 51:17,24 67:9,10	weekend [1] 266:24
typically [11] 23:10	unloading [1] 235:21	vaning [3] 53:20,22,24	6/:14 /8:10 88:5 138:19	weekly [2] 136:8 246:19
119:20 120:20 135:9	unsafe [1] 133:22	variety [3] 55:2,9 151:24	138:21 139:25 140:16,22	weeks [11] 10.13 12.21
140:6 141:8,23 142:1	unserviceable	various 1241 19:6 14	141:1,13,22 142:18 143:1	128.24 130.0 15 131.24
146:7 147:20 246:10	183.23	25.17 27.4 28.12 40.6	143:17 144:1,22 146:6	120.24 130.9,15 131.24
	105.25	50.21 54.22 64.9 10	146:11,16,25 14/:1/	268.12
-I]-	unsure [1] 251:24	75.17 78.21 81.20 92.7	148:4,9,13,20 149:1,5	200.12
	untouched [1] 20:12	111.20 110.3 125.5	149:18 151:8,21 152:21	weighs [1] 43:15
Uh-hm [2] 138:17 190:24	up [72] 1:11 6:11.21 8:16	1/120 117.5 125.5	157:24,25 158:12 159:24	weight [7] 89:9 105:12
UK [5] 29:15 70:15 71:5	15:19 16:16 19:13 25:25	222.4 220.5 250.4 263.22	160:21 164:24 165:1,6	155:22 179:16,22 193:10
71:7 86:15	26:10 31:16 40:5 42:8	222.4 230.3 239.4 203.22	167:3,10 168:6,15,21	208:25
	43:21 48:25 54:1.14 60:1	vary [1] 157:19	169:23 170:13 171:11,18	weights [2] 240:20
	60:14 61:24 65:25 66:1	vast [3] 260:7 266:4,8	171:22 172:2,25 173:15	243:24
ultimately [10] 12:15	83:19 104:10 107:10	vehicle [3] 194:3 216:14	173:25 175:15,21,25	well-designed (1) 07.14
22:7 133:12 173:11 181:8	114:19.23 116:12.25	216:15	176:8,13 177:2,8,12,24	well-designed [1] 97.14
195:13,20,22 196:13	117:20 124:8 127:7	vonture (2) 33.22 34.1	178:5,15,25 180:18	well-trained [1] 116:18
240:3	136:12.13.22.150:20	51 .0	181:18,25 182:10,18	wellhead [1] 48:5
umbrella [1] 4:13	154:5 155:5 156:22	51.9	183:12,19 185:21 186:8	wellheads [5] 47:3.21
unable [2] 53.17 102.20	164.23 174.17 182.16	venturers [5] 33:23 35:3	186:20 188:13,19,25	47.24 48.2.7
	184:1 188:21 190:15.17	59:19 60:3,16	189:5 190:9,25 191:23	woller 1 47.5 14 15
	198.16.21.214.5.215.20	ventures [3] 31:21 33:15	192:3,8,12,16,24 194:5	Wells [7] 47.3,14,13
225:1	215.21 216.21 234.4	33:21	194:14,19 198:12,20	48:22,25 34:10 145:2
unconscious [2] 217:22	238.21 241.17 242.12	verified (2) 217.2 240.16	201:5 210:17 230:21	west [4] 29:7,10,16 45:17
241:4	244.8 250.2 3 7 251.5	vorify (1) 174.22	238:9 268:17	Western [3] 28:3,6 29:14
under [21] 14:12 32:5	253:3.7.10.22.25.254:7	verity [1] 174:25	volume [1] 115:8	Westland [2] 193:8.20
36:9 50:10 61:7 63:9	259:12.14 265:20 266:23	versus [1] 39:4	vovage [1] 206:1	wheel (1) 155.6
80:5 82:15,16,23 85:19	266:25 267:4	vessel [34] 22:24 46:25	, oj ugo [1] 20011	
90:10 125:14 138:7	undate (2) 246-16 262-11	49:10,17,21,23,25 50:4		wheels [12] 184:1 250:3
149:11 218:15 219:14,16		50:5,7 52:14,15 53:13	- VV -	251:5 253:3,7,10,22,25
236:25 237:15 265:7	updates [1] 121:22	54:1 56:12 58:1,3 80:10	wait 11 199:12	254:6 266:23,25 267:4
undergo (11, 160:11	upper [1] 166:21	101:7 147:9 149:9 150:22	welking (1) 24.14	whereabouts [1] 40:20
undersigned (1) 271.2	unwards 121 50.10 16	150:23 151:4,13,15,24		whereas (2) 39.12
		152:10 158:1 205:20	wall [1] 47:23	150:22
understand [22] 2:25	177.19 179.1 14 107.9	206:1,7 240:4 265:11	wants [1] 239:11	whorehy (a) 125.6
8:16 19:15 93:13 103:1	1/7.10 178.1,14 197.8	vessel's [1] 248:17	warning [1] 201:6	WHELEDY [2] 155.0
113:1 129:14 154:3	used [28] 14:25 47:9	VOCCOL E [45] 29.19 21 55.2	worront (1) 94.11	130.11
159:11 163:17,19 176:6	55:17,20,22 59:1 71:18	57.5 7 58.4 11 72.0 91.4		wnerever [1] 53:25
181:11,12 183:14 191:20	93:16 103:15 109:23	138.24 144.6 140.21	wasn [1] 38:21	Whiffen [1] 49:5
203:25 204:3 214:1	110:13 117:15 151:9,10	130.24 144:0 149:21	watch [2] 80:10 222:13	whilst [2] 92:21 118:21
254:25 256:22 259:20	167:17 169:9 189:19	150.5,15,10,10 151.7,9	watching 111 20:4	White [15] 22.10 24.6
understands [1] 154:16	190:8 194:8 196:6 202:5	151.11,12152.7,9,13,10 $153.2510167.1102.12$	water 1371 22.17 22.2	35.1 40.3 40.2 0 51.9 10
understatement	213:14,15 219:9 220:4	247.10 21 25 248.2 15	35.20 38.21 42.17 43.4	86.15 87.17 175.6 150.20
28:17	224:2 243:21 259:19	247.17,21,25 240.5,15	A3.7 20 A7.10 21 A8.1 2	186.5 101.1 107.17
understood 121 7.13 14	useful [2] 11:8 202:9	265.6 10 266.12 14 15	48.10 14 24 51.20 55.6	
213.21	user [3] 208:12 218:11	265.0,10 200.12,14,15	115.6 8 18 122.14 20	willea [1] 15:2
undentelse se col o	266:8	207.10,21 Nosta (1) 202.2	123.4 6 10 152.17 20 22	WHMIS [1] 234:24
undertake [1] 221:3		vests [1] 203:3	123.7,0,10 132.17,20,22	

Discoveries Unlimited Inc., Ph: (709)437-5028

Multi-Page[™]

		Onshore mene	opter safety m
whole [4] 12:15 174:4	261:7 262:8,12,19,23		
213:19 214:3	workings [1] 118:24		
wide [1] 49:20	workplace [2] 20:22		
widely [1] 110:13	102:18		
wider [2] 115:21 116:14	WORKS [18] 21:4 24:5,6 51:25 56:16 02:1 04:22		
wife [1] 28:18	99:8.10 104:13 126:16		
wiggle [1] 173:4	127:18,19,21 172:22		
willing [1] 133:16	181:7 225:25 235:2		
wind [6] 35:16 37:18 50:6 53:25 147:8 153:14	world [21] 14:19 25:19		
windows (1) 223.20	40.1 49.16 56.23 57.1		
winds [2] 172.11 175.3	59:9,10 87:25 92:23 93:3		
wing $[2]$ 265.4 21 266.14	110:18 173:6 193:22		
wing [5] 203.4,21 200.14	220:1,3		
winter [1] 147.21	world's [1] 28:9		
with $[1] 30.19$	worldwide [3] 87:8,10		
WISH [6] 9:9 10:2 19:11 54:5 131:14 132:22	268:15		
wishes 11 241.13	wrist [1] 232:3		
withdraw (1) 179.12	written [1] 79:16		
within [10] 2.16 10.0 20	wrong [2] 159:2 199:10		
14:23 15:20 59:22 78:18			
82:1 91:7 126:20 142:2	-X-		
155:16,17,18 250:2,3	X [2] 150:15 157:11		
259:6 263:24 266:9	x-ray [1] 241:7		
without [5] 89:5 109:24			
153:13 220:12 222:25	-Y-		
withstand [2] 39:11			
44:11	year [26] 2:4 25:18 24:2 25:22 26:1 29:22 69:23		
witness [1] /:21	83:14 84:13 107:20		
WILDESSES [4] 18:23	128:22 133:25 135:20		
Wolff (1) 22:6	140:6 150:7 153:8 156:5		
wonder (2) 9.7 225.2	160:12 170:8 173:20		
wonder [2] 8:7 223:3	251.4 267.5		
231.22	vearly (1) 121.17		
word [9] 15.1 127.13	vogr s (27) 21.15 23.25		
152:19 167:4.4 168:10	24:9.24 25:17 26:7.22		
192:6 224:1	27:4,24 28:5,8,22 42:20		
worded [1] 184:14	51:7 74:17,24 77:13,19		
wording [2] 176:12,14	92:18 93:1 116:21 145:9		
words [4] 111:12 169:13	1/0:12,15 193:4 197:14		
211:20 259:13	223:15 232:19.20.21.25		
wore [1] 214:21	233:14,16 239:3		
worked [18] 21:17,19	yellow [5] 111:21 202:16		
22:5,20 23:24 28:3,12	202:18 260:24 261:24		
28:24 59:10,13 116:20			
120:5 150:10 170:11 195:20 211:19 236:10	-Z-		
268:9	zip [2] 242:12 244:8		
worker [10] 22:8 77:9.9	zipper (1) 212:10		
107:10 133:5,20 159:15	ZODE [1] 38:12		
232:12 262:20,21			
workers [21] 78:2 100:4			
100:5 107:2 120:2 124:15			
124:16,25 125:7 129:14			
155:11,15 135:4 159:3			
201.2 223.17 220.3,22			
workforce (21) 20.11			
20:14,20.22 29:25 78:7			
98:11 106:13,21 107:15			
108:17 135:12,14 156:14			
198:15 199:6 211:23			
213:4 220:10 226:15			
221:3,0,13,20 231:22,24			